Oracle Case Management Solutions

Downloaded by [5.154.190.33] at

Oracle Case Management Solutions

Léon Smiers Manas Deb Joop Koster Prasen Palvankar



CRC Press is an imprint of the Taylor & Francis Group, an **informa** business AN AUERBACH BOOK CRC Press Taylor & Francis Group 6000 Broken Sound Parkway NW, Suite 300 Boca Raton, FL 33487-2742

© 2016 by Taylor & Francis Group, LLC CRC Press is an imprint of Taylor & Francis Group, an Informa business

No claim to original U.S. Government works Version Date: 20150813

International Standard Book Number-13: 978-1-4822-2384-2 (eBook - PDF)

This book contains information obtained from authentic and highly regarded sources. Reasonable efforts have been made to publish reliable data and information, but the author and publisher cannot assume responsibility for the validity of all materials or the consequences of their use. The authors and publishers have attempted to trace the copyright holders of all material reproduced in this publication and apologize to copyright holders if permission to publish in this form has not been obtained. If any copyright material has not been acknowledged please write and let us know so we may rectify in any future reprint.

Except as permitted under U.S. Copyright Law, no part of this book may be reprinted, reproduced, transmitted, or utilized in any form by any electronic, mechanical, or other means, now known or hereafter invented, including photocopying, microfilming, and recording, or in any information storage or retrieval system, without written permission from the publishers.

For permission to photocopy or use material electronically from this work, please access www.copyright.com (http:// www.copyright.com/) or contact the Copyright Clearance Center, Inc. (CCC), 222 Rosewood Drive, Danvers, MA 01923, 978-750-8400. CCC is a not-for-profit organization that provides licenses and registration for a variety of users. For organizations that have been granted a photocopy license by the CCC, a separate system of payment has been arranged.

Trademark Notice: Product or corporate names may be trademarks or registered trademarks, and are used only for identification and explanation without intent to infringe.

Visit the Taylor & Francis Web site at http://www.taylorandfrancis.com

and the CRC Press Web site at http://www.crcpress.com

Contents

Fore	eword.			xvii
List	of Acı	ronyms		xxi
Pref	ace			XXV
Ack	nowle	dgment	S	xxix
Aut	hors			xxxi
SEC			HE CASE FOR CASE MANAGEMENT	1
1	Basi	cs of Ca	se Management	3
	1.1	Introd	uction	3
	1.2	Unexp	ected Surprises (or Shocks)	4
		1.2.1	"Are You Being Served?"	4
		1.2.2	"Doctor, I Have a Headache"	5
		1.2.3	"Look! There's a Dead Man in the Closet!"	7
		1.2.4	Goal-Driven, Knowledge Work Freedom, On-Demand Collaboration	9
	1.3	Handl	ing of Credit Card Rewards Disputes	10
	1.4	Case N	Management: Basics	14
		1.4.1	What Is Case Management? What Is Adaptive Case Management?	14
		1.4.2	Case Management Ecosystem	18
		1.4.3	Brief History of Case Management	20
		1.4.4	Formal Definition of (Adaptive) Case Management	21
		1.4.5	Application of Case Management Solutions	24
	1.5	Creati	ng Čase Management Solutions	27
		1.5.1	High-Level Solution Architectures	27
		1.5.2	Build or Buy?	30
		1.5.3	Building a Fit-for-Purpose Case Management Solution	33
	1.6	ACM	and Business Agility	40
	1.7	Summ	ary	45
2	Adar	ntive Ca	se Management (ACM)–Related Tonics	47
-	2.1	Introd	uction	
	2.2	Adapti	ive Systems and ACM	48
	2.2	2.2.1	Nile Crocodiles and Cuttlefish	
		2.2.2	Sense-and-Respond, Anticipate-and-Preempt, and Complex Adaptive	10
			Systems (CAS)	49
		2.2.3	ACM Solutions as Sense-and-Respond (Complex) Adaptive Systems	
		9	(SR-CAS)	55

Dynamic, Production, and Adaptive Case Management (DCM, PCM,				
and AC	CM)	61		
2.3.1	Basic Debates	61		
2.3.2	Digging Deeper into DCM, PCM, and ACM	. 63		
Process	Management and Case Management: Friends or Foes?	. 70		
2.4.1	Anatomy of a Journey	. 70		
2.4.2	Process Management vis-à-vis Case Management	. 71		
2.4.3	Case Management as Part of a BPMS?	76		
Adopti	on of Case Management: Capability Maturity Considerations	79		
2.5.1	"Generalized" BPM	. 80		
2.5.2	Maturity Levels	81		
2.5.3	Maturity Model Capabilities	. 82		
2.5.4	Adoption of Case Management: A Case Study	. 88		
2.5.5	Maturity Is More Than Technology	91		
Making	g the Business Case for Case Management	. 92		
2.6.1	General Considerations for Creating and Presenting a Business Case	92		
2.6.2	Specific Considerations for a Case Management Business Case	. 94		
Summa	ary	. 97		
	Dynam and AC 2.3.1 2.3.2 Process 2.4.1 2.4.2 2.4.3 Adoptio 2.5.1 2.5.2 2.5.3 2.5.4 2.5.5 Making 2.6.1 2.6.2 Summa	Dynamic, Production, and Adaptive Case Management (DCM, PCM,and ACM)2.3.1Basic Debates2.3.2Digging Deeper into DCM, PCM, and ACMProcess Management and Case Management: Friends or Foes?2.4.1Anatomy of a Journey2.4.2Process Management vis-à-vis Case Management2.4.3Case Management as Part of a BPMS?Adoption of Case Management: Capability Maturity Considerations2.5.1"Generalized" BPM2.5.2Maturity Levels2.5.3Maturity Model Capabilities2.5.4Adoption of Case Management: A Case Study.2.5.5Maturity Is More Than TechnologyMaking the Business Case for Case Management.2.6.1General Considerations for Creating and Presenting a Business Case2.6.2Specific Considerations for a Case Management Business CaseSummary.		

Case	Manag	ement Fu	Inctional Design	101			
3.1	Introd	Introduction					
3.2	Princip	oles for th	e Solution Design Components	101			
3.3	Desigr	n Compor	ients	102			
	3.3.1	Design	Components Overview	102			
	3.3.2	Case Lif	fecycle Design Category	103			
		3.3.2.1	Aim	103			
		3.3.2.2	Relation with the Principles	104			
		3.3.2.3	Case Management Lifecycle Design Components	104			
	3.3.3	Case Int	formation Design Category	108			
		3.3.3.1	Aim	108			
		3.3.3.2	Relation with the Principles	108			
		3.3.3.3	Case Management Information Design Components	108			
	3.3.4	Case Int	teraction Design Category	112			
		3.3.4.1	Aim	112			
		3.3.4.2	Relation with the Principles	112			
		3.3.4.3	Case Management Interaction Design Components	113			
	3.3.5	Case Cr	oss-Functional Design Category	114			
		3.3.5.1	Aim	114			
		3.3.5.2	Relation with the Principles	114			
		3.3.5.3	Case Management Cross-Functional Design Components	114			
3.4	Desigr	n Templat	e	116			
3.5	Conclu	usion		116			

3

4	Case	Manag	ement Classifications	119
	4.1	Introd	uction	119
	4.2	Forres	ter Case Management Classifications	119
	4.3	Comn	non Principles for the Classifications	120
	4.4	Service	e Request Classification	120
		4.4.1	Context	120
		4.4.2	Determination	120
		4.4.3	Alignment with Industry Processes	121
		4.4.4	Business Challenges	122
		4.4.5	Principles for the Solution	123
		4.4.6	Case Management Design Components	123
			4.4.6.1 Service Request Design Category Overview	124
			4.4.6.2 Case Lifecycle Design Category	124
	4.5	Incide	nt Management Case Classification	133
		4.5.1	Context	133
		4.5.2	Determination	134
		4.5.3	Alignment with Industry Processes	134
		4.5.4	Business Challenges	134
		4.5.5	Principles for the Solution	135
		4.5.6	Case Management Design Components	135
			4.5.6.1 Incident Management Design Category Overview	135
			4.5.6.2 Case Lifecycle Design Category	135
	4.6	Investi	igative Case Classification	143
		4.6.1	Context	143
		4.6.2	Determination	145
		4.6.3	Alignment with Industry Processes	145
		4.6.4	Business Challenges	145
		4.6.5	Principles for the Solution	145
		4.6.6	Case Management Design Components	146
			4.6.6.1 Investigative Handling Design Category Overview	146
			4.6.6.2 Case Lifecycle Design Category	146
	4.7	Concl	usion	153
5	Dete	rmina	Solution Manning	155
9	5 1	Introd	uction	
	5.2	Techni	iques Used to Determine the Impact of Functionality onto the	
).2	IT Lar	adscape	155
		521	Common Reference Architecture Model	1))
		5.2.1	COP A Model Lavering and Canabilities	1))
		5.2.2	Delivering a Solution with the COP A Model	190
	5 2	J.2.5 Doutou	m a Europianal Decomposition Using COPA Application Lawre	1)/
	5.5	5 2 1	Haw to Man the Case Management Design Areas ant st	1)ð
		י.כ.ו	COP A Application Lawre	150
			5.3.1.1 Casa Lifacuela Design	1)ð 150
			5.2.1.2 Case Intervetien Design	159
			9.9.1.2 Case Interaction Design	101

		5.3.1.3 Case Management Information Design	162
		5.3.1.4 Cross-Functional Design Areas	162
	5.4	Determine Required Technology Capabilities	165
		5.4.1 Case Flows and Processes Capabilities	165
		5.4.2 Event Handling Capabilities	165
		5.4.3 Business Rules Capabilities	165
	5.5	Decide What Technologies to Use	168
	5.6	Conclusion	168
6	Con	cluding Remarks	171
SEC		N III ORACLE CASE MANAGEMENT TECHNOLOGIES	173
7	Orac	le BPM/ACM	177
	7.1	Introduction	177
	7.2	Oracle BPM/ACM/SOA Suite Overview	177
	7.3	Oracle BPM/ACM Usage Characteristics	179
	7.4	Case Management Lifecycle Functionality with Oracle BPM/ACM	179
		7.4.1 Case Flow and Process Design	180
		7.4.1.1 Case Flow Design with Oracle Adaptive Case Management	180
		7.4.1.2 Process Modeling	181
		7.4.2 Event Design	182
		7.4.3 Rules Design	182
		7.4.4 Life Cycle Design Product Mapping to Oracle BPM/ACM/SOA Suite	183
		7.4.4.1 Case Flow and Process Mapping	183
		7.4.4.2 Event Handling Mapping	185
	7.5	Case Management Interaction Functionality with Oracle SOA Suite	185
		7.5.1 Portal Design: 360-Degree View of the Case	186
		7.5.1.1 Out-of-the Box Functionality via the Oracle BPM/ ACM Portal	186
		7.5.1.2 ADF, Customized Functionality with the Aid of the	
		Case Management/BPM and Human Task APIs	186
		7.5.2 Channel Design	187
	7.6	Case Management Information Functionality with Oracle BPM/ACM/	
		SOA Suite	187
		7.6.1 Case Execution Data: Detailed Audit Trail	188
		7.6.2 Document and Media Design	189
	7.7	Cross-Functional Case Management Capabilities	189
		7.7.1 Case Management Integration Functionality with Oracle SOA Suite	190
		7.7.2 Case Management Security Functionality with Oracle	190
		7.7.2.1 Case Stakeholders	190
		7.7.2.1 Permissions	190
	7.8	Conclusion	191
8	Sieb	el	193
2	8.1	Introduction	193
	8.2	Oracle Siebel Overview	193

	8.2.1	History	
	8.2.2	Horizontal and Verticals	194
	8.2.3	Configuration versus Customization	194
	8.2.4	Extensions of Functionality	194
	8.2.5	Siebel Specific Terminology	194
	8.2.6	Market Example: Law Enforcement and Policing	
		Operations Solution	195
8.3	Oracle	e Siebel Usage Characteristics	195
8.4	Case N	Management Information Functionality with Oracle Siebel	198
	8.4.1	People Relationship Design and Data Design	198
	8.4.2	Document and Media Design	200
8.5	Case N	Management Life Cycle Functionality with Oracle Siebel	200
	8.5.1	Case Flow and Process Design with Oracle Siebel	200
		8.5.1.1 State Model	201
		8.5.1.2 Workflow Management	202
		8.5.1.3 Assignment Management	202
		8.5.1.4 Extending the Life Cycle Management, with the	
		Law Enforcement and Policing Operations Example	202
	8.5.2	Rules Design	203
	8.5.3	Event Design	203
	8.5.4	Life Cycle Design Product Mapping to Oracle Siebel and Related	
		Products	203
8.6	Case N	Management Interaction Functionality with Oracle Siebel	205
	8.6.1	Portal and Channel Design	205
8.7	Cross-	Functional Case Management Capabilities	206
	8.7.1	Case Management Integration Functionality with Oracle SOA Su	ite206
	8.7.2	Case Management Roles and Authorization Functionality with	
		Oracle Siebel	206
8.8	Conclu	usion	207

9	Orac	Oracle Service Cloud		
	9.1	Introd	uction	209
	9.2	Oracle	Service Cloud Overview	209
		9.2.1	History	209
		9.2.2	Basic Functionality	210
		9.2.3	Cloud-Based Management and Usage	210
		9.2.4	Extensions of Functionality	210
	9.3	Oracle	Service Cloud Usage Characteristics	211
	9.4	Case N	Aanagement Interaction Functionality with Oracle Service Cloud	212
		9.4.1	Portal Design	212
		9.4.2	Channel Design	213
		9.4.3	Dashboard Design	214
	9.5	Case N	Aanagement Information Functionality with the Oracle Service Cloud	214
		9.5.1	Data Design	215
		9.5.2	People Relationship Design	217
		9.5.3	Document and Media Design	217

	9.6	Case M	Ianagement Life Cycle Functionality with Oracle Service Cloud	
		9.6.1	Case Flow and Process Design	
		9.6.2	Rules Design	
		9.6.3	Event Design	
		9.6.4	Life Cycle Design Product Mapping to Oracle Service Cloud	220
	9.7	Cross-H	Functional Case Management Capabilities	220
		9.7.1	Integration	220
		9.7.2	Case Management Roles and Authorization Functionality	220
	9.8	Conclu	ision	222
10	Intel	ligence i	in Relation to the Case Execution	223
10	10.1	Introdu	action	223
	10.1	Case Ir	ntelligence Design	
	10.2	10.2.1	Intelligence Determined by Rules Design	
		10.2.1	Intelligence Determined by Event Design	224
		10.2.2	Intelligence Determined by Dashboard Design	225
		10.2.9	Intelligence Information Levels	225
	10.3	Oracle	Intelligence Products	225
	1010	10.3.1	Oracle Business Analytic Monitoring	225
		101011	10.3.1.1 Functionality	
			10.3.1.2 How Does BAM Relate to Case Management?	
			10.3.1.3 Technical Architecture	
		10.3.2	Oracle Event Processing	
		- • • • • • •	10.3.2.1 Functionality	
			10.3.2.2 How Does CEP Relate to Case Management?	
		10.3.3	Oracle Policy Automation	
			10.3.3.1 Functionality	
			10.3.3.2 How Does OPA Relate to Case Management?	
		10.3.4	Oracle Endeca Information Discovery	
		-	10.3.4.1 Functionality	
			10.3.4.2 How Does Endeca Information Discovery Relate to	
			Case Management?	
		10.3.5	Oracle Real-Time Decisions	
			10.3.5.1 Functionality	
	10.4	Summa	ary	
11	Con	Judina	Domostro	220
	Cond	Juuing I	NV111a1 RJ	

Mort	gage Request Handling	245
12.1	Introduction	
12.2	Reading Guide	
12.3	Mortgage Request and Business Challenges	
-	12.3.1 Mortgage Request	
	12.3.2 Business Challenges	246
	Mort 12.1 12.2 12.3	Mortgage Request Handling 12.1 Introduction 12.2 Reading Guide 12.3 Mortgage Request and Business Challenges 12.3.1 Mortgage Request 12.3.2 Business Challenges

		12.3.2.1 Serving the Customers	246
		12.3.2.2 New Regulations	246
		12.3.2.3 Minimizing the Probability of Default	247
12.4	Derivir	ng a Solution Based upon the Case Management Solution Framework	247
	12.4.1	Introduction	247
	12.4.2	Determine Case Management Classification	247
	12.4.3	Determine Functional Components and Needs	247
		12.4.3.1 Case Flow and Process Design	248
		12.4.3.2 Case Information Design	251
		12.4.3.3 Case Interaction Design	254
	12.4.4	Determine Solution Mapping	255
		12.4.4.1 Impact of a Mortgage Request Operating Model on	
		Solution Capabilities	255
		12.4.4.2 Mapping Mortgage Request Operating Models to	
		Oracle Products	258
12.5	Conclu	sion	262

13	Citiz	en Services in the Public Sector	263
	13.1	Introduction	263
	13.2	Reading Guide	263
	13.3	Welfare Request and Business Challenges	264
		13.3.1 Welfare Request Handling	264
		13.3.2 Business Challenges	264
	13.4	Deriving a Solution Based upon the Case Management Solution Framework	265
		13.4.1 Introduction	265
		13.4.2 Determine Case Management Classification	265
		13.4.3 Determine Functional Components/Needs	265
		13.4.3.1 Case Life Cycle Design	265
		13.4.4 Determine Solution Mapping	271
		13.4.4.1 Determine Solution Capabilities	271
		13.4.4.2 Mapping the Welfare Request to Oracle Products	273
	13.5	Conclusion	276

From Crime to Court			
14.1	Introduction	. 277	
14.2	Reading Guide	. 277	
14.3	Police Investigation and Business Challenges	. 278	
	14.3.1 Police Investigation	. 278	
	14.3.2 Business Challenges for the Police Force Related to Crime Investigation	279	
	14.3.2.1 Globalization	. 279	
	14.3.2.2 Effectiveness of the Police Organization	. 280	
	14.3.2.3 Technology	. 280	
	14.3.2.4 Social Values Are Changing Rapidly	. 280	
	14.3.2.5 Adaption to Social and Crime Patterns	. 280	
	14.3.2.6 National Budgets Are Being Reduced	281	
	From 14.1 14.2 14.3	From Crime to Court 14.1 Introduction 14.2 Reading Guide 14.3 Police Investigation and Business Challenges 14.3.1 Police Investigation 14.3.2 Business Challenges for the Police Force Related to Crime Investigation 14.3.2.1 Globalization 14.3.2.2 Effectiveness of the Police Organization 14.3.2.3 Technology 14.3.2.4 Social Values Are Changing Rapidly 14.3.2.5 Adaption to Social and Crime Patterns 14.3.2.6 National Budgets Are Being Reduced	

	14.4	Solution Design Aspects	281
		14.4.1 Introduction	281
		14.4.2 Determine Case Management Classification	281
		14.4.3 Determine Functional Components/Needs	281
		14.4.3.1 Case Life Cycle Design	281
		14.4.3.2 Case Information Design	284
		14.4.3.3 Case Interaction Design	285
		14.4.4 Deriving a Solution Based upon the Case Management Solution	
		Framework	288
		14.4.4.1 Full-Blown or Umbrella Type of Solution	288
		14.4.4.2 Solution Mapping	290
		14.4.4.3 Mapping Police Investigation to Oracle Products	292
	14.5	Conclusion	294
15	Fnd-	to-End Insurance Claim Management	297
15	15 1	Introduction	297
	15.2	Reading Guide	297
	15.3	Insurance Claims and Challenges	298
	1).5	15.3.1 Insurance Claims	298
		15.3.2 Business Challenges Related to Insurance Claims	298
		15.3.2.1 Financial	298
		15.3.2.2 Customer Facing	299
		15.3.2.3 Supporting IT Landscape	299
	15.4	Deriving a Solution Based upon the Case Management Solution Framework.	299
		15.4.1 Introduction.	299
		15.4.2 Case Management Classification	299
		15.4.3 Determine Functional Components and Needs	300
		15.4.3.1 Case Life Cycle Design	300
		15.4.3.2 Rules Design	301
		15.4.3.3 Case User Interface Design	301
		15.4.4 Determine Solution Mapping	303
		15.4.4.1 Determine Solution Capabilities	303
		15.4.4.2 Mapping the End-to-End Insurance Claim Handling	
		to Oracle Products	304
	15.5	Conclusion	307
16	Cond	luding Remarks	309
SEC		V TUTORIAL	311
17	Orac	le BPM/ACM Tutorial	313
	17.1	Introduction	313
	17.2	Credit Card Charge Dispute: Use Case	313
		17.2.1 Use Case Design Topics	313
		17.2.2 Use Case Description	313
		17.2.3 Credit Card Charge Dispute Life Cycle Design	317

		17.2.3.1 Case Flow and Process Design	317
		17.2.3.2 Rules Design	319
		17.2.3.3 Event Design	321
	17.2.4	Credit Card Charge Dispute Information Design	322
		17.2.4.1 People Relationship Design	322
		17.2.4.2 Data Design and Document Management Design	323
17.3	Installa	ation and Tutorial Files	324
17.4	Key El	ements of Oracle Adaptive Case Management	325
	17.4.1	Case	325
	17.4.2	Case Activities	326
		17.4.2.1 Activity Implementation	326
		17.4.2.2 Activity Properties	326
	17.4.3	Case Data	326
		17.4.3.1 Data Definition	326
		17.4.3.2 External Case Data	327
	17.4.4	Case Stakeholders	328
	17.4.5	Case Documents	328
	17.4.6	Rules and Policies	328
	17.4.7	Case Events	329
17.5	Implen	nent the Case with Oracle ACM	329
	17.5.1	Create the Case Management Project	331
	17.5.2	Define the Case Properties	333
		17.5.2.1 Title	333
		17.5.2.2 Summary	333
		17.5.2.3 Priority	334
		17.5.2.4 Category	334
		17.5.2.5 Due Time	334
		17.5.2.6 Milestones	334
		17.5.2.7 Outcomes	334
	17.5.3	Define the Data Model	335
		17.5.3.1 Identify the Case Data Objects	335
		17.5.3.2 Define the Case Data Objects	336
		17.5.3.3 Create Data Forms	340
		17.5.3.4 Define Flex Fields	341
	17.5.4	Content Configuration	343
	17.5.5	Define User Events	344
	17.5.6	Define the Stakeholders and Permissions	344
		17.5.6.1 Application Roles	344
		17.5.6.2 Application Policies	345
		17.5.6.3 Stakeholders	345
		17.5.6.4 Permission Labels	345
		1/.5.6.5 Define Koles	
		1/.5.6.6 Define Stakeholders	346
	1757	1/.5.6./ Define Permissions	34/
	17.5./	Define Case Activities	348
		1/.)./.1 Create a BPMIN-Based Activity	
		1/.5./.2 Create a Human Task–Based Case Activity	

	17.5.8	Define the Rules	
		17.5.8.1 Activity Event Rules	
		17.5.8.2 Milestone Event Rules	
		17.5.8.3 User Event Rules	
	17.5.9	Resulting Application	363
17.6	Deploy	and Test	
	17.6.1	Prepare Server for Deployment and Test	
		17.6.1.1 Start the Integrated WebLogic Server	
		17.6.1.2 Enable Development Mode	
		17.6.1.3 Create Test Users	
	17.6.2	Deploy Case Application	366
		17.6.2.1 Deploy the Case Composite	366
		17.6.2.2 Deploy Case Data Forms Application	366
		17.6.2.3 Deploy Task Form Applications	
	17.6.3	Create a Case Instance	
	17.6.4	Working on the Case	368
		17.6.4.1 Case Workspace	369
		17.6.4.2 Update Process Roles	
		17.6.4.3 Grant Investigator Access Permission	
		17.6.4.4 Scenario #1	
		17.6.4.5 Scenario #2	
17.7	Summa	ary	
17.A	Append	dix	
Siebe	l Tutori	al	
Siebe 18.1	l Tutori Introdi	al	379
Siebe 18.1 18.2	e l Tutori Introdu Siebel I	al uction	379 379 379
Siebe 18.1 18.2	e l Tutori Introdu Siebel I 18.2.1	al action Inplementation Approach Choosing the Siebel Solution Type	379
Siebe 18.1 18.2	el Tutori Introdu Siebel I 18.2.1 18.2.2	al action Inplementation Approach Choosing the Siebel Solution Type Familiarization	379
Siebe 18.1 18.2	! Tutori Introdu Siebel I 18.2.1 18.2.2 18.2.3	al iction Implementation Approach Choosing the Siebel Solution Type Familiarization Design	379 379 380 380 381
Siebe 18.1 18.2	el Tutori Introdu Siebel I 18.2.1 18.2.2 18.2.3	al action Implementation Approach Choosing the Siebel Solution Type Familiarization Design 18.2.3.1 Functional Design	
Siebe 18.1 18.2	el Tutori Introdu Siebel I 18.2.1 18.2.2 18.2.3	al Iction Implementation Approach Choosing the Siebel Solution Type Familiarization Design 18.2.3.1 Functional Design 18.2.3.2 Solution Mapping	
Siebe 18.1 18.2	el Tutori Introdu Siebel I 18.2.1 18.2.2 18.2.3 18.2.4	al iction Implementation Approach Choosing the Siebel Solution Type Familiarization Design 18.2.3.1 Functional Design 18.2.3.2 Solution Mapping Implementation	379 379 379 380 380 381 381 381 385 385
Siebe 18.1 18.2	el Tutori Introdu Siebel I 18.2.1 18.2.2 18.2.3 18.2.4 Using t	al Inction Implementation Approach Choosing the Siebel Solution Type Familiarization Design 18.2.3.1 Functional Design 18.2.3.2 Solution Mapping Implementation the Police Investigation Application	379 379 379 380 380 381 381 381 385 385 385 385
Siebe 18.1 18.2	El Tutori Introdu Siebel I 18.2.1 18.2.2 18.2.3 18.2.4 Using t 18.3.1	al Inction Implementation Approach Choosing the Siebel Solution Type Familiarization Design 18.2.3.1 Functional Design 18.2.3.2 Solution Mapping Implementation the Police Investigation Application Storyline	379 379 379 380 380 380 381 381 385 385 385 386 386 387
Siebe 18.1 18.2	el Tutori Introdu Siebel I 18.2.1 18.2.2 18.2.3 18.2.4 Using t 18.3.1 18.3.2	al Inction Implementation Approach Choosing the Siebel Solution Type Familiarization Design 18.2.3.1 Functional Design 18.2.3.2 Solution Mapping Implementation the Police Investigation Application Storyline Recording a Lead	379 379 379 380 380 381 381 381 385 385 385 385 387 387 387
Siebe 18.1 18.2	el Tutori Introdu Siebel I 18.2.1 18.2.2 18.2.3 18.2.4 Using t 18.3.1 18.3.2	al Inction Implementation Approach Choosing the Siebel Solution Type Familiarization Design 18.2.3.1 Functional Design 18.2.3.2 Solution Mapping Implementation the Police Investigation Application Storyline Recording a Lead 18.3.2.1 Creating the Lead and Entering the Generic	379 379 379 380 380 381 381 381 385 385 385 385 387 387
Siebe 18.1 18.2	el Tutori Introdu Siebel I 18.2.1 18.2.2 18.2.3 18.2.4 Using t 18.3.1 18.3.2	al Inction	379 379 379 380 380 381 381 381 385 385 385 385 387 387 387
Siebe 18.1 18.2	E Tutori Introdu Siebel I 18.2.1 18.2.2 18.2.3 18.2.4 Using t 18.3.1 18.3.2	al Inction Implementation Approach Choosing the Siebel Solution Type Familiarization Design 18.2.3.1 Functional Design 18.2.3.2 Solution Mapping Implementation the Police Investigation Application Storyline Recording a Lead 18.3.2.1 Creating the Lead and Entering the Generic Lead Information 18.3.2.2 Entering the Person, Object Location, and Event	379 379 379 380 380 381 381 385 385 385 386 387 387 387
Siebe 18.1 18.2	E Tutori Introdu Siebel I 18.2.1 18.2.2 18.2.3 18.2.4 Using t 18.3.1 18.3.2	al Inction Implementation Approach Choosing the Siebel Solution Type Familiarization Design 18.2.3.1 Functional Design 18.2.3.2 Solution Mapping Implementation the Police Investigation Application Storyline Recording a Lead 18.3.2.1 Creating the Lead and Entering the Generic Lead Information 18.3.2.2 Entering the Person, Object Location, and Event Information	379 379 380 380 380 381 381 385 385 385 386 387 387 387 387 387
Siebe 18.1 18.2	E Tutori Introdu Siebel I 18.2.1 18.2.2 18.2.3 18.2.4 Using t 18.3.1 18.3.2	al Inction Implementation Approach Choosing the Siebel Solution Type Familiarization Design 18.2.3.1 Functional Design 18.2.3.2 Solution Mapping Implementation the Police Investigation Application Storyline Recording a Lead 18.3.2.1 Creating the Lead and Entering the Generic Lead Information 18.3.2.2 Entering the Person, Object Location, and Event Information 18.3.2.3 Recording the Operational Information	379 379 379 380 380 381 381 385 385 385 385 387 387 387 387 387 387 387 390 390
Siebe 18.1 18.2	el Tutori Introdu Siebel I 18.2.1 18.2.2 18.2.3 18.2.4 Using t 18.3.1 18.3.2	al Inction	379 379 380 380 380 381 381 385 385 385 386 387 387 387 387 387 390 390 391
Siebe 18.1 18.2	E Tutori Introdu Siebel I 18.2.1 18.2.2 18.2.3 18.2.4 Using t 18.3.1 18.3.2	al Inction	379 379 379 380 380 381 381 385 385 385 386 387 387 387 387 387 390 390 390
Siebe 18.1 18.2	E Tutori Introdu Siebel I 18.2.1 18.2.2 18.2.3 18.2.4 Using t 18.3.1 18.3.2	al Inction	379 379 380 380 380 381 385 385 385 385 386 387 387 387 387 390 390 390 391

18

	18.3.3	Recording an Incident	
		18.3.3.1 Creating an Incident	
		18.3.3.2 Working on an Incident	
		18.3.3.3 Offence Management	
		18.3.3.4 Coercive Measures	
		18.3.3.5 Incident Overview	
	18.3.4	Recording a Case	396
		18.3.4.1 Information Held as Part of the Case File	
		18.3.4.2 Assignment of Cases and Ownership	
		18.3.4.3 Case Serialization	398
		18.3.4.4 Evidence Management	399
		18.3.4.5 Involved Persons	
		18.3.4.6 Report Out	
	18.4 Conclu	ision	401
19	Concluding 1	Remarks	403
Арре	Appendix A: CORA Model Capabilities		
Арре	Appendix B: A Solution Characteristic 413		
Inde	Index		

Downloaded by [5.154.190.33] at

Foreword

There were many lessons I learned while doing my MBA in the beginning of the twenty-first century. Two of the topics we learned kept puzzling me for a long time:

- 1. The importance of operational excellence by measuring and then optimizing the individual elements of a process.
- The X and Y theory: Can you trust your employees and should you give unlimited freedom or are employees per definition not to trust and should you therefore install fixed procedures to support and control them.

Working in the services industry, it seemed that creating operational excellence was not really applicable since it all depends on the knowledge of the individual, of course supported by frame-works and best practices. Would it be useful and possible to break up the processes in a consulting engagement and optimize them accordingly?

In the later case you would also apply more control to the consultant workforce. Would that bring the best value to the customer, or should one apply the X theory and give full trust and freedom to the individual consultant in the hope they will act on the best behalf of the customer with the highest impact and productivity?

After several years managing P&Ls within our organization, I became convinced that probably a combination of both theories should apply. On the one hand, industrialize the way of working, meanwhile preserving the creativity of the individual. As once somebody told, each tailor uses the same procedures to create a suit; it is the fabric and the elegant touches that make him/her unique.

For me the big question is how can you facilitate the decision process of the individual, making sure he or she has all the information required to make the best decision for the company.

Current technology has finally come to maturity to deal with these questions. Case management allows you to gather information, industrialized or from creativity to guide knowledge workers in their decision-making. Not only in guiding but also allowing us to capture the Case in a structured way, making it possible to learn and adapt our decision making for future cases. In other words, to really make us a learning organization.

Case management is too often seen as a toolset to allow document handling in the public space. Case management can bring so much more.

The authors demonstrate in this book that if you want to be leading the digital space, if you want to provide the best customer experience, you need to start with facilitating the decision

making process. This is what case management is all about. It is not about registering the transaction—it is to allow a better interaction with the end user.

With some great examples, this book will give you a better perspective of the art of the possible. How you can achieve a superior customer experience. How you can allow your employees to leverage their creativity to the max, based on thorough information and analysis. Yes, that can go hand in hand!

In other words, this book gave me the answer to the questions that puzzled me since the beginning of 2000. Can we further industrialize and optimize the services business while giving employees the maximum freedom to use their intellect? Yes we can; we call it case management.

Frank Wammes

CTO Application Services Continental Europe Capgemini, the Netherlands

In 1959, Peter Drucker, an American management consultant whose work contributed to the foundations of the modern business, coined the term "knowledge worker" to refer to "employees such as data analysts, product developers, planners, programmers, and researchers who are engaged primarily in the acquisition, analysis, and manipulation of information as opposed to in the production of goods or services."

Fast-forward to today's modern IT landscapes, case management solutions aim to support the knowledge worker in his decision-making process and find the way into software vendor product portfolios. In fact, a business process management (BPM) product is not considered complete without some support for case management. BPMN 2.0 very well addresses the requirements for structured and human-centric business processes and has become quite successful for both process analysis and execution.

However, in situations with a large degree of process variance (a number of channels to consider, product groups, regions, etc.) and where new situations must be considered while the process is running, it has turned out to be quite hard or even impossible to implement in Business Process Modeling Notation (BPMN). The combinatorial explosion considering all variants of a process typically ends in bloated process diagrams with many decision gateways and/or event subprocesses with the process engine in full control of the execution. For an optimal support of the knowledge worker, the user should be in the driver seat of the business process though.

The upcoming Object Management Group (OMG) standard for case management model and notation (CMMN) addresses those requirements by providing the ability to "plan" work and in general follows an event and condition (considering the case data) paradigm rather than a strict sequence flow model.

This book is a "must read" for people who want to learn more about the area of case management and how case management functionality is supported in various Oracle products, including Oracle BPM Suite. It motivates with real-world examples and goes into details all the way to concrete implementation. The case management solution framework (CMSF) carved out in the book is technology agnostic and applicable in general. The case diagrams of the CMSF are modeled on CMMN, which makes this book a good complement to the CMMN specification and, thus, is among the first books using CMMN.

In our roles of Oracle BPM Suite product development and as contributors to the OMG BPMN 2.0, CMMN 1.0, and DMN 1.0 standards, we welcome this book as it nicely introduces

the field of case management and then shows how the Oracle BPM Suite with its comprehensive set of functionality from low-level service orchestration and mediation to business processes with BPMN 2.0 and case management components can be used to implement adaptive case management solutions.

Ralf Mueller

Architect Oracle BPM Suite

Vaishnavi Sashikanth

Vice President Oracle Product Development

Presently, as machines and software are providing all the necessary means to automate routine work to their maximum possible extent, automation efficiency is receding in value as a competitive differentiator. Instead, a company's ability to use its organizational knowledge and experience to solve customer problems effectively is becoming more and more a favored strategy for excellence and competitiveness. Knowledge workers, those who actually translate the organizational knowledge to problem-solving tricks and techniques, need highly flexible work tools that traditional BPM, with rigid process models, do not provide; event-based adaptive case management (ACM) approach fills this gap. Of course, the world is typically not black-and-white, and, as we also see in my own company's work, most solutions aimed at organizing business activities naturally demand that there be a combination of case management and traditional BPM to deliver the right balance of effectiveness and efficiency. This book calls this combination "Generalized BPM."

Oracle provides a wide range of software tools to create Generalized BPM solutions, the most notable being the Oracle Unified BPM Suite, which uses essential features of SOA Suite as its foundation. ACM, BPM, and SOA are a natural match. Together, they facilitate the next phase in the evolution of digitization and automation of the next-generation business process management and derive higher value from SOA services, including better reuse, composability, and manageability. With this combined approach, business process management will no longer be about hard coding functions in a rigid process model but will rather become an agile platform helping to cope with continuous business changes and operations improvements.

However, having only the tools does not help with the whole process of taking user requirements and building solutions. Adequate and practical guidance on solution design philosophy as well as solution architecture and the implementation of best practices are necessary for the realization of a quality solution. This book, the first of its kind, provides this guidance starting at highlevel considerations through appropriate technology choices from Oracle's big basket of products right up to solution design details, thus making the dream of creating business applications to manage structured and unstructured business activities an achievable one.

> Geoffroy de Lamalle CEO eProseed

In the early years of web services and service-oriented architecture (SOA), I had the pleasure to work in my role at Oracle with some key architects at leading system integration companies. Across companies, we started to exchange information and best practices to deliver successful SOA customer projects, for example, one of the first European Oracle Service Bus or one of the first Oracle BPEL projects. This group of experts became the "masons of SOA." Based on the mission to share knowledge, leading industry experts like Thomas Erl invited us to speak at conferences, for example, the SOA Symposium. We also contribute to *SOA Design Patterns* or *Next Generation SOA* and the *SOA Magazine* with the Industrial SOA series.

From the early days of web services, we were part of the SOA journey and became more mature by adopting the service bus technology, orchestration like BPEL, business activity monitoring, pattern matching, and API management—all the solutions that became Oracle SOA Suite. The next step for Oracle was to adopt BPMN as an executable engine in the middleware portfolio. In 2013, it was obvious to extend BPM Suite with adaptive case management (ACM) functionality. ACM benefits from the whole Oracle middleware technology stack, including SOA, traditional BPM, document management, business intelligence, and identity management solutions.

In my role in Oracle, where I am responsible for Fusion Middleware Partner Adoption in Europe, Middleast, and Africa, I have the privilege to talk to industry experts as well as to Oracle product management and product development teams. It is wonderful to see that many of them contributed to this book. Oracle is organized by key product areas like database, middleware, and applications. Since more than 10 years, I have focused on the middleware technology set and this book is an eye opener for me to leverage this technology set as well as a few other Oracle application products for different case management solutions. It defines the starting point of an ACM journey. The solutions framework helps customers to design and classify your project. Based on the solution characteristics and the functional design, the technology capabilities are determined. The CORA framework is an excellent tool to determine an ACM solution that fits the use case. Customers benefit from selecting the right case management solution based on their need from the broad Oracle portfolio.

Like SOA, ACM, as we can see, continues to evolve. With the rise of the Internet of Things (IoT), the amount of information explodes. Big data make this information available. Knowledge workers can leverage ACM systems to make better decision. My daily work and the huge success of the Oracle Middleware Partner Community is an excellent example for knowledge workers working on cases—I would like to congratulate all the writers of this book to have achieved another milestone in our case.

Jürgen Kress Fusion Middleware Partner Adoption EMEA

List of Acronyms

A2A	Application to application
ABC	Activity-based costing
ACM	Adaptive case management
ADF	Application development framework
AIIM	Association for Information and Image Management
AP	Accounts payable
API	Application programming interface
AR	Accounts receivables
B2B	Business to business
BAM	Business activity management
BI	Business intelligence
BO	Business object
BPEL	Business Process Execution Language
BPM	Business processes management
BPM Suite	Oracle Middleware Suite delivering BPM on top of the SOA Suite (contain-
	ing ACM functionality)
BPM/ACM	Combination of business processes management and adaptive case
	management
BPMN	Business Process Modelling Notation
BPMN 2.0	Business Process Modelling Notation version as defined by OMG
CAS	Complex adaptive systems
CDO	Case data object
CEP	Complex event processing
CES	Complex evolving system
СМ	Case management
CM L1, L2, L3	Case management design levels
СММ	Capability maturity model
CMMN	Case management model and notation
CMS	Content management system
CMSA	Case Management Society of America
CMSF	Case management solution framework
CORA	COmmon Reference Architecture
CQL	Continuous Query Language
CQS	Continuous query service
CRM	Customer relationship management

xxii List of Acronyms

CRP	Conference room pilot
CSR	Customer service representative
CSS	Cascading style sheets
CX	Customer experience
DCM	Dynamic case management
DILO	Day in the life of
DO	Data object
EAI	Enterprise application integration
EDA	Event-driven architecture
EDN	Event delivery network
ERP	Enterprise resource planning
Four-Cs	Context, capabilities, collaboration, and coordination
GIS	Geographic information system
GPS	Global positioning system
H2A	Human to application
H2H	Human to human
HCM	Human capital management
IDE	Integrated development environment
INSEAD	Business school
J2EE	Java 2 Enterprise Edition
JSF	Java server faces
KPI	Key performance indicators
KRI	Key risk indicators
LDAP	Lightweight Directory Access Protocol
LOB	Line of business
LOV	List of values
M&A	Mergers and acquisitions
MAF	Mobile application framework
MDS	Meta data store
MQ	Magic Quadrant
MS	Microsoft
OA&M	Operations, administration, and management
OASIS	Organization for the Advancement of Structured Information Standards
OEP	Oracle event processing
OLAP	Online Analytical Processing
OMG	Object management group
OOTB	Out of the box
OPA	Oracle policy automation
Open UI	(Siebel related) Open user interface
OPSS	Oracle platform security services
Oracle SOA Suite	Oracle Middleware Suite delivering service-oriented architecture capabilities
PCM	Production case management
PDF	Acrobat portable document format (PDF)
PoC	Proof of concept
POLE	Person, object location, event
PoV	Point of view
PWT	Package walkthrough

REST	Representational state transfer
RFID	Radio frequency identification
RTD	Real-time decisions
SAAS	Software as a Service
SCA	Service composite architecture
SCM	Supply chain management
SI	System integrator
SLA	Service level agreement
SOA	Service-oriented architecture
SOAP	Simple object access protocol
SQL	Sequential Query Language
SR	Sense and respond
SR-CAS	Sense and respond complex adaptive systems
STP	Straight through processing
T&Cs	Terms and conditions
TCO	Total cost of ownership
TEI	Total elemental interactions
UI	User interface
VIP	Very important person
WF	Workflow
XLS	Microsoft Excel
XML	EXtensible Markup Language

Downloaded by [5.154.190.33] at

Preface

"A crop farmer, motivated by the success of his friend's chicken farm, decided to 'grow' chickens himself. He tilled and fertilized his land nicely and 'planted' eggs in the soil. When no chickens emerged from the ground, utterly frustrated, he sent some soil sample to his regional agricultural office who promptly tested the soil and advised the farmer to increase the watering frequency...". I (Manas Deb) had once read a little narrative along these lines where the author was trying to illustrate (albeit using an absurd example) the difference between "doing the right thing" and "doing things right"—the difference between "effectiveness" (the former) and "efficiency" (the latter). Effectiveness is related to how useful some work is while efficiency is a measure of how well a "given task" is done. It is conceivable that a given task is not very useful for the ultimate goal; this was the case with the actions of this crop farmer or the regional agricultural office. At their simplest definition, effectiveness and efficiency may be seen as "perpendicular" concepts; that is, one can be achieved without worrying about the other and that one does not necessarily lead to the other. However, it should be immediately obvious to a sane mind that a simultaneous achievement of both effectiveness and efficiency would be highly desirable in most situations. The real question then is "How difficult is it to attain high effectiveness at high efficiency simultaneously?"-this is an "optimization" problem.

Tempted by the efficiency of machines that resulted from the various chapters of the Industrial Revolution and due to the fact that efficiency is relatively easy to measure and improve in most situations, we are often inclined to exclusively focus on it especially when trying to boost organizational performance. Is this enough? Say, we are trying to improve customer service. Rapid access to customer care representative (CSRs) or some helpful information is surely a good thing, but will they relieve the customer's agony when he or she is faced with a complex issue? Probably not. On the other hand, if the CSR is able to quickly navigate through the details of the customer complaint and is able to come up with creative solutions to resolve the issue, then that would certainly be a rewarding experience for the customer resulting in customer loyalty and competitive advantage for the company. Such actions (of the CSR) require the benefit of their experience and knowledge. Peter Drucker, one of the most notable management gurus of modern time, had been discussing the nature and importance of "knowledge work" since 1954. In his 1999 book, Management Challenges for the 21st Century, Drucker notes, "The most valuable assets of a 20thcentury company were its production equipment. The most valuable asset of a 21st-century institution, whether business or nonbusiness, will be its knowledge workers and their productivity." So, how can we help knowledge workers do their jobs better?

Knowledge work is typically directed toward the nonroutine, that is, where not all work activities are preordained and require the use of human experience either acquired through formal education or on-the-job learning. Often, collaboration of peers or consultation with subject matter experts features as one of the ways to resolve complex issues. A knowledge worker requires more flexibility in the way work activities are selected and sequenced. Timely and proactive supply of helpful information can guide and facilitate a knowledge worker's performance. A traditional business process management (BPM)-based application has a fully predefined process model that captures the details of all work activities, logic for their sequencing, information about process worker assignments to specific tasks, and the necessary business rules. Such applications handle routine work very well but fall short in supporting knowledge work. In fact, even small deviations from the routine path are treated as "exceptions," that is, singled out as special cases typically dealt with by human experts at relatively higher costs; this strategy may be workable when the number of exceptions are small, however, from the description of knowledge work that we just provided, knowledge work modeled via traditional BPM is guaranteed to generate excessive amount of exceptions. Clearly, we need another approach to handle nonroutine knowledge-based work. This is where the idea of case management comes in.

In case management, instead of a fully predefined process model, work is organized more loosely under the umbrella of a "case" (as opposed to a "process") using "events," "phases," "milestones," and "activity lists" where some of these may be "mandatory" and the rest are selected by the case workers (who are essentially playing the roles of knowledge workers). Details of case progression are recorded in a "case folder" and case workers refer to this folder to learn about the history of the case progression and to decide what actions to undertake next. In certain implementations of case management, case workers can also add new activities or invite additional participants for contribution in the middle of a case progression—this is often termed as "adaptive" case management (ACM). One the other hand, if the case worker's flexibility to choose activities to be executed in a given case is limited to preset lists of available activities, then this variant is termed as "production" case management (PCM). Of course, it is conceivable that some of the case activities can be well described by processes (as in traditional BPM); conversely, starting from a process, we may find that certain process activities are better handled as cases; these hybrid solutions, which are likely to be quite common, is what we have termed here as "Generalized BPM."

While the idea of case management has been around for a long time and case workers have utilized things like paper trails, phone calls, and e-mails to manage the case progression, in recent years, with the increasing maturity of event-based application architecture, content management, real-time decision management, and supporting technology platforms, it is now possible to create comprehensive case management applications. Since a large portion of business activities contain some level of unpredictability and knowledge work, especially in the high-value-add situations, this technical feasibility of creating case management solutions has now pushed the interest in case management to a significantly high level. As can be expected, where case management is applicable, it brings higher efficacy, but how do we maintain high efficiency as well? Also, how do we continue to benefit from our knowledge of traditional BPM even in the presence of unpredictable knowledge work? In other words, how do we conceptualize, architect, and build Generalized BPM solutions that achieve the right balance of flexibility and efficiency and evolution and transformation of human behavior and expectations? The book is dedicated to this exploration, both from theoretical and practical points of view, starting with key concepts and observed examples to solution architectures and specific implementation strategies. While we have chosen a set of



Figure P.1 The writing process of case management with Oracle Technologies.

Oracle technology products for the implementation examples, the rest of the book is technology platform–agnostic and, thus, should be helpful for anyone interested in going deeper into case management applications.

Looking back in retrospect, now that the book is finished, we realized that the writing process of our book was actually a case on its own. Figure P.1 gives a high-level overview of the activities, milestones, and events that formed the process of writing this book. The overall process is predictable: A set of high-level activities aimed at delivering a book. The process of writing this book started with creating a blueprint and finding authors. Once the team was composed, we all decided to go for it. This was the first milestone in the process. Then we started looking at where to publish our book. We were supported by the Oracle Publishers Program, which brought us in contact with different publishers. Different proposals were sent to the recommended publishers. Taylor & Francis Group were enthusiastic about the proposal and the contract was signed with them, marking the next milestone. A continuous loop of investigations, discussions with experts, writing, and reviewing took place followed by a one-and-a-half-year-long quest. Some unpredictable events occurred during this period that made us deviate from the desired path. *New findings* during the investigation and some of the *review feedback* from our esteemed reviewers made us change the content and the structure of the script. As this book was written next to our day-today busy jobs and in different time zones, our jobs required us to rearrange the writing priorities. At the same time we also had an arrangement with our publisher to deliver the book within a certain timeframe. The *investigate* and *write material* phases were the milestones accomplished next, in which the draft material was delivered. The last set of activities dealt with finishing the book, aligning the different book parts, getting all the pictures together, and reviewing the language. The final milestone was reached where the book was delivered to the publisher, and the case was closed.

In order to support this book a website has been created, see: http://oraclecasemanagementsolutions.com/. At this website code examples that are used in this book and errata on the book material are provided. From time-to-time insightful blogs on the topic of Case Management with Oracle technologies will also be posted at this site. If an error is found in the book or more information is needed, we request the reader to contact us via the website.

> Léon Smiers Manas Deb Prasen Palvanker Joop Koster

Acknowledgments

This book required a lot of insight and energy from many of our learned colleagues. Our sincere thanks to all those who were involved in reviewing, discussing, challenging, cowriting, and testing the book material even if we somehow missed to list them here explicitly. Since the book covers knowledge from three areas—architecture, industry, and Oracle technology—we needed support from a variety of people in different phases of the book writing process. We are especially indebted to them for giving us their tremendous support despite us being grumpy at times!

In the architecture area, we would like to thank Joost van der Vlies (think about meta-model!!) and Theo Elzinga for extensive (typically nightly) discussions on the case management solution framework. We thank Danilo Smiedel for collaboration on the generalized BPM maturity model and the ACM adoption case study. We gratefully acknowledge Tom Barrett for doing a very thorough review of the case management solution framework! We also enjoyed our discussions with Sybren de Hartog and Tim Arkin around the patterns. Last, but not the least, it was always a pleasure discussing with Roeland Loggen different aspects of case management.

In the example area in the book, we required specific industry knowledge in mortgage, public sector, police and insurance area. We thank Onur Balcak and Nicholas Kitson for numerous discussions around case management and the mortgage and insurance industry. Special thanks go to Hans Telgen for all the knowledge he brought in for the police and the public sector cases. Jeroen van Essen did a tremendous job contributing to the police case information! We thank Peter Kuijvenhoven and Maurice Vuijk for enriching our insurance knowledge.

In the Oracle technology area, we thank Ruben Spekle for collaborating on Oracle Service Cloud. We thank Hans Telgen, again, for his Siebel knowledge and for his help with the tutorial material. We also thank Martijn van der Kamp, Steven Boon, Jon Petter Hjulstad, and Alexander van der Woude for discussing and reviewing the material.

Thanks to Joost Smiers and Roeland Jansen for checking the book contract.

In the writing and finalization phase a special thanks goes out to Hiske van Haren for reviewing all the material and removing all the Denglish (Dutch-English) in the work.

We would like to thank the publishing team at Taylor & Francis Group, specifically John Wyzalek and Jessica Vakili, for their patience, cooperation, and help with the book delivery.

A last thanks is for all the coffee places in Cape Town, Shanghai, Rotterdam, Utrecht, Terschelling, Leeds, London, Sao Paulo, Singapore, Saigon, Hanoi, San Francisco, Santa Cruz, Sydney, Osaka, and Tokyo for providing the right amount of caffeine and atmosphere to write this book. Downloaded by [5.154.190.33] at

Authors



Léon Smiers works as a solution architect for Capgemini, The Netherlands, in the area of Oracle technology and architecture, where he is one of the leading Oracle specialists. He has done a lot of work and research in the field of integration and new technologies, like RFID, SOA, EDA, BPM, and case management, on which he wrote articles and presented in international conferences. Currently Léon is setting up Solution Architectures for large Oraclebased projects and is Oracle BPM Thought Leader for Capgemini. Léon is coinventor of the Common Reference Architecture mode or CORA model (www.coramodel.com), which helps in getting control over the IT landscape in hybrid environments and deliv-

ers a predictable, repeatable, and risk-aware solution design. Based upon his close cooperation with Oracle product management and field experiences, Léon was awarded the Oracle ACE title in 2010.



Dr. Manas Deb is president and COO of eProseed, a well-recognized Oracle technology partner. Previously he was a senior director in the Fusion Middleware/SOA and BPM Suite products group at Oracle HQ, where he had been leading outbound product management globally with a special focus on strategic customer engagements. Dr. Deb has worked in the software industry for more than 25 years, most of which was spent in business development, sales enablement, product development and management, and on architecting and leading many enterprise-level application development and business integration projects in a wide variety of industries. Dr. Deb has coauthored three other books on SOA and BPM and has been a speaker

in many industry conferences. Dr. Deb attended postgraduate studies at The University of Texas at Austin. He has a PhD in computer science and applied mathematics, and an MBA.



Joop Koster is an experienced Oracle solution architect, CRM, and front-office consultant. He is one of the founding fathers of the Capgemini t-Police solution (https://www.capgemini.com/public-sector/transform-police-t-police), which entails a standardized solution for the police force. He has 15 years of IT consulting experience and more than 10 years of (management) experience in the printing industry. He is familiar with front-office and contact center, multichannel customer interaction, and CRM technologies. He has a broad experience in marketing, sales and services processes and technologies, and has worked in both B2B and B2C environments.



Prasen Palvankar is a director of product management at Oracle and currently leads development of industry solutions using Oracle's various PaaS offerings. Prior to taking on this role in February 2015, he was the product manager for Oracle Adaptive Case Management, Business Activity Monitoring, and Oracle Business Rules products in the Oracle BPM Suite product family. Prasen has more than 20 years of work experience in all aspects of software application development such as consulting, solution and enterprise architecture, business integration and process management, and advanced training development and delivery.

THE CASE FOR CASE MANAGEMENT

Introduction

Case management is a particular style of business activity management that has attracted significant attention in recent years. In the traditional business process management (BPM) discipline, all work activities, their execution sequences, and the rules governing them are predesigned via a "process model." A "process worker" interacts with an executable form of the process model and is mainly concerned with handling a subset of these work activities, and the process model guarantees the desired outcome of the whole process when the relevant activities are completed. All situations, known or unknown, that are not covered by the process model are tagged as "exceptions" and are handled in a special manner, typically outside the main process and often by humans.

In case management, where a case is some problem to be solved or a transaction to be completed, the primary goal is to reach a resolution for the case. The actual (case) work activities, their execution sequences and the involvement of case participants (i.e., the "case workers"), etc., while kept within what is permissible and reasonable, are secondary to the ultimate goal attainment (i.e., resolution of the case). Thus, case management provides a more flexible work activity management framework than traditional BPM; in its "adaptive" variation, adaptive case management (ACM) even provides the ability to modify work lists and case participant lists. In many instances, for example, those involving unstructured and knowledge work, turns out to be a better paradigm to manage business activities.

In the first section of this book, we provide a general treatise on case management, using mostly nontechnical verbiage, to cultivate an appreciation for case management philosophy and discipline. Chapter 1, titled "Basics of Case Management," establishes the usefulness of the case management discipline through a series of day-to-day work examples and sketches out what a basic case management system would look like, and then provides a historical background, discusses formal definitions, and illustrates typical industry applications of case management. The chapter concludes with a short discourse on how case management helps with increasing business agility.

Chapter 2, titled "Adaptive Case Management (ACM)–Related Topics," explores several concepts that are interesting when considering the "adaptive" aspect of ACM. These concepts include sense-and-respond, anticipate-and-preempt, and the characteristics of complex adaptive systems (CAS), and indicate how case management style of problem solving in fact mimics the way humans solve problems on a day-to-day basis. Often a bit philosophical, these discussions are

2 Oracle Case Management Solutions

aimed at helping case management system solution architects better imagine the overall solution characteristics in a manner best suitable for target unstructured and knowledge work. A small discussion on the case management system classifications like dynamic case management (DCM) and production case management (PCM) along with ACM is presented here as well with the goal to help solution architects pick the most appropriate style of case management system for their problem at hand. Next, case management is compared and contrasted with essentials of BPM. A notion of "Generalized BPM" that combines capabilities of both is then introduced as an approach likely to find broader usefulness in capturing business activities that display both structured and unstructured aspects. A "capability maturity model" for Generalized BPM is also discussed. The chapter ends with a short discourse on considerations relevant to the creation of business cases for case management.

Chapter 1

Basics of Case Management

1.1 Introduction

Home proponit, sed Deus disponit (or in English: Man proposes, but God disposes) first appeared in a published work in the collection named *Die Imitaione Christi* (or *The Imitation of Christ*) by Thomas Kempis in the early 1400s. The *Imitation of Christ* is claimed to be the second most read Christian text after The Bible, and the quote itself is well known and well appreciated by all regardless of their religious beliefs. A secular interpretation of the quote would simply state that we (the humans) cannot foresee it all. Hence, the wiser among us must "plan for the unplanned"! The fact that uncertainty could hit our otherwise planned activities is not limited to our personal lives—the same is true in many business situations. Clearly, some suitable strategy has to be in place if organizations want to track and improve business activities that include unplanned actions.

Uncertainty is not the only variable introducing complexity in managing business activities. We can be certain that there has always been a standing struggle between two distinct work patterns: those that can be done by preset "routine" activities requiring relatively low skills typically perfected by mere practice and those that require "knowledge work" performed by experienced personnel requiring some level of freedom in a way how the activities are chosen and executed. Routine work can be made highly efficient and often automated, thus speeding up the work and driving down the cost. Routine work is also more easily managed, monitored, and improved than knowledge work since knowledge work involves a variety of "loosely connected" tasks undertaken in sequences that depend on the evolving context of the business problem. Thus, predetermined routine work patterns are simpler and preferred in many ways; however, they have limited capability of handling complex contextual variation and emergent events—knowledge work pattern is much better suited in these situations.

If we took a close look at the nature of work that businesses perform, we would find that a significant portion of the business activities are not of routine and deterministic type; instead, they are either done by knowledge workers or contain some nondeterministic characteristics. Work flow (WF) and business process management (BPM) have been traditionally used as the chosen disciplines to handle preset deterministic routine work. The necessary technology, solution, architecture, and practices supporting WF and BPM are now quite mature. Unfortunately, they do not provide a suitable approach to design, implement, and execute business applications that are knowledge work driven and need to handle uncertainties due to emergent events. This is the

4 Oracle Case Management Solutions

job of "case management" where work is done (i.e., the case is progressed) primarily to achieve a set of goals for the ultimate benefactors, for example, customers. A customer may contact a business with a request for resolution of a complaint that he or she may have about a product or a service, which may lead to starting of a "case" with the goal of delivering the requested resolution satisfactorily. The "case owners" and "case workers" will progress the case via completing certain milestones by arranging their activities best suited for achieving the goal of the case. We will be delving into more details of what a case is made of, how it is progressed, how work during the case progression is "adapted" when necessary, etc., later in this chapter and throughout the rest of the book as opportunities arise. First, let us look at a few real-life problems where it should be obvious that the solutions to the problems involve dealing with unpredictable or knowledge work.

1.2 Unexpected Surprises (or Shocks)

1.2.1 "Are You Being Served?"

Sometime in the 1970s a customer in Fairbanks, Alaska, walked into a Nordstrom store, returned a tire, and got some refund. Ordinarily this would sound like nothing special unless of course the store, in this case, Nordstrom, was an "apparel and shoes" store and did not sell tires! Founded in 1901, Nordstrom is now a large fashion retail chain and prides itself in providing extraordinary customer care and experience; "A relentless drive to exceed expectation" is a quote from the "About Us" page at their website (shop.nordstrom.com). While there are many variations that exist as to the exact details of the aforementioned tire incident (and the curious reader is welcome to do some fun research). The "tire refund" tale has become legendary in the realm of high-quality customer service. In fact, there is a well-established perception in the buyer community that Nordstrom staff is knowledgeable and is keen on providing a lasting positive buying experience to their customers whether it is to find the right product, arrange custom orders, or provide insight into latest fashion trends. Corporate resolve to serve customers excellently must be matched by the knowledge and freedom of action of the customer care people that are needed in order to handle tricky customer situations. Exceeding customer expectation is generally not trivial. However, a pleasantly surprised customer often becomes a loyal follower of the brand, providing the company a strong source of competitive advantage. In fact, a company's reputation as a bad customer service provider can cost them dearly in many ways and really fast with today's customers constantly posting reviews online, which prospects almost always look into before making buying decisions; bad reviews are posted and shared much more frequently compared to good reviews. Companies serious about healthy growth and margin therefore take the delivery of good customer service very seriously and sometimes employ interesting and creative means to recover from bad service delivery. Take, for example, the Canadian health care company "Nurse Next Door." As reported in an online article by Justin Martin accessed at cnn.com in January/2014 (http://money.cnn.com/galleries/2007/fsb/0709/gallery.where_customers_come_first.fsb/) when this company fails to deliver good service it delivers a "humble pie," which is a real fresh-baked apple pie, to the affected customer with an apology note. The company estimated that the value of customer retention via this innovative and proactive is the range of 60–70 times the cost of apology.

Are you being served? was a highly popular (and remains popular to date to select audiences) British sitcom (http://www.bbc.co.uk/comedy/areyoubeingserved/; accessed December/2013) that ran from 1972 to 1985. Set in the early-twentieth-century culture and etiquette and the day-to-day on goings at Grace Brothers, a fictional department store in London, the episodes of this sitcom humored viewers by parodying customer service and innuendo filled exchanges among the shop
floor workers across gender and class lines. Often, the customers in this sitcom were put in awkward and unpleasant situations by the doings of the shop floor employees and the customer would vow never to come back to the store. This was fine since it was a sitcom and the only goal there was is to provide entertainment to the viewing audience. Had this been a real store, it would have had to close its doors rather rapidly with such bad customer service.

In fact, in the real world the task of acquiring and retaining customers is a very big deal to companies. It is rather expensive to acquire new customers, and companies spend significant resources to marketing their products and services. By certain accounts (http://economicsofadvertising. com/?page_id=44; accessed January/2014) in the United States, advertising alone could range between 2% and 5% of GDP (depending on whether nontraditional advertising is included in the estimate) with top spenders such as financial and health care industries spending above 30% of their sales in ads (http://www.econlib.org/library/Enc/Advertising.html; accessed January/2014). Thus, once acquired, sensible companies that aspire to continue to sell to the existing customer base would put in the extra effort by providing quality customer service to retain the customers.

Business dictionary (http://www.businessdictionary.com/definition/customer-capital.html; accessed January/2014) defines "customer capital" as "the value of relationship that a firm holds with its customers, and which is reflected in their loyalty to the firm and/or its products." Customer capital is one of the three components of intellectual capital (the other two being human capital and structural capital) and is a measure of customer intimacy that a firm develops in its customer base and is a major contributor to firm's sustained competitive advantage (Jan Duffy, Measuring customer capital, Strategy and Leadership, 28(5), 10–15, 2000). If you have ever received a fair and smooth claims settlement from an insurance company, you are likely to stay with that company even if their premiums are higher; if you have had a pleasant experience at a clinic, you are far more likely to stick to them even if the charges there are higher; if you have felt that one of your investment bankers is going the extra mile to protect and grow your investment, you will probably channel all your life savings to him/her—examples of customer satisfaction as a result of superior customer service and the obvious benefits to the firms delivering those services are easy to find in all walks of life. Figure 1.1 provides a qualitative relationship between the ability to provide superior customer service, customer capital, and sustained competitiveness of firms stating essentially that the better the customer service the higher the competitive advantage generating customer loyalty. Superior customer service is particularly important and often a positive differentiator in mature markets. As we had remarked earlier in this chapter, in order to exceed the customer's expectation on the customer service front all those that are involved in providing the service must have the required knowledge and adequate freedom to exercise their judgment (based on their knowledge) to take a few nonstandard steps, if necessary.

1.2.2 "Doctor, I Have a Headache..."

A middle-aged man shows up at a hospital with severe headache due to high blood pressure. The check-in clerk does the necessary paperwork, and the patient sees the doctor in due course. In many instances, such a visit would end in the doctor conducting some basic investigations, asking a few pertinent questions, and prescribing a set of common medicines, and all will be well. However, in this particular instance, during the basic investigation phase, the assisting nurse reports to the doctor that while the patient's blood pressure is little high it seems unlikely that the headache might have been caused by it (typically, high blood pressure may cause some headaches in situations of hypertensive crisis, i.e., systolic pressure higher than 180 and/or diastolic pressure higher than 110 but not when the blood pressure is only mildly high; ref.heart.org).



Figure 1.1 Customer capital, customer service, and competitiveness of firms.

This information causes the doctor to look for other symptoms. She notices, among other things, that the patient is exhibiting occasional speech disability. Using her years of education and work experience, the doctor suspects a possible brain tumor and proceeds with the diagnosis activities and treatments accordingly. What could have been a common case of headache suddenly involves sophisticated tests like CT scans and MRI, and upon confirmation of the existence of a tumor requires complicated chemotherapy, radiation, or surgery as part of the treatment. From a general practitioner adequate in dealing with most common headache complaints, a headache due to brain tumor will now involve a group of collaborating radiation experts, specialized oncologists, and neurosurgeons that are now required to handle the treatment of this patient.

More than 10% of the overall human population complain about headaches and in the majority of the cases they are the "vanilla-type" that are easily diagnosed and are generally treatable by common medicines. However, on occasions, there are situations that are similar to the aforementioned one and require vast amounts of specialized knowledge and myriads of activities including cross-discipline collaboration that need to be adjusted based on the result of a previous diagnosis or the reaction of the patient to a particular treatment regimen. Being experts in their trade, these collaborators are also granted a lot of freedom as they pick the next best action during the diagnosis or treatment phases. Coordination of such a wide variety of activities is not easy, and given the large number of participants, many and possibly large financial transactions, and potential liability issues and activities must also be tracked for immediate and future reference. It is not hard to imagine that management of the complexity of such activities will grow rapidly as the amount of collaboration increase and as the level of freedom granted to the collaborators gets high unless, of course, a suitable framework is put in place to enable and track these activities; Figure 1.2 states this concern graphically.



Figure 1.2 Collaboration, work freedom, and management complexity.

1.2.3 "Look! There's a Dead Man in the Closet!"

"You may succeed in silencing me but that silence comes at a price. You have shown yourself to be as barbaric and ruthless as your most hostile critics have claimed." Reported in a New York Times article in 2006 (http://www.nytimes.com/2006/11/24/world/europe/25spycnd.html; accessed January/2014), these remarks aimed at the then Russian President Vladimir Putin were made by Alexander V. Litvinenko, a 43-year-old former Russian K.G.B. officer, in his deathbed in London. Litvinenko died of poisoning from Polonium 210, a rare and highly radioactive isotope—the poisoning is alleged to have been carried out by the Kremlin since Litvinenko was apparently considered an enemy of the Kremlin due to his anti-Russian government activities after he left the K.G.B. The British authorities called the circumstances unprecedented and launched an investigation. One can only imagine how complicated such investigations must have been for the police-so many facts to uncover, so many allegations to weed through, and so many geopolitical aspects to consider. Who can foresee or preplan all activities in such investigations, or for that matter predict as to how long it will last? In fact, given the involvement of multiple governments who are not the best of friends and possible unveiling of government secrets during investigative activities, making progress in the Litvinenko has proven to be difficult—to our knowledge, as of summer of 2013 a satisfactory resolution was still pending.

One does not have to look for a Litvinenko-type situation to appreciate how a police investigation can get easily complicated real fast. Finding a dead body unexpectedly in the clothing closet when a police visits a residence following a neighbor's complaint due to loud music, correlating signals from multiple channels to pursue a bomb threat investigation, or discovering an international drug cartel while chasing a few counterfeit notes are the type of things that are not at all uncommon in police work. In this type of work, often there is a great distance between where the investigation starts and where it ends with the middle riddled with steps continuously adapting to the situation at hand. In fact, police investigations involving substantial detective work would often be characterized by disjoint spells of high activity, many creative explorations by the detectives where keeping within legal bound the detectives are likely to undertake whatever investigation activity that would seem appropriate, some of these activities may be ad hoc and some would be well formulated like collecting evidence and running forensic tests, and there would be plenty on demand information exchanges among many connected organizations. Further, the investigation may be suspended one or more times if pertinent information flow dries up and will be reopened again on receipt of new information and the whole thing could take months or years to complete.

In Figure 1.3, for a fictitious yet realistic nontrivial criminal investigation by a police organization, we chart a time-lapse evolution of four related but distinct entities relevant to the investigation. Going from bottom to top in the chart, the first entity "number of events processed" represents sum of the number of interviews the police conducts, the tips they received, the testimonies that they collected, the forensic reports that are generated, etc., with each of these generically representing an "event," as a function of time after the police investigation begins. While it is typical that such events would be high as the investigation starts, their distribution over time, however, depends on how the case progresses and it is generally nondeterministic. The entity charted above number of events processed termed "impact of processed events" shows the impact or usefulness of the events and indicates that the impact or usefulness is not necessarily proportional to the amount of such events processed. For example, the police might receive a lot of tips at some point (i.e., a lot of events processed as per our terminology here) but these tips may not be very useful (i.e., may not have much



Figure 1.3 Time-scale variations of events and work in a criminal investigation.

impact on the case being pursued) while an isolated event, for example, receipt of some key information, may suddenly provide a deep insight into the case and thus trigger a lot of follow-up action from the police. The entity next up is "amount of investigative work" that quantifies the amount of activities the police is undertaking, which is a complex function of the events that they are processing and the impact of those events on the case. The topmost entity, "amount of cross-team collaboration," indicates the amount of activities where collaboration among many individuals, teams, and departments is needed for reasons like leveraging of knowledge or geographical proximity for on-site investigations. The main observations that we want to make here, with the current example, is that in situations like a police investigation there are many internal and external factors that introduce lack of predictability in the course of actions that are undertaken to solve the case. Thus, software solutions intended for helping with police investigations need a framework that has great access to all related current information, relevant knowledge from the past, and can process and adapt to a large variety and volume of intrinsic and extrinsic events; a framework that provides broad freedom yet enforces law and governs extremes; a framework that keeps strong audit trail.

1.2.4 Goal-Driven, Knowledge Work Freedom, On-Demand Collaboration

In the three distinctly different situations we just explored, and it is very easy to think of many such situations in all types industries, we notice a common set of characteristics in the purpose of the work and the way work gets done. First, we see that the work is motivated by some end goal that has the primary importance and the actual actions and the sequence in which they are executed are of lesser importance. High customer satisfaction is the real goal of superior customer service and not the specific actions by the customer service representatives; alleviation of ailment is the ultimate goal for health care givers and not the dispensing of diagnostic tests or medicines; a police investigation will have achieved its goal when the alleged crime situation is resolved and the perpetrators have been brought to justice, if indeed there was a crime committed. These types of work are identified as "goal-driven" work. Given that there could be many variations in work planning, scheduling, and execution, such goal-driven work generally has an "owner" (this is a role) assigned to it and it is the owner's responsibility to make sure that work activities contribute to the goal. This goal-driven notion of work is different than a series of activities done as part of traditional WFs or BPM where the "goal" for a given process worker is to properly (and repeatedly) execute one or more activities assigned to him/her, and it is the process model that ensures that the ultimate purpose is achieved.

Second, we note that the goal achievement relies on the workers being able to use their knowledge and experience to make decisions. In a complicated customer service request, the servicing representative may request from his/her supervisor special considerations in order to meet customer expectation; in treatment of complex diseases or injuries, the medical team members are expected to use their experience to adapt the course of treatment; police investigators will undertake the necessary investigation activities depending on emerging leads and facts. In other words, to benefit from the knowledge and experience of the workers they cannot always be restricted to follow preset directions in every step and must be allowed to decide on what next best action to be taken; judgment of these knowledge workers is also relied upon to create the next step when there are no preset directions available.

Finally, we see that the workers tend to or need to collaborate among themselves in order to reach the ultimate goal. While the workers may generally know what they may be collaborating on, the exact instances, amount, and details of the collaboration are usually dependent on the specific situation. Hence, we term this as "on demand" collaboration. The idea is to provide the participating knowledge workers the ability to bring their knowledge and experience to solve the problem at hand as a team, all contributing to reaching the target goal.

The real questions can now be asked: What type of solution framework would best facilitate goal-driven work by knowledge workers who may want to engage in just-in-time collaboration? Most of the work our businesses do today is made more efficient and effective by information technology. What flavor of information technology would enable this type of knowledge work? Historically WF, either embedded in packaged applications or implemented independently in middleware, and more recently BPM solutions have been utilized to organize and execute business activities. However, WF or BPM disciplines assume that we are able to "model" the work, i.e., prescribe how the work activities will be sequenced and how the "agents" will perform those activities before the work begins. As per WF or BPM models, simply by knowing which activities have been executed at a given time, we are able to state the exact status of the overall work and how close we are to completion. While WF and BPM strategy are obviously well suited for many classes of business work patterns, for the three types of situations we described earlier in this chapter, i.e., complex customer service, critical patient care, and police investigation, and many more similar to them, these (WF and BPM) approaches will clearly not be suitable. A different paradigm called "case management" where the "case" represents the overall problem to be solved, engaging case owner and case workers (aka the knowledge workers) endowed with the necessary flexibility to "work the case" so as to reach the goal of resolving the problem is much better suited for the problem classes we explored earlier in this chapter. Given that this book is about case management, we will get much deeper into the nitty-gritty of case management from many different points of view as we progress through the book. To begin with, in the next section, we will use a relatively simple business situation to illustrate how predefined business process (BP) models can quickly turn inadequate when a bit of real-life deviation gets added to mix of preplanned work; this example will also show some preliminary attributes and capabilities of case management solutions.

1.3 Handling of Credit Card Rewards Disputes

A customer of Diamond Credit Card Company has been a member of that company's credit card loyalty program for some time. When reviewing his last credit card statement, he suspected some error in the reward points—even though he had been charged more in the last billing cycle than in the previous ones he had received fewer reward points. So, he promptly called a customer service representative (CSR) at the Diamond Credit Card Company and filed a reward dispute claim. The CSR collected some basic information from the customer and told him that their reward dispute claims processors will look into the situation and get back to him as soon as possible. Using the information collected, the CSR opened a "ticket" for this case and the dispute resolution activities got kicked off; a case owner or manager got assigned to the ticket or the case who would be ultimately responsible for a proper and fair resolution of the case so that the customer is satisfied with the handling of the dispute by the credit card company.

The major activities that can be expected in such a dispute resolution at the Diamond Credit Card Company, as may be with most credit card companies, are collection of necessary information from the customer, a quick analysis of the situation to assess whether the reported claim has any merit, an investigation into what caused the alleged error and identification of possible remedies, and finally deciding on a resolution that could be handed back to the customer. The roles that would perform these activities could be: "CSR" mainly communicating with the customer, "case owner" making a preliminary check that enough information has been obtained (by the CSR) to start an assessment, "analyst" doing the initial assessment, "investigator" doing a deeper analysis and suggesting recommendations, and the case owner then taking the final decision regarding the resolution. Here we assumed these roles and responsibilities just to keep the situation simple—in reality, there may be multiple roles that could share a given responsibility, for example, under certain circumstances a CSR may be permitted to close such a case. Figure 1.4 shows a traditional BPM model sketched using BPMN (Business Process Modelling Notation, see: omg. org for details on BPMN) involving around 10 activities (rectangular boxes with rounded corners) and the flow controls or gates (crossed circles) and transition lines. The swim lanes are used to indicate primary assignment of activities to the roles participating in the dispute resolution. The highlighted trace over this process model diagram shows the "happy path" where information collected at the start was adequate for the analyst to validate the merit of the claim, the investigator did not need any more information to conduct the necessary deep dive into the situation, and the case owner ended the dispute resolution activities with concrete and unambiguous decision about the remedy that the credit card company would offer the customer.

Of course, a few small real-life variations can easily make the planned progression challenging. For example, let's say that when the case owner gets his assignment (of the dispute resolution) and does a quick review of the customer information collected and entered by the CSR, he (the case owner) notices that the customer has multiple credit cards from the Diamond Credit Card company and suspects a mixup of credit card numbers. He requests the CSR to confirm the details; the CSR contacts the customer back for some information updates and verification (and this could take more than one iteration). When a deeper investigation is undertaken, the investigator may feel the need to double check some of the transactions within the scope of the dispute. Thus, additional contacts are made with the customer and more documents are obtained or more information is collected. In the meantime, the anxious customer may call the CSR to get an update on the progress, and these could also happen more than once. The investigator's recommendation on the resolution may turn out to be not so pleasing to the customer and the case owner may override that with a more favorable conclusion since he notices that the customer has been with the credit card company for a long time and has always been in good standing.

Thus, a few little deviations from what is expected to happen in the "happy path" (per the process in Figure 1.4) introduces a good lot of criss-cross conversations and out-of-sequence activities among the involved parties, frequent updates to information attached to the work, and requires some allowance in decision making be granted to those trying to resolve the dispute. The progression of the dispute resolution in the presence of the deviations mentioned in the last paragraph is shown with the highlighted segments and dotted arrows in Figure 1.5. Note that even though parts of the process in Figure 1.4 are still used, these parts are now disjointed. Also, the dotted lines only indicate logical connections or activity initiations and are not intended to show a directed activity to activity transition (as in traditional process flow diagrams). What would be a good way to treat such interactions in a digital solution? How would we ensure that the participants have easy and timely access to data and documents they need in order to complete their work? Further, good visibility into the progress of the resolution, archiving of information, and activity audit trails for future reference is also necessary.

Could BPM models, nowadays most popularly created using the modeling language BPMN 2.0 (ref.omg.org), adequately handle the requirements for a suitable solution? BPMN 2.0 is quite rich in the number and variety of modeling elements and perhaps it would be possible to create a model to handle the scenario with the real-life variations described. Of course, the BP model







Figure 1.5 Credit card reward dispute resolution with real-life variations (#1).

would look quite different than the "happy path" BPMN process model depicted in Figure 1.4, and it will be a significantly more complicated model due to the presence of interaction loops with dynamic counters, participants reaching out to each other as and when necessary, new (or newer versions of existing) documents being uploaded as the resolution progresses, and the ability of the case owner to override the standard progression of activities and enforce a decision, if necessary. To make matters further challenging, we can anticipate that the next dispute could easily introduce or generate a different set of variations. For example, after the initial assessment by the analyst, the case owner may feel that he/she has seen several similar situations before and may conclude the dispute resolution process abruptly (with a decision, of course) bypassing the investigation phase and simply inform all the participants about the decision (Figure 1.6 provides a visual for this variation). This will require yet another version of a traditional BPM process. Continuing in this manner, we can conclude that in order to take care of all possible variations we will need an inordinate number of (complicated) business processes, thus making the prospect of solving such a business problem using traditional BPM technique quite daunting; and, this is clearly not the most difficult business problem that we can imagine. Using traditional BPM approach of creating a software solution to this problem would be highly tedious, expensive, and difficult to manage at best and thus would be generally impractical; any of the readers who may have tried to use traditional process models in these types of problem situations will readily agree with this statement.

Just to be clear, the aforementioned conundrum is not a shortcoming of BPMN or any similar process modeling language or a specific BPM practice—it is the set of solution requirements where the solution goals are of primary importance and the specific activities are, generally speaking, less important, where the progress of the solution is difficult to measure by simply observing the status of the activities, where content often plays a significant role and in those situations actions and content are closely related, where internal and external events can influence the actions and outcomes easily and significantly, and where knowledge workers are granted freedom to make decisions that are not already preordained, that make an approach based only on traditional BPM inadequate. Clearly, there has to be another approach that is capable of handling such challenges. The answer, of course, is "yes," and as we describe the art and science of designing and building solutions using case management (or in some cases a hybrid approach using traditional BPM and case management) in rest of the book, this affirmation will become clearer.

1.4 Case Management: Basics

1.4.1 What Is Case Management? What Is Adaptive Case Management?

From all the examples and introductory remarks so far, it should be apparent that even though BPM and Case Management (CM) are both ways to solve business problems, they approach the problem and the solution very differently. BPM relies on predefined process models where activities are connected together using a set of activity to activity transition rules. In BPM, a given business problem is captured in an "instance" of the process model that "executes." The (business) process workers are typically engaged in completing one or more activities that they might be assigned and are not generally responsible for the outcome of the total process; it is the process model that progresses the solution. All the necessary knowledge and experience needed to drive the solution is captured at designtime (i.e., when the process model is designed)—sometimes small variations from the most likely paths through the process are handled by an exception handling strategy. In case of CM, the problem to be solved is captured in the notion of "case" (we will describe the key



Figure 1.6 Credit card reward dispute resolution with real-life variations (#2).

16 Oracle Case Management Solutions

attributes of a typical case later) as opposed to a process, and a case manager or a case owner role is made responsible for ensuring a closure of the case that usually corresponds to achieving a set of end goals signifying solution to the original business problem. Besides the case owner there may be many others (aka "case workers") who may put in effort and contribute toward the closure of the case through their actions, i.e., by performing "tasks." In order to keep track of the progress of the case, a set of "milestones," spread over the collection of tasks, are used—milestones typically correspond to start or end of something significant to the case. A set of "phases" grouping closely related tasks and milestones are used to provide a logical flow of activities from start (or "opening") of the case to its closure. A logical "case file" or "case folder" concept is used to keep track of information and status related to the case.

Figure 1.7 displays the essential CM concepts described in the preceding paragraph. It is important to note that phases in a case are not necessarily connected to each other as activities in processes and that, when necessary and permitted, case workers can jump from tasks to tasks, forward and backward, crossing milestones and phases. Also, the notion of the task in the context of CM is quite generic—it can be a simple phone call, making a decision, following a process (which could be predetermined) or even physically creating a new task. The tasks may be done by humans or could be automated, i.e., executed by the software system once all the necessary preconditions have been met. The tasks could be either "mandatory" or "optional" (aka "discretionary"). The tasks could be "planned," i.e., suggested prior to the start of the case or could be added by the case owner or a case worker during the progression of the case (this is an important qualifier for "Adaptive Case Management" or "ACM"—we will delve deeper into this later). The concepts



Figure 1.7 Essential concepts related to case management.



Figure 1.8 Start and progression of a case in a case management system.

of "events" i.e., signature of something that happened and "event handling," i.e., sensing and responding to events, and "rules" that guide how to handle events are used to manage the loose coupling of all the case tasks. As an example, using the mechanism employed in Oracle's Case Management (a comprehensive set of functionalities packaged in Oracle Unified BPM Suite) in Figure 1.8 we present a simple chronology of how a case management (software) system could handle the start and progression of a case; this is only to demystify the basic working of such systems, and more general situations will be dealt with in later chapters.

As we just indicated that ACM furthers the flexibilities and the capabilities of simple case management solutions by letting case workers and owners modify the tasks to be executed as well as their execution sequence at runtime without the need of solution developers to change the design of the solution and then having to redeploy the solution (as would be the case for most other software application including traditional BPM). Thus, in principle, ACM includes CM and in this book we will use CM and ACM interchangeably and will point out if we specifically mean one or the other when the situation demands that.

Let's revisit the issue of knowledge workers needing to have adequate freedom to take actions that they deem most fitting at a given point in the progression of a case. Note that case management is after all a strategy to solve business problems. This means that knowledge worker freedom will still have some boundaries that are defined by prevailing organizational and industry business practices and government rules and regulations and any applicable laws. For example, in case of a police investigation involving a lot of detective work (i.e., a lot of knowledge work) will generally start with some guidance as to what the investigation needs to accomplish and how such investigations are usually approached by the concerned police organization. As the investigation is launched the detectives, within these high-level guidelines, will have reasonably high amount of freedom to undertake a variety of activities many of which will be determined depending on how the investigation evolves. From time to time, there will be activities such as gathering and recording evidences or testimonies or running forensic tests that would have to follow strict discipline for the results of such activities to be admissible in the court of law, eventually. Similarly, we can consider the situation of credit insurance processing for businesses where large (often in tens and hundreds of millions of dollars' worth) credit applications from businesses are vetted and decisions regarding credit eligibility, terms and conditions (T&Cs) and premiums are taken.



Figure 1.9 Knowledge work freedom and case activity levels.

The basic review of the credit application, the key phases and milestones that need to be covered before approving the requested credit, the formal contracting during the credit grant, etc., are all tightly regulated yet the risk analysts and the underwriters will typically be expected to and will have the necessary freedom to exercise their knowledge and experience in order to dig deeper into the supplied and related information and to hold negotiations with the customer. Generalizing these observations, we can state that in case management involving knowledge work restriction will commonly apply when case activities are "high level," i.e., those activities that help enforce overall guidance for handling the case or when the activities are "low level," i.e., where they are usually governed by the organization's standard operating practices, and that most of the freedom for the knowledge workers will be needed when the activities are between high-level and low-level ones (see Figure 1.9); such realizations are helpful for solution architects as they go about designing case management solutions.

1.4.2 Case Management Ecosystem

Figure 1.10 shows the core and supporting systems of a full-featured general purpose case management platform. As this figure indicates, the core system contains those features that are deemed essential to provide all the standard capabilities and expected flexibilities of a case management platform. Typically, case activities that can range from a simple task to firing of a business process are governed by a set of rules and policies associated with a particular "case type" or "model." These activities are triggered by internal or external events where the internal events could be generated either by the case engine or by a case worker. Thus, rules engine, and event handling needs to be an integral part of the core of case management platform. Concept of a "case folder" is commonly used by the case engine to keep track of various activities as the



Figure 1.10 Core and supporting components of a case management system.

case progresses. The case folder can either be a concrete one or could be a virtual one, and this should not be of material importance on the usability of the case engine. Cases also frequently deal with some case specific content which can be either structured (as in formatted documents) or unstructured (such as emails or notes from a phone call). The core functionality of a case management platform must be able to handle at least the case specific content. Since case workers may choose to alter the tasks to be done or the sequence in which they are done, some task management capability that includes task presentment and audit is necessary. In many complex case resolutions, team collaboration is highly beneficial and the case management platform has to facilitate that. Basic security features are also mandatory for case management platforms as with most other platforms targeted for business solutions. Certain amount of dashboard capability is essential for visibility into tasks and their status. These dashboards can have a "case worker view" that gives the details of the cases a particular case worker is involved in and a "manager" view that provides higher level summaries such as how many cases are in the system, how many have been completed, how many are in progression, how many are in the queue, how many are (potentially) in jeopardy, statistics on completion and wait times, and on case worker load distribution. Case management generally involves significant amount of human work and thus the case management platform must provide the ability to create and manage appropriate user interfaces (UIs) on desktops and on mobile devices.

A full-featured enterprise-grade general purpose case management platform needs more capabilities than what we just listed for the core system. It needs to be able to handle a large variety of inbound and outbound content that could be either structured or unstructured including (at least in some cases) the ability to automatically derive and use information embedded in such content, for example, deciphering identifying information from the scan of an invoice via image processing, or extracting routing information from a PDF form, or outputting formatted communications based on case resolution. Besides the set of rules that govern the basic progression of case activities, the system needs to facilitate access to various context and industry specific "expert systems" or at least to knowledge capturing rule bases should the knowledge workers need such support. Analytics and business intelligence (BI) are special types of knowledge that can be also helpful in certain case progression, for example, a case manager may want to quickly check the historical and current level of underwriting risks before deciding on a high-risk loan. Since case management often involves dealing with humans who, in an organization, have certain organizational privileges due to their assigned position and responsibilities and since such privileges need to be referred to when assigning or sharing work, case management system needs to have adequate access to organizational structure (often stored in "directories"). Information being handled within a case can be sensitive from privacy or regulation compliance points of view; hence, additional security considerations beyond the basics can be critical when putting together a case management solution. As the case progresses, information typically held under the supervision of an organizations enterprise applications such as ERP, financial, inventory, or even HR systems may need to be referenced or updated; in some situations information exchange could extend beyond the boundaries of the organization; also case management system may need to access content repositories external to the case management system. Thus, a case management system often requires the support of high quality application (EAI) and business to business integration (B2B) framework, and integration with content management systems (CMS). Finally, a portal framework may have to be included in the case management solution either for convenience of the case workers or to provide information about the cases to a broader set of stakeholders including periodic updates to and self-servicing of certain steps in the case by the ultimate customer. How much of these additional capabilities beyond the core (as we have described earlier) are integrated and packaged as part of the case management platform offering or what needs to integrated on a case-by-case basis varies from software vendor to vendor; what is optimal for the buyer of course depends on the actual scope of the case management solution.

1.4.3 Brief History of Case Management

While there is a lot of buzz these days about case management, and some project that the market for case management to be 5–10 times as large as that of BPM, we can say quite confidently that there is a high chance that you have heard of case management prior to this latest round of excitement. In fact, by some accounts the concepts of case management were already in action over a century ago in countries like England and the United States (see, for example, the 1989 book Case Management: Historical, Current and Future Perspectives edited by Mary Hubbard Linz, Patricia McAnally, and Colleen Wieck from Brookline Books, and the 2002 article "Towards a Typology of Case Management" by Philip Fleisher and Mark Henrickson accessible at ftp://ftp.hrsa.gov/ hab/Typology.pdf; accessed January/2014). Most of the early case management like activities were around church and social workers trying to help the poor in need. These caregivers had to be aware of what services were available via public and private sources and coordinate them for the benefit of the recipient. In the beginning, such work was organized in intuitive and commonsensical fashion; there was no real methodology, framework, or training that would help prepare these caregivers and consequently these caregiver actions often lacked efficiency and efficacy. In early twentieth century, Western governments started organizing social care via federal programs (e.g., the Social Securities Act of 1932 in the United States, and the like) and naturally there were efforts to evaluate how effective these programs were and how efficiently they could be run.

While government funding surely helped the social service cause, many organizations, public and private, emerged controlling or specializing in specific services and roles. Situations that needed more than one service became harder to deal with. The concept of "service integration" emerged whose goal was to treat the problems more comprehensively (like a "case"), increase the ability of the caregivers to address the problems wholesomely by giving them appropriate training (i.e., making them sort of knowledge workers), and organize the necessary service components in a manner that they would be easily accessible to the caregivers. In some ways, the discipline of case management was taking root through the need for meeting higher standard of efficiency and effectiveness in relatively complex service offering. Outside of social work, this case management discipline was first visible in nursing and other medical care operations. The case management approach had other benefits as well—it organized information in a manner that facilitated collaboration among multiple participants (i.e., managed shareable information via case folders and content repositories accessible by the participants) and it kept track of "who did what and when" (i.e., audit trails). These characteristics made case management concept welcome in conducting police investigations and in organizing legal work. Over time, case management techniques spread into wider varieties of problem solving work, typically done by knowledge workers, and often by several of them on a given case, sharing experience and best practices.

As noted by Keith Swenson in "Historical perspectives" in the book Mastering the Unpredictable (Meghan-Kiffer Press, 2010), one of the early applications of computers was in the area of office work automation and related information sharing. Use of emails instead of handwritten memos passed around from desk to desk for coordination of work is now taken for granted. However, organizing and tracking more complex work needed something more than emails. Thus, the concept of "workflow" emerged and became popular as a paradigm to specify work that human "agents" did toward achieving some target outcome. The so-called middleware-based WF software systems helped create and execute digital models. In the meantime, packaged enterprise applications (like ERP, HCM, SCM, etc.) that had emerged and had become popular as systems of record also offered specific WFs embedded in them. About a decade ago, BPM emerged as a more generalized version of this WF concept and combined human and machine work including handling of documents where necessary. WF or BPM approached the problem to be solved as the end result of a preset sequence of predefined activities and proved to be quite efficient in delivering expected results when the work indeed followed a planned path; using a mechanism called "exception handling" deviations (i.e., exceptions) from the planned path were dealt with. Such exceptions typically handled relatively small deviations efficiently and effectively, in some cases used automated alternatives but often diverting the work to capable human workers. Work like complex service deliveries, complicated investigative work, monitoring and reacting to emerging threats, and overseeing compliance activities needed human knowledge and experience to figure out the next set of actions as events emerged during the execution of such work. Often such actions or their execution sequences could not be planned conveniently beforehand, thus making the WF or BPM approach less useful as applicable paradigms to construct software systems to handle such work. Case management, by definition, is designed to handle these types of work. In recent years, advances in construction of software systems have made it possible to design and implement efficient and effective case management systems.

1.4.4 Formal Definition of (Adaptive) Case Management

The pursuit of case management today is quite general, spans applications in many industries, and is supported by all applicable technical advancements. Here we need to recognize two things: (1) as briefly discussed in the preceding section, case management originated and spread its roots in a variety of organizations (albeit the biggest emphasis had been in the social and healthcare arenas) and these organizations have "customized" the meaning of case management as best suited for their purposes and (2) the flexibility afforded by case management is in part due to fewer solution design constraints imposed on it (as compared to say, WFs or traditional BPM). These two aspects of case management make it difficult for all to agree on a common definition that is unambiguous and sufficient to preserve all the intended characteristics of case management. To appreciate what we are talking about here, let us look at a few definitions put forth by industry analysts and different types of organizations or industries.

The analysts firm Forrester, in their 2009 report on dynamic case management or DCM (and Forrester has clarified that by DCM they mean the same thing as what others mean by ACM), put forth this definition: A highly structured, but also collaborative, dynamic, and information-intensive process that is driven by outside events and requires incremental and progressive responses from the business domain handling the case. Examples of case folders include a patient record, a lawsuit, an insurance claim, or a contract, and the case folder would include all the documents, data, collaboration artifacts, policies, rules, analytics, and other information needed to process and manage the case.

A couple of years later, another prominent analyst firm, Gartner, offered this definition in their 2011 BPM Conference: *Case management is the optimization of long-lived collaborative processes that require secure coordination of knowledge, content, correspondence, and human resources and require adherence to corporate and regulatory policies and rules to achieve decisions about rights, entitlements, or settlements. The path of execution cannot be completely pre-defined; human judgment and external events and interactions will alter the flow.*

Association for Information and Image Management (AIIM, www.aiim.org/what-is-case-management) offers the following definition: A "case" is any project, transaction, service or response that is "opened" and "closed" over a period of time to achieve resolution of a problem, claim, request, proposal, development or other complex activity. It is likely to involve multiple persons inside and outside of the organization, with varying relationships to each other, as well as multiple documents and messages.

CMSA or the Case Management Society of America focused on healthcare sector defines case management as (http://www.cmsa.org/Home/CMSA/WhatisaCaseManaget/tabid/224/Default. Aspx; accessed January/2014): Case management is a collaborative process of assessment, planning, facilitation, care coordination, evaluation, and advocacy for options and services to meet an individual's and family's comprehensive health needs through communication and available resources to promote quality, cost-effective outcomes.

In a September 23, 2009, post at the site http://social-biz.org/2009/09/23/what-iscasemanagement/ (accessed January/2014), Keith Swenson, a contemporary and big proponent of case management, also collected case management definitions from a variety of sources that can be used do some contrast and compare exercise. Below are a few excerpts from his post on the definition of case management:

From a legal site: Techniques used to process cases from one stage of the proceeding to another, such as setting deadlines for discovery or scheduling a series of pretrial conferences. Case management calls for different approaches from one case to the next and is the primary responsibility of judges, assisted by lawyers and clerks' office personnel.

One from the healthcare sector: *Coordination of services to help meet a patient's healthcare needs, usually when the patient has a condition which requires multiple services from multiple providers.*

Another one from the healthcare sector: Offers a single point of entry to the aging services network. Managers assess clients' needs, create care plans, and coordinate and monitor services. They may operate privately or may be employed by social service agencies or public programs. Typically, case managers are nurses or social workers.

A constructive definition that utilizes the what and the how of a "broker type" case management model: The linking of a consumer to the service system and coordinating the various system components in order to achieve a successful outcome. The five case management activities are: (1) assessment, (2) planning, (3) linking, (4) monitoring, and (5) advocacy. Case management's primary goal is service provision for the consumer, not management of the system or its resources.

While there are many common elements in the aforementioned definitions of case management, there are also notable variations. Besides, the style and scope of the definitions are focused on the specific needs of the industry that produced it. In an attempt to both simplify the definition and to make it more generic, Keith, in the same post (as noted earlier), offered his own version: *Case management is the handling of non-routine work processes.* He explained his proposed definition further as: "One might argue that this is too broad and not precise enough, but it is hard to find examples of work that fit this definition and could not be considered case management. Patient care, in general, is not routine. Legal cases are in general not routine. Law enforcement investigations are not routine. The types of work that executives and knowledge workers perform are not routine either. It is a useful definition for that conversation in the elevator."

Indeed, the definitions are either too specific or too broad, and in some instances deficient in a set of high level and precise constructs that could be expanded on to make the definition actionable. Also, some definitions, at least in our opinion, have added restrictions that are not helpful. For example, we do not believe "highly structured" in Forrester's definition is a useful requirement for case management. Of course, in this book we are not aiming to settle the debate around the definition of case management nor do we expect that the definition we will adopt will be universally accepted. Here our goal is to work with a definition that will adequately cover a broad set of case management use cases that can be well tackled by the set of technologies and approaches that this book will be focusing on. Keeping these goals in mind, we adopt the January 2013 conceptual explanation of a "case" from the ongoing efforts of the standards organization OMG (the Object Management Group) to define a standard for "Case Management Model & Notation" aka CMMN (see http://www.omg.org/spec/CMMN/1.0; accessed January/2014) as the working definition for the purpose of this book. It stated:

 $A \ case$ is a proceeding that involves actions taken regarding a subject in a particular situation to achieve a desired outcome. Traditional examples come from the legal and medical worlds, where a legal Case involves the application of the law to a subject in a certain fact situation, and a medical Case involves the care of a patient in the context of a medical history and current medical problems. The subject of a Case may be a person, a legal action, a business transaction, or some other focal point around which actions are taken to achieve an objective. The situation commonly includes data that inform and drive the actions taken in a Case.

Accordingly, we can define case management as the overall handling of cases (per above definition). Drawing a parallel from the relationship between business processes and BPM, we can then characterize case management as *a discipline that guides the activities related to solution conception and design, deployment, management, and improvement of handling of cases.*

The definition uses "proceeding" (which taken in a general sense just means "a course of actions," see, for example, http://www.thefreedictionary.com/proceedings), which does not impose any restriction of a process (it does not exclude it either). Also, since a reference to "process" is eliminated we do not need to distinguish between the so-called "structured" and "unstructured" processes. As we will see later in the book, in order to offer adequate flexibility during case management solution design we need to free the designer of such restrictions as starting from a process framework. Of course, it is quite likely that some of the case activities or tasks may actually be fulfilled by the execution of processes, structured or not. In most proceedings, actions are documented as they happen—this is a desirable quality in case management solutions that need to produce audit trails for retrospection. The broadness of the ultimate beneficiary, i.e., the "subject" in the OMG definition is also quite appropriate since in application of case management the ultimate beneficiary, generally speaking, can be something or someone different than an individual customer in the traditional sense such as a patient or a plaintiff, and can be something broader or

more abstract such as a business (as in company litigations) or a whole country (as in a terror threat resolution case). This definition also identifies the importance of "data" as an important component of case management that coincides well with the solutions required in real-life situations. In our view, data in case management need to be interpreted more generally to include events, messages, content, and knowledge-bearing information.

While our adopted definition does not highlight it nor it is essential to include this in a basic definition, analysis of typical case management requirements shows that on demand collaboration among the "actors" in a case (i.e., the case workers or the knowledge workers engaged in the case) is of critical importance if we want to ensure higher levels of efficacy and efficiency in the solution. Another aspect our definition is silent about is whether the case workers have the freedom to choose the actions and the sequence in which these actions are executed during the proceedings of a case are to be necessarily defined at design time (i.e., predefined) or can also be created at runtime (i.e., adaptive). As we will see later, many case management solutions will require that certain activities be created by the case workers at run time, hence, they (the case workers) will need the freedom to either alter predefined sequence of activities or create their own actions in order to best serve the resolution of the case. As we have already mentioned, the abilities to handle runtime changes in case worker actions are some of the key aspects of what are known as "adaptive" case management (ACM) solutions. We will embark on a detailed discussion about "adaptivity" in the context of case management later in the book.

1.4.5 Application of Case Management Solutions

Work done by employees of an organization, whether it is designing and manufacturing products or providing services, is the essential mechanism that ultimately delivers value for its customers; it is the payments for this delivered value to the customer that translates to the revenue for the organization. For public sector organizations, the revenue is replaced by government allocated budget, which is apportioned from taxes and other income that the government generates typically from citizens and businesses. Since work costs money and resources, optimization of work has always been a key goal of organizational improvement efforts. Due to significant progress made to date in work design, it has been possible to make a large portion of work fairly "routine" (i.e., predictive and repetitive) that often requires low skill (or at least skills that can be learned via simple trainings and applied without much variation). Much of the routine work has also been automated and this automation is well supported by a wide variety of technology currently available to solution designers. With most of the routine work is automated, the remaining work required to fulfil an organization's promise to its customers has now become increasingly more "knowledge"-oriented, i.e., require expertise of the workers and in many cases information from external sources. Given that a lot of the customer care has been steadily moving to self-service models, when a customer does need to get hold of a (human) CSR the problem that needs to be solved is usually more complex than what could already have been handled by the self-service model (e.g., via a self-service web portal). Thus, a CSR, faced with an anxious and troubled customer, needs to quickly assess the problem, do some research into possible solutions, get help from others if necessary, override a few of the rules of organization's standard operating practices (within reason, of course), etc., to deliver a high-quality response back to the complaining customer. Moreover, for legitimate reasons, the complaint may not be resolvable in one interaction, hence, it is important that proper record of what went on in each of the interactions be kept and utilized in subsequent contacts to expedite the resolution process, as well as to maintain an audit trail. Such CSR activities

are clearly not routine. And, then there are activities that naturally tend to be nonroutine such as investigations looking to validate or uncover facts or masterminding a threat response. Solutions based on ACM, as we have already mentioned (and more details will follow), have the necessary capabilities to handle all these types of work that cannot be preplanned, nonroutine, or cannot be automated. Since the need for such unpredictable collaborative knowledge work is present across industries, ACM-based solutions are now spreading well beyond the social work, healthcare, and legal sectors.

From the point of view of value that is delivered, ACM solutions can be distinguished by the type of services that are delivered, the nature and expectation of the ultimate beneficiary (i.e., the "subject" in OMG's case management definition), and the type of end goals that are to be met. For example, in the social and healthcare realm there exist case management solutions where the case worker's responsibility is largely consultative, i.e., the case manager analyzes the assistance required, locates appropriate services, informs the care recipient of the options, and in some cases coordinates the recipient's efforts to avail the necessary services. This is called a "Broker Model" (see, for example, US National Institutes of Health or NIH manuscript "Effectiveness of different models of case management for substance-abusing populations" at http://www. ncbi.nlm.nih.gov/pmc/articles/PMC1986794/ as applied to helping substance-abusers; website accessed January/2014)—these are light-touch and relatively cheap service models. On the other hand, there are more comprehensive and more expensive case management models also directed to helping substance abusers such as "assertive community treatment and intensive case management," "strength-based case management" and "clinical case management," which are hightouch and the case workers not only recommend remedial services but also help with access and administration of the same. Clearly the applicability of any of these models would depend on the acuteness of the need of the substance abuser and availability of budget for such care operations. It should be apparent that the case management software systems targeted to serve the lightertouch models can be a lot simpler than the higher-touch models. Although we cited these case management models for a very specific type of care giving, one can imagine usefulness of similar models in other industries as well.

As we mentioned, case management solutions can also differ vastly in terms who they are serving (i.e., the end customer, the ultimate beneficiary, or the subject) and what type of end goal will signify a successful closure of the case. When a customer contacts a business with a complaint about a product or service, a case may be opened in order to pursue the activities that would collect, review, analyze, and remedy the complaint; here both the ultimate beneficiary, i.e., the customer who complained and the end goal, i.e., a remedy of the complaint is well defined and easily understood. However, for example, when a case is opened by a regulatory agency in order to either monitor or investigate some regulatory compliance of certain prevailing practices of a business, say, accuracy of financial record keeping or use of gifts during promotion of products. In this case, the subject of the case is compliance and not any particular individual and in some sense all stakeholders of that business may be impacted by the outcome. In certain situations, the goal of a particular case may be defined only very broadly. For example, if an incident is reported to the antiterrorist agency of a government and the agency suspects a possible terror threat, it may open a case to keep track of all the follow-up investigations. Such investigative actions may or may not be able to confirm the actual threat even if it were to exist; however, prompt investigative actions like these are in general beneficial to all concerned citizens; here we notice the difficulty in specifying some concrete end goals or in identifying a set of individuals who would be the ultimate beneficiary of the investigations every time a case is opened following an incident reporting. The type and amount of interactions between the case workers and the rest of stakeholders of the case, and

the kind of information that is expected to be exchanged between those who trigger the opening of the case and those who pursue the case are clearly quite different in these examples and these differences must be adequately considered when designing the software systems for handling each of these types of cases.

In a December 2009 report by Craig Le Clair and Connie Moore of the analyst firm Forrester titled "Dynamic Case Management—An Old Idea Catches New Fire," the authors classified case management solutions as "dynamic business applications" for a new generation of people-driven processes that would offer all the flexibilities that we have been discussing thus far. The report provided several examples of business problems where case management was either being used or could potentially be used. The examples had a broad range and included applications for highquality customer experience, multichannel customer support, overseeing audit processes, fraud investigations, dealing with request for sensitive information from government agencies, adverse incident reporting during medical trials, handling of potential terror threats, and the like. Several case management "case studies" contributed by customers, technology vendors, and system integrators have been compiled in the books Taming the Unpredictable and How Knowledge Workers Get Things Done published by Future Strategies, Inc. in collaboration with Workflow Management Coalition in 2011 and 2012, respectively, and both edited by Layna Fischer. These case studies briefly described the business problems they solved, technologies that were used, benefits that were obtained, and the challenges that were encountered and handled. While most of these case studies focused on the traditionally popular areas of application case management such as advanced customer care and patient care, claims processing for insurance and social benefits and handling of legal proceedings albeit powered by modern technology platforms there were also somewhat nontraditional use cases like handling of investments, collection of biometric information by police, airport operations tracking landing to takeoff of aircrafts, and driving brand and product promotions. Almost in all cases some level of runtime flexibility in selection of case worker tasks and execution sequence based on human judgment, on-demand collaboration among various stakeholders of the case, and appropriate handling of content and events were important. The primary benefits were increased effectiveness of the solutions driving customer satisfaction while maintaining good efficiency and providing comprehensive visibility into case progression.

In the 2009 Forrester report already cited, based on functional and operational goals targeted by specific implementations, the authors predicted emergence of three broad categories of case management solutions (see Figure 1.11): investigative, incident management, and service requests. Investigative use cases include activities to find and disseminate facts and could range from criminal and fraud investigations to handling requests (e.g., researching, selecting, complying with provisions of the law, etc.) for release of information from government agencies critical to public policy and economy. Management of incidents typically starts with one or a few triggering events (or the incidents) that may be received by agencies, organizations, or businesses, and often the responses to these incidents would involve timely notifications to all concerned, sometimes a very select set of people (e.g., covertly informing the local police of a possible bank heist, formally informing regulatory authorities of adverse drug reactions during a drug trial) or a broad population (e.g., all travelers in certain regions about a possible terrorist attack, broadcasting possible danger due to contaminated food supply or spread of an emerging epidemic). Communication and sharing of vital information among all stakeholders is critical to handling incident management type of cases. In case management problems involving complex customer service situations, whether the service is sought from a government agency or from a commercial business, exchanges between the customer and the case workers and the ability of the case workers to select and provide the components of the service necessary to satisfy the customer are critical to successful closing of the cases.



Figure 1.11 Forrester's classification of case management solutions (2009).

While these three categories of case management solutions display many distinct characteristics, we should not be surprised by overlaps of certain common functionalities that are at the core of each of these case management solutions such as work flexibility, on demand collaboration, content and event management, and overall visibility to case progression, as we have already pointed out earlier. Also, many case management solutions may embody characteristics from more than one of these three solution categories. In Table 1.1, we provide a list of case management applications for several industries and indicate how their primary characteristics are aligned with one or more of the three categories we have just discussed. This table is a shorter version of a bigger collection that we have analyzed; we present the longer version in the Appendix. Please note that the case management solutions listed for any given industry are only just examples and are not meant to be exhaustive, and through the information in Table 1.1 (or the table in the Appendix) we are also not saying that the industries not listed there do not have use cases where case management solutions are applicable; in fact, with the broad capabilities included in the case management paradigm, there are likely applications in practically all industry sectors.

1.5 Creating Case Management Solutions

1.5.1 High-Level Solution Architectures

In an exercise to typify ACM solutions, especially in the context of this book where one of the primary goals is to assist the readers gain insight into ACM solution design, besides understanding the business problems that ACM is capable of solving as described in the last section, we must also

Public Sector	Financial	Insurance	Healthcare	Energy and	
	Jervices	msurance	Treattricare	Ounty	
Benefits eligibility	Loan origination	P&C claims processing	Payer claim processing	Process safety management	
Grants management	Investor servicing	Underwriting	Policy and procedure mgmt	FERC eTariff	
Tax processing	New account opening	New account opening	Virtual patient record	Transmittals process	
Unemployment	Dispute resolution	Policy management	Legal contracts mgmt	Well log file management	
Welfare services	Wealth management		Member enrollment		

Table 1.1 Examples of Case Management Solutions in Various Industries

look into the main construction blocks of the ACM solutions. We present core and supporting technology components of case management solutions in Figure 1.10. As should be expected, not all components are important in every ACM solution. In order to focus on the design considerations and to identify relative importance of such technology components in a particular ACM solution, a pattern's approach that relates to high-level solution architecture can be helpful. The key question that follows is "what are the distinct solution architecture patterns that we should analyze and understand" so as to make knowledgeable choices during a solution design exercise. One option and the one that we have chosen to pursue here is to start with the three ACM solution types enumerated by Forrester. These types, as already described earlier in this chapter, are service request, incident management, and investigative. Of course, there are many pertinent questions that still need to be addressed before settling down on these three types of ACM solutions as targets for high-level solution architecture patterns. For example, how useful are these types in characterizing any given ACM solution? Are these types sufficiently distinct to warrant a solution architecture pattern? Does a given ACM solution fall purely under one of these solution types? To answer such questions, we analyzed over 50 ACM solutions across more than six industries and observed the following:

- 1. Most real-life ACM solutions would display characteristics that would be found in a combination of two or more of the three types (i.e., service request, incident management, and investigative) although in some instances one of these three types may assume a dominant role. For example, a new bank account opening case is predominantly a service request type solution while a loan origination case can include, besides the characteristics of service request—type solutions, some flavors, and amount of investigation-type activities.
- 2. Investigative type of case work is typically triggered by either a service request or an incident reporting. Examples include sorting through causes of an insurance claim as a result of filing of a claim (a service request–type ACM solution) and searching for the culprits after receiving a report of a murder (an incident management type ACM solution). Even though investigative case types may be triggered by other case types, we feel that it is distinct enough that it can be treated as a high-level solution architecture pattern just as service request or incident management types.

- 3. There are wider variations in the types of case work (i.e., activities) that could be associated with incident management and investigation while service request type of case work would typically have narrower variation. This is mainly because in service request type of solutions use cases where most of the actors (i.e., the participants) are usually well defined and the activities to be undertaken are generally known. On the other hand, a fraud investigation or handling of sudden epidemic can easily require many unexpected steps that only reveal themselves as the cases progress.
- 4. A significant portion of case work tends to require human touch and these could vary between a simple human task like a phone call to complex approval chains or fact finding missions. Of course, wherever possible routine work tends to get automated and executed by machines. Work that requires a lot of conversation, substantial exchange of information and ideas, subjective decision making or consultations with topical experts are best done by humans or under human supervision.
- 5. Use of rules plays a big part in construction of ACM solutions. Rules (along with case worker knowledge) typically determine triggering of case activities, either human tasks or automated machine tasks, as events emerge during case progression. For example, in a loan origination use case upon receipt of all the necessary application material a rule may cause notifications to be fired in order to inform the loan processor and the underwriter to take the next steps toward closing the loan application. In addition, many business logics such as risk indices associated with a credit application and calculation of insurance premiums can also be conveniently encoded in rules.
- 6. Many ACM solutions need to adhere to regulations or stipulations imposed by governmental authorities or by prevailing industry practices. This fact adds a stronger requirement of capturing detailed audit trails of actions and the corresponding actors associated with the case activities. Examples of such situations can be easily visualized in complex financial, legal, or medical cases.
- 7. Handling of documents and content is an integral part of most ACM solutions; however, large variations exist in terms of type of documents handled, i.e., whether they are structured such as a form or unstructured such as emails or scanned images, and in their role, i.e., either driving or supporting, in the progress of the case. Take, for example, a case that may be dealing with a contract dispute. As the case workers try to resolve the case, they may have to reference all the related contract and the terms and condition documents, and in this situation documents mainly support the case activities. On the other hand, imagine a case to handle possible terror threat–case-related information may be gleaned from all sorts of sources including postings on the Internet, news, and social media, etc., none of which are well structured and a lot of the case activities are likely to be driven by such information. There are also certain cases management problems where extensive pre- or post-processing of documents is needed either to trigger the case or to disseminate the case resolutions. Deciphering of information in an invoice scan, verifying a signature on a contract or creating notifications in multiple formats for multiple recipients are examples of pre- or post-processing of case-related documents.
- 8. Application integration shows up as a major component where information exchanges are needed between the case solution system and other systems of record such as ERP, HCM, or SCM. In some situations, information may have to be exchanged between companies or relatively independent business units and there B2B-type integrations between private (i.e., within the company or business unit) and public (i.e., shared among the companies or business units) processes are needed.

- 9. While case management systems are built around the concept of sensing and responding to emerging events, in some special cases, more specific treatment for handling events may be necessary. Commonly termed as complex event processing (CEP) where through a variety of techniques such as filtering, aggregation, and pattern matching important insights are derived out of a large number of events and the ultimate actions or responses reflect these insights or conclusions. For example, during a communication network or a power grid outage many monitoring devices could generate large number of events even if the outage is caused by failure of a single network element—CEP can be used to sort through the collection of all such events to identify the root cause and initiate or assist a case to help resolve the problem.
- 10. While standard access control and security requirements are relevant in most case management solutions, these requirements are quite similar to typical enterprise software solutions. In situations where the solution involves sensitive information, for example, in financial transactions or adverse incident reporting, access to information needs higher levels of control and more complicated policies.

In Figure 1.12 for the three types of ACM solution types, i.e., service request, incident management, and investigative, we present a summary highlighting the similarities and differences among them of the technology or enabling components that are likely to be needed to construct them. In Part 3, we discuss these solution types in more technical detail and address implications of some of the observations listed earlier.

1.5.2 Build or Buy?

Whenever there would be a need for a case management solution, it would be natural to ask the question: "Build or buy?" How should we answer this? Of course, there are organizations that simply prefer, for whatever reason, one option versus the other, i.e., either choose one of "build" or "buy" as part of their standard operating procedure, no matter what the consequences are. However, if instead we sought to find some rationale for our choice, then how should we go about it? First of all let us consider the three basic approaches that we can take to realize a case management solution: *bespoke* (i.e., custom developed or tailor-made software), *packaged application*, or a *middleware-based application*. Bespoke approach by definition has the advantage of being able to deliver a solution that exactly fits the requirements and this is a positive. However, often bespoke approach leads to a large collection of handwritten code typically utilizing low-level programming languages and scripts that are difficult to understand, maintain, debug, and enhance, thus incurring high cost and long project times. If the business problem is truly unique and the solution is really critical to the business, a bespoke approach may be justified, at least to start with.

Use of packaged applications sit on the other end of the spectrum from bespoke approach. The vendors of such applications package functionalities that serve most common use cases, provide some ability for customization, and generally carry the responsibility for bug fixing and regular upgrades to the software. As will be discussed later in this book, one packaged application from Oracle called Siebel provides a set of out-of-the-box (OOTB) case management functionalities for some of the public sector use cases. If the solution requirements are such that OOTB functionalities of a packaged application, perhaps with light customization (say, changes in configuration), are good enough to solve the problem at an acceptable level of satisfaction, and if the business has already invested in and is familiar with the packaged application (for example, a company already

ACM Solution Type												
	Automated Tasks	Human Interactions	Integration (EAI, B2B)	Business Rules/expert system help	Regulations/ statutes	Structured Content	Unstructured Content	Special Event Handling/CEP	Special Security Considerations	High Customer Interactions	External Communication	Expert Help
Service Request	L											
	M					1						
	н		li i		1				li il i			
									1			
Investigation	L											
	M					.[J.			
	н											
Incident	L		2			1						
	Μ		ii ii			Ĩ.						
	н								1	[
	Mostly supporting	All	SR	SR and Investigation	All	SR and Investigation	A11	Specific for Incidents		SR	Incidents	Investigation

Figure 1.12 High-level solution patterns—dominant components.

using Siebel for CRM) or the initial license and implementation costs of the target packaged application are acceptable, then this (i.e., the packaged application approach) is quite viable. It should be noted that if heavy and invasive customization are needed in order to make a package application fit the requirements adequately, then one can expect very high maintenance and upgrade costs for that application down the line—in fact, these difficulties can sometimes surpass those of bespoke software, hence in such situations use of packaged applications should be avoided.

The third approach uses a suitable middleware, for example, Oracle BPM Suite Case Management from Oracle Fusion Middleware family of products, as the basis for creating case management solutions. This approach mostly involves visual programming, can connect to packaged applications if needed, and can be also made as exactly fitting to requirements as bespoke software. In fact, most of the leading case management vendors offering that are targeted as general case management platforms fall under this category. Middleware-based case management platforms essentially combine the core case management functionalities (see Figure 1.10) such as case rules and policies, basic event handling and security, management of case activities (i.e., tasks and processes) and case content, UIs for case worker collaboration and task execution over multiple channels (including mobile where appropriate), and management dashboards. In addition, these platforms provide ways to connect to other supporting components such as integration backbone, analytics, and organizational directories. Due to componentized architecture of these platforms and the visual programming approach, change management solution creation is usually cheaper and can be done faster. It should be noted that handling of middleware platforms does require some specific training and expertise and in absence of adequate middleware competency any middleware-based solution, case management or otherwise, faces possible unsatisfactory delivery of functionalities and benefits.

Figure 1.13 provides a visual of how these three approaches would generally fare when we consider how well they deliver on functional requirements, time taken for initial deployment, and cost



Figure 1.13 Comparison of bespoke, middleware-based and packaged application choices for ACM.

of ongoing change management (either for bug fixes or enhancements). This depiction assumes that we are applying each of the approach using its best practices and not stretching reasonable limits of its as designed capabilities such as undertaking massive invasive customization for a packaged application in order to meet solution requirements adequately or resorting to large and complex code plug-ins in a middleware-based solution to achieve fulfillment of not so important requirements. In some situations, a hybrid approach such as using a combination of packaged and middleware-based application may also be useful. As an example, we can envision extending the base case management capabilities of Oracle Siebel with Oracle BPM Suite Case Management in order to accommodate a set of special processing or tasks or to enhance the overall adaptiveness of the solution.

1.5.3 Building a Fit-for-Purpose Case Management Solution

So, how does one go about building an ACM solution? Where does one begin? Is there something characteristically different about creating an ACM solution as compared to other traditional software solutions, e.g., a standard BPM-based application? In this section, our goal is to discuss the essential methodology concepts related to creation of an ACM solution, in particular using a middleware product (e.g., Oracle BPM Suite or similar). We will defer the finer and more technical details till later in the book after prebuilt case management solutions in Siebel and ACM-related features of BPM Suite have been presented and discussed. A pertinent question at this point of course is what type of information we need to collect and sort through as we embark on designing an ACM solution—the mind map in Figure 1.14 displays a set of information dimensions that can be conceptually handy in answering this question. Also, such mind maps can provide a flexible format to record notes and highlights as we collect input from various parties during the solution design process. Before proceeding further, we would like to note that unlike traditional BPM projects, methodology for ACM projects are still being developed by concerned vendors and SIs, hence, a universally agreed approach is not yet available; what we are discussing in this book are based on our experiences working with customers.

As with most software application development project, an ACM project will start with a "project discovery and requirements gathering" phase. Of course, when using iterative or agile



Figure 1.14 Key information required to construct a case model (in Oracle BPM Suite).

methods of project delivery additional requirements are gathered, whether to capture newer capabilities or to modify existing ones, in each cycle of the iteration. These phases are vital since majority of large defects or shortcomings of software applications can generally be traced to not having done a good job in these phases. In fact, much of the differences in handling of an ACM project compared to a traditional BPM project will show up in these phases. Let us recall that ACM solutions are aimed at providing much greater flexibility to the stakeholders of the case by conceptualizing the solution as a composite of events and the response to those events via rules and case activities. This means that as we go through the discovery process we need to have the discussions free from constraints that, for example, a traditional BPM approach may impose. We will also need to use more free formatted documentation styles to capture the information that we would collect during the discovery phase; in our experience, simple mind maps and suitably organized spreadsheets serve well for this purpose. The main idea is to capture enough information to be able to construct a "case model" that would be used to instantiate individual cases and will guide the progression of those case instances. In this context, we evaluated the proposed CMMN standard (ref. omg.org) and concluded that it was too technical for many typical case management solution stakeholders and thus may pose unnecessary burden during the discovery phase.

With a goal to decompose the business problem so as to identify the necessary solution components, we start with very loosely coupled events that describe the high-level interaction among the case stakeholders and the information that they interact with which may involve external sources like the customer or internal supporting systems (e.g., packaged applications like ERP, HCM, or CRM). Then we delve into the case activity details and we start to recognize pieces of the solution that display familiar solution component patterns, e.g., human tasks like uploading or reviewing a document or an approval, BPM processes like making payments or settling credits, etc. As shown in Figure 1.15, we have arranged the progression of the project discovery and requirements gathering activities in three levels, "CM L1" through "CM L3", and have listed the type of information that should be collected at each level. These have been designed in a manner that by CM L2 it should be possible to grasp how the target case model would look like, and by CM L3 there would be enough information to complete the construction of the case model along



Figure 1.15 Case management solution creation and discovery phase and CM levels.

with some level completion of the associated case activities and UIs. Next we present a high-level description of information that needs to be collected in each of these levels and the reasons for doing that.

1. *Case Management Level 1 (CM L1)*: In CM L1, our goal is to collect enough high-level information such that completion of CM L1 will provide adequate contextual information about the target ACM solution that would be normally sufficient for the solution owner to feel satisfied about the expected outcome and for the solution designers and developers to grasp the "big picture." As we have stated before, case management paradigm is result or goal oriented and the ultimate result or the goal that would characterize the resolution of the case is a key input under the entity "key business and solution goals." In addition, key business and solution goals should also include the overall capabilities of the solution as well as any major performance considerations including any service-level agreement (SLA) information. The "use case" along with "special situations" is meant to capture the essential business problem, the expected solutions strategy, and the possible known exceptions.

The "stakeholders" (aka case participants) are the main "actors" (in use case terminology) these are roles that are interested in the resolution of the case and some of these roles are also the ones that perform the case activities. As an example, for a loan origination case it is quite common that there will be a loan officer (often this is the person that provides the quotes and T&Cs), a loan processor (who does most of the document checks and some of the calculations), an underwriter (who accepts the risk associated with the loan, i.e., has the final say on the approval of the loan) and a closing agent (someone who does the title checks, gets the buyer and the seller to sign-off on their paper work, and collects or disburses funds from or to all parties). Case stakeholders interact with the case through human activities (e.g., review, approve, etc.). A variety of computer applications may also execute case activities such activities are usually termed automatic activities (as they are automatically fired when all the preconditions for the activities are met). "KPIs or KRIs," i.e., the key performance indicators or key risk indicators, would be the measures that the "solution owner" and any concerned management and stakeholders would inspect in order to gain knowledge about how the solution is performing against expectations.

2. Case Management Level 2 (CM L2): Having captured the overall solution context in CM L1, next we proceed to gather information required to create the structure of the case that we have been referring to as the case model (aka case type or case template). A solution designer, through the process of discovery and requirements gathering, will collect the necessary information about the attributes described under CM L2 column and, using the facilities of the case management tool that he/she is using, will create the case model. While ACM provides great flexibilities in terms of what activity to undertake when and by whom, often leveraging the knowledge and experience of the participants, some sort of a structure like a case model is still needed to kickoff a case instance. The level of detail for this case model could of course vary from solution to solution depending on the nature of the problem being solved and the amount of *a priori* knowledge about the paths the solution is likely to take. At one end of the spectrum of required case model detail, when the *a priori* knowledge about the solution cannot be established, the case model may have just a name and one stakeholder, and everything else could be added as the case progresses. On the other end of this spectrum where everything about the solution is known *a priori*, the case model will have many attributes and will be endowed with a lot of detail; in fact, in this situation an ACM solution and a traditional BPM solution will become functionally equivalent

(although the solution complexities of the two approaches may not be the same). As always in real life typical ACM solutions will fall somewhere in between; as per our experience, the list of attributes given in column "CM L2" in Figure 1.16 would be a set of information that would be required to create most case models.

A typical case progresses by going through a set of phases and by completing a set of milestones within those phases (see Figure 1.7 for a schematic). A phase is a coarse-grain group of activities and all the phases taken together should provide a fairly good idea of what all things, at least at a high level, that need to get done in completing the case. For example, if a loan origination for home mortgage is modeled as a case, it may have phases like document collection for eligibility, quotation of interest rate and other T&Cs, underwriting of the loan, and the closing. Each of such phases may contain, besides a start and an end (of the phase), intermediate milestones to signify important activity completion points in the case progression. For example, the document collection phase may have intermediate milestones like completion of loan application, collection of income documents, etc., as intermediate milestones. We want to point out that phases and milestones do not have to be necessarily traversed in a fixed sequence; after completion of a milestone or a phase, if the need arises and based on either machine analysis or human judgment, case workers should be able to revisit some of the activities summing up to that milestone or contained in that or another phase unless prohibited by some company rules or government regulations. This ability to move back and forth between activities depending on case status is an important difference



Figure 1.16 Credit card rewards dispute resolution and discovery mind map.

between the natural flow of activities in ACM solutions compared to that in traditional BPM where activity flows are fixed at design time.

"Events" in a case signify occurrence (or nonoccurrence) of something of interest, e.g., uploading of required documents needed for the case or reporting of an accident by a rental car customer or an excessive delay in making a particular decision. Using case "activities" that are typically triggered by a set of "rules" upon receipt of pertinent events, a case model provides the necessary response to such events. Case activities are either completed by case workers (i.e., human tasks) or by machines (i.e., automated tasks); case activities may be mandatory or optional and rules can be specified to indicate this fact.

Case "data" normally refers to the basic information that triggers the start of case and then supports the case progression, possibly getting modified or enriched through the life of the case. Case data is mostly structured information. In case of loan origination, the loan application could be a key part of the case data. Case "documents," on the other hand, can be an arbitrary mix of structured and unstructured data; involvement of such documents in a case can range from a purely support role to a driving force. In case of loan origination, images of recent tax returns may be one of supporting documents while the scan of the contract signature page could trigger, i.e., drive, the next suitable case activity. Note that besides the events and the activities that the solution designer may include in the case model, during the execution of a case, case workers may add events and activities at run time and may alter the sequence in which these activities are completed (as pointed out before these runtime modification capabilities make a case management solution adaptive).

Generally speaking, all ACM solutions should be able to provide, for intermediate and postmortem reviews, an audit trail of all the events that occurred and actions taken (as a result of those event occurrences) as the case progresses. However, certain types of cases, for example, those dealing with law enforcement or critical patient care or adverse incident management may have special regulatory audit or compliance requirements—such concerns should be recorded under the "audit requirements."

3. Case Management Level 3 (CM L3): CM L3, as shown Figure 1.15, continues with additional discovery and requirements gathering process mainly to make the case activities and UIs more concrete. Requirements like security and permission that tend to be more technical in nature can also be handled at this level. At CM L3, the case activities should be sufficiently detailed for us to recognize if they are simple or complex tasks, if they are business processes, or perhaps they are something so unique that special supporting applications may have to be custom developed. Once we have identified the exact nature of a case activity, we can then use any suitable existing methodology to further develop the design and eventually implement that activity. For example, if a case activity turns out be, say, standard credit check, we can use an appropriate business process to model and implement this activity. Similar comments can be made for simple human tasks or complex human approval chains. CM L3 can be seen as a level where hand-offs from ACM to a variety of standard modeling techniques can take place. As an example, in situations where a case activity has been identified to be a business process, as we have indicated in Figure 1.15, it would correspond, depending on the granularity of the activity, to some generally accepted process decomposition level (BPM L1, L2, L3, or L4). And, once we have recognized the case activity to be a business process at some level, we can then simply follow an appropriate process modeling technique (see, for example, http://www.oracle. com/technetwork/topics/entarch/oracle-pg-bpm-bus-proc-eng-r3-0-292099.pdf; accessed January/2014) to realize that activity.

38 Oracle Case Management Solutions

Recall the credit card rewards dispute resolution case that we had mentioned in Section 1.3 when discussing the consequence of the occurrence of even a small amount of real-life variation in an otherwise simple business process. From that discussion, a general conclusion is that such situations are better modeled with the ACM approach. As a constructive example of some of the preliminary outputs of ACM discovery and requirements gathering process, in Figures 1.16 through 1.20 we present the mind map capturing the essential outcomes from the discovery interviews and the spreadsheet listing the key case activities highlighting whether an activity is automatic (i.e., executed by machines) or knowledge driven (i.e., handled by humans). Of course, in order to create an ACM solution one will have to get to much finer nitty-gritty of all the attributes in various CM levels. We defer such finer details later in the book, keeping this early chapter focused more on familiarization of the key concepts.









Figure 1.17 Credit card rewards dispute resolution and discovery mind map (detail #1).



Figure 1.18 Credit card rewards dispute resolution and discovery mind map (detail #2).



Figure 1.19 Credit card rewards dispute resolution and discovery mind map (detail #3).

1.6 ACM and Business Agility

In today's highly competitive environment, "agility" is a very desirable quality of all successful businesses, and consequently, the phrase "business agility" receives a lot of coverage in many ranks of organizations; there are often many initiatives to improve agility of businesses, and management consultants and industry analysts enthusiastically play their part in pushing the importance and urgency of businesses being agile offering many strategies to achieve such agility. Business agility can mean different things to different people: at the highest level, say, at an LOB or higher level, speed at which new or variants of existing products or services are delivered or customer disputes resolved can be critical; operational divisions would be more interested in speed of execution in particular when work activities are added or modified; faster speed of application implementation,
Phase	Milestone	Pre-condition	Activities	Predefined Execution	Knowledge Worker Driven
Assessment phase	Initial assessment started	Case owner claims the case for processing	Update complaint details		X
			Conduct assessment/review		Х
			Send email		Х
			Create and assign ad-hoc task		Х
			Prepare case	Х	
	Initial assessment completed	Conduct assessment completed			
Investigation phase	Investigation started	Initial assessment completed and complaint deemed to be a valid complaint	Ad-hoc task for expert analysis		х
			Investigate complaint		Х
			Notify customer that the complaint is being Investigated	х	
			Request more information from customer		Х
			Review received documents	Χ.	
			Send Email		
	Investigation completed	Investigation activity completed	Catalog uploaded documents	Х	
Resolution phase	Complaint resolution started	Investigation completed or initial assessment found that the complaint had no merit	Notify customer		х
			Initiate process to credit missing rewards (If type if of complaint is rewards related)		х
			Initiate process to revert finance charges (If type of complaining is finance charge related))	х
			Initiate process to send a gift card to address a loyal customer's dissatisfaction due to this complaint		х
			Initiate fraud investigation		Х
			Blacklist a merchant (if type of complaint is merchant experience related)		х
			Close the case		Х
	Complaint resolved		Notify customer regarding the final resolution	X	

Figure 1.20 Credit card rewards dispute resolution and case activities and their classification.

improvement, and maintenance would be of primary importance of IT groups. Business capabilities that would provide such agilities would be enabled by better business and IT solution architectures, ability to quickly understand signals of change and make fast decisions with respect to appropriate responses, end to end visibility and control of business activities, easier access to right information at the right time, higher productivity, and so on. Given that ACM advocates certain departures from the traditional management of business activities, (i.e., via predefined business processes or BPM paradigm) it may be worthwhile to understand how the adoption of ACM influences business agility. However, before we can address ACM's role in business agility, it would be useful to embark on a short discourse on key characteristics of agile businesses.

As per its basic definition, agility of an organism or an organization or a system indicates its ability to react or move fast and with ease. Often the ability to be "adaptive," i.e., the ability to incorporate change efficiently and effectively and on demand, is also seen as an essential characteristic of being agile. An organization may need to react quickly to a wide variety of emerging events caused by changes internally to the organization or by the external business environment. A company may need to offer different types or quality of products and services due to changing customer expectation; it may need to adjust its internal working leading to a different way it delivers value to the end customers. Any combination of drivers like cost optimization, market penetration and product or service leadership can demand agile reactions from an enterprise. Citing work done by Kishore Sengupta from INSEAD and Andrea Masini of the London Business School a November 2007, a report by Henry Peyret of Forrester points out that organizations need to manage two types of agilities based on the nature of the change imperatives: *Range* and *Time*. As explained in this report, range agility refers to an organization's ability to react adequately to expected change events, for example, fast assimilation of merging organizations in an M&A (mergers and acquisitions) situation; such change-forces can be anticipated reasonably well in advance although considerable organizational capabilities are needed to effect such changes efficiently. Time agility, on the other hand, concerns an organization's ability to respond quickly to a single or perhaps a few events, such as a sudden price drop or a sharp demand for a new feature, where the organization may need to fast modify its current way of working to accommodate reactions to such challenges. For example, if car buyers start to strongly favor built-in GPS systems, an automaker would have to find ways to make quick design and production changes to respond to this customer demand. As we will see later in this section, the ACM paradigm can help with both these kinds of agility.

While essentially an organization must be able to quickly and accurately "sense" the applicable change imperatives and then effectively "respond" faster than competition in order to demonstrate a beneficial level of agility, a little more detailed structure to capture the key characteristics of agile enterprises is helpful in judging whether a given paradigm would promote agility. For example, in 2012 a blog article (http://strategicfrontier.wordpress.com/2012/01/05/business-agility-defined/; accessed January/2014), Eric Jansen, a VP at ThoughtWorks (a software consulting company), describes business agility with six key components (see Figure 1.21): "Adaptive planning," i.e., not restricted to sequential planning but embracing frequent adjustments to the original plan, "focus on time-to-value," i.e., adopting a strong focus on frequent and on-demand deliveries of products and services, "decoupling," i.e., practicing loosely coupled business processes where departmental or functional processes are designed to be relatively independent from each other, "low latency," i.e., faster execution, "economic efficiency," i.e., promoting "lean" style of execution, and "rapid adaptation" i.e., allowing for fast course correcting of execution whenever necessary. In September 2013, Craig Le Clair et al. of the analyst firm Forrester Research published a report titled "The 10 Dimensions of Business Agility" (http://www.forrester.com/The+10+Dimensions+Of+Busi ness+Agility/fulltext/-/E-RES98503; accessed January/2014) that follows a similar approach to



Figure 1.21 Six key components of business agility. (Adapted from Eric Jansen, 2012, http:// strategicfrontier.wordpress.com/2012/01/05/business-agility-defined/. Accessed on January 2014.)

Jansen's and defines 10 key dimensions as yard-sticks for an enterprise to measure and grow agility. Forrester report grouped these 10 dimensions under three major categories: market, organizational, and process; for market, the dimensions are channel integration and market responsiveness; for organizational, they are knowledge dissemination, digital psychology, and change management; and for process, they are business intelligence, infrastructure elasticity, process architecture, software innovation, and sourcing and supply chain. These key components or dimensions of business agility cover activities and capabilities spanning from planning to execution; the ACM paradigm, as we will soon see, impacts favorably a similar spread of activities and capabilities.

In a quest to provide a system-theoretic approach to identification of the key characteristics of agile enterprises, one may look for similarities between such enterprises and complex systems; work of P. Cilliers ("What Can We Learn From a Theory of Complexity?" published in Emergence, Vol. 2, Issue 1, 2000) provides such a comparison. Cilliers uses concepts from complex systems like "interactions," "self-organizing," "co-evolution," and the "edge of chaos" to characterize agile enterprises. Easy and meaningful interactions among all concerned participants with a common goal to achieve, as Cilliers observes, is the most important attribute of complex systems that can enhance enterprise agility. The self-organizing characteristic refers to the feedback-driven governance that a participating set of individuals would naturally adopt and leads to faster delivery of products and services leveraging experience and creativity of the individuals yet staying within the guard rails of applicable rules and norms. Organizational learning and adaptations based on such learning to improve organizational performance are captured in the coevolution attribute and are important for an agile enterprise as they the enterprise stay aligned with (or sometimes even ahead of) the market trends and demands. The edge of chaos is an organizational behavioral notion that relaxes rigid rules of traditional bureaucracy so that the agility enablers like self-organization and coevolution can be sustained and fostered. This system-theoretic view of the key components of an agile enterprise turns out to be the most helpful approach to investigate and establish how the ACM paradigm impacts business agility.

As we have described previously in the context of case model, ACM approach decomposes the problem in terms of very loosely coupled phases, milestones, and activities that are triggered by events and controlled by associated rules. This approach of solution construction is inherently rapid and allows easy modifications. Thus, solution development time can be significantly reduced, release of frequent versions of the solution can be handled easily, and evolving solution capabilities



Figure 1.22 Solution management complexity versus solution component granularity.

can be promptly incorporated. In other words, the ACM approach to implementing solution is naturally amenable to agile style of software development. Throughout the life of software solutions, often significant costs are incurred in trying to keep the solution up to date as the problem definition continues to evolve over time. Optimal componentization (see Figure 1.22) is a strategy, first well studied within the purview of component based development (CBD) and then later through the practice of service oriented architecture (SOA), that can be utilized to reduce the time and effort costs for ongoing changes in ACM-based solutions since they are inherently composed (via events and rules) of solution components whose granularities are easier to control. Due to the inherent event-driven design of ACM solutions they are also suitable for rapid and agile extension of packaged applications or for quickly creating umbrella applications that facilitate collaboration between disparate business processes as may be the case in M&A situations. Thus, we see that the ACM approach of creating and maintaining software solutions can boost both range and time agility of an enterprise in the context of handling business activities while maintaining economic efficiencies.

Next we examine the working styles associated with ACM. As we have indicated before, ACM targets unstructured and unpredictable business activities that often require involvement of multiple individuals (aka participants or stakeholders) and need to leverage their collective experience and knowledge to attain the goal of solving business problems. Case models are designed such that they can handle expected and unexpected events, and are endowed with rules that can provide the necessary checks and balances while still affording high degree of autonomy of action by the participants. The adaptive aspect of ACM allows the list of activities and participants to be modified at runtime depending on the need of the case as it progresses. In ACM-based solutions, these runtime modifications can be done by the participants themselves without having to create and deploying a new version of the solution as would be the case, for example, in a traditional

BPM-based solution or legacy or packaged applications. In earlier sections, we also noted that an ACM solutions can (and ideally should) include solution components which lend themselves well to predefined (i.e., predictable) models, for example, case activities that can be modeled via routine human tasks or well-known business processes. As ACM solutions are developed, in early stages, many of the case proceedings can be treated as unpredictable and nonroutine some of which could be converted or reimplemented later as predefined tasks and processes as the organization learns from handling many cases. Learnings from handling of past cases can also be used to provide better guidance to case workers as they tackle new instances of cases, thus improving case resolution speed and reducing the amount of work needed to do so. While the event-based structure of ACM solutions provide great flexibility as to what activities can be used to achieve the final goals and who should execute those activities, rules that are inherent part of a case model can be utilized to keep the required level of order as the case progresses so that the whole process does not turn unmanageably chaotic. Good management and visibility of information related to the case, whether these are documents or messages or status of the progress, are also strengths of ACM solutions, and thus reduces delays associated with access to right information. It should be clear by now that indeed ACM paradigm is capable of providing products and services quickly to the end customers (i.e., organizational agility as seen by the end customers) and is well aligned with the necessary key elements of an agile enterprise as we have stated earlier (i.e., boosting organizational agility systematically); in fact, there is almost a one-to-one correspondence between the key characteristics of ACM and those of an agile enterprise when viewed through the systemtheoretic lens.

1.7 Summary

Our key goals for this chapter were to provide adequate motivation behind pursuing the ACM approach and to familiarize you with some of the basic but important concepts involved in ACM. Traditional ways of handling business activities in organizations have largely focused on predictable and often routine actions. Over the years, packaged applications like enterprise resource planning (ERM), human capital management (HCM), supply chain management (SCM), and workforce management (WFM) have captured and streamlined the essential record keeping and repetitive tasks. Customer relationship management (CRM) systems, nowadays quite a popular type of packaged applications, tended to go beyond the routine flavors of packaged applications in trying to match idiosyncrasies of customer behavior. While all packaged applications provide some capability for customization, customizability of CRM applications is usually broader. Also, most packaged applications embed a set of business processes that define and execute sequences of automated and human tasks in order to complete information queries or business transactions for internal and external customers. When the work set spans multiple applications or the capabilities of packaged applications need substantial extensions, use of external business processes that are typically realized in middleware type of software platforms (e.g., in BPMS) is usually the common practice. Both packaged applications and external (to packaged applications) business processes have well-defined models and thus are easy to monitor and optimize, which has led to their pervasive adoption. Unfortunately, and as we have illustrated in this chapter, a significant amount and variety of business activities do not lend themselves well to such predefined approach that do not allow runtime modification to the definition of the solution; work patterns that need to deal with lack of predictability or need to facilitate knowledge work are all examples of such business activities and this is where ACM approach shines.

46 Oracle Case Management Solutions

Case management paradigm is goal-driven; it is the attainment of desired results that is of primary importance and the actual activities performed are of secondary consideration. Often cases are strongly driven by data and documents, and collaboration and knowledge sharing are frequently leveraged in resolving a case. We should note that a select set of packaged application such as Oracle Siebel do offer some prebuilt case management solutions, which, when these solutions fit the requirements well, should be duly considered as opposed to building solutions from scratch in middleware; we discussed this topic in the context of build versus buy decisions for case management solution. Even if case management functionalities of packaged applications either for enhancing their functionality or for making them adaptive at run time, using middleware-based ACM capabilities and hence should also be duly considered during solution architecture creation. Additionally, as we have explained in this chapter, fundamental characteristics of ACM paradigm and solution are natural agility boosters; when packaged applications are customized or extended with agile ACM, the agility of the end-to-end solution increases as well.

As we have also mentioned in this chapter, the idea of viewing a business problem as a case has been around for a long time and its application by social workers predates even the emergence any type of modern technology. Today, with all the advancements of software technologies, in particular the loosely coupled event-driven type, case management is seen as a highly applicable computer-aided business problem solving strategy across many industries. At a high level, we cited three types of case management flavors: service request, investigative and incident management; in reality, many case management solutions turn out to be some mix of these three types. Through simple examples, we have touched upon the nature of things, the attributes of a case model, that need to be discovered as the ACM solution design starts; in fact, we have introduced a three-level methodology to help with organization of discoveries needed for ACM solution design. While the building blocks of case management solutions across industries tend to be similar, it is the details, the types of events that are of importance, the activities that need to be performed as response to the events, the type and amount of documents that are involved in the solution, and the breadth and depth of collaboration among the case participants that can be quite different from industry to industry. As the book progresses, we will systematically visit many finer aspects of designing and building ACM solutions.

Chapter 2

Adaptive Case Management (ACM)–Related Topics

2.1 Introduction

If we reflect on the relationship between human behavior and technology, we are certain to agree on at least one thing: they influence each other, bidirectionally. Need for a better quality of life is often a major driver for emergence of new technologies. On the other hand, technology can make new things possible or old things easier and better, and through use of such technology we allow changes to our behavior. If we consider technology in a broad sense, examples supporting this bidirectional relationship would require a very long list; even if we restrict ourselves to information and computing technologies, the list of such examples would still be quite long. Just consider technologies or systems like the Internet, the worldwide web, and social media—these continue to evolve and improve as we adopt them more and more in our lives, and as we realize things we can do with these we continue to change our socio economic behavior in order to derive bigger benefit and higher gratification from these technologies. Interestingly, adaptive case management (ACM) turns out to be a great example of this two-way human-technology coevolution. "Interactionism" (see, for example, http://www.slideshare.net/jquemada/socialprotocolbaseddesign; accessed February/2014 for some basic information on interactionism) is a currently popular sociological theory about the drivers and modes of interaction that is used to explain a variety of human behavior. In interactionism, actors have the ability to alter their behavior to best suit the needs of the environment they are in. These notions have immediate relevance in today's social media-dominated world-digitally sharing of knowledge and experiences in our private lives has become a daily habit of most of us. With easy and up-to-date information readily available, we tend to take decisions quickly and accordingly adjust our actions. Such habits in private life invariably influence the way we conduct our work day; often, we find it easier to solve difficult problems collaboratively, and we desire such abilities from the tools that we use at work. ACM solutions nicely fill this need. On the flip side, on demand collaboration tools with great flexibility to arrange work activities in real time, as offered by ACM solutions, also strongly encourage knowledge workers to engage in team collaboration and in finding individualized solutions to tough customer problems. Thus, ACM and emergent human needs are nicely aligned.

When faced with a complex problem, we typically tend to devise a high-level strategy to solve the problem and then as we work on the problem we evaluate the progress and course-correct our actions in the direction that would provide a higher, better, or faster chance of resolution. Besides facilitating spontaneous on demand collaboration, it is the adaptive nature of ACM solutions where the case workers are able to make choices about what next steps to take that make them (the ACM solution) a more natural fit for the way we humans solve complex problems. In fact without the ability for the case workers to enjoy the freedom of choosing a context-appropriate activity, whether it is already made available as part of the packaged solution or needs to be created at run time, how would the added benefit of on-demand collaboration and expert advice be incorporated into the problem resolution exercise? Given the importance of adaptivity of ACM systems, it is fitting that we delve a little deeper into the essential properties and considerations for adaptive systems. This will be one of the key topics we will discuss in this chapter, and our expectation is that such a discussion will strengthen our grasp of the fundamentals about designing ACM solutions. Additionally, in this chapter, we will take up short discussions on a few other topics related to either the concept or the implementation of ACM. In the literature, there has been a fair amount of discussion around different types of case management solutions—we will compare and contrast ACM, dynamic case management (DCM) and production case management (PCM). Since traditional BPM is another dominant way of designing and managing business activities, we will extend the comments we have already made regarding the differences and similarities between BPM and ACM paradigms.

2.2 Adaptive Systems and ACM

2.2.1 Nile Crocodiles and Cuttlefish

"It is not the most intellectual of the species that survives; it is not the strongest that survives; but the species that survives is the one that is able best to adapt and adjust to the changing environment in which it finds itself."—Leon Megginson, a management professor at Louisiana State University, made this remark (see: http://www.darwinproject.ac.uk/one-thing-darwin-didnt-say; accessed February/2015) in 1963 when reflecting on Darwin's work on evolution in Origin of Species. Darwin or Megginson are not the only ones upholding the importance of the ability to adapt. Famous scientists like Albert Einstein and Stephen Hawking have proclaimed this ability as a sign of intelligence; the well-known playwright and essayist, George Bernard Shaw, had remarked that all reasonable people adapt to their environment; notable writer H.G. Wells has also alerted, "adapt or perish." The word "adapt" means to "change something" so that it functions better in or is better suited for a given situation or environment (ref. Merriam-Webster dictionary). Conceptually it seems easy to understand the meaning of and the reason for adaption; however, it may not be as easy to create adaptive capabilities or systems. Evolution of life forms provides us abundant example of adaptivity, most of which have occurred over a really long time and the end results show a wide range of variations. The crocodiles of the Nile, one of the most ferocious and predatory animals of Africa or the thousand pounds plus sea turtles, appear to have essentially retained their image and characteristics from prehistoric times yet they have had to adapt to changes in climate and environment; in a sense these creatures have adapted to stay "the same." Evolution of land animals (e.g., tetrapods) from aquatic creatures or *cetaceans* (i.e., whales, dolphins, porpoises) from land mammals introduced significant amount of very slow morphological changes, ultimately resulting in large differences in their forms and structures. It is theorized that these evolutions were triggered by a desire for the concerned animals to have a better survival. And, then there are creatures like the cuttlefish—it can

alter its body shape, color, and skin texture in the blink of an eye to make itself essentially indistinguishable from its surrounding whether that is a smoothly patterned sandy sea bottom or highly rough coral reef (if interested search for cuttlefish camouflage videos on the Internet to see cuttlefish in action), and thus can easily fool a predator or a prey. Cuttlefish's adaptation demonstrates very high agility—if we were to wish a type of adaptiveness for a modern business or a business application, we are sure to favor the cuttlefish type of adaptivity over the resilience of the Nile crocodile or the slow adjustments that the land mammal made on its way to becoming a whale.

Cuttlefish is a special breed of creature—it is a mollusk with three hearts and greenish-blue blood. It displays its mastery in active camouflage with the help of a large brain that manipulates its tens of millions of color and reflector cells. How many camouflages can a cuttlefish do? For an external observer, it may seem uncountably large. However, as Dr. Roger Hanlon of the acclaimed Marine Biology Laboratory in Woods Hole, Massachusetts, explained in a 2008 interview in *New York Times* (see: http://www.youtube.com/watch?v=mW4PbW893ik; accessed February/2014), cuttlefish and most other animals that display active camouflage to adapt to their environment rely on three basic camouflage templates, namely *uniform*, *mottled*, and *disruptive*, to rapidly construct the shapes and colors needed for adaption. Reflecting on cuttlefish's strategy and ability to adapt in the context of a business or business system, we can note the following:

- 1. In order to adapt, we must first decide when to adapt, i.e., recognize the triggers that demand adaption.
- 2. Given an adaption imperative, we must identify what aspects of the system, i.e., structure or behavior, need to change.
- 3. We must be prepared, prior to the instance when we need to adapt, with the necessary modular building blocks or capabilities that we could manipulate or derive from in order to adapt.
- 4. The central ability to adapt is intrinsic, i.e., an inherent property of the system and result of adaption can change the system itself even if that change is temporary.
- 5. We need to adapt in an agile fashion, i.e., the system response times need to be adequately fast as the system adapts.

So, what model or at least the key structural and behavioral characteristics of a business system should we focus on in order to evaluate and improve its ability to adapt in an agile way? How would we translate these observations from a general adaptive and agile system to an ACM solution? We will explore answers to these questions in the next two sections.

2.2.2 Sense-and-Respond, Anticipate-and-Preempt, and Complex Adaptive Systems (CAS)

As we concluded in the previous section, the type of business systems we are focused on here needs to have properties that are a union of those from adaptiveness and agility, i.e., we are shooting for systems with "agile adaptivity" or for "adaptively agile" systems. In particular, solutions we would like to design and build through ACM paradigm need to be capable of responding to the emerging and often unpredictable case events; they need to provide a good balance between case worker freedom and overall governance; they need to have mechanisms to learn from the past to be better functioning in the future. In a quest to better define the philosophy and the underlying structure and mechanism of such systems, we can start by visiting the work published by Stephan Haeckel (*Adaptive Enterprise*, Harvard Business School Press, 1999). Haeckel's work targeted the



Figure 2.1 System characteristics for handling of predictable and unpredictable events.

topic of adaptive enterprise, which we have reinterpreted here in the context of adaptive systems as may be relevant to ACM solutions. Accordingly, given our current system requirements, we would conclude that our solution would include, as necessary, properties an "open and adaptive" system (see Figure 2.1—this is a slightly modified version of what Haeckel presented in his book) capable of handling unpredictable events. Such a system would recognize the goals to be achieved within a broader context; it would have built-in capability to both assess external signals (i.e., events) as they arise and evaluate (and correct, when necessary) its own actions via feedback (i.e., adapt); various subsystems would collaborate within a framework of commitment and coordination so as to deliver the overall system capabilities; the system (and its subsystems) would learn from its own actions and evolve to become more effective (i.e., "open" definition of the system). We can compare these characteristics with those of standard closed systems (e.g., packaged applications like ERP or HCM, or solutions built using traditional BPM) designed to handle at most predictable change (shown in the left part of Figure 2.1), where the system is focused on achieving its immediate goals via a set of predefined steps monitored closely by a "command and control" management framework. Effectiveness of such closed systems is limited by their preplanned structure and behavior and working of such systems is analyzed typically to identify opportunities to enhance operational efficiency (e.g., higher throughput, faster response time, lesser resource need, etc.).

An adaptive system interacts with its environment in a *sense-and-respond* manner, i.e., the system dynamically senses external signals and using its capabilities provides a response back. A four-phase adaptive loop (Haeckel, 1999) of *sense, interpret, decide,* and *act* captures the internal mechanism of such a system (see Figure 2.2 where we present a slightly enhanced depiction of Haeckel's adaptive loop diagram). These four phases are as such quite intuitive: in *sense* phase, the system captures external (or even internal) signals or stimuli, does the necessary filtering and classification so as to make a preliminary decision whether to process or reject these triggers; in *interpret* phase, using collective experience of its subsystems the triggers selected for further processing are analyzed to gain insight; in *decide* phase, possible system responses are explored and choices are made to select a final action plan that is what is executed in *act* phase. One can argue that (at least in a broad sense) most systems go through these four phases in order to display their capabilities.



Figure 2.2 Essentials of sense-and-respond mechanism of an adaptive system.

However, it should be noted that for traditional closed systems where external signals and internal behaviors are predictable much of the sense, interpret, and decide activities are already done in the design phase and the system simply acts, using one of the prespecified actions plans that best fits the triggering event; in the event the situation at hand does not have a corresponding prespecified action plan, a closed nonadaptive system will classify that as an "exception" and rely on some type of special handling, often outside the system, to find and execute the appropriate actions. Exception handling typically is very expensive, and depending on the situation a transaction with one or more exceptions can easily incur several times to several orders of magnitude higher time and cost than a normal transaction. Hence, exceptions are a killer of efficiency and can adversely impact quality of service. Adaptive systems are designed to handle variations and unpredictability as a rule instead of as exceptions, and using the assets of its subsystems and suitably adapting its structure and behavior construct appropriate actions plans at run time to meet the demands of the situation at hand, i.e., create unique solutions to unique problems. Thus, adaptivity helps reduce exceptions. In fact, adaptive systems can go a step further than sense-and-respond, which has the connotation of being reactive—it is possible to design adaptive systems that can interact with their environment proactively, i.e., anticipate-and-preempt, as Haeckel points out in his earlier cited book referring to a comment made by Adrian Slywotzky in the preface of the book. The system stakeholders, those who have *bona fide* interest in proper functioning of the system, play a major role in enabling the adaptivity of these systems. Given all available input using their experience and applicable policies, they may form many conclusions, stakeholders evaluate merits and demerits of possible for and against arguments and construct suitable action plans (some readers may find it interesting to note that such actions of the stakeholders parallels the dialectic triad, thesisantithesis-synthesis, usually associated with the philosophical school of German Idealism promoted by Hegel, Kant and Fichte). Such knowledge work by the stakeholders endows a key property that distinguishes adaptive systems (of the type we are interested in here) from static systems.

Since ACM, the central theme of this book, aims to deal with unpredictability and knowledge work leading the requirement of the ability to provide runtime change in work activities and participating workers, we want to note the following with respect to the desirable properties of our typical target adaptive system:

- 1. It will "empower" the system users and their organizations to not only handle unpredictability but also to leverage it whenever possible in order to derive or provide better value, for example, handling of difficult customer situations makes the service team better at rendering customer service or solving a hard design problem pushes an engineering team to innovate, etc. This property of a system that turns unpredictability into an opportunity to provide better than expected system performance is essentially what Nassim Taleb calls "antifragility" (interested readers can access a healthy discussion on antifragility in Taleb's 2012 book *Antifragile: Things That Gain from Disorder* from Random House; see also Keith Swenson's presentation "Antifragile Systems for Innovation and Learning Organizations" at *2013 BPMNext Conference*, http://www.bpmnext.com/bpmnext-2013-presentations/antifragile-systems-for-innovation-and-learning-organizations/ where he mentions antifragility in the context of adaptive systems including ACM).
- 2. It will be "robust" (i.e., will not break down or be rendered dysfunctional) so as to provide enough resilience when faced with the challenge of handling change in structure or behavior.
- 3. It will "eliminate" or at least significantly reduce bottlenecks that lead to delays either in making decisions or in taking actions.

Complex systems theory has been used to better understand adaptive system (Haeckel also refers to this fact). Our interest here in exploring the properties of appropriate variant of complex systems is to recognize the underlying structure and mechanisms of systems that, in our opinion, most ACM solutions do or will represent. With such understanding, it is our belief that solution architects would be able to better focus on and design the right elements of the solution so as to endow it with the aforementioned qualities (we will deliberate on this aspect in the next section).

While humans have been dabbling in exploration of complex systems for quite some time, organized scientific quests for modeling and analysis of complex systems are relatively recent (see, for example, http://www.health.org.uk/public/cms/75/76/313/2590/jIq8CP.pdf for a quick review; webpage accessed Februray/2014). The scope of this book will not allow us to present a detailed description of complex systems and their variants; instead, we will be content with just enough information about them so as to achieve our ultimate goal here, i.e., developing a better understanding of systems represented by ACM solution. At a basic level, complex systems can be defined as a set of interacting parts (or agents) where behavior of the system cannot be fully deduced by simply aggregating the behavior of the individual parts, i.e., there is always some level of unpredictability and nonlinearity in the overall behavior of such systems. Such agentbased interpretation of complex systems is usually attributed to J.H. Holland (ref. Hidden Order, Addison-Wesley, 1995). Figure 2.3 shows a schematic depicting the essentials of a complex system. The system exists within an environment and through system facades or interfaces it interacts with that environment, i.e., receives input from and provides output to that environment. The components inside the system are indicated as agents who interact with each other in any manner that may be necessary. These interactions result in summary responses from the system to the outside environment and the reaction of the environment due the system responses are fed back to the agents who in turn react to such feedback, and the cycle goes on. Generally speaking the inter agent interactions are more spontaneous and less regular than the "system to environment"



Figure 2.3 Simple schematic of complex adaptive systems. (Adapted from ref. http://www.trojanmice.com/articles/complexadaptivesystems.htm.)

interactions. As a simple example consider a system consisting of a family and their real estate agent and the internal and external behavior of this system during the process of making offers to a seller of a property; in this case, the seller will represent the external environment with which the system will interact. Various members of the family and the real estate agent will engage in several brainstorming sessions and discussions (i.e., nearly ad hoc agent interactions in our parlance) in order to decide on the details of the offer to be made to seller. Once these decisions are made, an appropriate interface of this system, most likely the real estate agent, will respond to the seller with the offer using proper protocol and terms and conditions (i.e., following a more disciplined interaction style than the brainstorming style used by the system agents to come up with the offer). The seller may or may not accept the offer. The system interface will receive seller's decision and relay it back to the concerned agents (i.e., the family members) who, depending on the feedback received (i.e., acceptance or rejection of offer by the seller), will either move towards finalization of a sales contract or consider a revision of the offer.

Complex systems must not be confused with complicated systems where the latter can always (at least theoretically) be reduced to a combination of simple (linear) systems for analysis purposes and where cause and effect relationships always hold. Most things in nature are actually nonlinear complex systems even though often many of them are approximated as (at most complicated) linear systems for modeling, analysis, and prediction exercises. Complex systems that have the ability to adapt, i.e., change their behavior due to internal and external stimuli or feedback are termed "Complex Adaptive Systems" (CAS). In fact, most naturally occurring complex systems are of CAS type and hence the motivation for the study of CAS. Currently there is rising interest in applying CAS ideas to problem solving in many industries including IT (see, for example, http:// emergentpublications.com/(S(gkhrfo5szaqgz0waagkcfijg))/ECO/submissions/manuscript_85. pdf, ftp://ftp.cordis.europa.eu/pub/ist/docs/directorate_d/st-ds/emitleton.pdf, etc.; accessed February/2014). It should be noted that although CAS have a lot of flexibility in the way they act and react, in real world they are ultimately bound by some higher order laws; in case of natural systems, these higher order laws can include speed of light, laws of thermodynamics, etc., and for business systems similar disciplines may be imposed by industry practices and applicable regulations. Some CAS may also have the capability to evolve, i.e., become different in structure or behavior as these systems or their parts interact with their ecosystem. Some researchers simply

54 Oracle Case Management Solutions

include this capability to evolve as part of CAS properties, wherever applicable, while some others refer to such CAS as Complex Evolving System; in this book, we will adopt the former strategy.

The work of Paul Cilliers (Complexity and Postmodernism: Understanding Complex Systems, Routledge, 1998) discusses a set of major characteristics for CAS—some that are of direct interest to us are paraphrased as follows:

- 1. Multiple agents-These are the interacting parts that make up the CAS.
- 2. Rich interactions—Elements of the subsystem affect and are affected by elements of many, the system as a whole, and the environment in which the system exists.
- Nonlinearity—Small changes in input can cause large and nonproportional changes in the output.
- 4. Openness—CAS are open systems, i.e., defining exact system boundaries are not always easy.
- 5. Nonequilibrium operation—CAS often operate away from equilibrium (or average) conditions.
- 6. History-Present behavior of CAS is affected by the past.

CAS can have multiple levels of subsystems. CAS are "emergent" (i.e., display spontaneity) and "self-organized" (i.e., loosely connected seemly chaotic actions governed or guided by higher level order and goals). Typically the interaction rules between CAS are relatively simple. Many CAS can learn from the past and evolve to something of better quality or more suitable to its environment. In some situations, CAS can also anticipate possibilities using its endowed and acquired knowledge. Given these characteristics of CAS and the ACM discussions, we have already had so far we should start noticing strong similarities between them. A pertinent question at this point would be: Can CAS be modeled, and if yes, what modeling artifacts would be needed? In general, thorough modeling of nontrivial CAS can be difficult although representative structures with elements, subsystems, and guiding rules for interaction and evolution can be assembled and simulations of CAS can be done to gain reasonable understanding of the overall system. It is therefore important that whenever a CAS is designed the construction of the necessary building blocks is given proper attention. In their book, *Harnessing Complexity* (Basic Books, 2001), Robert Axelrod and Michael Cohen have discussed a set of modeling components for CAS that could be summarised as follows:

- 1. Strategy-A conditional action pattern that indicates what to do in which circumstance.
- 2. Artifact—A material resource that has a definite location and can respond to the action of agents.
- 3. Agent—A collection of properties, strategies, and capabilities for interacting with artifacts and other agents.
- 4. Population—A collection of agents, or in some situations, collection of strategies.
- 5. Type—All the agents (or strategies) in a population that have some characteristics in common.
- 6. Variety—The diversity of types within a population or system.
- 7. Interaction pattern—The recurring regularities of contact among the types within a system.
- 8. Space (physical)—"Location" in a geographical space and time of agents and artifacts.
- 9. Space (conceptual)—"Location" in a set of categories structured so that "nearby" agents will tend to interact.

- 10. Selection—Processes that lead to an increase or decrease in the frequency of various types of agents and strategies.
- 11. Success criteria or performance measures—A "score" used by an agent or designer in attributing credit in the selection of relatively successful (or unsuccessful) strategies or agent.

While some of the items in the list may seem a bit theoretical, we have retained them for the sake of completeness.

As we have discussed already, the type of business systems that we would like to design and build through ACM paradigm needs to be capable of responding to the emerging needs of cases in a *sense-and-respond* (and possibly even in an anticipate-preempt) manner; they need to provide a good balance between case worker freedom and overall governance; they need to have mechanisms to learn from the past to be better functioning in the future. Will the set of goals and modeling components be enough for the type of CAS we are interested in here? Haeckel's book (see citation earlier in this section) points out that while CAS characterizations as given here do provide a lot of insight and guidance for building our target CAS, they either miss or do not emphasize enough a few key properties: *intentionality* and *purposefulness*, i.e., desire of the system to reach certain goals that also in turn influences the behavior of the agents, and the fact that agents working within the system can redefine (or evolve) the system itself (this is a defining aspect of *social* systems). As we compare ACM solutions with CAS more closely in the next section, we will keep Haeckel's observation in mind.

2.2.3 ACM Solutions as Sense-and-Respond (Complex) Adaptive Systems (SR-CAS)

Having discussed various types and aspects of complex systems so far in this section, we would now like to investigate how that knowledge relates to ACM solution, and if indeed the notion of complex systems is relevant to ACM then how that knowledge could help us derive some design philosophies for ACM solutions. In this section, we first introduced complex systems where multiple agents or subsystems interact with each other with high level of freedom and spontaneity to bring about the overall response characteristics of the system, which is a nonlinear complex function of the individual agent or subsystem characteristics. Next we focused on those complex systems that can adapt, i.e., change their structure and behavior due to internal or external forces—these were termed complex adaptive system (CAS). We then explored the key concepts behind adaptive organizations that behave like CAS and in that context deliberated on the senseand-respond (SR) behavior of such organizations. It turns out that ACM systems have strong similarities with SR adaptive organizations when we view these organizations as systems and thus we will pay closer attention to the properties of SR adaptive organizations that henceforth we will term "SR-CAS." We can establish these similarities if we can validate a strong correspondence between constructive elements of SR-CAS and a typical ACM solution. While we have made occasional remarks about ACM while discussing SR-CAS earlier in this section, in Table 2.1 we present a more thorough comparison by noting relevance of key SR-CAS elements (referring to work of Cilliers, Axelrod and Cohen, and Haeckel, as discussed earlier in this section) with the elements of typical ACM solutions.

The remarks in Table 2.1 should easily convince the reader that the structure and essential properties of SR-CAS are highly relevant to ACM solutions. This means that knowledge of how we can construct or influence the behavior or functioning of SR-CAS should provide us useful clues as to how to design ACM solutions as well. Comparing properties that have high influence

Sense-and-Respond CAS Attribute	Relevance in ACM Solutions	
Cilliers (1998)		
Multiple agents	Typical ACM solution system will have more than one stakeholder	
Rich interactions	Collaborations among stakeholders	
Nonlinearity	A single or few events can potentially change the case progression significantly	
Openness	An ACM system can evolve with learning	
Nonequilibrium operation	Flexibilities to move away from an "average" execution path, when necessary	
History	Often the next best step depends on the case progression thus far	
Axelrod and Cohen (2001)		
Strategy	Org. procedures, industry norms, and regulations guiding stakeholder actions	
Artifact	Case data, documents, information	
Agent	Stakeholders; software applications	
Population	A typical ACM solution involves multiple agents	
Туре	Roles and responsibilities of the stakeholders	
Variety	A typical ACM solution involves multiple roles and responsibilities	
Space (physical)	Organizational affiliation of individual stakeholders	
Space (conceptual)	Expertise of individual stakeholders	
Selection	Periodic evaluation and consequential modification of case models	
Success criteria/performance measures	KPIs/KRIs; goal achievement	
Haeckel (1999)		
Sense-and-respond	This is the style for inter participant collaboration and customer interactions	
Intentionality and purposefulness	Goal-driven nature of ACM solutions	
Evolution	Progressive improvement in system strategy and structure due to past learning	

 Table 2.1
 Relevance of Elements of Sense-and-Respond i.e., SR-CAS in ACM Solutions



Figure 2.4 The "Four-Cs" of ACM solutions.

on the behavior of SR-CAS and the key functionalities of typical ACM solutions, we have identified four high-level considerations, "context," "capabilities," "collaboration," and "coordination," that we feel are critical for conception and design of ACM solutions (we would term these as the "Four-Cs" of ACM—see Figure 2.4). We would like to point out that at some level most of these Four-Cs considerations are also generally relevant to the process of designing many other systems; however, in our opinion, these are fundamental considerations for typical ACM solutions. The meaning and relevance of these Four-Cs are as follows:

1. *Context*: For a SR-CAS like an ACM, solution context is the highest-level description of the system. Following Haeckel (1999) we can define this to be a combination of purpose of the system, i.e., why does it exist, essential governing principles, i.e., the guard rails that define what must or must not be done, and the architecture, i.e., a prescription of the system components and their interaction styles and protocols. Of course, during the design of any system the designer considers the goals of the system and the means to achieve them through his/her design. So, why is this special for ACM? The answer lies in the realization that the systems where the execution path is fully predefined and the activities are prescheduled and assigned, the workers who use the system to resolve a business problem are primarily (if not solely) focused on completing their tasks; these workers may be aware of the design goals and context of the system but those facts are not generally essential for completion of their tasks.

Take, for example, the processing of a typical employee expense report. Once an expense report is submitted via an ERP or financials application (this is quite a common approach) a

manager approves or rejects it; if approved the expense reports proceeds to the next step and if rejected it is sent back to the employee for revision and resubmission. Next, the approved expense report is accessed by an accounts payable person who checks every line item in the report for its legitimacy following all applicable rules, verifies the reported amounts against expenditure receipts, and finally clears the expended sum for payment. These fixed steps of expense verification, the set of applicable rules, and any additional supervisory checks on the proceedings were all put into the design of the expense payment application part of the ERP by some application designer starting from a higher level context. However, for the accounts payable person processing a particular expense report that high-level context is not pertinent. In fact, by limiting the scope of what the worker does in this example, the work activity can be tuned to be as simple as possible and the required skill level of the worker as low as possible, both of which enhances the efficiency and reduces the cost of the execution of the expense reimbursement process.

Things can be quite different in case of a transaction going through an ACM system. For example, let us consider a case representing the processing of a high-value and high-risk credit application from a business by a credit insurance company. From the time that such an application is accepted (for processing) by the credit insurance company, several workers with a variety of expertise (e.g., analysts who analyze viability of companies, risk assessors, underwriters, lawyers, etc.) will undertake many predetermined tasks and many case specific tasks not necessarily preordained, sometimes individually and in some instances in collaboration with others, in order to come to a conclusion for the case.

Clearly, these proceedings of high-value and high-risk of credit insurance application processing are lot more complex than the expense report processing example and from time to time case workers, given some unique characteristics of the case, may find themselves with questions like "What should be done next?" or "Who could help with the next step?" or "Will the next step violate some rules or regulations?"; resolving questions like these at runtime (i.e., when the case is being executed and not when the ACM system was being designed) require the case workers to be aware of the full system context as formulated above. In other words, for ACM knowledge of context is not a "good to know" thing, it is essential to proper working of the system.

Contextual statements can be described at a high-level however must be done without ambiguity as the context becomes the ultimate reference for the SR-CAS agents and subsystems when they need guidance or inspiration as to what action to take in a given situation. In Chapter 1 as part of the discussion of CM Level-1 discovery phase for ACM solutions, we have recommended a set of attributes that should be included as the solution designers create the necessary context for their solutions.

2. *Capabilities*: In multiagent systems like SR-CAS, the system capabilities are a fairly complex function of agent or subsystem capabilities. One of the key mechanisms via which the overall system delivers its functionality is by on-demand composition of appropriate component capabilities, i.e., the capabilities of its agents or subsystems. This means that the capability components have to be modular and composable. Haeckel (1999) uses modular and composable capabilities as the foundation for the design of the underlying structure and strategy creation of sense-and-respond adaptive enterprises, and the same philosophy is also valid for ACM solutions. Again, creating larger and complex systems from more manageable smaller and less complex modules is a fairly standard design strategy for most systems, whether we are talking about a software system or an automobile. In these cases, however, the users of the systems deal primarily with the overall system capabilities and are not concerned with

the modules that were used to create the system. If there are two versions of a software system to handle bank transactions providing the same user experience and performance, one written as a million lines of code and the other assembled from several hundred subroutines or methods, from the point of view of the bank and its users the two systems are equivalent even though the time and cost of developing and maintaining the systems may be quite different. In ACM solutions, however, the agents utilize subsystem capabilities and ultimately deliver the system capability, at least in part, through inter-agent interactions. Hence, the design of the capability modules is a critical consideration as the inter-agent interactions tied together selects subsystem capabilities at runtime shape the behavior of the overall system. Since ACM solutions must be capable of handling unique situations on a case-by-case basis, and since these variations cannot always be predicted upfront, the solution designer cannot aim to program the complete problem solving strategy along with its execution steps that would be sufficient for all situations; at least a part of the problem solving strategy will evolve along with the progression of the case. Instead, the agents responsible for executing the emerging parts of the problem-solving strategy will combine suitable capability components at run time. This consideration has a direct implication on how individual tasks in ACM solution are designed from modularity and composability points of view, i.e., what granularity they should have and what kind of functional independence they should possess; readers conversant with the key concepts related to design of "services" with a goal to create agile application building foundation in the context of Service-Oriented Architecture (SOA) should find these granularity, functional independence, and composability concerns as familiar topics.

- 3. *Collaboration*: As we have already noticed, a key aspect of SR-CAS is the interaction among the individual agents that is central to capability composition and ultimately results in the overall system behavior observed externally. Whenever there are human agents (i.e., workers) engaged in solving a problem, some collaboration among the workers is likely. In fixed non-adaptive systems dealing with routine work, activities are either designed to be performed by individual workers or through some fixed collaboration pattern of a set of workers—this leads to increased efficiency. In adaptive systems like an ACM solution, runtime collaboration. In ACM solutions, participants engaged in knowledge work are expected to collaborate with their peers and subject matter experts to resolve case related issues quickly and satisfactorily. A good ACM solution design, therefore, must facilitate easy and quality collaboration among its participants; as a corollary it must also adequately integrate required information sources and applications as timely access to right information is a key factor in to quality collaboration.
- 4. *Coordination*: Since agents and subsystems of SR-CAS enjoy high levels of freedom in the way they behave, surely some mechanism needs to be in place that would prevent the system from becoming awry. The trick is to adopt a governance style that would provide the right balance of level of freedom and control over agent actions. In traditional nonadaptive systems, a *command-and-control* style of governance is commonly employed. Such governance structures are typically hierarchical, upper levels specifying to lower levels what to do and what not to do for all anticipated situations; a department head giving specific instructions to his/her managers, the managers in turn instructing the workers, and so on. This hierarchy of commands also provides the basis for a mechanism by which the superior members of the organization aim to control the overall behavior of the organization; performances measures are usually aggregated upwards level by level to reflect the overall performance

of the organization or the system. While command and control governance style has been around for a long time, is easily understood and implementable, and thus has been quite popular, such governance styles restrict the freedom of action as would be required by agents of adaptive systems. Some CAS researchers have pointed to *self-organization*, a type of endogenous discipline, as the paradigm that maintains the higher level order of adaptive systems. However, as Haeckel (1999) illustrated for CAS with higher levels of complexity and multiple levels of agents only self-organization is not enough to guarantee purposefulness of the system, i.e., the requirement that the system ultimately achieve some stated goal. While command-and-control style of governance works for closed systems where the execution strategies are completely predefined, for multiagent adaptive systems dealing with unpredictability and knowledge work a governance strategy based on "coordination of commitments" is a better model (Haeckel, 1999), one that would provide the required balance between freedom and control of agent actions. For a sense-and-respond governance system, Haeckel defines commitment as "an agreement between two parties to produce a defined outcome and to accept that outcome if it meets the agreed-to conditions"; the governance structure then is set as a coordination of such commitments, i.e., specifying a set of norms and protocols as to how different parties should interact with each other. Thus, coordination of commitments essentially articulates the expected behavior of the agents, i.e., how agents of the adaptive system will react to a given stimulus, and the rules for inter-agent interactions. In ACM systems, this strategy and style of governance translates to appropriately specifying privileges, tasks and interaction protocols for various stakeholder within the high-level context of the solution; typically a case owner assigned to the case has the responsibility of making sure that this coordination of commitment properly happens and the case ultimately advances to a resolution.

As we have indicated earlier, case models embody most of the essential structure of an ACM solution; solution architects can help ensure proper adherence to most of the mentioned Four-Cs principles by appropriate design of the case models. The mentioned four are not the only considerations when designing ACM solutions; however, in our opinion this is an important set of considerations that need special attention so as to build in the desired adaptive characteristics. As we have already explained when discussing the basics of case management, adaptivity is neither required nor desired in every level or every component of an ACM solution. This is due to the fact that as we decompose the ACM problem some of the tasks and processes emerge as deterministic where either due to efficiency or regulatory reasons prefixed optimized designs that are best governed in the traditional command-and-control manner. In other words, a complete ACM solution is likely to be a hybrid (see Figure 2.5) sitting somewhere between rigid closed systems and flexible adaptive ones.

We would like to close this section by revisiting the three primary characteristics that we had stated in Section 2.2.2 as essential for the type of adaptive systems we are after: our system needed to be robust, empowering, and would have means to eliminate (or at least significantly reduce) bottlenecks that hinder its agility. A rigid system tends to be fragile, i.e., malfunctions when impacted by out of design forces; systems that are malleable, i.e., have the flexibility to adjust to such forces avoid frequent breakdowns and thus are more robust than rigid systems. The adaptive nature of ACM solutions provides the desired robustness in our target solutions. Emphasis on knowledge work and provisions to exercise freedom to do the right thing at the right time encourages ACM system stakeholders to think out-of-the-box when necessary and deliver better solutions than would be possible via a rigid solution framework. As we had illustrated in Chapter 1, the way ACM solutions are built and operated is well aligned with key agility enablers. Since agility



Figure 2.5 Hybrid nature of real-life ACM solutions.

requires a system to demonstrate faster response, particularly in unpredictable situations, it would be useful for us to consider aspects of an ACM solution that, if proper care is not taken, could slow down the system response. From a customer point of view, as they raise unique requirements in a case, the ACM solution, along with the actions of its participants, needs to be able to respond quickly and correctly—the adaptive nature of the solution is at the heart of ACM's fast response capability. However, there are many other aspects of the solution that influence the speed of system response. For example, if the information or documents needed by the case participants are hard to find, or the participants do not have comprehensive visibility into the progression of the case including relevant KPIs or KRIs, their ability to act quickly will be hindered. If the inter-participant interaction protocols or the dos and don'ts are ambiguous, then collaboration among participants, one of the key enablers of ACM solutions, will be poor and slow. Automated tasks, i.e., those executed by machines are typically much faster than if they are done by humans (of course, not all tasks are doable by machines). This means that if we fail to recognize opportunities for task automation we will slow down the system response of ACM solutions along with possible degradation in operational efficiency.

2.3 Dynamic, Production, and Adaptive Case Management (DCM, PCM, and ACM)

2.3.1 Basic Debates

"Better, cheaper, faster—pick any two," sometimes also called the designer's "Holy Triangle" or "Iron Triangle," is commonly taken as a guiding principle when scoping a product or service offering. Although there are those like Daniel Goldin (http://history.nasa.gov/dan_goldin.html; accessed February/2014) who ran NASA between 1992 and 2001 and famously promised to achieve all three simultaneously. Conclusions like "picking better and cheaper will typically slow you down," "combination of better and faster will generally be more expensive," and "doing things fast and cheaply will cut into the quality of the ultimate outcome" surely sound intuitively correct. Two types of questions should be asked at this point: when we say better, cheaper, or faster what we are comparing these with, and would it indeed be foolish to "have it all," i.e., aim to achieve success on all three measures? Firstly, better, cheaper, or faster are relative either to status quo or to what the competition is doing. This assumes of course that the one that is not picked (i.e., as part of the "pick any two" premise) is at least adequate for the purpose for which the system is being designed. The second question, unfortunately, is much harder to answer although anecdotally "pick any two" mantra seems to get a fair amount support from "experienced" project and product managers and developers. The problem is that even though these three measures are fairly well understood in our day-to-day lives, for complex projects these metrics are not always easy to assess. We know that Goldin (of NASA) made the promise of succeeding on all the measures at the same time-did he succeed? It really depends on how one defines success (for a slightly pro-Goldin summary of successes and failures of Goldin's initiatives see: http://www.nationaldefensemagazine.org/archive/2012/April/Pages/Faster,Better,CheaperWhyNotPickAllThree.aspx; accessed February/2014). The facts are that many of Goldin's initiatives did fail but he attempted really ambitious goals and as a result surely pushed the frontier of contemporary knowledge and experience. So did Goldin deliver on his promise? Clearly, the answer will depend on whether one would measure success by the fraction of successful projects or by advancement of science and technology. Regardless of whether this "pick any two" is indeed some unbreakable law of projects, a good solution design must settle the debate of how much to focus on for each of the quality, time, and cost dimensions (see Figure 2.6) as clearly and as "early in the game" as possible; design of a case management solution will not be an exception!

Another equally important question that is pertinent when laying out the plan for creating a solution is "where to start from"? Should one start from "a clean palette" or leverage some existing solutions possibly through modifications and extensions. In the case management solution arena, we see both approaches being used by software vendors and end customers. "Pure play" vendors as well as some customer projects have created their case management offering or solutions "from scratch" that have had the benefit of having fewer restrictions in the way the products or solutions could be designed and implemented—this strategy gave them opportunities for easier innovation. However, these products or solutions generally faced the challenge of having to create a lot of infrastructure either to provide the basic case management capabilities or to make the outcome fullfeatured and mature. On the other side of the spectrum, established software vendors in related fields, such as those in BPM or content management arena, recognized that they already had much of the necessary infrastructure and created their offering by expanding on such base lines.



Figure 2.6 Competing design considerations.

Customers with existing projects like CRM or document management felt similarly and extended such implementations with added features to handle case management. Such efforts by established ISVs or by existing application teams have had varied levels of success in matching the innovative features of the pure play efforts. For example, case management solutions, particularly ACM, relies heavily on an event-based engagement and on-demand collaboration of stakeholders that requires the software application to have a strong event handling and collaboration foundation—this is not something that can be easily inserted into existing products or applications unless of course the necessary foundations for event handing were already present in them. Practically all industrial grade software dealing with business activities and transactions need management and monitoring tools, dashboards and altering capabilities, security and other policy enforcements, broad integration support, high performance and reliability standards, and so on-an innovative "built from scratch" case management solution can relatively easily demonstrate advanced functional features yet can find itself to be inadequate in supporting the necessary qualities needed for industrial deployment. Fortunately, over time, consolidation of ideas and technologies from both camps are producing enriched and robust capabilities for the end customer and case management solutions are currently benefiting from such consolidations.

These "quality-cost-time" considerations and the choice of the starting point of a particular case management solution (i.e., whether pure play or legacy extension) have resulted in meaningful distinctions between certain categories of case management solutions like production case management (PCM) and adpative case management (ACM) that would be useful for us understand in the context of case management solution design (see: Keith Swenson's 2012 post at http://social-biz.org/2012/09/12/ case-management-contrasting-production-vs-adaptive/; accessed February/2014). Mainly triggered by Forrester's coinage ("Dynamic Case Management—An Old Idea Catches New Fire," Craig Le Clair and Connie Moore, Forrester report, December 2009) there is also the meaning of DCM to be considered. Although it is remarked that DCM is the same as ACM, it would be worthwhile to closely inspect this statement; for a discussion on DCM versus ACM by Max Putcher (see: http://acmisis. wordpress.com/2010/11/18/the-difference-between-dynamic-and-adaptive/; accessed February/2014). In an expanded blog article posted in 2012, Michael Poulin suggested an umbrella category called "Purpose Case Management" where he proposed that the overall purpose of the system be broken down in subpurposes (i.e., subgoals), which would then dictate what type modeling paradigm, i.e., the type of case management or process management would be the most suitable. While this was an interesting proposal and Poulin's analysis as presented in the article is a good read, there has not been much uptake of this purpose case management framework (interested reader can access Poulin's article at http://www.infoq.com/articles/purpose-case-management; accessed February/2014). We should note that "static versus dynamic" or "deterministic versus nondeterministic," or "production versus adaptive" case management is essentially a system "classification" strategy—the universe of case management solutions are being grouped into disjoint sets as per some given points of view. Given a business problem to be solved by case management, which class of solution would be the most appropriate? The motivation and characteristics of such system classifications are generally useful as background knowledge for designers to make the right choice for picking the solution class, and we delve into more details on these next.

2.3.2 Digging Deeper into DCM, PCM, and ACM

Our main goal in this section is to understand better the meanings of and motivations behind labels like DCM, PCM, and ACM as these phrases are currently in active circulation in the case management practitioner circles. Our hope is that with a clearer understanding of these phrases



Figure 2.7 A generic system.

we will avoid confusion as we try to absorb industry literature and also will be able to make better choices for our case management solution architectures. Per our earlier remarks, such labels can be identified as system classifications treating case management solutions as systems. We, therefore, will take a small digression into standard notions of system classification in order to lay the basic foundation for the exercise to explore case management system classifications.

Starting at the very basic, a system is simply an encapsulation of a set of interacting components with a visible interface to the world outside the system, which is essentially the environment where the system exists. Most systems interact with their environment by an input-output mechanism: when the system receives some input or trigger from the environment, it responds by exhibiting a corresponding output (see Figure 2.7). When we select a particular point of view (PoV) and observe the universal set of systems from a particular PoV, we create "classification," i.e., nonoverlapping subsets of systems that provide further detail with respect to the chosen PoV. Accordingly, and as shown in Figure 2.8, we have chosen three different PoVs resulting in three different system classifications:

- 1. *Static or dynamic*: A system is "static" if the system components and their assigned behaviors stay fixed over time, and if the system response at any given time is independent of what may have happened in the past i.e., it is "memory-less." A system that is not static is "dynamic."
- 2. Deterministic or nondeterministic: A system is "deterministic" if it always produces the same output for a given input set, i.e., the system response is fully predictable. Accordingly, "non-deterministic" systems may exhibit different behaviors corresponding to identical input sets applied at different times.



Figure 2.8 Different classifications of a system.

3. Adaptive or nonadaptive: A system is "adaptive" if it is able to change its composition or behavior as a result of external stimuli. In software systems, this would mean changes in the system at "runtime," i.e., such changes will not require "design-time" decisions, reimplementation or redeployment of the system. "Nonadaptive" systems belong to the complement subset of adaptive systems and their composition and ability to respond to external stimuli remain fixed over time.

Note that since these classifications correspond to different and independent PoVs, i.e., there is a clear separation of concern for each of the PoVs leading up to the classifications, a system can be simultaneously characterized by as many classifications as necessary. For example, a system can be qualified as "static, nondeterministic and nonadaptive" or "dynamic, nondeterministic and adaptive," and so on. We should also point out that a system can be classified in many more ways than the three we have chosen here—our choice is based simply on the characteristics; we will need to continue with the explanations later in this section and elsewhere in the book.

Let us consider a few simple examples to make the above system classifications more concrete. Take the behavior of an ideal metal spring—when we pull (aka "load") the spring it extends and the extension diminishes as we let go of the pull (aka "unload"). The extension of a metal spring is simply proportional to the force on the spring at any given time as Figure 2.9a depicts; it is immaterial as to how we arrive at that force. If we load or unload a piece of rubber, however, the behavior is quite different. As Figure 2.9b indicates, the extension for rubber does depend on whether we are loading or unloading and how fast we are doing so. Thus, for rubber, unlike the metal spring, the deformation at some load will depend on how we arrived at that load. In other words, a metal spring is not affected by loading history, i.e., it is memoryless and hence it is a static system while response from rubber depends on the loading history and thus it is a dynamic system. Both the metal spring and the rubber are also deterministic systems since their behavior is fully predictable at all times as long as we know the input or the input history. The "throw value" from rolling of dice is an outcome of a memoryless, i.e., static system as every roll is independent from all previous



Figure 2.9 Load and deformation characteristics of metal springs (a) and rubber (b).

rolls; however, we cannot predict with certainty the value of the throw—this makes it a nondeterministic static system. Similarly, a pendulum is a deterministic dynamic system while weather system is both dynamic and nondeterministic. Adaptive and nonadaptive systems differ in the way they handle intrinsic and extrinsic stimuli—adaptive systems are capable of changing their constitution and response protocols while in nonadaptive systems these stay fixed. The pendulum is a nonadaptive system and so is a simple pencil sharpener. Of course, nonadaptive systems do not have to be "simple"; in his 2010 book *Management 3.0: Leading Agile Developers, Developing Agile Leaders* published by Addison-Wesley Professional, author Juergen Appelo mentions a whirlpool in a bath tub as an example of a complex nonadaptive systems. Presence of feedback loops is an important characteristic of adaptive systems. Most biological systems, human organizations, and some robots with feedback controls are examples of adaptive systems.

We can easily draw similar examples in software systems as well. A service that, upon receipt of a query, selects and retrieves information from a fixed knowledge base is a deterministic static system. Google's (or similar) search engine learns from the searches that it handles in order to find content with higher relevance faster from an ever increasing pool of information and thus will produce different outputs, both in terms of content and their order, at different time instances even if the exactly same search parameters are used at the input. This is a nondeterministic dynamic system. Consider an online portal that presents catalogs of a fixed set of products and their prices that a customer can browse and choose from, provide the necessary payment and shipping information, and close the buying part of the transaction by executing the "check out" step. Here the customer interaction with the portal provides the input to the system, which in turn provides output responses by displaying prices and shipping confirmations, the portal's response only depends on the customer's order at that instance. As per definitions stated earlier, this portal would correspond to a static system. However, if functionalities like promotional pricing based on the purchase history of the customer are added to this sales portal it will become a dynamic system supporting dynamic pricing. If, based on the customer profile and past purchase history, the portal modifies the content it displays so as to present customized pages to the customer i.e., changes one or some of its constituent parts then the portal system becomes an adaptive dynamical system. We may also note here that while all adaptive systems are dynamic systems not all dynamic systems are adaptive as we just illustrated in the portal example; when specifying the type of system that we are aiming for; it is therefore clearer to state "adaptive" explicitly if that is indeed the intent.

Having refreshed ourselves on a set of system classifications in general terms, we will now turn our focus back to classification of case management systems. In particular, we are interested in exploring PCM, ACM and DCM further. From our experience as well as from published case studies and from discussions in industry circles, we see that PCM is a style of case management that has been and likely to be in broad use. Quoting from Keith Swenson's work (2012), we can define PCM as:

Production Case Management (PCM) is an approach to supporting knowledge workers which is programmed by specially-trained technical people (programmers) to produce a case management application. That application is deployed for use by knowledge workers to get their work done. The application offers collections of operations that the knowledge worker can select to use or not use depending on the specific needs of the case.

In the same article, Swenson defines ACM as:

Adaptive Case Management (ACM) is an approach to support knowledge workers who need the most flexibility to handle their cases. ACM allows the knowledge workers themselves to create and modify all aspects of a case at any time. There is no distinction between design time and run time: the designing and running are done at the same time by the same people.

While these definitions nicely capture the essence of PCM and ACM, in our view, Swenson takes a purist approach to ACM—in real projects we find the case management solutions, even the ones that are being termed as "adaptive," are rarely as adaptive as Swenson envisions them to be via his definition.

PCM and ACM are both designed to provide higher degree of flexibility of task execution by case workers than what traditional ERP-like or BPM-based solutions would. They are both goal-driven solution paradigms and share the idea of case folder as a key artifact for collecting the progression of a case. Both approaches facilitate a certain amount of knowledge work for case resolution although in ACM, where the work freedom for knowledge workers is much higher than in PCM, knowledge workers bear a higher level of responsibility for their judgments and actions. As Swenson (2012) points out, PCM systems are targeted for situations where level of unpredictability is significantly less than those where ACM approach is strongly required. We can envision that at one end of the PCM spectrum where essentially all work steps and their execution sequences are fully predefined the solution would correspond to a deterministic case management system. Even though PCM systems are less flexible than ACM systems, activity execution sequences in PCM do not need to be completely preordained, i.e., the execution path can move around through the steps as necessary. Also, in PCM, case workers can be presented with lists of possible actions at each step so that at a given step the case workers, using their knowledge and experience, can pick the right subset of actions based on the case goals and the case progression till that step. Thus, the PCM solution spectrum can stretch from deterministic into the realm of nondeterministic case management as judgments of two humans can sometimes differ even when all the other parameters are the same. The main difference between PCM and ACM though is that in the former the system composition, i.e., the structure and the assigned behavior of the subsystems stay fixed through the case progression. Let us consider an example where a case is being used to coordinate the work by a communications service provider for designing a network service for a business involving a fairly complicated set of steps requiring several different types of expertise. At some given step, as might be the situation with most of the steps, the expert (case) worker is presented with a set of possible actions. The expert worker is expected to review the overall design goals and the status of the design thus far and select a suitable subset of the actions from the possible actions list that he/she is provided. Once the actions at this step are completed, the design status is updated and the case progresses. It is possible that during the review the expert worker at a particular step detects possible errors done in some previous step-the case progress could then be regressed back to that faulty step and reworked. A PCM system would be fine for implementing a solution that would handle case progression through these steps. Now suppose that during this case progression at some step the assigned case worker feels the need for conducting a special survey or pursuing a special permit and he/she desires to accomplish this by adding a new activity and a new participant with the required expertise as the case is progressing then we would need the capabilities of ACM; here we would not only be changing the execution options but will also be altering the definition of the case management system itself by adding new activities and participants at runtime.

In Table 2.2, we have listed some of the key characteristics of PCM and ACM solutions that could be used as a starting point when deciding which style of solution would fit better for a

	РСМ	ACM
Type of work	Largely routine and repetitive	High knowledge and creative problem-solving work
Decision making	Usually from predesign sets	Can go beyond the anticipated
Type of collaboration	Mostly known ahead of time	Special enterprise/services may be needed
Work ownership	Adequate involvement	Sense of responsibility
Case volume	Typically large	Typically moderate

Table 2.2 Applicability of PCM and ACM-Type Solutions: Key Characteristics

given business problem. If we were to consider systems involving human workflows, implemented either via traditional BPM or delivered as part of a packaged application and ACM at two ends of a spectrum, PCM systems would fall somewhere in between. PCM systems would be targeted towards higher volume transactions where knowledge work is not too deep so that much of the activities can be predefined at design time; essentially repetitive work requiring somewhat special skills like that of an appliance repair person or a benefits administrator would be examples of such work. Many of the building blocks of PCM systems are often found in existing packaged solutions or are easily created using full-featured BPM systems—this allows vendors and customers to put together a PCM style case management solution quickly. It is relatively easy to train workers on repeatable deterministic activities; such activities are also easier to tune for efficiency and monitor for visibility and management purposes. As we noted before, although limited, there is some opportunity in PCM solutions for exercising human judgment for resolving certain unique issues related to a case rendering PCM solutions the ability to handle somewhat nondeterministic i.e., unpredictable situations. These properties make PCM systems an attractive solution proposition in many case management problems; we need to be careful though that in the zeal of landing on a quick solution we do not force ACM used cases into a PCM framework as that would guarantee highly suboptimal system performance and efficacy.

Let us take a closer look at "deterministic versus nondeterministic" system classification in the context of case management systems. If the activities are routine, required demand on human judgment is low, and the available decision options are quite limited, a case management system will behave deterministically. In the presence of high levels of knowledge work, human discretions and ad hoc collaborations a case management system will show nondeterministic characteristics. For example, an engineering design process handled as a case will be lot more deterministic than processing of a case dealing with an application for immigration into a country. How does it help us to know if a particular case management system is likely to be more deterministic or otherwise? Deterministic systems are easier to manage and improve especially from an efficiency point-ofview. Hence, a solution designer should consider keeping the solution as deterministic as possible so long as the effectiveness of the system is not driven below what may have been set by the goals of the system.

By following standard definitions of static and dynamic systems, a typical case management system will be classified as a dynamics system since an evolving case history and dependence of case worker actions on the case history are part and parcel of the case management way of working. In other words, the "dynamic" qualifier in DCM coinage seems superfluous in the sense that it does not add more information to what we might glean by the simply saying "case management," and hence we will not use the DCM abbreviation further in this book. We do want to remind the reader though that since the Forrester report in 2009 on DCM where they described several use cases that would tantamount to a combination of PCM and ACM as we have described here, the acronym DCM has been in circulation—we do see it occasionally appearing in customer specifications and in communications from a few system integrators and ISVs.

In Figure 2.10, we have graphically summarized the case management solution classifications we have been discussing so far. The figure essentially states that all case management solutions fall under the class of dynamic systems but can be either deterministic or nondeterministic; PCM and ACM together cover the spectrum of case management solutions; PCM solutions lean more towards deterministic systems; and, ACM solutions are specifically designed to handle unpredictable situations. We should reiterate that from our experience with customer projects we often see the need of creating solutions that combine features from PCM and ACM styles of case management. For example, a service request type of case management is likely to be dominated by PCM features while handling of incident management cases are likely to demand significant amount of ACM features. Of course, it is ultimately the solution designer's responsibility balance the "cost-quality-time" considerations, select the appropriate features from the relevant classes of solution, and choose the right starting point, i.e., opt for a fresh new development project or modify or extend existing systems.



Figure 2.10 Cross-mapping of PCM, ACM, and dynamic systems.

2.4 Process Management and Case Management: Friends or Foes?

In Chapter 1, while discussing the history of case management, we remarked that the basic idea of case management is over a hundred years old-church-affiliated social workers were already using similar strategies trying to keep track of diverse activities as they tried to help the poor and the needy. Since then case management has evolved significantly gaining a formal recognition as a paradigm of organizing, executing, and managing work activities; application of case management has been spreading across diverse industries and organizations as an enabler of knowledge work that can in some situations be even unpredictable; with advances in applications that are capable of handling a wide variety of events and content interest in case management is increasing very rapidly. In the meantime, BPM, a discipline to structure and optimize work activities with the promise of providing great efficiency and competitive advantages to businesses, has also risen to a high popularity level—we will be referring to this as "process management" in this section. As can be expected, there are strong proponents of each of process and case management paradigms. We would not be surprised if you either read articles or saw presentations by leading industry practitioners discussing the strengths and future development outlook of process and case management and concluding that we just need one of process or case management—you may be confused as to which paradigm to put your bet on. Our goal in this section is to compare and contrast process management with case management paradigms, to seek answers to questions like "do we in fact need both process and case management?" and "must we have to choose one or the other, or is there a meaningful middle ground?," and to explore how such answers could benefit concerned solution architects. We want to point out that in this section by process management solutions, unless otherwise mentioned, we would mean both process-based capabilities embedded in packaged solutions or traditional BPM applications built using middleware BPM tools.

2.4.1 Anatomy of a Journey

"Life is a journey not a destination," a phrase that is often part of popular quotes, first used by theologian Lynn Hough in the *The Christian Advocate* periodicals in 1920 but commonly misappropriated to the nineteenth century American essayist and poet Ralf Waldo Emerson, encourages us not to rush through life but to savor every moment of it. Proceedings to solve business problems are also well comparable to journeys except that destinations, i.e., the goals that need to be achieved as part of solving the business problem, are important. Journeys are often planned and sometimes such plans need to be modified to accommodate unexpected events. This "journey" metaphor seems quite suitable for capturing and comparing some of the essentials characteristics of process and case management.

A few years ago on an early winter afternoon in Germany, I (Manas, one of the authors of this book) was scheduled to fly from Munich to Dusseldorf. It takes about an hour in a nonstop flight to reach Dusseldorf from Munch, there are some 25 daily connections between these two cities, and a large volume of passengers travel in this route that include a sizeable number of daily commuters. Flying from Munich to Dusseldorf normally is a routine activity especially for business travelers like me. Ordinarily there would be nothing worth mentioning about such a trip except that on that day due to heavy snowfall Munich airport had been closed! It was all well planned—my travel agent had booked and confirmed my flight many days in advance, I was at the airport well before the scheduled departure with a boarding pass that I had obtained the night before via mobile check-in and I was just looking to go through the rather mechanical process of

passing through security checks, boarding the plane and hitting Dusseldorf as planned—there wasn't anything that I was to do specially for this trip—then the airport closed! It was important for me to accomplish my goal of reaching Dusseldorf that evening so undeterred by happenings at Munich airport I started exploring other options. I found out that some of the trains were still running, that I could get to Frankfurt by train and then could reuse the value of my original ticket to fly from Frankfurt to Dusseldorf. A bit more investigation told me that while Frankfurt airport was still operating there could be flight cancellations there as well if the snow continued to fall heavily. I decided to take the chance of traveling via Frankfurt and catching a flight from there, however just to have another option I also booked myself a rental car at Frankfurt. By the time my train arrived in Frankfurt, as feared, flights out of Frankfurt were cancelled due to snow but then I had my rental car booked already that I could use to drive to Dusseldorf. It was not an easy drive and I reached Dusseldorf several hours later than I had originally hoped but I achieved my goal as good as it could be achieved under the circumstances. If we analyze my travel from Munich to Dusseldorf, the planned version would correspond to executing a traditional BPM process—I was to execute a fixed set of routine activities and the plan would have taken me to my goal of reaching Dusseldorf. However, the actual travel corresponds to what happens in a typical case management scenario. Due to unexpected events I had to improvise; I used my knowledge and experience of traveling in Germany to change my plans several times, setting a different set of subgoals and finally achieving my target goal. Interestingly, for each subgoal I was picking from options that already existed, hence, this modified travel would correspond to a PCM style case management (Figure 2.11).

Now, consider a little more exotic travel experience—the one that "Pi" Patel, a 16-year-old boy from Pondicherry, India, experienced as he drifted across the Pacific Ocean in a lifeboat with a bunch of zoo animals after the ship he was traveling on from Philippines to North America sank. This is described in a now famous fictional novel by the Canadian author, Yan Martel, in the book Life of Pi whose adaptation directed by Ang Lee won several awards in 2012 including four Academy Awards. In short, Pi Patel escaped the sinking ship in a lifeboat where he found himself sharing the tiny vessel with a hyena, a zebra, an orangutan, and a tiger—these were some of the zoo animals that were also on board the ship that sank. Through a series of events including the hyena eating the zebra and the orangutan only to be finally devoured by the tiger, Pi was left to find ways to deal with a tiger in a lifeboat while fighting extremely harsh elements of the ocean. Pi had no experience in survival at sea or taming a tiger but using his basic knowledge and experience, and some trial and error, he figured out many strategies to beat the odds to keep him alive until circumstances led to his rescue, and he was ultimately able to reach Canada. Clearly, in this journey while the final goal, i.e., that of Pi reaching North America was ultimately achieved, the path to that goal bore little similarities to the original plan. Pi had to set many new subgoals and learn or discover new tricks to achieve them before being able to arrive at the final and target destination. This is in the same spirit as what knowledge workers would do in ACM-style case management.

2.4.2 Process Management vis-à-vis Case Management

"Plans are worthless, but planning is everything"—this was a statement that Dwight D. Eisenhower, 34th president of the United States had made in 1957 in a speech to the *National Defense Executive Reserve Conference* in Washington, DC; Eisenhower made this statement in the context of preparing war strategies where one is expected to encounter many unexpected events. This statement essentially points out that, unlike predictable situations where well-designed plans work great, in the face unpredictability holding onto a predesigned plan is not useful instead



Figure 2.11 (a) Planned versus (b) actual trip (after airport closures).

focus must be given to the ability to plan based on the evolution of events. Interestingly, this comment is also quite relevant in the context of a comparison between process management and case management. Process management excels in implementing a "plan" to execute business activities efficiently; case management facilitates knowledge workers to make or alter work plans depending on how the case evolves. Of course, as we have discussed earlier, the degree to which knowledge workers will able to or will be allowed to modify their work plan depends on whether we restrict ourselves to PCM or embrace all that ACM can offer.

While comparing process management with case management solution, we have already mentioned that they differ greatly in the flexibility of work execution that they offer—traditional BPM being quite rigid and ACM being the most flexible. In fact, Michael Poulin's 2012 article (http:// www.infoq.com/articles/purpose-case-management, 2012) goes through a little logic exercise to establish that "flexible process" is an oxymoron-the "structured" process model that is at the heart of process management and the one that is the key to the efficiency that process management offers is too restrictive to provide work flexibility. Of course, this is a relevant concern if we indeed need that kind of flexibility. Here, we want to explore the modeling implications of another important aspect of knowledge work—"collaboration." Collaboration among case participants is critical in many case management applications, and collaboration implies interaction among these case participants. If we want to keep track of the details of such collaborations, then the solution platform needs have a mechanism that can follow the interactions that underlie these collaborations. In case of a traditional process-based solutions, whether embedded in a packaged solution or built using a middleware BPM tool, this would mean creating additional explicit traces, i.e., paths in the graph that govern the flow of execution of activities to mirror these interactions. Will this be an issue? Let us consider a situation where there are three interacting parties, "A," "B," and "C" (see Figure 2.12). If we explicitly count the possible ways basic interactions can take place amongst these parties, what we have denoted as "Elemental Interactions" in Figure 2.12, we get a count of "15." In fact as the figure shows, we can come up with a general formula for "Total Elemental Interactions" (TEI) using permutation formulas as a function of "N," where N is the number of interacting parties. It is not important for us to derive the stated TEI formula (although readers with a bit of math inclination should find this easy to do); instead, what is important is for us to appreciate how fast TEI grows with N (see the chart at lower right-hand corner of Figure 2.12); the TEI count is already 64 for N = 4! What we are trying to show through these calculations is that if we use a directed graph to model all these interactions, as would be the case with traditional process-based solutions, the process graph would become exceedingly complicated. While TEI count reflects the complexity of possible interactions in a particular instance, only a few of these interactions would be required during the resolution of a specific instance of the business problem. If, however, we cannot predict which few interactions would be required or if the set of required interactions vary from instance to instance, in a process-based solution we will have to model all or most of the interactions a priori, i.e., we have to carry the burden of this complexity. Now, if we construct the solution in such a way that it will have the ability to create the necessary connections between interacting parties at run time, then we will avoid this interaction complexity and, in particular, the rapid growth of this complexity as the number of interacting parties grows. Case management solutions are based on an event-driven paradigm and the interactions are triggered by



Figure 2.12 Rapid growth of interaction complexity.

and result in events. Thus, case management style of solutions are much better suited for tracking collaboration activities where the interaction ability is inherent in the solution and explicit design-time modeling of all possible interactions are not necessary.

The radar chart in Figure 2.13 compares traditional BPM with case management along eight dimensions. In this figure, by case management we mean the combined features of both PCM and ACM styles of case management solution—this is not unreasonable since many real-life case management solutions will end up combining features from both PCM and ACM styles. Essentially this figure depicts the following:

- 1. In process management, generally speaking, runtime changes to the system definition are not allowed, unlike as may be typical for case management especially the adaptive variety. Whenever the process needs to be changed, it will have to be done as a design revision and a new version of the solution will have to be created and deployed. In some BPM tools, for example Oracle's BPM Suite, certain amount of change in business rules are allowed on the runtime version of the process model that can take effect without an IT-level project deployment—this adds a limited amount of runtime modification within an otherwise traditional BPM environment. Of course, and as we have discussed before, an ACM implementation will be expected to handle run time changes to many more solution features besides the business rules.
- 2. Case management solutions are designed to handle nonroutine and unpredictable work that can become necessary due to some emergent events as the case progresses while process management solutions specialize in routine work that are all fixed at design time with worker roles pretty rigidly assigned.
- 3. In process management, all critical decisions about work flow and work assignment are taken during the design phase and are captured in the process model via different types of activities and their sequence augmented by business rules. In business problems targeted by



Figure 2.13 Differences between process and case management (PCM+ACM) solutions.

case management, case workers are expected to utilize their knowledge (hence the coinage "knowledge worker") to decide what actions to take next and who would be best candidates to execute them.

- 4. While some use cases require somewhat long running processes, most process management solutions are designed to provide fast responses; case management use cases, on the other hand, generally span relatively longer time spans typically in the order of days and weeks, and in some special cases like in immigration or legal cases could also be in the order of months and years.
- 5. Data and content are important to both process and case management solutions; however, in process management these are mostly structured, are "tagged along" with the process, and generally provide supporting role. On the other hand, in case management data and content have much wider variety and often drive the case progression.
- 6. Collaboration enablement is an essential feature of case management solutions and case workers are expected to utilize collaboration to resolve complex issues at runtime. In process management, such runtime collaboration is usually not needed since process workers are mainly focused on completing their assigned activities.
- 7. Case management solutions are fundamentally based on event-driven framework and are equipped to handle a wide variety of events including the "emergent" ones, i.e., events that appear as a consequence of peculiarities of the case some of which may even be unexpected. Advanced process management solutions do handle a limited variety of events, which are typically provisioned for at design time.
- 8. The problem-solving style employed in case management solutions are "goal-driven," i.e., the activities that the case workers execute and the decisions that they take are aimed at reaching some goals set by the type of case they are working on—goal achievement thus becomes the thing of primary importance with the activities playing supporting role. Process management solutions are "activity driven" and the process model ensures that target goals are achieved as long as the activities are completed correctly in a timely manner.

While the previous transcript has identified many categorical differences between process management and case management solutions, there are also many similarities between the two styles of solution paradigms (see Figure 2.14)—this is what we will focus on next.

The ultimate purpose of both process management and case management solutions is to solve business problems, albeit targeting different problem domains and solving them in different styles. Regardless of the actual mechanism used in solving a problem, businesses have certain generic

Similarities
Task/process modeling
Rules and policies
• Uls and dashboards
• Analytics
Governance and audits
System integration

Figure 2.14 Common features and capabilities in process and case management solutions.

requirements that they expect out of enterprise-grade software applications. Visibility into status and activity-level details of the problems that are being solved or have been solved in the past, for example, is the key to a business making sure that the business is functioning at the right level of efficacy and efficiency. Thus, a variety of dashboards and analytics appropriate for workers and managers are just as essential in case management as they are in process management. The requirement of maintaining adequate traceability is also common in both styles of problem solution. Audit trails to provide traceability are created as work activities steer a problem to a solution—this is done either for regulatory compliance purposes or as reference should someone, and this can be an employee or a customer, need to revisit work done up to a certain instance. Business rules are an easy way to capture organizational knowledge and mandates, and whether they are used for providing guidance or for enforcing policies, rules are considered important and essential components of both process and case management solutions. Typically, case management solutions involve higher amount of human work than process management solution, however, high-quality user interfaces (UIs) and easy to understand modeling of tasks are desirable in either solutions whenever human work is involved. As we have discussed earlier in this book, often a case management activity may turn out to be a process (in the traditional sense) leading to the same requirements as would be with process modeling for process management solutions. System integration is another area of commonality between process management and case management solution styles. Adequate integration of information sources is needed to provide users of both process and case management solutions with access to right information at the right time. Additionally, from time to time as a result of activities being performed in either type of solution, information in systems of record applications (e.g., applications like ERP, HCM, SCM and CRM that are custodians of business records) need to be updated in a timely manner—application integration capabilities are essential in such situations as well.

While we see many categorical differences between process and case management solution styles, we also note a large overlap in functionalities between them. So, how would we answer one of the key questions that we asked ourselves in the beginning of this section: Which one of these two do we really need? In a 2012 blog post (see: http://isismjpucher.wordpress.com/2012/01/04/ acm-and-bpm-a-battle-of-hemispheres/; accessed February/2014) titled "ACM and BPM: A Battle of the Hemispheres?" Max Putcher used the contrast of analytical versus creative traits of left and right brains to compare BPM and ACM finally remarking that we need both. Putcher's conclusion seems almost obvious, even intuitively. The real question for a solution designer is how much of each style to include in a solution to create the right balance of efficiency and effectiveness while keeping in mind all the applicable quality-cost-time constraints. Another related question we need to consider is whether process management and case management styles are so different that separate software platforms are needed to support these solutions. Case management purists strongly support separation of platforms proclaiming that any attempt to create platforms that would support both styles could only do so poorly. What would pragmatists say? We will explore this question next.

2.4.3 Case Management as Part of a BPMS?

Ever since there have been profit or growth motivations and competitive pressures, businesses have been after strategies and technologies that could help with their chores more efficiently and effectively. Process management solutions in the form of workflows embedded in ERP, HCM, SCM, or CRM like packaged applications focused on making routine business work faster and less error prone have been in use now for more than four decades. Process-based applications
using middleware BPM tools have been enjoying increasing interest and adoption for the last 15 or so years-these solutions targeted cross application, cross departmental, and "white-space" (i.e., functionalities not covered by packaged apps) activities. Occasionally middleware-based process management solutions compete with workflows in packaged solution but mostly they are complementary; they have also strengthened each other in terms of evolution of their functionalities. Case management is the current surge in technology paradigms aimed at handling business activities that are goal driven, involve substantial amount of content handling, collaboration, and knowledge work that can sometimes be unpredictable. Given the shared focus of both process and case management on handling business activities, it would be a natural question to ask if traditional process management tools and platforms are relevant for building out case management platforms. The way of working with case management has many differences compared to process management but does this mean we have to create brand new software platforms for case management or could we suitably extend process management platforms to handle these differences? After all there are also many similarities in requirements and functionalities of these two solutions as we have described earlier. Will existing process management platforms provide a better start for case management platforms? This is where the "purists," i.e., those who believe that case management solutions especially ACM are so different that they must have their own platforms, and the "pragmatists," i.e., those who are more reconciliatory in their approach are at severe odds with each other (Figure 2.15). The purists point at poor attempts of creating ACM solution using existing platforms to bolster their conclusions. However, good understanding about ACM is still immature in the industry and most tools are in their early versions begging the question if these conclusions by the purists would bear out even a few years from now. There is a popular idiomatic expression that says "a bad workman always blames his tools"-could it be that the way these "poor" case management solutions were conceived and used were faulty to begin with and no tooling should take the full blame for the ill outcomes? Besides is there a strong craving for pure ACM solutions in the marketplace? Surely there are some situations where the complete case management solution



Figure 2.15 Process and case management platforms: Merging and separating forces.

is fully adaptive. However, from our experience in working with customers majority of the case management solutions that we will view as mainstream at this time require a relatively small amount of adaptivity, almost as extensions to the rest of nonadaptive parts of the solution—should the functionalities of software platforms used to build these case management solutions not reflect a similar mix of adaptive and nonadaptive functionalities.

If we were to consider starting a platform for adding case management features, a full-featured Business Management Suite (BPMS) would be a good choice. BPMS is a coinage by the analyst firm Gartner popularized during 2009 and 2010 in the context of their Magic Quadrant (MQ) reports for BPM vendors. A BPMS is an integrated environment with broad capabilities to model and manage all aspects of BPM. Figure 2.16, excerpted from a 2010 post by Janelle Hill of Gartner (see: http://blogs.gartner.com/janelle-hill/2010/04/22/do-you-understand-the-differencebetween-workflow-and-bpm/; accessed February/2014), lays out the key components of a BPMS; clearly the capabilities of a BPMS as depicted exceeds what are needed to support "traditional BPM." Regardless of whether it is for faring better in Gartner's MQ or in response to customer needs, leading BPM software vendors have been pushing their BPM platforms along the lines of what is expected of a BPMS that has resulted in pretty rich offering for BPM platforms. How relevant are the BPMS components to case management platform that would include both PCM and ACM capabilities? If we recall the description in Chapter 1 regarding core and auxiliary system components for case management as well as reflect on various discussions so far about what is important for successful case management solutions, project the support necessary from a software platform for case management, and compare the result with the BPMS components in Figure 2.16, we would conclude that a full-featured BPMS will provide nearly every feature that case management solution would require. We would like to point out that content management requirement in case management solution is usually much stronger than process management.



Figure 2.16 Key components of a Business Process Management Suite (BPMS). (Courtesy of Gartner, 2010.)

In fact, a good lot of the features of enterprise content management systems including cataloging, check-in/check-out and revision management facilities, content sharing, access control management and archiving frequently show up as critical requirements in case management solution. This would mean that if we were to use a BPMS as a foundation for case management solutions, then it needs to either include or be seamlessly integrated with a full-feature content management system. We would also like to note the criticality of event handling, rules and collaboration in case management especially in the ACM realm. As opposed to a graph-based sequencing for activities as in process management, in ACM activities are usually triggered by emergent events and rules are used, besides encoding business logic, to guide case worker involvement such as which activities are possible at a given instance and who are allowed to execute them. Knowledge work at times needs the capability of on demand collaboration, for example, a case worker reaching out to a subject matter expert to resolve a unique issue related to the case. Thus, a BPMS targeted for building out a case management platform needs to have an event-handling infrastructure at its core, an integrated rules engine, and adequate facility for on demand collaboration.

From a pragmatist PoV, there are additional benefits of being able to extend a BPMS to handle case management. As we have already, seen there is a lot of overlap of functionalities between process and case management solutions. Also, a complete case management solution is likely to need a certain amount of human tasks and prebuilt processes as well as will require a mix of PCM and ACM capabilities. In certain cases, we may even include case management features of packaged applications like CRM for part of the overall solution. A case management platform that is part of a BPMS will natively handle most of these challenges. In addition, many of the BPMS platforms are now quite mature from a deployment perspective, i.e., they are able to handle large loads with high reliability and good performance characteristics—these properties will directly benefit case management solutions as well if the case management platform is built on top of an established BPMS. Some of the purists have feared an ACM platform built on BPMS would cause the resulting solutions to be too complex. This is a valid concern; however, it is the tooling environment of a particular BPMS and the design of a specific ACM solution that will dictate how easy and useful the solution will be for its end users. Considering all of these arguments, we can see that with proper strategy it is possible to create a quality case management platform on top of a full-featured BPMS; later in the book we will be discussing how this has been accomplished in Oracle's BPM Suite platform.

2.5 Adoption of Case Management: Capability Maturity Considerations

Let us say that you are a good swimmer and do impressive laps in the biggest pool in your city. Would you set out to swim across the English Channel? And, if you are contemplating on doing so, you would surely ask yourself if your swimming "capabilities" like skill and endurance are "mature" enough to give it a shot. You will ask such questions to eliminate (or at least greatly reduce) the chances of you drowning in the English Channel. Successful adoption of a technology or solution paradigm demands that we ask similar capability maturity questions relevant to the specific technology or solution else we run up the risk of failure much like the (naive) swimmer drowning in the English Channel. In the software engineering realm, work done at the Software Engineering Institute (SEI) at Carnegie-Mellon University in Pittsburg, Pennsylvania since 1980s supported by U.S. Department of Defense focused on improvement of software quality. This work has resulted in the development of a five-level Capability Maturity Model (CMM) followed by the more recent Capability Maturity Model Integration (CMMi) aimed at articulating the ability of an organization to develop, deliver, maintain, and improve software applications and platforms; these have been commonly at the heart of most (capability) maturity models in the context of SOA and BPM including those from Oracle. These maturity levels are designed in such a way that the higher the maturity level the more complex projects an organization is capable of handling and yielding higher benefits from them with better certainty.

In the context of technology adoption, maturity models are useful for an organization to assess their "as is" and "to be" competencies in the required capabilities, identify the gaps between the two (i.e., as is and to be states), and plan activities in order to close those gaps. The understanding of maturity levels and the associated capabilities are also useful in analyzing the competencies needed to execute projects successfully and within accepted risk levels. For example, it is conceivable that an organization is, on an average, at level 2 of an applicable maturity model yet an important project requires certain features that demand a few of the capabilities to be at higher levels in order to assure project success. Such information if available in the early stages of a project can be immensely helpful to plan appropriate contingencies in a timely manner and prevent project failures. The idea and use of maturity models are now fairly common in more mature technology areas such SOA or traditional BPM. For case management, however, this is an area of active research. Given the scope of this book, we are not going to critique ongoing efforts in this area instead we will use Oracle's five-level BPM Maturity Model (see: http://www.oracle.com/technetwork/topics/entarch/oracle-pg-bpm-roadmap-r3-0-176693.pdf; accessed February/2014) along with the individual capabilities that have been identified for this model to explore how it can be leveraged for case management. At this point, we also want to thankfully acknowledge significant benefits relevant to content of this section that accrued from extensive discussions and collaboration between one of the authors of this book (Manas) and Danilo Schmiedel of Opitz Consulting, an Oracle partner in Germany.

2.5.1 "Generalized" BPM

As we have discussed earlier in this book, both traditional BPM and case management are strategies to organize, execute, and improve business activities in an organization. We have also seen that while these two strategies have certain clear differences in the way solutions are designed and used they do have a lot of similarities. This consideration leads us to approach case management maturity model as something that could leverage established knowledge assets of traditional BPM maturity models. To start with we want to restate the fact that the collection of business activities fall under two distinct categories (see Figure 2.17):

- 1. Routine work-done well with structured and completely predefined process models
- 2. Knowledge work—requires handling of unstructured and unpredictable tasks

In order to capture both of the varieties of business work, we introduced here the notion of "Generalized BPM" that is meant to include both the styles. (Ideally we would have liked to call this "Business Activity Management;" however, the resultant acronym, "BAM," would conflict with the prevailing notion of BAM that describes the monitoring of business activities, hence, the suggestion of Generalized BPM.) This Generalized BPM is both an approach and a formula for designing and implementing business solutions; in this context the word "process" is used loosely, i.e., we do not impose any restriction of structure or predictability as would be the case in traditional BPM rather it simply indicates all the "proceedings" required to solve the business



Figure 2.17 Generalized BPM to cover traditional BPM and case management.

problem. Thus, as an approach Generalized BPM will explore a given problem without any preconception as to whether the solution would be a traditional BPM one or a case management one and will let the discovery process answer this question. As a solution paradigm Generalized BPM will facilitate design and implementation of applications that would combine features of traditional BPM and case management, as necessary (see Figure 2.17). With this background, we are now ready to investigate capabilities and maturity levels for case management as a component of Generalized BPM.

2.5.2 Maturity Levels

Capability maturity levels are meant to indicate a measure of competency—the higher the level the higher the competency, i.e., the ability to do something well. Figure 2.18 is a slightly modified version of Oracle's BPM maturity levels that was developed following CMMi's maturity model. Keeping the "strategic goals" essentially the same as in the original version we have introduced small changes, mostly additive, in the tactical plans' column to reflect some of the important case management adoption goals, thus turning these maturity levels appropriate for Generalized BPM. Maturity levels are additive with respect to competencies in individual capabilities—a given level will include all the competencies in the lower levels. The five levels of maturity reflect:

- 1. *Ad Hoc*—Suitable for doing small or pilot projects this level indicates mainly exploratory activities, often done mainly by small IT initiatives, to "kick the tires," i.e., to "get a feel" for the BPM technologies. At this level of maturity, there is no established or generally agreed methodology or approach for BPM adoption.
- 2. *Opportunistic*—At this level an organization looks for a few real projects, selected in an opportunistic manner, to prove delivery of business value. Typically both IT and business groups are involved in selecting the projects however approach and methodologies to plan and execute such projects are not fully standardized.
- 3. *Systematic*—By this level an organization has selected and standardized their approach and methodologies for planning and executing projects that are generally followed in all projects.



Figure 2.18 Generalized BPM maturity model (based on Oracle's standard BPM Maturity Model).

Technology adoption is on a successful growth path by this level and complex and mission critical projects can be safely undertaken within acceptable risk limits.

- 4. *Managed*—At this level, capabilities are measured via a set of agreed metrics so as to evaluate how they compare with expected efficacy and efficiency so that any improvement can be systematically planned and executed. Technology adoption is quite significant at this level.
- 5. Optimized—This is the final level of maturity where appropriate balance of effectiveness and efficiency in approach and methodologies for planning and execution have been achieved and are stable. All assets related to technology adoption are proactively monitored, kept current and improved, as needed. At this level, an organization does not worry about technology adoption instead is focused on deriving strong competitive advantages from it.

As we have already alluded to, it is of course conceivable and in fact quite likely that an organization may want or need to undertake projects where certain activities demand a higher level of maturity than the current maturity level of the organization for the relevant technologies. For example, a case management project that from a basic functionality PoV may fit well at the opportunistic maturity level yet may require monitoring of a few system characteristics and KPIs that may correspond to a managed maturity level—this would imply a special attention for those higher level capabilities within the scope of the project.

2.5.3 Maturity Model Capabilities

Besides maturity levels a maturity model covers a set of capabilities that enable the organization gather competencies to accomplish its goals, say, technology adoption in this case. In fact, it is the maturity of these capabilities that aggregate up to the overall maturity level. Here we will adapt Oracle's traditional BPM maturity model capabilities—we will extend them as necessary to account for case management and thus making the enhanced set valid for Generalized BPM (in fact in this section all references to BPM pertain to Generalized BPM unless otherwise stated).



Figure 2.19 Generalized BPM capability domains. (Adapted from Oracle BPM Maturity Model.)

These capabilities can be grouped together using their thematic similarities. In the Oracle BPM maturity model, such a grouping is done by creating eight capability domains half of which are technology dominated and the other half concerns organizational disciplines (see Figure 2.19). These capability domains address

- 1. *Business & strategy*—contains capabilities that provide the high-level constructs that allow the BPM initiative to proceed. This includes such things as business motivation, expected benefits, guiding principles, expected costs, funding model, etc.
- 2. *Architecture*—contains capabilities concerning the definitions of the overall architecture and guidelines for various practitioners to ensure adherence to relevant architectural principles.
- 3. *Infrastructure*—contains capabilities related to the service infrastructure and tools that provide the technical foundation for the BPM initiative.
- 4. *Information*—contains capabilities focused on the information aspects of BPM, e.g., providing Information as a Service (IAAS). This includes shared data models, message formats and schemas, master data management, content management, etc.
- 5. *Projects, portfolios & services*—Contains capabilities concerning the planning and building of BPM applications and required services, and the service usage guidelines of service consumers.
- 6. Operations, administration, and management (OA&M)—Contains capabilities relevant to post deployment aspects of solutions based on a service-oriented architecture, i.e., the operations, administration, and management aspects of Generalized BPM solutions.
- 7. *Organization*—Contains capabilities required for the development of corporate competency around BPM including the organizational structure and its evolution (for example, from siloed to integrated and collaborative organization), and skills development.

8. *Governance*—Contains capabilities concerning the governance structures and processes that support and guide the BPM efforts. Generalized BPM governance includes end-to-end process/case owner roles as well as goal/KPI-driven solution development. Maturity and adoption of an adequate amount of governance is a leading indicator of the overall Generalized BPM success.

Even though these capability domains were originally constructed from the individual capabilities applicable to traditional BPM adoption, we have left the domains unchanged here due to their full applicability to Generalized BPM.

The eight capability domains depicted in Figure 2.19 have included around 70 individual capabilities in Oracle's original BPM maturity model. Since Generalized BPM combines both traditional BPM and case management, all of these capabilities remain equally relevant for Generalized BPM. However, in order to address competency requirements unique to case management, we have added a few more capabilities to the original list—the final list is displayed in Table 2.3; these additions are pointed out with a "**." Besides these new additions, there are some capabilities from the original list that need special attention in the context of case management—these capabilities are marked with a "*" in Table 2.3. Assuming the reader's general familiarity with capabilities related to traditional BPM, we will skip explaining those instead we will visit each of the capability domains and devote some comments regarding capabilities that are either new additions or have been marked as needing special attention due to inclusion of case management.

- 1. Business & strategy-In this domain, we have added three new capabilities: "Guiding Principles," "Process Mining," and "Pattern Detection and Template Definition." The Guiding Principles competency feeds case management solutions with rules and helpful tips that knowledge workers may need in order to pursue case resolutions; generally speaking these are a mix of mandated and suggested actions that the case participants will have to act on or will choose to act on to drive cases to completion. Process Mining is a capability that is useful in the discovery phase of Generalized BPM solutions where, in absence of complete and coherent knowledge of exactly how an organization solves business problems, prevailing activities are "mined," i.e., collected and analyzed in order to trace actual execution paths that are typically used (see www.processmining.org for some more description and links to the underlying philosophies are techniques of process mining); these traces are then used as input to the solution design process. In Pattern Detection and Template Definition, an organization matures its ability to extract solution patterns at various levels (of the solution) that is helpful in deciding what mix of traditional BPM and case management a solution should contain plus additional details that need to modeled or provided for in the solution. In this domain, we have also marked "Business Process Analysis" and "Process Improvement Approach" as those needing special attention in a case management solution setting. In case of Generalized BPM, Business Process Analysis capabilities need to approach business problems first without any preconceived bias towards either traditional BPM or case management and should allow for the naturally existing balance of the two to emerge. In Process Improvement Approach, in order to cater for case management, besides traditional BPM improvements of attributes like levels of automation, elapsed time or resource requirements, improvements in decision making and stakeholder collaboration need to be included.
- 2. Architecture—This domain is concerned about various architectural issues including usage of patterns and standards (see examples of case management patterns later in the book) and modularity and style of solution construction (in the context of case management this

Business and Strategy	Architecture	Infrastructure	Information	Projects, Portfolios, and Services	Operations, Admin, and Management	Organization	Governance
Business driven	Diverse process types*	Access to enterprise functionality	Process data*	Enterprise functionality portfolio mgmt	Activity monitoring*	Business and IT collaboration	Enterprise functionality
Business process analysis*	Expose enterprise functionality	Service discovery	Message exchange data	Coordinated modeling	Process control*	Business and technical modeling skills	Business rules governance
Business justification	Process participants interaction*	Unified repository	Business process performance management	Business rules abstraction	BPM infrastructure dependency tracking	Executive sponsorship	Business process change management
Business architecture linked	Closed loop	Human interaction	Process definition storage*	Business process identification	Business process release mgmt practices	BPM competency center	Standards vitality
Business automation	Versioning strategy*	Document management*	Process content management*	Business process change mgmt*	Business process exception handling*	Education and training	BPM Infrastructure governance
Process improvement approach*	Accessability of business functionality	Process modeling*	Data permission management**	Process portfolio mgmt		Work structure	Business process lifecycle governance

Table 2.3 Generalized BPM Capabilities (Extended from Oracle BPM Maturity Model)

(Continued)

Business and Strategy	Architecture	Infrastructure	Information	Projects, Portfolios, and Services	Operations, Admin, and Management	Organization	Governance
Value chain	BPM security architecture	Process execution*	Case documents**	Process- centric approach		Process change authority	Knowledge (data) governance**
Measurement	System integration and collaboration**	Process views and round- tripping*		Process capture		Process change delivery	Stakeholder governance**
Innovation	Event architecture**	Modeling tools*				Process worker collaboration protocols**	Activity governance**
SOA	Decision management	Business rules**					Policy governance**
Guiding principles**	Stakeholder change mgmt**	Event processing**					
Process mining**	Rules and policy management**	Identity management**					
Pattern detection and template definition**	Data change management**	Predictive analytics**					
	Dynamic activity configuration**						

Table 2.3 (Continued) Generalized BPM Capabilities (Extended from Oracle BPM Maturity Model)

would also mean moving from large complex processes to simpler "atomic" and "reusable" processlets that are invoked as guided by case progression). In this domain, we have added "System Integration and Collaboration" and "Event Architecture," "Decision Management," "Stakeholder Change Management," "Rules and Policy Management," "Data Change Management," and "Dynamic Activity Configuration" to traditional BPM capabilities. Typical case management systems need to facilitate much higher levels of collaboration among case participants and provide them access to higher varieties and sources of information—this makes the System Integration and Collaboration a necessary capability. Ability to conceive and implement event-based solution architecture is critical for case management since most of the case management activities are usually responses to some internal or external business event-typically events trigger execution of rules, which result in initiation of case activities that in turn generate further events, and so on. Suggestions like "next best action" are provided by a "Decision Management" capability. In adaptive variety of case management, case participants are given the flexibility to dynamically modify stakeholder list, some of the rules and policies, selected data elements, and the list of activities-these are provided by the "Stakeholder Change Management," "Rules and Policy Management," "Data Change Management" and "Dynamic Activity Configuration" capabilities. "Diverse Process Types" and "Process Participant Interaction" capabilities from the original list are marked here for special attention since in Generalized BPM processes, unlike traditional BPM, can be unstructured, process activities can be unpredictable, and participant interactions can be more significant and on demand. Case management solutions, especially in the initial stages, may undergo frequent revisions to reflect rapid learning that would take place from the feedback from actual case resolutions—this makes the "Versioning Strategy" capability especially important in case management situations.

3. Infrastructure—Here the word infrastructure encompasses both hardware and generic software capabilities, i.e., the computation enablers. "Business Rules," "Event Processing," "Identify Management," and "Predictive Analytics" are the new capabilities that we have added to this capability domain on account of case management. Business rules are a way to change the behavior of a business application based on a set of preconditions. While business rules are being used increasingly more and more to alter flow control logic for traditional BPM, for case management these have an even greater importance-they practically control "who is doing what and when." These rules are "fired" as certain events occur during case progression; thus, "Event Processing" capability becomes a necessary component of the infrastructure supporting case management solutions. As more experience is gathered around how to solve typical cases, a case management system can suggest "next best action" to case participants—this is supported, in part, by the "Predictive Analytics" capability. We have also marked "Document Management," "Process Modeling," "Process Execution," "Process Views," and "Modeling Tools" capabilities in this domain as "special attention" items. As we have described before, in case management solutions a wide variety of documents, typically much more than traditional BPM, are often involved in supporting and driving case participant activities. In case management, the modeling takes the form of specifying phases, milestones, and activities along with events and rules, at least at the higher levels of solution construction. This requires special support of modeling paradigm, tooling, and visualization. In presence of case management, process execution engine needs to handle execution of case activities and handling of all applicable milestone status.

- 4. *Information*—Since case management typically involves a significant amount, source, and variety of data and structured and unstructured content (includes traditional formatted and scanned documents, messages, etc.) handling along with management of privileges that case participants need to have as they interact with such data and documents, we have added "Data Permission Management" and "Case Document" capabilities in this domain. It is in the same spirit we have flagged "Process Data" and "Process Content Management" from the original capability list for special attention. We also want to note here that the capability "Process Definition Storage" in the original list now needs to also contain case definitions (aka case templates or case models).
- 5. *Projects, portfolios & services*—Capabilities in this domain remain generally valid for case management. We have marked "Business Process Change Management" capability mainly to indicate that in case management change management will also include changes in decision making rules and protocols.
- 6. Operations, administration, and management (OA&M)—All the capabilities in this domain are generally valid for case management as well. We have marked "Activity Monitoring," "Process Control" and "Business Process Exception Handling" particularly to remind us that in case management we are required to pay extra attention to human activities or human interaction aspects.
- 7. Organization—All capabilities in this domain are also applicable to case management. Additionally, we have added "Process Worker Collaboration Protocols" capability here since in case management often worker collaboration is important and organizations need to set the necessary guidelines as to the boundaries and expectations in such collaborations.
- 8. *Governance*—Case management requires all the governance capabilities of traditional BPM. However, since knowledge work and ability to modify stakeholder and activity lists and update certain rules and policies are also important to case management, we have added "Knowledge Governance," "Stakeholder Governance," "Activity Governance," and "Policy Governance" to the original capability list of this domain.

As per our definition, Generalized BPM is a union of traditional BPM and case management functionalities. By augmenting the traditional BPM capabilities with those needed especially for case management, we have created here a capability list that would now be able to support the maturity model for Generalized BPM and would be helpful in guiding related technology adoption.

2.5.4 Adoption of Case Management: A Case Study

Here we will briefly describe the ACM adoption journey at a midsize insurance company in Europe (henceforth labeled the "Company"). This Company has been in existence for a fairly long time, provides nearly a dozen different types of insurance coverage ranging from auto to life, and over time has amassed a sizeable customer base. The Company, until recently, relied on use of a set of "siloed" legacy applications to capture and process insurance claims and to handle associated customer interactions. These legacy applications had grown in complexity due to addition of features catering to emerging business functionalities and changes in rules and regulations. While some of the "old timers" who grew up with the legacy systems had reasonable proficiency in using these systems it was becoming quite difficult to train newer employees on them. These legacy applications were not integrated, thus forcing repeated manual data entry. Collaboration among workers handling claims was done over emails, phone calls, and hallway conversations that were very hard to trace back. Lack of information integration and poor collaboration led to frequent data errors,

prolific and expensive exceptions, service delays, and consequent customer dissatisfaction. Also, continual and often ad hoc modifications to large code bases of these systems had rendered them vulnerable to hard to discover coding errors as well as were difficult to change. The Company was hurting in terms of agility and quality in providing new and improved services—it was concerned about losing current and new business to competition. This concern led to explorations into possible technology adoptions that could drastically improve the current situation with their business supporting software systems.

The Company first did a set of experiments with traditional BPM tools and methodologies and found it to be generally difficult and often impossible to capture their claims management activities in a "BPM model." Through such experimentations, they realized that there were two main root causes that limited their success with traditional BPM approach:

- Lack of complete knowledge—Claims management, generally speaking, can be fairly complex and can involve many activities that need to abide by all applicable contract policies, business rules, and relevant government regulations. A fair amount of these policies, rules, and regulations were "hard coded" in their legacy applications. When queried about the execution sequences or paths followed by a set of business activities needed to resolve a claim, different claims processors provided different articulations. It was thus impossible to "sketch" a traditional BPM model with activities, flows, and rules with great confidence.
- 2. High variability—The Company offered insurance for a large variety of risks like home, auto, life, boats, travel, and precious stones and artifacts. For a given risk, they also offered many options with regards to claims admissibility criteria, deductibles, and so on; these choices were introduced to help the customers fine-tune their insurance policies. While the wide range of offers was certainly helpful to the customers, and in fact, contributed to the Company's attractiveness, they also created over a hundred sufficiently distinct types of claims to be processed. Within each such claim type, the actual actions taken by the claims processors and their managers varied, sometimes significantly, due to unique situations associated with the claim instances. Claims processors and their managers made many decisions based on their judgment and experience bypassing or ignoring what the standard operating procedures and the outcomes from their legacy applications would dictate—this made their work activities somewhat unpredictable. Thus even in situations where the Company employees could fully articulate all the actions that were ever taken to process claims the details were too overwhelming to model using standard BPM syntax and tools within reasonable modeling efforts and complexity.

In looking for suitable strategies to cope with these challenges, the Company decided to adopt process mining and ACM, the former to gain more comprehensive knowledge about the actual activities by analyzing past activity logs and the latter to handle the high variability and unpredictability in work activities.

Given that the Company already had a whole set of legacy applications and a "way of working," albeit not satisfactory anymore, many stakeholders, ranging from technical development teams to business owners started wondering ways a different paradigm such as ACM should be adopted. In trying to minimize significant interruptions to day-to-day business and to have the work force ease into a different technology and work style, a "phased" approach of technology adoption made the most sense. Besides, it was felt that a phased approach would allow the Company to become steadily "mature" in the capabilities needed for a successful ACM adoption. Partnering with a trusted implementation partner well conversant with Oracle SOA, BPM, ACM and UI



Figure 2.20 An ACM adoption example—from application silos to ACM in four steps.

technologies, the Company put together a phased adoption plan with four steps (Figure 2.20). Major work activities and technology adoption targets for these steps or phases were:

- 1. Step #1—this was a preparatory step. The Company wanted to understand better and create a knowledge base of how they actually performed their day to day work, how much variability in work actually did exist, and how often there were exceptions and special attention was needed. This was done by a variety of methods of process mining such as interviews, analyzing application logs, and monitoring pertinent events and their handling; they also used a process mining tool. Once they were sufficiently comfortable with the "as is" knowledge level, a round of analyses were done to identify groups of work activities that could be fully predefined and thus could be easily and efficiently handled by traditional BPM, and the rest. A high-level case management style discovery was then undertaken to make a tentative roadmap as to how functionalities should be moved from either packaged or legacy applications to more middleware-based BPM and ACM; a cost-benefit estimate for such moves was also done. The outcome of this discovery along with the cost-benefit estimates helped the Company to identify the "low hanging fruits" that could be undertaken in the near term deferring the rest to be distributed over future phases when the Company expected to gain higher levels of maturity. The Company also got a taste of event-based problem decomposition and case management solution formulation by conducting a few simple PoCs with the help of their implementation partner. Comparing with the Generalized BPM maturity model this step or phase would correspond closely to a "Level 1" maturity.
- 2. *Step #2*—In this step, the Company selected a small group of activities where the notion of case management was deemed to be helpful. These were not PoCs as in the previous step but represented real business problems; business users were involved along with the technical teams to articulate the problems and designing the key features of the end solutions. The activities typically represented some of the "white spaces," i.e., activities there were not (fully) covered by the Company's current applications and required ad hoc actions by the claims processors or their managers. The focus in this step was mainly on organization of work activities around the concept of case folders and to provide a new set of UIs to the claims processors to better access work items and supporting documents, and to have better visibility of the overall progress of claim resolutions. Certain levels of application and UI integrations were needed in order to achieve these goals. The Company chose a "service oriented"

integration strategy keeping in view possible reuse of the integration services in later phases. The key benefits of activities in this step were incremental improvement of work efficiency that could be demonstrated by using simple KPIs and the Company was now getting closer to a "Level 2" maturity as discussed in the Generalized BPM maturity model.

- 3. Step #3—While some of the key concepts of case management were already in place, the projects in the previous step did not include provisions for any runtime adaptivity, i.e., the abilities to either alter the work activities, their sequences or the work assignees. This was done on purpose since the Company wanted to limit the amount of change in the tools and work practices to a level that would not be outright uncomfortable for their existing workforce. In this step, however, the Company started extending the scope of work activity groups to include facilitation of some knowledge work, i.e., started providing ACM-style flexibilities so that in some situations the claims processors could use their experience or expert advice to choose work activities that would resolve their cases better or faster. Additional information and UI integrations were needed in this phase to provide the claims processors with broader and timely access to a variety of documents, work-in-progress reports as well as up-to-date case resolution status, and a more easy-to-use work environment. This phase also introduced collaboration portals to facilitate better team resolution of difficult problems. In addition to extending existing application capabilities, the Company was also targeting a few of the small and new projects using ACM; these projects helped the Company gain deeper experience in conducting case management discoveries and planning and executing case management projects using iterative and agile methods, and in training claims processors in using the runtime adaptivity and on demand collaboration to their advantage. By the end of this step the Company was climbing into "Level 3" of the Generalized BPM maturity model.
- 4. Step #4—Having successfully conquered the key challenges of adoption of case management technology and practices, the Company was now ready to undertake more complex and mission critical projects using ACM. In fact, they were now creating hybrid solutions that employed a mix of traditional BPM and ACM to achieve optimal level of efficiency and flexibility for business problems in scope, i.e., they were using the Generalized BPM approach. The Company started putting in more detailed measures and KPIs to monitor and improve the process of creating and using Generalized BPM solutions. Per Generalized BPM maturity model described earlier, at this phase the Company was operating somewhere between "Level 3" and "Level 4."

It should be pointed out that a maturity model is only a guiding framework to help an organization arrange development and fine-tuning of relevant competencies related to adoption of a paradigm. This sense also applies to Generalized BPM maturity model where required competencies for adoption of case management are included. As was the case with the company we discussed here, project requirements often drives the technology adoption scope and pace; however, being cognizant of required competencies to wholesomely raise organizational maturity can be very helpful in merging the tactical and the strategic goals.

2.5.5 Maturity Is More Than Technology

Since after all we are considering overall organizational maturity, in this case for adoption of Generalized BPM, becoming experts in relevant technologies only is not enough to raise the maturity—behaviors of participating humans individually and the relevant sectors of the organization must also evolve to embrace the associated practices. This concern is reflected in the spread of the

capability domains in our maturity model where about half the focus is technology driven and the rest involves organizational and business concerns. For example, when adopting case management, the participants need to change the way they assess and progress a resolution as compared to traditional BPM—in case management progression of the resolution is deciphered from information collected in case folders along with status of milestones (that collectively lead to the end goal) whereas in traditional BPM the process model and the current state is enough to indicate progress of the business transaction. Since some level of knowledge work is involved in case management, a typical participant, i.e., a case worker needs a relatively deeper understanding of the overall problem being solved and the organizational practices for solving them as compared to a typical participant, i.e., a process worker in traditional BPM setting where the process worker specializes in executing the types of (individual) tasks that he/she is normally assigned to. Typically, case management also assigns a "case owner" role that is responsible overall coordination of participant activities, interventions when necessary, and ultimately for a successful resolution of the case instance. In adaptive version of case management implementations, there can be a lot of on-demand team collaboration and just in time expert decisioning-these activities along with their governance cannot be automatically assumed to be present in a traditional BPM driven organizational culture and must be systematically developed through enablement, rewards, and control. Thus, alongside tools and applications, a set of human and organizational capabilities need to be also developed in order to reap higher benefits from case management or Generalized BPM paradigm.

2.6 Making the Business Case for Case Management

"There is no such thing as a free lunch"—a quote often attributed to the late Nobel Prize–winning economist Milton Friedman well captures the sense that everything that we do has a cost, whether direct or indirect. Why should the adoption of case management be any different? In fact, and like any other nontrivial initiative, adoption of case management has to be well justified in all but obvious situations. Since traditional BPM has now been a popular efficiency strategy for some time that is aimed at improving organization and execution, "Why should we adopt yet another paradigm such as case management to manage business activities?" is a legitimate question. The answer, of course, has two parts: technical enablement and business case. So far we have discussed many of the essentials behind technical construction and capabilities of case management; in this section, we will briefly visit the key considerations for creating a business case for case management.

2.6.1 General Considerations for Creating and Presenting a Business Case

Adoption of some new (or newer) paradigm to create capabilities in an organization will always partition people into groups who "like," those who "dislike," and those who are "indifferent" to the idea. Some of these people are likely to be the ones making the decisions about policies and resource allocation that would control whether the initiatives leading to the adoption will be undertaken. A well-composed business case is often critical to obtain a positive decision towards adoption commitments. Figure 2.21 shows the interplay of the dominant forces or the key considerations that directly impact the creation of business case that justify the necessary organizational commitment to initiate and execute an adoption journey. The capabilities resulting from the target adoption are motivated by a set of business requirements and hold the promise of delivering a set



Figure 2.21 Business case justification for capability creation and the key forces.

of business values that could include cost reduction, cost avoidance, revenue growth, compliance management, or creation of a better image for the organization. Those who propose or like the idea of creating such capabilities are usually excited about getting on with the related initiatives as quickly as possible and may sometimes not pay enough attention to the possible risks of failure of such initiatives; the naysayers are, of course, mostly focused on the risks ignoring the possible positive implications of the initiatives if the risks are managed adequately. A well-formed business case aims to rationalize the adoption decision process using objective arguments related to costs, benefits, and potential risks.

Effective creation and presentation of a business case should follow a well-understood format and must contain succinct information useful for deriving conclusions easily. While there are many formats that can be adopted for putting together a business case, the following topics are commonly included:

- 1. A description of the initiatives and their final goals
- 2. Business needs served by the proposed initiatives
- 3. The consequences of staying with the status quo
- 4. Key business benefits of the initiatives, both tangible and intangible, direct and indirect
- 5. Resource requirements including type, amount, and duration of resources
- 6. Financial metrics such as payback period, ROI, IRR, or NPV calculations
- 7. A comparison of merits and demerits of possible alternatives
- 8. Key assumptions and their impact on the analysis and arguments presented
- 9. Possible risks, their sources, and impact along with recommendations for their remedies
- 10. A list of critical success factors (CSFs) that would help ensure better success rate
- 11. A list of key performance indicators (KPIs) that can be used to measure and tune progress
- 12. A list of key risk indicators (KRIs) that could help tune or abandon execution, if necessary



Figure 2.22 A cost-benefit-risk representation of possible options.

In the event there is more than one option that needs to be considered, a cost-benefit risk representation of these options provides a helpful visual for decision making process. As shown in Figure 2.22, options 3, 4 and 5 are not desirable either due to high cost, low benefit, or higher than acceptable risk; option 2 represents a good balance of high benefit, moderate cost, and relatively low risk; option 1 indicates a low level of risk and cost but also provides low benefit—such options may be useful when an organization is starting to climb the learning curve of competencies required to plan and execution adoption initiatives.

2.6.2 Specific Considerations for a Case Management Business Case

With the outline of general considerations for creating and presenting a business case just presented, let us turn our attention back to case management, including the production and adaptive variations, to see what type of business problem indicators, expected outcomes for the business, and likely risk factors that may exist or should be considered for adoption of case management initiatives. Since we have already had several discussions around the technology-related concepts underlying case management, our focus here would be more from a business and organizational angle; we will focus mainly on business needs and financial benefits, costs and risk aspects of case management adoption since the rest of the items required for creating a business case for case management will follow naturally from these:

1. Business needs and financial benefits: Here we need to map a set of business imperatives against the capabilities afforded by adoption of case management and then translate the fulfilment of these business needs to changes in the top-line (i.e., revenue) or bottom-line (i.e. revenue-cost) of the organizations that can be reasonably attributed to case management adoption. We have already discussed how production and ACM solutions provide

significantly higher work flexibility than traditional BPM and facilitate knowledge work; PCM solutions will typically allow case workers to choose the right subset of activities, their execution sequence and work assignee from broader predefined sets while ACM will additionally allow the case workers to modify the work list and introduce stakeholders on demand. As mentioned already, it should be expected that many solutions will benefit from a combination of traditional BPM and case management (recall "Generalized BPM") and that with the right solution architecture and technology platform such hybrid solutions are quite feasible. The types of business problems that these case management or Generalized BPM will solve depend on the type of industry the business is in and further the type of product or services it offers.

For example, in financial services industry and in regulatory agencies most of the supervision of compliance adherence, fraud investigations, and special services for high net-worth investors are great candidates for case management. In insurance companies, complex claims management are better dealt with using case management. In utilities industry, while some of the activities in the area of outage and work order management can be handled by traditional BPM, more complicated situations, say, involving damage claims are more suitable for case management based solutions. With relatively higher familiarity with traditional BPM that currently exist solution designers may have high temptation to somehow fit business activities into traditional BPM framework and hope that a seemingly well structured workflow will provide business excellence even for these use cases. However, inherent rigidity in traditional BPM does restrict the ability for knowledge workers to exercise their experience and solve difficult problems creatively. Trying to adhere to fully predefined activities of traditional BPM can lead to solutions that are not effective thus leading to proliferation of exceptions, expensive rework, long delays in resolution of issues and loss of customer satisfaction. Also, in situations where the process definition evolves rapidly, an event-based decomposition of the business problem, as would be in case management paradigm makes it much easier to change the solution definition rapidly. For the purpose of creating a business case for case management, one needs to estimate, using any suitable calculation method, the financial benefits of being able to do work more effectively, being able to handle rapid change in business need or raising customer satisfaction for applicable use cases like the ones we just mentioned. Of course, translation of some of the case management benefits to financial gains like cost reduction due to more accurate and effective issue resolution would be easier to calculate while the financial value of generating long-term customer loyalty would be trickier to compute. In Section 1.4.5, we have presented a discussion on application of case management across many industries-this could be a good starting point when looking for the types of operational benefits that case management can provide and then use the relevant ones for financial benefit calculation.

2. *Costs*: The process of adoption of case management, like other technology adoptions, can and will incur a variety of costs, some of which can be calculated in a straight forward manner while the others may be more nebulous. Generally speaking, costs can be separated into fixed and variable categories, and some of these costs can be directly attributable to specific activities while the others will be more general in nature, i.e., would be associated with many activities.

In the context of case management adoption, costs would arise from acquisition of needed software, hardware, and training to get technology platforms operational and personnel knowledgeable about using such platforms and following suitable methodologies. There will also be the compensations of the personnel involved in pushing the adoption. Depending on the scope of various phases of adoption, the shares of fixed and variable costs will vary. For example, if the early phases of adoption are designed as exploratory, it may be possible to construct initiatives with mostly fixed cost. For the purposes of cost estimation, "activity-based costing" (aka "ABC", see: http://www.accountingcoach.com/activity-based-costing/explanation for a primer; website accessed February/2014) can be generally used for the concrete activities. To these costs overall management and support costs plus any relevant opportunity cost should be added to reflect the overall cost. It should be noted that other than activities like well-established manufacturing of products, cost estimates will always carry a decent amount of uncertainty; the level of uncertainty should reduce as additional information becomes available with progression of activities in the initiatives (see: Figure 2.23, courtsey of the US General Accountability Office aka GAO). Sometimes three cost estimate figures, "likely," "pessimistic" (i.e., upper bound) and "optimistic" (i.e., lower bound) are presented in order to provide an idea of the level of uncertainty—these estimates tend to converge as estimation uncertainties reduce. A strategy for measuring costs and course correcting (if necessary) should be in place to guide the adoption process.

3. *Risks*: All real-life activities have some level of risk associated with them—adequate upfront planning along with strategies to adjust the planned course in response to evolving situations is the key to successful navigation through risks. In order to plan for handling risks, we have to first recognize that risks may arise either from "known unknowns" or "unknown unknowns." For the first category, we will typically know the sources of risks but may not be able to fully assess the timing or the impact of their occurrence while in the second category, we would be "caught off guard." Generally speaking, we can be better prepared for the first type of risks with specific remedies already identified while for the second category we have to be "agile" and "adaptive," i.e., rely on the organization's ability to quickly sense and analyze the risks and come up with a suitable remedy on demand. It may be sensible to set some acceptable risk criteria of a given adoption, which when exceeded the adoption may be abandoned in the spirit of "cutting losses" in a timely manner so as to prevent more damage



Figure 2.23 Cone of uncertainty in cost estimation. (Adapted from US GAO.)

downstream. Specifically for case management, adoption risks are likely to arise from things like incorrect use case or adoption roadmap (i.e., wrong scope or pace for projects undertaken), inadequate competency, inability to effect required change in work culture, out of balance and noncomprehensive governance, lack of suitable KPIs and KRIs, and inability to appropriately course-correct when necessary. The Generalized BPM maturity model and capability domains that we have previously discussed can provide useful guidance in identifying major sources of risks as well in planning for their remedies.

There is no argument that business cases are generally important for "go" or "no go" decision making for projects; they are even more important when new a type of undertaking has to be initiated due to higher levels of associated uncertainty. Here lies an apparent paradox: just when facts and arguments of a business case are most important and useful to combat uncertainty the difficulty in creating such facts and arguments are the highest due to the same uncertainty. Thus, creating a good business case for adoption of a newer technology adoption like case management will certainly involve both "art" and "science"; proposer's knowledge of current state of the art of case management and of conducting projects of similar complexity in his/her organization would be invaluable assets in creating credible business cases for case management.

2.7 Summary

In this chapter, we have covered a variety of topics that are mutually distinct but are directly related to either better understanding of case management concepts or its adoption in an organization. We started with comparing adaptive aspects of case management with characteristics of complex adaptive systems; this led to a method for identification of key attributes of adaptive case management. We then described different variations of case management solutions; specifically, we focused into the details of production and adaptive varieties of case management and their applicability in solving business problems. Next, we compared and contrasted traditional BPM with case management and noted that many solutions will require a suitable mix of both these paradigms. We introduced the notion of "Generalized BPM" to capture the combined problem solving strategy of using both traditional BPM and case management. From an adoption point of view, we defined a capability maturity model, its associated capability domains and individual capabilities in those domains for Generalized BPM. We advocated the practice of creating strategic adoption framework based on such a maturity model and building up of capabilities through proper scoping and design of case management projects. Finally, we discussed the challenges associated with and strategies for creating business cases for adoption of case management and suggested a "cost benefit risk" framework for evaluating such business cases. We are now ready to take up a set of real-life case studies of case management solutions and then get into the technical details of how to design and implement case management solution both from a solution architecture PoV and using case management features of relevant Oracle technologies, in particular, Siebel and Oracle BPM Suite—these are the discussion topics for the rest of the book.

CASE MANAGEMENT SOLUTION FRAMEWORK

Introduction

In this section, we introduce the case management solution framework. This framework enables the functional and technical design of a case management solution with a set of *case management design components*, *three case management classifications*, and the *case management solution mapping*.

Chapter 3 shows the case management functional design describing its functionality in three functional categories and one cross-functional category. The functional categories look at the life cycle of the case, information related to the case, and the interaction with stakeholders. The cross-functional category supports the three functional categories with a security and integration design.

Chapter 4 details the *case management classifications*_from a functional perspective. "Forrester's Classification of Case Management Solutions" is introduced earlier (Section I—Chapter 1), where the three classifications are described: *service request, incident management,* and *investigative handling.* Case management design components are used to describe the functionality of the three classifications with special emphasis on the life cycle design category.

Chapter 5 shows how functionality is mapped toward solution layers and technical capabilities with the help of the vendor-agnostic architecture model, the Common Reference Architecture (CORA) model, and how Oracle products can be mapped against these technical capabilities.

Chapter 3

Case Management Functional Design

3.1 Introduction

The first part in the case management solution framework is the identification of the design components needed to describe the case management functionality. We will use a design model that contains three functional and one cross-functional design categories. The functional categories describe the *lifecycle* of the case, *information* related to the case, and *interaction* with stakeholders. The cross-functional category supports the three functional categories with *roles and authorization* and *integration* design. For each design component, we will describe the high-level design functionalities.

The goals of this chapter are as follows:

- Describe the principles for the case management design components
- Describe the three functional design components; the *lifecycle* of the case, *information* related to the case, and *interaction* with stakeholders
- Describe the cross-functional category contains the *roles and authorization* and *integration* design
- Provide a design template for the determination of the required, high-level case management functionality

3.2 Principles for the Solution Design Components

In Chapters 1 and 2, we discussed the motivation, concepts, and the characteristics of a case management solution.

The principles for the case management design components are derived from these chapters and summarized in Table 3.1.

Principle	Statement	Rationale		
Goal-driven	The case management paradigm is goal-driven	Reaching the desired result is of primary importance. The actual activities performed, and the order in which these are executed, are of secondary consideration		
Generalized BPM	The business activities during case work fall under two distinct categories: routine work and knowledge work	The concept of Generalized BPM combines these two categories		
Special situation handling	Interventions interact with the flow of a case	During a case, interventions can originate from different sources that interact with the flow of the case		
Adaptable	Adaptive systems are designed to handle variations and unpredictability as a rule instead of as exceptions	Adaptivity helps reduce exceptions		
Driven by data and documents	Case management is strongly driven by data and documents	(Un)structured document and data are key components during the flow of a case		
Collaboration and knowledge sharing	High level of collaboration and knowledge sharing are required during case work	Collaboration and knowledge sharing are frequently leveraged in resolving a case		
One centralized view of the case	Direct insight into what activities are done, which activities are in progress, who are involved, and what incident information is available	Be able to make decisions based on case progress		
Spanning multiple applications	Case management work often spans multiple applications	Case work often spans different departments and/or companies and, as such, needs to integrate with multiple applications		

Table 3.1 Principles for Case Management Design

3.3 Design Components

3.3.1 Design Components Overview

The functional design of a case management solution contains three main *functional design categories* and two *cross-functional design components* as shown in Figure 3.1.

The functional categories each contain three design components and look at

• The *case lifecycle design category* with support for the coordination of activities in reaching one of the end goals.



Figure 3.1 Case management design categories and components.

- The *case information design category* dealing with all sorts of data and documentation flowing through the case, and gives insight into the stakeholders and their relations.
- The *case interaction design category* maximizing the support for the case work force.

The *cross-functional design category* supports the functional design components integration and roles and authorization:

- The *case integration design* related to the delivery of functionality and information across all participating applications and stakeholders.
- The *case roles and authorization design* ensures the execution responsibilities, visibility of data, and tracking and tracing across the case execution.

In the following sections, the case management design categories and components are described in more detail and contain the following:

- Aim of the design category/components
- Relation with the principles
- Description of the design components part of the design category
- A modeling notation for lifecycle design

In the description of the design components, topics are described that should be analyzed during the case design. The design component sections do not intend to describe how the design is executed. We touched on this topic in Section 1.5.3 and will show design methods later in the Oracle Adaptive Management tutorial description in Section 17.2.

3.3.2 Case Lifecycle Design Category

3.3.2.1 Aim

The case lifecycle design category describes the coordination of activities in reaching one of the end goals of a case and consists of three design components: *case flow/process design*, *rules design*, and *event design* (Figure 3.2).



Figure 3.2 Case management lifecycle design category.

3.3.2.2 Relation with the Principles

This design category relates to the following principles:

- Goal driven—The case flow is focused on reaching one of the end goals and also focuses on intermediate goals.
- Generalized BPM—The case can consist of both routine work and knowledge work
- Special situation handling—Interventions can enter at any point in the case flow and these interventions have the potential to alter the case flow
- Adaptable—The case model needs to be adaptable at both designtime and at runtime

3.3.2.3 Case Management Lifecycle Design Components

3.3.2.3.1 Case/Process Flows Design

The design of the case execution can be based upon two types of work:

- 1. Knowledge work—Requires handling of unstructured and unpredictable tasks.
- 2. Routine work-Based upon structured and completely predefined process models.

In order to support the design of the case flow, process and events, two modeling notations are used for the description of the architectural patterns in this section: a high-level, case flow model, and a case state model. The *case flow model* gives insight into the interaction of stakeholders and an (high-level) activity flow. The *case state model* focuses on state changes.

The *case flow model* is used to model the case lifecycle from an activity flow perspective and is used to give a high-level overview of the stakeholders, the flow of activities, and the events intervening in the execution of the case flow.

Every activity in the case flow, as shown in Figure 3.3, consists of more detailed work and can be a simple workflow or a complex case functionality involving investigation activities. Lines show the flow of activities and the interaction among the stakeholders. The *stakeholders* are modeled in this diagram as horizontal swim lanes. The stakeholders relate to the roles as will be described in Section 3.3.5 where the stakeholders related to a case and their responsibilities are described. During the design phase, these roles need to be specified further, and related to the more detailed



Figure 3.3 Case flow model example.

work in the activities. *Events* are shown in the right hand side of the case flow diagram and represent manual or automated interventions in the case flow.

The *case state model* shows the case lifecycle from another viewpoint related to state flow, activities, and events (Figure 3.4). Stakeholders are not included in this view. This view is based upon the *case management model and notation* (CMMN) and is described in more detail at http://www. omg.org/spec/CMMN/.

The CMMN model is focused on state changes during the case lifecycle. Each case state change determines what activities can be used at a particular moment. This model describes the different activities or groupings of activities and gives insight into entry and exit criteria of the different phases in the state.

The entry criteria determine when activities or a set of activities are allowed to start. Exit criteria determine when a phase or activity can be considered closed. Milestones, which signify important activity completion points in the case progression, determine when a business goal or target is met. Typically, a set of tasks needs to be executed in order to reach a milestone.

3.3.2.3.2 Rules Design

The following rules types can support the case management execution:

- Case lifecycle decision-making—This relates to the ability to make decisions based on a change that happened in the lifecycle of a case. For instance, when a milestone is reached a decision is needed to determine what follow-up activities are required and which ones are optional.
- Entitlement determination—This functionality relates to the decision made in a customer related case, where the entitlement of the customer on a specific service or claim is determined. The entitlement decision entails the eligibility of the customer on the service or claim, and can contain money involved, duration, restrictions, and specific treatments. Examples are the decision in a mortgage request, a welfare request decision, and an insurance claim decision. This typically is guided by policies and regulations, and can change on a regular basis.
- Progress measurement—In order to be able to show the progress of a case and be able to make decisions in relation to the case progress, a decision process around "measuring the progress" is needed. This involves measuring the case state against predefined progress indicators.
- Data entry validation—This is required in situations where customers or other stakeholders enter data into the case that need to be validated against for instance regulations and policies.
- Information discovery—Different data types and sources can relate to case. In the information design of a case, all predefined data and document information objects are designed that determine the input and output of a case. Another type of data that can determine the output and behavior of a case is unstructured in nature, such as data streams, video, and social media. Searching through this unstructured data, in relation to structured data, can deliver some new or changed insights in the case execution.

3.3.2.3.3 Event Design

Events intervene in the case activities in a nonpredictable way and can interact with the case at any moment in the case lifecycle. Summarizing the characteristics from events found in real life cases, events can be interrupting in nature or noninterrupting. Interrupting events have the potential to



Figure 3.4 Case state model: notation and example.

alter the flow of activities; for instance when in a police investigation new evidence is found at the crime scene the investigation can change into a new direction. A noninterrupting event starts some activities parallel to the case flow, but does not interrupt the flow itself. A request for an update during an insurance claim assessment starts activities to gather the update information and present the update to the customer.

3.3.3 Case Information Design Category

3.3.3.1 Aim

The case information design deals with all sorts of data and documentation flowing through the case and maps the relationships between all persons and/or assets involved in the case. The case information design category contains three design components—*data design, document and media design,* and *people relationship design* (Figure 3.5).

3.3.3.2 Relation with the Principles

This design category relates to the following principles:

- Driven by data and documents—Case management is strongly driven by data and documents which can be of structured and unstructured nature.
- Collaboration and knowledge sharing—In order to support collaboration and knowledge sharing, the involved persons and the relation between all case stakeholders need to be understood.

3.3.3.3 Case Management Information Design Components

3.3.3.3.1 Data Design

Case management supports a large group of stakeholders within an organization with the execution of case related activities, all with different intelligence needs related to different information levels:

• At an *operational* level in real time, supporting the work on a single case with the right knowledge.



Figure 3.5 Case management information design category.



Figure 3.6 Case management information levels. (Courtesy of Ard Jan Vethman.)

- At a *tactical* level aimed at improving the efficiency, customer satisfaction, and controlling case related key performance indicators (KPIs).
- At a *strategic* level looking back in time and investigate process and product improvements and forecasts.

The different information levels for a case supporting the different stakeholders are summarized in Figure 3.6. In this figure additional indicative timing levels are added. These time levels only need to be seen in relation to one another; actual time amounts depend on the case situation and duration.

3.3.3.3.1.1 Operational Level The operational level is where the actual work on the case is done. The execution is lead by the cases owners and executed by the case workers. In the *service request* type of cases, see Chapter 4 for details, also customers have a large stake in what decisions are made and in what direction the case is going. In order to support a case from an operational perspective, the following data functionalities are required:

- Case execution data—This includes all data related to the case execution, such as the case status, activities, decisions, and the history of the case execution.
- Case supporting data—The case execution is supported with details of the case, history of all cases and insight in laws, regulations and policies to which the case must abide. The history in both the case execution and case supporting data area is used for the actual case execution and could be of help for future cases as a reference or a source of investigation.
- Risk parameters for monitoring—In Section 4.5, *the monitoring characteristics* of the incident management case classification are described. The likelihood of an incident can be predicted by monitoring certain risk parameters. These risk parameters are re-evaluated on a regular basis.

3.3.3.3.1.2 Tactical Level Multiple cases can be open and be worked on at the same time. The activities that are related to these cases are distributed by the case owners among the case workers. In environments with hundreds, thousands, or more cases, an overview is needed to check the state of all these cases and related tasks and be able to drill in to the level at which

operation activities can be executed. In this area, the following information functionalities are needed:

- *Workload information—This level of inform*ation is aimed at delivering intelligence to ensure *efficiency* in the case *execution*. The case owners and stakeholders need to be able to act upon a changing workload. This can be due to campaigns being executed, a problem with a product, which raises the amount of complaints, or a problem with the workforce such as multiple persons getting flu at the same time. The workload then needs to be relocated to different case workers, extra people being called in, or customers informed that certain activities are deferred.
- Key performance indicator (KPIs)—Case owners and departmental stakeholders have KPIs assigned to the case and the related processes. The case owners have KPIs on an end-to-end level. Departmental stakeholders have KPIs more on a micro level, for instance at a departmental level. Case management activities are for a larger part driven by human interactions, which in time and money can make the execution of the case expensive. Both the case owners and the departmental stakeholders have KPIs related to the result and efficiency of (sub) results of the case.
- Customer satisfaction information—In some industries, the way a case is handled for a customer can make or break the relation with that customer. In these situations, thorough measurement of the customer satisfaction after (or during) the case execution is crucial.
- Audit trails—For compliance, regulatory, or contractual reasons activities, decisions and/or discussions need to be tracked during the execution of a case.

3.3.3.3.1.3 Strategic Level Strategic data involves gathering data in the long run, and identify what happened in all the case executions over the last months and years. This is a "good old" business intelligence (BI) and helps in identifying finding improvement areas, trends, and supports in making forecasts. The result of this intelligence is fed back into the operations level, from which either the case and/or the process lifecycle is adjusted, or the decision support is improved.

In the *improvement* areas, three main topics can be identified, case/process improvement, product improvement, and decision support improvement.

Case/Process Improvement The BPM lifecycle describes the various phases of a case and its related processes during its lifetime, as shown in Figure 3.7. It starts with the design and implementation after which the process "goes live" where it will be executed by the different participants. By measuring the execution time and the behavior of a case/process, we are able to optimize these case/processes. In the last phase of the lifecycle, a case/process can be retired or we can execute another cycle in the lifetime of the case/process. Up until the time that the process is retired the BPM lifecycle will be rerun in various cycles:

- Design—In the design phase, business analysts and participants design the solution.
- Modeling—In the modeling phase, business analysts and developers create the solution.
- Execution—In this phase, case and related processes are executed by participants.
- Monitoring—Monitoring delivers insight in the status and whereabouts of processes.
- Optimization—With the information from the monitoring phase, processes can be adjusted and new simulations can be created and tested.

Product Improvement Case handling is goal driven and deals with the execution of a certain product. This can be a customer supporting product such as a mortgage, or a more abstract product such as a police investigation. The behavior of the case handling for these products is determined



Figure 3.7 BPM process lifecycle.

by product features and policies and can be guided by internal rules and by laws, regulations, and policies. Recording the actual behavior of the case execution in relation to the way the product is dealt with gives insight in where to improve the product.

Forecasts, Trend Analysis, and Customer Behavior Strategic data supports the creation of forecasts, trend analyses, and insight in customer behavior.

Summarizing all the needs described earlier gives the following functionality overview related to the three information levels, as shown in Figure 3.8.



Figure 3.8 Information level-related case functionality.

3.3.3.3.2 Document and Media Design

Case workers can use a variety of documents and media during the case execution delivered as input to the case, and at the same time deliver documents as output. The input for a case is usually supported by a combination of structured and unstructured data such as social media input, video, or sound. The output of a case, as delivered by the case workers, is mostly of structured format, such as standardized documents.

The document and media design investigates the different forms and formats of the input and output material. Additionally, the sources of the input material and the amount of material that can be brought into the case needs to be analyzed.

3.3.3.3.3 People Relationship Design

During the execution of a case, multiple persons can be involved from both internal within the organization as well as external stakeholders. The people relationship design maps all the stakeholders involved in the case and enables interactions among the stakeholders.

Internal stakeholders:

- Case workers/owners
- Front/mid/back office
- Risk department
- Management

External stakeholders:

- Customers (and/or representatives)
- External specialists (such as fraud check specialists)
- Related companies and/or departments
- Victims, witnesses, culprits
- Press

3.3.4 Case Interaction Design Category

3.3.4.1 Aim

The case interaction design enables maximizing the support for the case work force. This design area contains three design parts: *portal design, dashboard design, and channel design* (Figure 3.9).

3.3.4.2 Relation with the Principles

This design category relates to the following principles:

- One centralized view of the case—Multiple activities can be executed by different persons, involving different data types during the execution of a case. A centralized overview is essential for the execution of a case.
- Collaboration and knowledge sharing—This encompasses the collaboration and knowledge needs to be supported by an interaction platform.


Figure 3.9 Case management interaction design category.

3.3.4.3 Case Management Interaction Design Components

3.3.4.3.1 Portal Functionality

The portal design supports all stakeholders in the case execution with a centralized view on case handling in two areas. The "one central view of the case" gives a complete overview of case details, activities, involved persons, history, decisions, and media. Depending on the role of the stakeholders, the full view or a subset of the case view is presented. Cases that contain customer interactions also need to have a "one overview of the customer" that provides the customer details, related parties, history of activities with the customer, and (optionally) a risk profile. Both the central view of the case and the central view of the customer can be partially shared with the customers.

3.3.4.3.2 Dashboard Functionality

Two types of dashboards are used in case management execution, the tactical related dashboards and strategic dashboards:

Tactical dashboards look at case progress and execution data from all running cases and test this data against predefined KPIs, check the operational performance management, and investigate the customer satisfaction related to the case execution. This functionality relates to the tactical related data as explained in Section 3.3.3.3.1. Dashboards give an insight in all these KPIs and workload related checks. Intelligence behind the dashboards combines data together and, based upon rules, visualize this data and provide insight when thresholds are exceeded.

Strategic dashboards sample data from all finished and still running cases and look at trend analysis, forecast predictions, and for improvement areas in the case execution. This functionality relates to the strategic-related data as explained in Section 3.3.3.3.1.

3.3.4.3.3 Channel Functionality

The case information and functionality can be exposed to the stakeholders via different channels, such as via browsers and mobiles apps. The channel design should be aimed at two communication types, internal and external communication. Internal communication specifically is necessary in cases where direct communication is needed to improve the quality of case work, and could be supported for instances by (secured) mobile access and messenger type of applications. External communication is aimed providing customers and other external stakeholders with case progress information and the ability to intervene with and deliver information to the case.

3.3.5 Case Cross-Functional Design Category

3.3.5.1 Aim

The cross-functionality design category contains the *integration design* and the *roles and authorization design* that both support the other functional design categories (Figure 3.10).

3.3.5.2 Relation with the Principles

This design category relates to the following principles:

- One centralized view of the case—*Roles and authorization: Different persons interact with* the centralized view. A security regime is needed to ensure that case participants can only perform activities and have access to information they are allowed to see based upon their role.
 - Integration: Supports the delivery of the data needed in the centralized views.
- Collaboration and knowledge sharing—Collaboration and knowledge sharing is restricted according to a roles and authorization model.
- Spanning multiple applications—Integration enables the sharing of information and functionality across applications within the scope of a case.

3.3.5.3 Case Management Cross-Functional Design Components

3.3.5.3.1 Roles and Authorization

Stakeholders who participate in the case have different responsibilities, ranging from working on the case, completing intermediate results, closing the case, auditing the case, and owning the case. In order to ensure that every stakeholder is working within the limits of his responsibilities and has only access to the information tailored for him, a security model is needed.



Figure 3.10 Case management cross-functional design category.

On a high level, the following stakeholder profiles can be identified in the context of a case:

- Case owners have the responsibility over the case or parts of the case. They are responsible for bringing the case, or parts of it, to a successful end. The ownership ranges from groupings on a case overarching level down to lower level process activity groupings and levels in between such as related to milestones.
- *Case workers* are the specialists in the field executing the work in case activity level. The case workers work in the different departments across the organizational structure of the companies involved, and are chosen because of their responsibilities and knowledge.
- Customers, the case is always aimed at delivering an end result for a customer. The customer can be an actual person, such as in a complaint case, another organization, or another case/ process needing extra context for a research.
- Management is interested in the outcome and the performance of the case. They have their KPIs related to costs, efficiency, and result, which are defined upfront and measured during the execution of the case. Another operational task of management is the allocation of the case workers to the cases.
- *Auditor*, the role description of an auditor encompasses many aspects of financial, organizational and IT related areas. The activities that need to be done in the context of a case are assuring that all tactical related controls (amongst others compliance with regulations, customer agreements, and contracts with third parties) are implemented. Once implemented these controls need to be tested to assure that these are operating with sufficient effectiveness.
- *Risk officer*, in the context of a case the activities of the risk officer are making a risk profile of the cases and keeping track of these case risk parameters. When a case enters a high-risk state, the risk officer is allowed to intervene in the case activity.

The stakeholder interaction can be summarized in a case stakeholder interaction overview as shown in Figure 3.11. Cases where work is executed are in an open state. Cases that have completed in an end result are in a closed state. Case owners, case workers, customers, and risk officers work on cases with an open status. Auditors and management want to measure results on open and closed cases. Management is involved in allocating persons for cases in an open status.

3.3.5.3.2 Integration

Case management usually operates in a hybrid landscape and communicates with other applications to exchange information and manage activities. Three main areas can be identified where integration is needed in the scope of a case:

- Application integration—Relates to the interactions between the case system and other applications, where information is exchanged and functionality is started, and managed to support the case execution.
- Case integration—Relates to all integration activities that interact with the case lifecycle. A case can start and manage external or internal processes. These processes can be a short-lived series of activities or a long-running process, externalized to a separate internal system or to an external provider. A special type of application integration is on event entry initiated from an external application. An event can have direct impact on the case lifecycle or an interrupting event. An event can activate some activities that do not interrupt the current



Figure 3.11 Case-stakeholder interaction overview.

case situation. Another topic related to case integration is interaction management between a case and an investigation started as part of a case.

Portal and channel integration—Relates to the data provisioning for the portals needed to support the case workers, such as the one centralized view on the case, the one centralized view on the customer, and the KPI dashboards. All communication and interaction with the channels, such as internal communication, application support, web support relates to another integration topic for which support is needed.

3.4 Design Template

In this chapter, the different case management design components are described by looking at the high-level functionalities. In Table 3.2, the high-level requirements for a case management solution are summarized. This table will be used as a template in the upcoming chapters (Chapter 4) and Part IV to give an insight into the nature of the case management solution.

3.5 Conclusion

In this chapter, the *case management design components*, as part of the *case management solution framework*, are described. This framework enables the functional and technical designs of a case management solution. The design components form the first step in this framework.

In the previous chapters, we described the motivation, concepts, and the characteristics of a case management solution. With this knowledge of a case management solution, we derived principles

Case Design Category	Case Design Component	High-Level Functionalities
Case lifecycle design	Case flow and process design	The case flow specific for a case
	Rules design	Case lifecycle decision making Entitlement determination Progress measurement Data entry validation Information discovery
	Event design	Interrupting events Noninterrupting events
Case information design	Data design	Case execution data Case supporting data Tactical-related information Strategic data
	Document and Media design	Case input management Case output management
	People relationship design	Internal stakeholders External stakeholders
Case user interface design	Portal design	One central view on the case One overview of the customer
	Channel design	Internal communication External communication
	Dashboard design	KPI management Performance management Customer satisfaction determination
Cross-functional design	Integration design	Application integration Case integration Portal/channel integration
	Roles and Authorization	Internal profiles External profiles Compliance

 Table 3.2
 Summary of Case Management Design Components

for the case management design. In this chapter, we use three functional design categories and two cross-functional design categories to describe the case management functional requirements.

The *functional categories* each contain three design components:

- The *case lifecycle design category* with support for the coordination of activities in reaching one of the end goals
- The *case information design category* dealing with all sorts of data and documentation flowing through the case, and gives insight to the stakeholders and their relations
- The *case interaction design* maximizing the support for the case work force

118 Oracle Case Management Solutions

The cross-functional design components support the functional design categories:

- The *case integration design* related to the delivery of functionality and information across all participating applications and stakeholders
- The *case roles and authorization design* ensures the execution responsibilities, visibility of data, and tracking and tracing across the case execution

We finished this chapter with a summary of the design components in a table format where a high-level overview of the functionalities is provided. This table will be used as a template in the upcoming chapters to give an insight in the nature of the case management solution.

Chapter 4

Case Management Classifications

4.1 Introduction

In this chapter of the case management solution framework, the case management classifications are discussed. The classifications are based upon the "Forrester's Classification of Case Management Solutions" that is introduced earlier in the book in Chapter 1 and contain the service request, incident management, and investigative solution.

The goals of this chapter are as follows:

- Provide characteristics for classification determination
- Provide alignment of the classifications with industry processes
- Determine principles for the classifications to guide the solution
- Use the case manage management design template to provide a high-level overview of the classifications
- Use the case flow and case state models to detail the case lifecycle design

4.2 Forrester Case Management Classifications

The "Forrester's Classification of Case Management Solutions" is introduced earlier in Section 1.4.5. In 2009, Forrester predicted the appearance of three broad categories of case management solutions: Service requests, incident management, and investigative solution. Looking through multiple industry processes, we identified alignment of the characteristics of these processes with the three Forrester classifications (Section 1.5.1). We will use the Forrester case management classifications as one of the key elements in the case management solution

framework. These classifications are elaborated in the rest of this chapter. The observations we identified in Section 1.5.1 are used as input:

- Most real-life solutions display characteristics that could be found in a combination of two or more of the three types (i.e., service request, incident management and investigative).
- Investigative type of case work is typically triggered by either a service request or an incident reporting.
- There are wider variations in the types of case work (i.e., activities) that could be associated with incident management compared with service requests.
- A significant portion of case work tends to require human touch.
- Use of rules play a big part in construction of case management solutions.
- Many case management solutions need to adhere to laws and regulations.
- Handling of documents and content is an integral part of most case management solutions.
- Application integration shows up as a major solution component.
- Event handling is an integral part of most case management solutions.
- Security has high relevance in the solution.

4.3 Common Principles for the Classifications

Principles are used to guide the determination of the solution for the three case management classifications: service request, incident management, and investigative handling. Some of these principles are shared across all three classifications and are summarized in Table 4.1.

4.4 Service Request Classification

4.4.1 Context

Service requests deal with the process of delivering a service that a customer requested from a commercial organization or government agency. These service requests include complex interactions between the customer and the service provider and often contain investigations to reach the end result. The ability of the case workers to select and provide the components of the service necessary to satisfy the customer is critical to successful closure of the cases. In the interaction with the customer, different channels are used such as mail, Internet, mobile, direct communication, and letter. Processing the request within the organization is completed with a variety of processes and actions that require support from different departments and applications.

4.4.2 Determination

The following characteristics can be used to decide when a case is a service request:

- A customer requests a service.
- The service provider is a commercial organization or government agency.
- Complex interactions exist between the customer and the service provider.
- Customer satisfaction is crucial.

Principle	Statement	Rationale	Implications
Standardize the process	Standardize the process flow to a reference model	Best practice in the market Improvement of time to market of new products	Every process with the same classification is handled in the same way The information model for every classification product is identical Insight in the efficiency of departments is available A standardized overview of outstanding requests is available
One centralized view of the case	Direct insight in what activities are done and which are in progress, who are involved, and what information is available	Be able to make decisions in relation to the case progress	Case steering is centralized Integration is key
Measuring the progress	Be able to show progress of the case	Be able to make decisions in relation to the case progress	Case progress indicators should be known Case real time state comparison against case progress indicators should be possible
Automate where possible	Automate the parts that speed up the efficiency of a service request product handling process via straight through processing (STP)	Lowering costs Standardized handling of a case classification process	Automation enables faster handling of a case product Integration is key Automation of complex rule verification avoids human errors
End-to-end auditing	All decisions and communication with the customer are recorded	Regulatory requirement Enables playback scenarios	Playback of why decisions are made is possible

 Table 4.1
 Common Principles for the Case Management Classifications

- In many situations, the process is predictable.
- (Optionally) Complex policies and/or regulations determine service request entitlement.
- (Optionally) Regulatory requirements are important.

4.4.3 Alignment with Industry Processes

In Section 1.5.1, we analyzed a nonexhaustive list of standard processes across the different industries. In this analysis, we cross-checked these industry processes against the determination characteristics for a service request type of case. These processes are summarized in Table 4.2.

Horizontal/Common	Employee onboarding Customer onboarding AP and AR processing Facilities management Order/complaint/service exception management Trade and contracts management Complaints
PS + Social Services	Benefits eligibility Grants management Unemployment/welfare services Patent application processing Permits and operator licenses Child protection/substance abuser help/HIV advisory
Financial Services	New account opening Loan origination Investor servicing Regulatory reporting
Insurance	New account opening Policy management
Healthcare	Member enrollment
Legal + Police	Legal practice management, time tracking and billing
Energy & Utilities	Project and transmittals management Work order management

 Table 4.2
 Service Request Usage across Industry Processes

It is clear from this overview that service requests can be found across many industries. For two service request-related examples, more detail is provided in Chapter 12, for the *loan origination* and the *benefits eligibility* in Chapter 13.

4.4.4 Business Challenges

For all the service request industry processes, similar business challenges can be identified. These business challenges are related to *cost, customer handling, regulations and policies,* and *risk management.*

- Customers want to be serviced better and faster. Case workers dealing with the service request need to have all customer information available to avoid making errors during the service request handling. Specifically for the commercial organizations the way the companies position themselves to meet the customer needs impacts their operating model. More information on this topic can be found in the book *The Discipline of Market Leaders* by Wiersema (http://www.basicbooks.com/full-details?isbn=9780201407198):
 - Should they focus on *operational excellence* and deliver a standardized product for a low price?
 - Should they go for *customer intimacy* and deliver customized products?
 - Should they aim at *product leadership* in which they provide state-of-the art products?

Making a choice among these operating strategies has an impact on the solution requirements.

- Stricter *regulations and policies* require end-to-end auditing for all service interactions. At the same time, these stricter regulations and policies need to be implemented at a faster pace by the service request supporting systems. This can be a challenge in the current system landscape.
- *Cost reduction* is an important driver for changes as a result of the financial crisis.
- More attention for fraud and financial risks requires improved *risk management* supporting for the case workers.

4.4.5 Principles for the Solution

The principles for a service request classification are based on the business challenges as defined in the previous section and the common principles described earlier in Section 4.3. Table 4.3 shows the service request-specific principles.

4.4.6 Case Management Design Components

In the first part of the case management solution framework in Chapter 3, we introduced the design components, three functional categories and two cross-functional categories that help describe the case management functionality. In the following sections, we will use the categories to describe the service request functionality. Specific focus will be given to the case lifecycle design category since this contains the core functionality of service request handling.

Principle	Statement	Rationale	Implications
One customer view	Provide a complete overview of the customer, personal information, and history of provided citizen services	No mistakes in handling the service request for the customer All case workers have the same understanding of the customer situation	Systems containing (and owning) customer information must be able to deliver customer information The ownership of customer details is clear Integration is key
End-to-end auditing	All decisions and communication with the customer are recorded	Regulatory requirement. Enables playback scenarios	Playback of why decisions are made and what options are offered to the customer is possible
Separation of know and flow	Keep the implementation of the process flow (the flow) and the decision management (the know) separate	Rules are related to laws and regulations and tend to change with a higher frequency than the case/process flow activities	Separate the implementation of decision management from the process flow

Table 4.3 Principles for the Service Request Classification

4.4.6.1 Service Request Design Category Overview

In Table 3.2, a design template is presented that will be used here. This template contains

- The case management design categories
- The design components that are related to the categories
- The high-level functional areas that are introduced

Table 4.4 is used to the provide an insight into the required functionalities for a service request, two extra columns are added to the original design template:

- The importance of a high-level functionality for the service request design
- The specific usage for a service request.

4.4.6.2 Case Lifecycle Design Category

The service request process can be broken down into two main parts, the triage and the case execution, as shown in Figure 4.1. In the triage, which is the first interaction with the customer, an assessment of the need for a service is investigated. When both parties have reached a decision that follow-up work is needed, the service request continues as a case containing investigations and complex interactions between the service provider and the customer. The triage and the service request handling have different characteristics; the triage is workflow-based and short-lived. The service request handling is case-oriented, containing investigations and customer interactions, and has a longer life span.

4.4.6.2.1 First Interaction with the Customer—The Triage

The customer raises a request for a specific product such as a request for a welfare service or a mortgage. In the first interaction with the customer, the actual need and priority for the service is investigated. When a customer wants to buy a house, without having made an offer on a house, the first interaction only involves providing information on the mortgage products and the mortgage loan amount that can be provided based upon salary and no follow-up activities are executed. In the situation where the customer actually has found a house that fulfills their needs, the assessment evaluates if the mortgage product is aligned with the needs of the customer. When the customer and the mortgage provider decide to continue, this stage is closed and follow-up work will be started.

Figure 4.2 shows the case flow model of the interaction during the triage. The case flow model is introduced earlier in Chapter 3.

In the triage phase, the following stakeholders participate:

- The customer who entered the service request
- The first line of contact who is responsible for all activities in this phase
- The case owner who registers the case after the decision is made to continue the service request

Case Design Category	Case Design Component	High Level Functionalities	Service Request-Specific Usage
Case lifecycle design	Case flow/process design	The case flow specific for a case	See Section 4.4.6.2 and Figures 4.2 and 4.3
	Rules design	Case lifecycle decision making	See Sections 4.4.6.2.1 and 4.4.6.2.2
		Entitlement determination	Determine the service request entitlement Agree/not agree Related value Periodicity Terms for which the decisions applies
		Progress measurement	Related to agreements with and expectancy of customers, regulatory demand, and/or internal performance management
		Data entry validation	Validating entry data for the request
		Information discovery	This relates to any investigation where the quality of the service request can be improved or the service request is checked against fraudulent behavior. This often includes verification against large combined (un) structured data sources
	Event design	Interrupting events	See Section 4.4.6.2
		Noninterrupting events	See Section 4.4.6.2
Case information design	Data design	Case execution data	Be able to follow the case execution across the various stakeholders and activities
		Case supporting data	Service request related data

Table 4.4Service Request Design Overview

(Continued)

	Oracle Ca
Service Request-Specific Usage	se M
st input, structured data	ana
est output, mostly documents	gem
vorkers, front, middle, back office, t	ent So
alists anies/departments	lutions
overview of the service request is required by all stakeholders	
ey elements in the determination of the	

126

Table 4.4 (Continued) Service Request Design Overview

Case Design Component

Case Design Category

Document and me		Case input management	Service request input, structured data	
People relationship design	design	Case output management	Service Request output, mostly documents	
	Internal stakeholders	Case owner/workers, front, middle, back office, management		
		External stakeholders	Customers External specialists Related companies/departments	
Case user interface Portal design design	One central view on the case A centralized overview of the service request proceedings is required by all stakeholders			
		One overview of the customer	One of the key elements in the determination of the service request is a full overview of the customer	
	Channel design	Internal communication	Interaction between case workers and first line of contact	
		External communication	Customer interaction	
	Dashboard design	KPI management	Cost driven	
		Performance management	Cost driven	
		Customer satisfaction determination	Avoid potential loss of the customer	

High Level Functionalities

(Continued)

Case Design Category	Case Design Component	High Level Functionalities	Service Request-Specific Usage
Cross-functional design		Application integration	Depends on the service request related IT landscape
		Case integration	During the entitlement, determination processes in different departments/systems can be started. For instance in the entitlement process for a welfare request the tax department is cross-checked for taxation history and behavior
		Portal/Channel integration	Delivering information for the portals and for determination of service request entitlement
	Roles and authorization	Internal profiles	Depends on the sensitivity of the service request case
		External profiles	Profiles determine the amount of information that is shared with the customer via different channels: Direct interaction with the first line of contact One central view on the case One central view of the customer

Table 4.4 (Continued) Service Request Design Overview



Figure 4.1 The service request classification—two interactions.

The service request triage contains the following activities:

- Create service request: The customer requests the service via a channel such as online, via a letter, or via direct contact with the service provider.
- Receive customer information: In this step, all information coming from the customer is received. This information is stored for later reference and routed to a person who can perform the first assessment.
- Assess and assign: After receiving the service request, the actual need and priority for the service is investigated by the "first line of contact" and, if needed, rerouted to another person.
- Create result: From the received information and the first assessment, preliminary results can be created for discussion with the customer.
- Share results-discuss results: In this two-step interaction, the first line of contact and the customer discuss the result of the first assessment.
- Create case and finalize: These two steps are the final steps in the triage activity. The customer and the service provider decide to continue working on the service request. Afterward, the service is registered as a case or the work done so far is deemed sufficient and no follow-on work is required. In this situation, the work is finalized.

The triage workflow can be interrupted by two events originating from the customer. These include interrupting and noninterrupting events:

Interrupting events:

Intervene

The customer can intervene in the activities when they want to stop the service request or extra information can be delivered that might change the assessment decision.

Noninterrupting events:

Ask for update

The customer can ask for an update if the triage assessment takes longer than expected. This is a noninterrupting event.

4.4.6.2.2 The Second Interaction with the Customer-Service Request Handling

Once the decision has been made in the triage stage to continue working on the service request, a case is created and a series of assessments, investigations, and discussions with the customer are executed until a result is reached that meets both the customer and the service provider expectations. One of the characteristic features of the service request pattern is that it has a predictable lifecycle. The service request case lifecycle contains one or more phases each of which ends in a milestone. A milestone signifies important activity completion points in the case progression and determines when a business goal or target is met.



Figure 4.2 The service request triage—case flow model.

When looking at the mortgage example (Chapter 12), three phases can be distinguished: *loan application (sales), mortgage origination,* and *mortgage fulfillment*. Each of these phases has the same high-level structure:

- An assessment to see if the milestone is met.
- Follow-up investigations working out the service request details.
- Discussions with the customer.
- The phase can be iterated multiple times until the milestone is met.

Figure 4.3 shows the details of the flow of activities in the service request case handling. In the case handling phase, the following stakeholders participate:

- The customer, who now is engaged in discussions related to the (in-between) results
- The first line of contact who is now acting as the discussion partner with the customer
- The case owner who is actively leading the service request case
- The case workers who are responsible for handling the service request details

The case handling work of the service request contains the following activities:

- *Create case*: If the triage results in a decision that follow-up work for the service request needs to be done, a case is created by the case owner.
- *Loop of investigation and discussion for different milestones*: This is a continuous loop of investigation and involves interactions with the customer until one of the (intermediate) goals is met.
 - Assess and assign: After each loop, the case owner makes an assessment if the milestone (or end result) is reached. When another loop is required, the case owner decides what activities need to be done.
 - Start investigation: In this stage, one or more investigations are executed in order to reach the required end result. These investigations are started as a separate case and can contain work such as determining the allowed mortgage amount for the customer, deciding the entitlement for a customer based upon laws and regulation, or executing a fraud investigation. This work can be done by internal case workers or by outsourced specialists. The output of each investigation is a report that is used by the case owner to determine follow-up activities in in the execution of the service request.
 - *Create result*: All in-between results and decisions are stored for later reference and end-to-end auditing.
 - Share results, discuss result, receive customer information: The (in-between) results are shared and discussed with the customer. This is followed by a decision from the customer on how to proceed with the service request.
- *Finalize*: Once an agreement is reached between the customer and the service provider, whether positive or negative, the final activities are executed. These include activities such as signing of the contract, delivery of products, and setting payment schedules in motion.

4.4.6.2.3 Service Request State Model

The state model of the service request pattern is a predictable flow of activities ordered by one or more phases as shown in Figure 4.4. This state model shows the significance of customer interactions during the case execution. The closure of every phase depends on the agreement with the customer. If no agreement is reached at the end of a phase, the phase is repeated or the case is terminated.



Figure 4.3 The service request case handling work—case flow model.



Figure 4.4 Case-state model for service request.

For figurative reasons in Figure 4.4, three phases and related (intermediate) milestones are shown. Real-life cases can have different amounts of phases and milestones.

A phase is a series of activities leading toward a (in-between) result. It is closed when a milestone is reached or when the activities are terminated. The activities within the phase align with the "loop of investigation and discussion for different milestones" in Figure 4.3. In Figure 4.4, these activities are summarized by the three main activities: *assess, investigate,* and *discuss.*

Events play a more important role in the case handling part than in the initial assessment workflow phase. This case handling part can be interrupted by events originating from the customer and the case owner.

Interrupting events

- Intervene:
 - This can be the same interrupting event for the customer as in the triage.
 - The case owner can intervene in the case execution when a situation occurs where the execution of activities of the case needs to be altered. For example, at one point in the mortgage service request handling, a suspicion of fraud is raised. The case owner makes a decision to suspend all activities, start a fraud investigation, contact the legal department, and send a formal statement to the customer describing the proceedings.
- Time-out checks:

The case owner is responsible for timely handling of all activities in the case. Every activity can have requirements around execution and/or communication timings. These may be related to regulatory requirements, agreements with customers, or internal performance agreements. Some of these timing requirements are translated into KPIs for the case owners or one of the other stakeholders. On an operational level, performance dashboards give an insight into the progress of these timing related KPIs.

Appeal and/or reopen:

Closing the case is the final stage for a service request handling. However, there are situations that require reopening the case after it is closed. The customer may not be satisfied with the result of the service request handling and may want to reopen the case for further investigation or better service. This can be escalated until the point that the customer files an appeal in court against the service provider.

Noninterrupting events

Ask for update:

This may be the same noninterrupting event for the customer as in the triage.

4.5 Incident Management Case Classification

4.5.1 Context

"Expect the unexpected" is a good way to describe incident management. We need to be prepared, but hope that the unexpected will not happen. Most of the time the connotation with incident management is not positive. The aim of an incident-related case is to alleviate a problem that occurred and includes providing help to persons and/or bringing situations to justice.

4.5.2 Determination

The following characteristics can be used to determine the case management classification for an incident management type of case:

- Incident handling is aimed at solving the problem and finding the root cause.
- Initiation of an incident may be unpredictable.
- The high-level execution of an incident is predictable, but certain parts of the execution are highly influenced by events.
- Both internal and external communication is key.
- Data related to an incident is both structured and unstructured by nature.
- Intelligence is needed to solve the case.
- (Optionally) Incident may be started by a customer.
- (Optionally) Regulatory requirements are often important.

4.5.3 Alignment with Industry Processes

The nonexhaustive list of industry processes that relate to incident management, as we analyzed in Section 1.5.1, is presented in Table 4.5. This list is much smaller then what was presented in the previous chapter for service requests. Two incident-related examples are detailed in Chapters 14 and 15.

4.5.4 Business Challenges

Three main business challenge areas can be identified that impact the way the incident management pattern is defined: the *effectiveness of the organization*, *costs reduction*, and *improved external facing communication*.

- Effectiveness of the organization:
 - Better effective internal communication is needed.
 - Improved supporting IT landscape for efficient and effective operation is required.
- Cost reduction can be an important driver for change as a result of the financial crisis. In both the public and financial market budgets are under pressure that impacts the way processes are executed.
- Improved external facing communication.
- The communication with all external stakeholders needs to be improved and formalized.

Table 4.5 Incident Management Usage across Industry Processes

Public + Social Services	Environmental safety monitoring Disaster relief Terror/Epidemic threat handling
Financial Services	Dispute resolution Corporate communications Fraud and money laundering investigations
Insurance	Property and Casualty claims processing
Legal + Police	Crime recording
Energy and Utilities	Process safety management

Principle	Statement	Rationale	Implications
Communication	Formalize the communication between internal and external stakeholders	Improved internal communication leads to more efficient and effective operation. In some incident cases, the external communication needs to be formalized, for example, to avoid premature coverage in the press, that might disrupt the investigation. When the incident relates to a customer problem, the communication should be brought to the level the customer accepts as fair.	Technology needed to support the internal communication All external communication is redirected to a specific department.

Table 4.6 Principles for the Incident management classification

4.5.5 Principles for the Solution

The principles for an incident management classification are based on the business challenges as defined in the previous section and the common principles described earlier in Section 4.3. In Table 4.6, an additional incident management specific principle is added:

4.5.6 Case Management Design Components

4.5.6.1 Incident Management Design Category Overview

Table 4.7 provides an overview of the functionalities need for an Incident management design.

4.5.6.2 Case Lifecycle Design Category

The incident case pattern can be broken down into different phases, each with their own characteristics and behavior, as shown in Figure 4.5. These phases include *being prepared*, *monitoring the incident risk parameters*, *acting upon the incident*, and *evaluate*. The work done in the "being prepared" phase is used to prepare the workers what to do when an incident happens. This phase has a short duration. The "monitoring the risk parameters" phase checks, on a regular basis, the likelihood of an incident happening in the near future. The duration of this phase is infinite since this is a continuous activity. When an incident happens, the workers use the templates prepared earlier and can process the information gathered during the monitoring phase, depending on the incident type and severity. This phase can be of short, medium, or long duration. After the work is finished, an evaluation can be used to improve the template and preparation work and also to help adjust the risk parameters used in the monitoring phase.

We start with a short description of the *being prepared*, *monitoring the incident risk parameters*, and *evaluate* phases. For the *acting upon the incident* phase, a case flow model and case state model will be provided.

Being prepared: This phase ensures that the preparation is done in case an incident occurs. The team is ready to react directly in an organized way with checklists, work templates, activity lists, responsibility schemes, communication schemes (for both internal and external),

Case Design Category	Case Design Component	High Level Functionalities	Incident Management Specific Usage
Case lifecycle design	Case flow/ process design	The case flow specific for a case	See Section 4.5.6.2
	Rules design	Case lifecycle decision-making	See Section 4.5.6.2
		Entitlement determination	Only in Incident cases where customers are involved
		Progress measurement	In order to reach the end goals in a timely fashion, progress measurement is key in an incident where a variety of persons can be involved. Certain parts of the incident process are influenced highly by events and investigations that are started as part of an incident handling
		Data entry validation	Only in incident cases where customers are involved
		Information discovery	The monitoring part in the incident process relies on information discovery. See Section 4.5.6.2
	Event design	Interrupting events	See Section 4.5.6.2
		Non-interrupting events	See Section 4.5.6.2

Table 4.7 Incident Management Design Overview

(Continued)

Case Design Category	Case Design Component	High Level Functionalities	Incident Management Specific Usage
Case Data design information design	Data design	Case execution data	Be able to follow the case execution across the various stakeholders
	Case supporting data	Incident management-related data	
		Tactical related information	Compliance-related
		Strategic data	Forecasts, trend analysis, improvement of incident related predication models
Document and media design	Case input management	Incident Management input, structured and unstructured data	
	Case output management	Incident Management output, mostly documents	
	People relationship	Internal stakeholders	Case owner/workers, front-mid-back office, management, risk department
design	External stakeholders	Customers External specialists Related companies/departments Victims, witnesses, culprits Press	
Case user Po interface design	Portal design	One central view on the case	A centralized overview of the incident management proceedings is required by all stakeholders
		One overview of the customer	Only in Incident cases where customers are involved

Table 4.7 (Continued) Incident Management Design Overview

(Continued)

Case Design Category	Case Design Component	High Level Functionalities	Incident Management Specific Usage	
	Channel design	Internal communication	Internal communication between the case workers is key	
		External communication	The communication with all external stakeholders needs to be improved and formalized.	
	Dashboard design	KPI management	Cost driven	
		Performance management	Cost driven	
		Customer satisfaction determination	Only in incident cases where customers are involved	
Cross- functional design	Integration design	Application integration	It depends on the Incident Management-related IT landscape. Often, an Incident is operated in a hybrid application landscape	
		Case integration	All events that intervene with the incident are entered here. In the context of an Incident, case many investigation can be started that interact with the case lifecycle	
		Portal/Channel integration	Combine data from disparate sources to provision the portals	
	Roles and authorization	Internal profiles	It depends on the sensitivity of the Incident management case. Often incidents handle sensitive situations	
		External profiles	Profiles determine the amount of information that is shared with the external stakeholders via different channels	

Table 4.7 (Continued)	Incident Management	Design Overview
-----------------------	---------------------	------------------------



Figure 4.5 Incident management classification—phases.

and access to the information sources. This phase is very structured and predictable and contains milestones aimed at delivering checklists, templates, monitoring preparation, team setup, and training. The preparation phase also includes determination of the risk parameters for the monitoring phase. The duration of this phase is relatively short.

- *Monitoring the incident risk parameters*: In this phase, the risk parameters related to the incident are monitored against "live" data. This information helps predicting the likelihood of an incident in the near future. This, in turn, helps lowering the impact of the incident by early rampup of the incident workers. The monitoring measurements are based upon both structured and unstructured information and are enriched with existing information to execute the analysis. All information gathered in this phase is stored for later usage in incident investigations. Business intelligence analysis of the monitored information can also be used to discern trends and improve what needs to be measured. The monitoring process can be implemented as an automated "sleeper" process that gathers and analyzes the information. When the "sleeper" process notices a deviation from the risk parameters, an event is raised and an incident case can be started. Examples range from monitoring the bank payments for getting an insight in fraud and money laundering incidents to monitoring the tectonic movement for an earthquake incident.
- *Evaluate*: Feedback and experiences from incidents are used to improve incident handling and monitoring. These experiences help improve the incident execution in four areas:
- Incident handling, by adding or changing activities, adjusting the work force, and/or outsourcing some activities to external specialist organizations.
- Monitoring, by improving the risk parameters, measurement, and monitoring calculation
- Improving intelligence that helps to drive decisions in the incident handling, by including new data sources, better intelligence methods, or tooling
- Performance management for driving down the costs by improving the workload of the incident workforce.

4.5.6.2.1 Acting upon the Incident-Process Flow

When an incident happens, an incident case is created and the case workers start processing according to the checklists and templates developed in the preparation phase. This work includes investigation of the incident, helping people or groups who are inflicted by the incident, finding potential culprits (depending on the industry process), and preparing work after the incident is closed. Communication, both internally and externally, during the incident resolution, can be crucial depending on the incident case. This phase is guided by the templates and checklists that are created earlier in the *prepare* phase. Events are a main driver for altering the flow of activities. Both structured content and unstructured content are used for the investigation.

Figure 4.6 shows the case flow for an incident where four main activity areas can be seen: the *loop of incident activities, close case, communication* (both internal and external), and *events*.

In the case handling phase, the following stakeholders participate:

- External stakeholders. Generally, these stakeholders want/need to be informed on the progress of the incident. In a police investigation, this can be the press and in a fraud investigation, this can be the central bank and/or shareholders.
- Depending on the incident industry process, external communication offers a formalized channel with the external stakeholders as needed.
- The incident reporter role depends on the incident triggering event and the incident industry process. In an insurance claim case, the incident is reported by the customer. In a criminal case, the incident can be started by a police officer. Incidents can also be started as a result of (automated) monitoring, for instance, by a bank payment monitoring process that initiated a potential fraud incident.
- The case owner is actively leading the incident case and assesses/prioritizes the current situation and assigns work to case workers. When the incident can be closed, they initiate the closing activities.
- The case workers are responsible for the investigations and/or execution of the high-level incident work.

The case handling work of an incident consists of the following activities:

- *Register case*: An incident is created as a result of it being reported by either an incident reporter or an automated event.
- *Loop of incident handling activities*: This is a continuous loop of investigation and high-level incident work.
 - Assess and assign: After each loop, the case owner makes an assessment if the end result is reached. When another loop is required, the case owner makes a decision on which follow-up activities need to be done, how these are prioritized and who is going to execute these activities. The case owner is also responsible for initiating external communication via the external communication role.
 - Investigate: Part of the incident handling is investigating root causes, responsible persons, and ways to solve the incident problem. The output of each investigation is used by the case owner to determine follow-up activities in the execution of the incident. It is also used by (other) case workers to decide how to execute the work in the incident work activities (amend and aid). The investigations are started as separate cases and can be executed by internal case workers or outsourced specialists.
 - Amend and aid: Depending on the incident industry processes, incidents might cause damages to property, personal belongings, financial situations of persons or companies, or personal (health) situations. The incident "amend and aid" activities are aimed at alleviating these problems, for example, by providing insurance compensation or direct aid to persons.
 - Create result: All in-between results and decisions are stored for later reference and endto-end auditing.



Figure 4.6 Incident management classification—case flow model.

142 Oracle Case Management Solutions

- *Close the case*: In the assessment phase, the case owner decides, based upon predefined exit criteria, if the incident can be closed. The closing phase of an incident contains three main activities:
 - Start follow up work: Depending on the industry, process follow-up work might be needed. This work might not be part of the incident handling itself. For instance, a fraud investigation in a bank results in fraud case against one customer that needs to be brought to trial. The fraud incident case initiates the trial case, but is not responsible for the actual proceeding of the fraud case.
 - *Communicate*: The final result of the incident case is communicated with the external stakeholders via the external communication role.
 - *Evaluate*: An evaluation of the incident handling (optionally) is started. This helps in improving the incident handling process.
- Communicate:

This activity is aligned with the communication principle and specifically to "improved internal communication leads to more efficient and effective operation." Faster insight into (in-between) results among the different teams helps improve the speed and quality of the incident handling work.

This incident case flow can be interrupted by multiple events. These include *external stakeholder–related events*:

Ask for update:

The external stakeholders can ask for an update to get informed about the latest status.

Case owner-related events:

Time-out checks:

The case owner is responsible for the timely handling of all activities in the case. Every activity can have requirements around execution and/or communication timings and are related to regulatory requirements, agreements with external stakeholders, or internal performance agreements.

Intervene:

The case owner can intervene in the case execution when a situation occurs where the execution activities of the case need to be altered. Intervention can also include termination of the work.

Case worker–related events:

Results from other case workers: Internal communication is the key for the good operation of an incident case. Any (in-between) results from other case workers can influence the work others in the team are executing.

For all three stakeholders, incident reporter, case owner, and case worker:

■ Special events:

A situation occurring that has the potential to alter the flow of the case drastically. For instance, in a police investigation, a special situation such as new evidence or witnesses giving new information can influence the course of the investigations.

4.5.6.2.2 The Incident Handling Case State Flow

The state model of the incident handling contains two main work phases: *handle incident* and *close incident*, as presented in Figure 4.7. A phase is a series of activities leading toward a (in-between) result and closes when a milestone is reached or when the activities are terminated.

The *handle incident phase* contains three internal phases. The *lead incident* activities, assess, prioritize, assign, and communicate, are executed by the case owner. In this phase, the case owner determines if the end result is met, if follow-up work needs to be done. In addition, they determine what type of work needs to be done. This work comes in two flavors: *investigate* and *amend and aid*. These flavors are described in more detail in the incident case flow. The lead incident phase has three milestones through which this phase can be finished:

- Incident handling ended: where the incident handling exit criteria are met. This closes the main phase handle incident.
- Continue work: where the incident handling exit criteria are not met and more work needs to be done. Based upon the decision of the case owner, more investigation work, and/or amend and aid work is needed.
- *Terminate*: where the incident handling exit criteria are not met and the decision is taken to stop the incident handling work.

Both the *investigate* and *amend and aid* end with the same milestones:

- *Work is finished*, after which the result is handed over to the case owner, who uses this result in a next iteration of the lead incident phase.
- Terminate, where the work is terminated by the case workers. For instance, when the case
 owner terminates the case, all investigation, and/or amend and aid work needs to be stopped
 as well.

The *close incident* phase contains three activities: start follow-up work, communicate, and evaluate already discussed in the previous section.

Different events intervene in the lifecycle of the incident lifecycle. These are attached to main case level and to the different phases. On the main case level, four events can be identified: *intervene, reopen, ask update,* and *special events.* Attached to the *investigate* and *amend and aid* phases, three events are attached. These include *results from other case workers, terminate,* and *special events.*

4.6 Investigative Case Classification

4.6.1 Context

In Section 1.5.1, we analyzed industry patterns for the Forrester case management classifications and identified that the investigative type of case work is typically triggered by either a service request or an incident handling case in order to deliver intelligence needed in the case flow.

In the case flow models for service request and incident management, we added investigation activities as part of the case flow, as shown in Figures 5.3 and 5.6. These investigations are executed by the case workers as part of the service request provider or incident handling party, but they can also be outsourced to external specialists and/or (partially) automated. The characteristics



Figure 4.7 Case state model for incident management.

of these investigations range from a predictable set of activities to an event-driven case execution supported by data originating from both structured and unstructured data sources.

4.6.2 Determination

The following characteristics can be used to determine the case management classification for an investigative handling type of case:

- They are started as part of a service request or an incident.
- Investigative cases are aimed at delivering intelligence.
- The high-level execution of an investigation is predictable.
- Certain parts of the execution are highly influenced by events.
- Internal communication is the key.
- Data related to an investigation can be both structured and unstructured.
- An investigation can also initiate new investigative cases.
- (Optionally) Regulatory requirements may play a role.

4.6.3 Alignment with Industry Processes

All industry processes mentioned in Sections 4.4 and 4.5 can include investigations in the case flow.

4.6.4 Business Challenges

Investigative cases are started from a service request or an incident case as discussed earlier. The business challenges related to these classifications also impact the investigative type of cases. Here, we will use the business challenges that are needed in both the service request and the incident management cases:

- Better effective internal communication is needed in order to improve the effectiveness of the organization.
- Cost reduction is an important driver for changes as a result of the financial crisis.
- Stricter regulations and policies require end-to-end auditing.

4.6.5 Principles for the Solution

The principles for an investigative handling classification are based on the business challenges as defined in the previous section and the common principles described earlier in Section 4.3. In Table 4.8, an additional incident management specific principle is added:

Principle	Statement	Rationale	Implications
Communication	Formalize the internal communication	Improved internal communication leads to more efficient and effective operation	Technology needed to support the internal communication

Table 4.8 Principles for the Investigative Handling Classification

4.6.6 Case Management Design Components

4.6.6.1 Investigative Handling Design Category Overview

In Table 3.2, a design template is presented that will be used here to give an insight into required functionalities for incident management.

4.6.6.2 Case Lifecycle Design Category

During the execution of a service request or incident types of cases, intelligence is needed to drive decisions. This intelligence is obtained by starting separate investigation activities from the service request or incident case flow. Multiple investigations can be started from a single service request or incident. Each investigation case is started as a separate case with its own control and responsibilities and has loosely coupled interactions with the service request or incident cases. Figure 4.8 shows the relationship between service request or incident cases and investigation cases. This figure also shows that investigation cases themselves can initiate follow-up investigations.

The flow of the investigative case is targeted at delivering intelligence needed by the initiating service request of incident. The main part of an investigation flow is based upon a loop of investigation activities, as shown in Figure 4.9. After every loop, the resulting research information is checked for completeness and quality. The investigation loop continues if more information is needed and the resulting information is handed back to the service request or incident when enough information is available enabling a decision in the calling of service request or incident case (Table 4.9).

In the case handling phase, the following stakeholders participate:

- Service request or incident is the investigation initiating parties.
- The case owner is actively leading the investigative case and assesses/prioritizes the current situation and assigns work to case workers. When the investigation can be closed, the case worker initiates the closing activities.
- The case workers are responsible for the investigation work.



Figure 4.8 Loosely coupled multilayered investigations.



Figure 4.9 The investigative case flow model.

Case Design Category	Case Design Component	High Level Functionalities	Investigative Handling Specific Usage
Case lifecycle design	Case flow/ process design	The case flow specific for a case	See Section 4.6.6.2
	Rules design	Case lifecycle decision- making	See Section 4.6.6.2
		Entitlement determination	Only in incident cases where customers are involved
		Progress measurement	In order to reach the end goals in a timely fashion, progress measurement is the key to an investigation where a variety of persons can be involved, certain parts of the investigation process are influenced highly by events and other follow-up investigations can be started as part of the investigation
		Data entry validation	Not used in investigation
		Information discovery	Used in finding trends, evidence, clues in a combination of data stored, both structured and unstructured data
	Event design	Interrupting events	See Section 4.6.6.2
		Noninterrupting events	See Section 4.6.6.2

 Table 4.9
 Investigative Handling Design Overview

(Continued)
Case Design Category	Case Design Component	High Level Functionalities	Investigative Handling Specific Usage
Case information	Data design	Case execution data	Be able to follow the case execution across the various stakeholders
design		Case supporting data	Investigative handling related data
		Tactical related information	Compliance related
		Strategic data	Case process improvement
	Document and media design	Case input management	Investigative handling input, structured, and unstructured data
			Investigative handling output, mostly documents
	People relationship	Internal stakeholders	Case owner/workers
	design	External stakeholders	Stakeholder that started the investigation, originating from the service request, incident management, or another investigation
Case user interface designPortal designOne centre view on t case		One central view on the case	A centralized overview of the investigation proceedings is required by all stakeholders.
		One overview of the customer	Only in investigative cases where customers are involved

Table 4.9 (Continued) Investigative Handling Design Overview

(Continued)

Case Management Classifications

149

Case Design Category	Case Design Component	High Level Functionalities	Investigative Handling Specific Usage
	Channel design	Internal communication	Internal communication between the case workers is the key
		External communication	Standardized communication with the service request/incident or investigation case that started this investigation
	Dashboard	KPI management	Cost driven
	design	Performance management	Cost driven
		Customer satisfaction determination	Not used in investigation
Cross- functional	Integration design	Application integration	Depends on the investigation-related IT landscape.
design		Case integration	This is the entry point for all events that intervene with the investigation In the context of an investigation case other investigations can be started that interact with the case lifecycle
		Portal/channel integration	Combine data from disparate sources to provision the portals
	Roles and	Internal profiles	Depends on the sensitivity of the investigative case
	authorization	External profiles	Profiles determine the amount of information that is shared with the external stakeholders via different channels

The case handling work of an investigation consists of the following activities:

- *Create case*: An investigative case is created when a service request or incident requests intelligence.
- Loop of investigation activities: This is a continuous loop of investigation work.
 - Assess and assign: After each loop, the case owner makes an assessment if the end result is reached. When another loop of investigation work is required, the case owner makes a decision, which follow-up activities need to be done, how these are prioritized, and who is going to execute these activities. The case owner is also responsible for communicating status updates to the initiating service request or incident case.
 - *Conduct research*: This activity is the core activity for the investigation case. Here, all
 needed information is collected to conduct the research needed to draw conclusions.
 During this activity, follow-up investigations can be started. The collected information
 used in this research activities can be based upon a variety of (un)structured data sources.
 - *Report out*: After the research activity is concluded and conclusions are drawn, a report is created and approval/sign off is requested. The resulting information is stored for later reference and auditing purposes.
- *Review analysis and confirm case closed*: When the investigation loop is finished, the resulting work is reviewed and handed over to the service request or incident.

The investigative case flow can be interrupted by multiple events.

Service request or incident-related events:

• Ask for update: The service request or incident case owner or case workers can ask for an update to get informed about the latest status.

Case owner-related events:

Time-out checks: The case owner is responsible for timely handling of all activities in the case. Every activity can have requirements around execution and/or communication timings, and is related to regulatory requirements, agreements with external stakeholders, or internal performance agreements.

Case worker-related events:

Results from other case workers: internal communication is the key for the good functioning of an investigation case embedded in a service request or incident cases. Any (in-between) results from other case workers, participating in both the service request or incident and other investigation cases, can influence the work somebody else in the team is executing.

Intervention events relate to both service request or incident and the case owner:

Intervene: The service request/incident or case owner can intervene in the case execution when a situation occurs where the execution activities of the case needs to be altered. Intervention can also include termination of the work.

For all three stakeholders, service request/incident, case owner, and case worker:

• Special events: A situation can occur that has the potential to alter the flow of the case drastically.

4.6.6.2.1 The Investigative Handling Case State Flow

The state model of the investigative handling contains two main work phases: *investigate* and *close case*, as presented in Figure 4.10.



Figure 4.10 Case state model for investigative handling.

The *investigate* phase contains two internal phases. The *lead investigation* activities, assess, prioritize, assign, and communicate are executed by the case owner. In this phase, the case owner determines if the end result is met and if follow-up work needs to be done. They also determine what type of work needs to be done. The *perform investigation* activities related to the work done by the case workers as described in the investigation case flow model.

The lead investigation phase has three milestones through which this phase can be finished:

- *Investigation finished* where the investigation exit criteria are met. This closes the main phase *investigate*.
- Continue work where the investigation exit criteria are not met and more work needs to be done. Based upon the decision of the case owner, more investigation work is needed.
- *Terminate* where the investigative exit criteria are not met and the decision is taken to stop the incident handling work.

The *perform investigation* phase has the following milestones:

- Work ended in which the result is handed over to the case owner who uses this result for assessment in the next iteration of the lead investigation phase.
- *Terminate* in which the work is terminated by the case workers. For example, when the case
 owner terminates the case, all investigation work needs to be stopped as well.

The *close case* phase contains two activities: *review analysis* and *confirm case closed*. These are already discussed in the investigation case flow description. Different events intervene in the lifecycle of the investigative lifecycle. These are attached to main case levels and to the different phases. On the main case level, four events can be identified, *intervene, reopen, ask update,* and *special events*. Attached to the *perform investigate* phase are three events: *results from other case workers, terminate,* and *special events*.

4.7 Conclusion

In this chapter, the case management classifications as part of the case management solution framework are described; the service request, incident management, and investigative handling. These three case management classifications have different characteristics that determine what components the case management solution should contain. In the description of all three classifications, a checklist and a list of aligned industry processes is provided that help determine the classification type of a case.

Service requests deal with the process of delivering a service that a customer requested, and contain complex interactions with the customer. The case flow for service requests is predictable, and contains two main parts; the initial interaction with the customer (the triage) that is workflow based, and the service request handling that is case oriented. Interventions in the case originate from the customer or the case owner.

Incidents are aimed at solving a problem and include providing help to persons and finding the root causes. The high-level flow of an incident is predictable, and contains four phases; *being prepared, monitoring, acting upon the incident,* and an *evaluation.*

154 Oracle Case Management Solutions

The actual incident handling, *acting upon the incident* is highly influenced by special events. Internal communication is the key to the operation of an incident since multiple case workers can work on an incident simultaneously. Communication to all external stakeholders needs to be formalized.

Investigations are started by either a service request or an incident and deliver intelligence needed by these cases. Just like in incident handling, this flow is highly influenced by special events and internal communication is the key for success in the operation.

Chapter 5

Determine Solution Mapping

5.1 Introduction

In the third part of the management solution framework, we will determine which technical capabilities are needed to implement a case management solution and how these can be mapped onto an Oracle solution. To apply this in a structured way, the common reference architecture (CORA) model is used.*

The goals of this chapter are as follows:

- Show how to structure a solution with the CORA mode
- Show how the case management design components relate to the CORA application layers
- Show how functionality can be mapped to application layers and technical capabilities
- Show how to derive a decision what Oracle products to use based upon technical capabilities

5.2 Techniques Used to Determine the Impact of Functionality onto the IT Landscape

5.2.1 Common Reference Architecture Model

Current IT landscapes are getting more complex and contain multiple technologies supporting all the functionalities required by the business. The business demands more flexibility, agility, and faster time to delivery. Every change required by the business has an impact on the IT landscape. In order to be able to determine the impact of a business change on the landscape and identify the risks a vendor, neutral reference architecture is required. The CORA delivers a reference model based upon a set of application layers and capabilities. This way it provides an insight into the way an IT landscape can support business functionality, flexibility, and compliance simultaneously.

^{*} http://www.coramodel.com/

The model can be used at both the enterprise architecture level and at the project implementation level to design and implement functions with a mixture of "architecture styles" on the best-fitted platform available (or planned).

5.2.2 CORA Model Layering and Capabilities

The CORA model uses a stacked solution layer to break down the complexity and define a clear separation of duties as shown in Figure 5.1. The definition of the layers and capabilities described in this chapter are obtained from the CORA model (www.coramodel.com).

The layering used in the CORA model to structure functionality is based upon six horizontally stacked layers: channel access, presentation, composition, integration, application, and data. Vertically, next to these stacked layers, are security and compliance and IT governance. Although the layering is stacked, individual layers can be skipped if not required. Here is a short description of the various layers:

- The *channel access layer* groups client specific capabilities to enable access to information systems.
- The *presentation layer* groups presentation specific capabilities for displaying information to the user and handling of user-initiated events.
- The *composition layer* groups composition specific capabilities clustered in Orchestrated and Composed elements.
- The *integration layer* groups integration specific capabilities clustered in "synchronous communication," asynchronous communication, and common elements.
- The *application layer* groups core business capabilities.
- The *data layer* groups data specific capabilities clustered in data access and data storage elements.
- For all horizontal layers, the security and IT governance layers and capabilities may apply.

Every layer within the CORA model contains technical capabilities that can be used within a solution. Groupings of related capabilities within these layers can help in identifying specific functionality required for a solution. In Figure 5.2, an example of the composition layer is used to show the usage of groupings and capabilities.

The following main groups can be identified:

• Orchestrated capabilities contain the execution of a predefined, multistep business process models and lifecycle management supporting case management.



Figure 5.1 The CORA model layers.



Figure 5.2 CORA model composition technical capabilities.

- Composed capabilities contain the composition of existing application logic to create additional logic with specific business value, such as composition of data from different sources to be used within a business rules engine.
- Business rule capabilities relate to the business rules used in cross domain processes.

The CORA technical capabilities are summarized in Appendix – CORA capabilities.

5.2.3 Delivering a Solution with the CORA Model

The goal of this chapter is to discuss the determination of the required technical capabilities for the implementation of a case management type of solution. These technical capabilities will be used in Part III to determine which Oracle products are best suited for a specific set of required functionalities. The way a solution is determined using the CORA model is explained in the blog http://www.coramodel.com/cora-methodology-project-level/. In this blog, an overview of the activities describing how to determine a solution based upon the CORA model is shown. We will use a tailored version of the CORA methodology in this chapter to determine the solution mapping and the technical capabilities for the case management solution as shown in Figure 5.3 and Table 5.1.

The activities described in this chapter, as shown in Figure 5.3, help derive the solution architecture for every case management solution and then map this to Oracle products. How to make



Figure 5.3 Working toward a solution architecture with the CORA mapping.

Input	Provided by	Description
Principles	Architecture	Principles are general rules and guidelines, intended to be enduring and seldom amended, that inform and support the way in which an organization sets about fulfilling its mission. (http://pubs.opengroup.org/architecture/togaf8-doc/arch/ chap29.html)
User stories/ DILO	Business and end-users	User stories or <i>The Day in a Life of</i> (DILO) method gives insight in the current situation and the business challenges. The objective of the DILO is to walk through the process and experience the day in the life of an "an invoice" for example
Process models	Business analysts	The process models for this chapter are based upon the three case management architecture patterns: service request, incident management, and investigative handling
AS-IS situation	IT	The as-is situation describes the current system landscape, which usually contains different categories of application types: package-based systems, custom made, legacy systems, and SAAS applications

Table 5.1 CORA Methodology Input

a mapping toward a technology will be explained in further detail in Sections 5.3 through 5.5. The actual mapping to different Oracle products will be done in Part III.

The left-hand side of Figure 5.3 shows that initial input is needed before going through the activities. These inputs originate from different groups within the organization.

5.3 Perform a Functional Decomposition Using CORA Application Layers

5.3.1 How to Map the Case Management Design Areas onto the CORA Application Layers

In Chapter 3, the design categories and components for case management were introduced, as shown in Figure 5.4.

Before we look at the case management design areas in detail (and how they relate to the application layers), we first show an overview of the design areas across the entire CORA model. In Figure 5.5, this high-level mapping has been done with regard to the case management functional and cross-functional categories. This mapping shows that each design area maps to only a few application layers, meaning that every design area relates to a specific set of technology capabilities.

As an example, the core of the case management design, the *case lifecycle design* is mapped to the composition layer. We will see later in Section 5.4 that this functionality requires *orchestration* and *business rules* capabilities. Looking at the specific orchestration/rules related requirements and required capabilities enables us to choose between the Oracle products BPM/ACM, Siebel, Service Cloud, business rules, Oracle policy automation (OPA), and/or Endeca (as will be explained in Section 3).



Figure 5.4 Case management design categories and components.

5.3.1.1 Case Lifecycle Design

The case lifecycle design category describes the coordination of activities in reaching one of the end goals of a case and consists of three design components: Case flow/process design, rules design, and event design.

Most of the case lifecycle functionalities are mapped against the composition layer, as shown in Figure 5.6. The case flows and processes for the three architecture patterns, service request, incident management and investigative handling, are each mapped in a separate group



Figure 5.5 Case management design areas mapped onto CORA.

Channel access			
Composition			Event entry
Case flows/processes Service request handling Triage Service request handling Spawn investigation Investigation handling Investigation handling Spawn investigation	Incident management Prepare Monitor Incident handling Spawn investigation	Rules support Case lifecycle decisioning Entitlement determination Progress measurement Data entry validation Information discovery	Event handling Interrupting events A Noninterrupting events A
Integration			
			Event entry

Figure 5.6 Case lifecycle design mapping against CORA layers.

inside the composition layer. The rules functionalities, such as case lifecycle, decision-making, and information discovery, are placed inside the rules support group in the composition layer. The only functionality that is mapped against three layers is the handling of events. While the actual *event handling* is placed in the composition layer, the event entry originates from two layers:

- When an event is entered via human interaction, the event entry is placed in the channel layer.
- An event originating from an application is automatically entered via the integration layer.

5.3.1.2 Case Interaction Design

The case interaction design enables maximizing support for the case work force. This design area contains three design parts: portal design, dashboard design, and channel design.

The *case interaction design* category is mapped on three layers: the channel, the presentation, and the composition layer as shown in Figure 5.7.

The case management portal design components are mapped onto the presentation layer. This contains the user interface functionality supporting the case owner/workers, customers, and management.

Channel access
Internal communication
Presentation
One central view One central view of the case One central view interface
Composition
Dashboard KPIs composed data Customer satisfaction determination

Figure 5.7 Case interaction design mapping onto the CORA layers.

- The *case management channel design components* relate to how the case supporting application is accessed, and contains the *internal and external communication*. The channel design is mapped onto the channel layer.
- The case management dashboard design components support the KPI management, performance management, and customer satisfaction measurement. It is mapped toward two layers: the dashboard data determination is part of the composition layer while the user interface part of the dashboard design is part of the presentation layer.

5.3.1.3 Case Management Information Design

The case information design deals with all sorts of data and documentation flowing through the case and maps the relationships between all persons and/or assets involved in the case. The case information design category contains three design components, data design, document and media design, and people relationship design.

The *case management information design* category is mapped onto the composition, application, and data layers, as shown in Figure 5.8:

- Data residing in the *composition* layer
- The case is supported with information originating from a composition of data from three different areas, as described earlier in Section 3.3.3.
 - Case execution data
 - Case supporting data
 - Strategic related information
 - Tactical related information
- This data originates from different sources, including the supporting case or BPM application.
- In the *data* layer, the access and storage capabilities are described that are needed to support the case management solution. These capabilities are described in Appendix. The composed case data that is plotted in the composition application has a direct relation with the storage capabilities. The case execution and supporting data is strongly related to transaction data and master data. Additionally, the decision management, as part of the case execution data, can relate to unstructured data. Tactical/Strategic data look at aggregated data. The supporting case management and BPM application require repository capabilities in the data layer.
- The *application* layer is the place where applications, such as package-based systems, custom made, legacy systems, and SAAS applications, deliver the specific data needed in the case management solution. Specifically customer relationship management (CRM) type applications are required for the people relationship design. Documents/media as part of the input and output management for the case management solution is placed within the application layer since specific products are tailored for this functionality.

5.3.1.4 Cross-Functional Design Areas

The cross-functionality design category contains the integration design and the roles and authorization design that both support the other functional design categories.

The cross-functional design areas (i.e., integration design and roles and authorization design) map directly to the CORA integration and security layers.

The three main integration functionalities are mapped onto the Integration layer as shown in Figure 5.9. All communication is based upon a canonical data model that is the key in the exchange of all information in all integration functionalities.



Figure 5.8 Case information design mapping to application layers.

Application integration Application connectivity Image: Content management interaction	Case integration Case/process interactions Case/event interactions Case investigation interactions	Portal/channel integration Portal data provisioning A Channel communication A	
Data Canonical data model (CDM)			

Figure 5.9 Case integration design mapping to application layers.

5.4 Determine Required Technology Capabilities

Every case management solution has its own specific characteristics and can be described with the help of the case management design categories and components. In the previous section, we have shown the relationship between the case management design categories and the CORA layers. The knowledge of the relationship between the design categories and these layers help identifying what technical capabilities could be used to determine the solution. And subsequently, understanding the need for a set of technical capabilities supports the identification of the need for a specific technology, such as making a decision between the Oracle BPM/ACM and Siebel products.

In this section, we will look at what technical capabilities relate to the case management *lifecycle* design category, and under what conditions these capabilities should be included in the solution. The lifecycle design is supported by "core" capabilities and "supporting" capabilities. The "core" capabilities for the case management lifecycle design originate from the composition layer, as we have seen in Figure 5.6. The "supporting" capabilities are additional to the solution, and help in the decision process what (Oracle) products to use. Figure 5.10 shows the mapping of the lifecycle design functionalities are shown. The right-hand side contains the capabilities, with a distinction between the "core" capabilities and the "supporting" capabilities. Groupings of related capabilities are denoted by white rectangles.

5.4.1 Case Flows and Processes Capabilities

The *case flows and processes* are mapped onto the composition *orchestration* capabilities. The orchestration capabilities facilitate the case management lifecycle with (automated) support for predefined multistep business process models. Table 5.2 gives an overview of the orchestration capabilities and the usage in a case management solution.

5.4.2 Event Handling Capabilities

The *event handling* function is related to multiple capabilities. First of all, the event handling interferes with the *lifecycle management* of the case:

- Interrupting events might impact the lifecycle execution of the case. From within the lifecycle
 management, new activities can be started or existing activities can be closed.
- Noninterrupting events do not impact the lifecycle execution of the case, but can trigger some human and/or automated activities.

Events originate from both human and automated interactions. Human interactions originate from within the channel layer (electronic channel capability). Automated events enter via the integration layer (message capability).

5.4.3 Business Rules Capabilities

As described in Section 3.3.2.3.2, different rule functionalities can be distinguished in a case management solution. These rule functionalities require different data types:

- Transactional data that relates to all dynamic data entered in the case.
- Aggregated data that is an assembly of transactional data often used in progress measurement and key performance indicator (KPI) monitoring.
- Unstructured data that relates to input data often used in investigations and can take any form, such as documents, media, and data streams.



Figure 5.10 CORA technical capability mapping explained for the lifecycle design.

Orchestration Capability	CORA Definition	Usage in Case Management
Lifecycle management	The end-to-end management of all activities in a business process and case and monitoring of the progress against predefined parameters. There is a close interaction between the lifecycle management and the orchestration capabilities A2A, H2A, H2H and B2B	This is the core functionality for the orchestration of all activities in a case. The case management paradigm is goal-driven. The lifecycle management supports the case execution with the management of activities and the monitoring of progress in order to reach one of the end goals
Application- to-application (A2A)	Capability to allocate tasks to applications within an organization and control the required routing of these tasks	All automated activities in a case relate to the A2A capability. The automated activity can be either a call to an external application or part of the case management solution itself In Section 4.4.4, the operating models are discussed for a service request classification. The operating model <i>operational excellence</i> is aimed at delivering a standardized product for a low price. The A2A capability enables the standardized delivery of tasks through automated handling of activities
Human-to- application (H2A)	Capability to allocate tasks to users and control the required routing of these tasks	Interactions between the case application and different stakeholders can be part of the case execution. When the interactions are predefined, these are modeled as standard BPM processes In situations where ad hoc interaction is required, these interactions need to be modeled at runtime
Human-to- human (H2H)	Capability to collaboratively use the same resource by different users	In Chapter 4, the three case classifications are described, the service request, incident management, and investigative. Each classification has specific needs for human to human interactions Service request: When the operating model is customer centric, direct interaction with the customer is a prerequisite for success Incident management: Both internal and external communication are the keys in the execution of an incident case Investigations: Internal communication between all the case workers is crucial in the execution of the investigation
Business to business (B2B)	Capability to allocate tasks to applications between organizations and control the required routing of these tasks	Both the service request and the incident handling classifications can have requirements for the inclusion of B2B activities. Outsourced work that falls under control of a case is a good candidate of a B2B interaction

Table 5.2 Orchestration Capability Usage in the Case Lifecycle Design

In Table 5.3, an overview is provided of the business rule functionalities, the relationship with the different data types, and an explanation of the usage in a case management context.

5.5 Decide What Technologies to Use

The first step in deciding what technology to use is choosing the required functionalities and the supporting technical capabilities. In the previous section, we described the relation between the functional areas and the technical capabilities for the case management life cycle design. Adding a functional area into the requirements can result in the addition of a certain set of technical capabilities needed in the solution.

After we have decided what functionality is included in the case management solution, and what technical capabilities are needed, we can choose the technology based upon the required capabilities. Every technology product has specific product features that relate to a specific capability. In Section III, we will show when to use one of the three main Oracle case management products: Siebel, Oracle BPM/ACM, and Oracle service cloud.

The diagram type as provided in Figure 5.10 will be extended to show the relation between the required functionalities, technical capabilities, and products to use in the case management solution in the remainder of the book. Figure 5.11 shows an example of the technology inclusion for Oracle BPM/ACM:

- On the left-hand side the required functionalities are shown.
- The next column shows the required technical capabilities.
- The remaining columns to the right show the products to use.

In Figure 5.11 the capabilities are shown for the case life cycle design category. This category contains functionality for case flows/processes, events, and rules support.

In the example, we left out the event functionality and all the rules support except for case management decision-making. This remaining functionality is the bare bones of the case management life cycle design and contains all the orchestration features that a case management solution requires. In Section III, an explanation is given for the product choices shown in Figure 5.11.

5.6 Conclusion

Determining the *solution mapping* is the third and last part of the *case management solution framework*. In this chapter, we used a vendor-agnostic reference architecture model, the CORA model to make the translation from requirements toward a decision as to what technology to use:

- Perform a functional decomposition onto the CORA layers.
 This shows the impact of a set of requirements on the application landscape.
- Determine the required technology capabilities.
 Based upon the required functionality a decision can be made what technical capabilities to include in the solution.
- Decide what technologies to use.
 The technical capabilities help make the choice for specific technologies to implement the solution.

We showed a mapping technology with related diagrams that enables us to translate the case management requirements toward a technical solution. In Parts III and IV, we will use these solution diagrams to show what Oracle products support specific functionalities.

Table 5.3 Business	Rule	Data	Usage
--------------------	------	------	-------

Functionality	Usage in Case Management	Transactional Data	Aggregated Data	Unstructured Data
Case lifecycle decision making	This rule functionality guides the lifecycle management in the decision process. Any change happening in the status of a case can result in changed behavior of the case lifecycle. The ability to make decisions on a change that happened in the lifecycle of a case is supported by business rules Examples of changes in the context of a case are: a milestone reached, an activity finished, a document entered, a person interacting with the case, an automated event informing the case about a changed situation, or a data object changed	X	X	
Entitlement determination	The determination of an entitlement is based upon regulations and policies and contains complex rules The data contained in the supporting business entities originates from transactional and aggregated data types	Х	Х	
Progress measurement	During the execution of a case insight is needed in the progress of the case and is calculated via business rules. The data contained in the supporting business entities originates from aggregated data types		Х	
Data entry validation	Data entry validation is required in situations where customers or other stakeholders enter data into the case. This data needs to be validated for instance against regulations and policies The data contained in the supporting business entities originates from transactional data types	Х		
Information discovery	Information discovery is needed in areas where large datasets of (un) structured data needs to be investigated for trends of direct impact on a case under investigation. The data contained in the supporting business entities originate from unstructured, transactional and aggregated data types	X	X	X

		Technical capabilities	ACM/BPM	Oracle rules-centric products	Oracle SOA	suite	Other oracle products
/	Case lifecycle design category						
	Case flows/processes	Lifecycle management	Oracle ACM			SOA suite 🔿 integration	
	Service request	H2A orchestration	Oracle BPM		Human tasks 💫	Mediator	
	Incident management	A2A orchestration 💫	Oracle BPM		Oracle BPEL	OSB Adapters	
		H2H collaboration 💫					Oracle communicator
		B2B orchestration			Oracle B2B		
	Rules support Case lifecycle decisioning	Business rules		Oracle business rules			

Figure 5.11 Combining functionality, capabilities, and technology in one overview.

Chapter 6

Concluding Remarks

We introduced the case management solution framework in this part. This framework enables the functional and technical designs of a case management solution with a set of *case management design components, three case management classifications,* and the *case management solution mapping.*

The *case management functional design components* and the *solution mapping* provide a means to create the functional design and the technical mapping for a case management solution. The *case management classification* gives an understanding of what functionalities are needed for a specific case management solution.

Figure 6.1 gives insight into the way the case management solution framework is used to determine a solution for a specific case. In the process of determining the solution architecture for a case management solution, the *case management solution framework* is used as follows:

- Determine the *case management classification*, service request, incident management, or investigative handling. This gives an insight into what functional requirements are needed. Specifically, insight is provided in the case lifecycle design category.
- Determine the *case management functional design*, based upon the chosen case management classification, and discover the high-level functional needs for the three functional categories and one cross-functional category.
- Determine the *case management technical solution* including the technical capabilities and mapping toward Oracle products based upon the required functionalities.



Figure 6.1 Creating a solution with the case management solution framework.

ORACLE CASE MANAGEMENT TECHNOLOGIES

Ш

Introduction

In the following chapters we introduce the three main products within the Oracle stack that support the case management functionality, Oracle BPM/ACM, Oracle Siebel, and Oracle Service Cloud. Each of these products has different characteristics that determine how it is (best) used in a case management scenario. Integration capabilities can be delivered additionally via the Oracle SOA Suite for all three scenarios. Intelligence in relation to case execution, the Oracle products that support the delivery, and the usage of intelligence will be discussed in Chapter 10.

Product Chapters

The product chapters will be structured according to the case management functional design presented earlier in (Section II—Chapter 1—Case Management Functional Design). Figure S.1 shows the functional design areas that will be used to describe the main features of a product.

Additionally for each technology we will map the life cycle design toward the technology capabilities according to the technique we defined in (Chapter 5 - Determine Solution Mapping), as shown in the example in Figure S.2.

In the usage characteristics description for the products we will look at the main characteristics of the product:

- 1. Main focus on the functionality in the product:
 - a. What is the focus of the design of the product?—Life cycle design, information design, and/or interaction design.
 - b. What classifications does the technology support?—service request, incident management, and/or investigative solution.



Figure S.1 Case management design categories and components.

- 2. *Delivery type*: The implementation type determines how the product is delivered: as a single entity that contains the full case management solution or an umbrella type of solution, an end-to-end control that interacts with other applications in the landscape.
- 3. Product type: Is the product package based, middleware based, or cloud based?
- 4. *Agility at run-time*: In (Chapter 2 Adaptive Case Management (ACM)–Related Topics) we discussed the case management types, production case management (PCM) and adaptive case management (ACM). Both types are an approach for supporting knowledge workers with the case execution. As was concluded in that section, the main difference between PCM and ACM is that in the ACM type of systems, all aspects of the case can be changed by the case worker at run-time, while in the PCM type the structure and the assigned behavior of the subsystems stay the same throughout the case progression. In summary, ACM systems offer more agility at run-time than PCM systems (Table S.1).

Intelligence in Relation to the Case Execution

In the chapter, the usage of intelligence within the context of case execution is described. Intelligence functionalities are related to three intelligence information levels—operational, tactical, and strategic—and are supported by different Oracle products. Oracle intelligence products support all three case products: Oracle BPM/ACM, Oracle Siebel, and Oracle Service Cloud.



Figure S.2 Lifecycle management Siebel mapping.

Key Characteristics	РСМ	ACM
Type of work	Largely routine and repetitive	High knowledge and creative problem- solving work
Decision making	Usually from predesigned sets	Can go beyond the anticipated
Type of collaboration	Mostly known ahead of time	Special services may be needed
Work ownership	Adequate involvement	Send of responsibility
Case volume	Typically large	Typically moderate

 Table S.1
 Key Characteristics of a PCM and an ACM Type of Case

Chapter 7

Oracle BPM/ACM

7.1 Introduction

In this chapter, we will describe the Oracle Adaptive Case Management (ACM) product as part of the Oracle Business Process Management (BPM) suite. The combination of Oracle ACM and BPM is built on top of the Oracle SOA Suite, which is a middleware product with a strong set of integration and interaction capabilities.

The goals of this chapter are as follows:

- Give a high-level overview of Oracle BPM/ACM and the SOA Suite
- Describe the usage characteristics of Oracle BPM/ACM
- Describe how Oracle BPM/ACM product supports the case management lifecycle functionality
- Show how the Oracle SOA Suite supports Oracle BPM/ACM with integration and interaction capabilities
- Give an overview of supporting Oracle products

We start this chapter with an overview and usage characteristics of the Oracle BPM/ACM and related SOA Suite products. Then we describe the product features of Oracle BPM/ACM/SOA Suite according to the case management functional design as presented earlier in Chapter 3, with the main focus on the lifecycle functionality.

In Chapter 17, a tutorial is presented on Oracle BPM/ACM where the product will be discussed extensively.

7.2 Oracle BPM/ACM/SOA Suite Overview

The Oracle SOA Suite is the middleware product within the Oracle architecture to support all integration in an enterprise wide landscape and is based upon a service-oriented (SOA) and eventdriven architecture (EDA). The strong capabilities of the SOA Suite are in the area of integration and interaction, and can nowadays be delivered on premise and in the cloud. The Oracle SOA Suite is created based on the principles complete, open, integrated, and reuse to the max. The last principle enables simplification and fast delivery of a new functionality in the SOA Suite:

Reuse of components across the Oracle SOA Suite stack, for instance a database adapter, is essential. It improves stability and predictability of the solution. BPM and ACM are just one of the components plugging into the stack and both reuse all other components.

Many books have already been written on the Oracle SOA Suite, so we will not dive in detail in the behavior of the Oracle SOA Suite.

Oracle BPM was added to the Fusion Middleware 11g stack with the Oracle SOA Suite 11.1.1.3 release in 2010. With this release, Oracle BPM provided a BPMN 2.0 compliant process execution engine complemented with a complete process modeling and development environment that addressed both developers and business analysts. Oracle BPM is layered on top of Oracle SOA Suite and hence benefits from the powerful integration infrastructure; which is seamlessly available to a BPM process.

Customers of Oracle BPM have been effectively using the product to manage a large part of their business processes. They have been able to derive a number of benefits from having successfully automated and streamlined core business processes, such as real-time insight into their processes, quick response to changes in business conditions, reduced costs due to processes optimization, etc.

Although Oracle BPM with its BPMN-based engine has successfully addressed a large part of routine and structured processes such as invoice processing, order processing, etc., there still remain areas of a business that have not been automated using BPM.

In Chapter 2, we discussed the need for "adaptively agile" systems. Solutions in such a system have to dynamically sense external events and respond appropriately. Variations and unpredictability are key attributes of such solutions. Traditional BPMN can be used to build solutions that can interact in a sense-and-respond manner but the responses are based on predefined actions plans. Any unforeseen variations cannot be handled by these applications without having to go back and modify the process models. We also discussed the important role stakeholders play in enabling adaptive behavior of such systems. Oracle BPM Suite extends the scope of BPM to address these highly variable, event driven, knowledge-work based processes.

You may wonder why should we use the Oracle BPM Suite to handle adaptive case management?

We have seen that the nature of work addressed by adaptive case management is quite variable in nature and depends significantly on the case workers' knowledge and expertise to progress the case. The individual tasks that case workers perform could be routine in nature. For instance, a task may be a simple approval workflow or an automated process to complete a set of actions. This means even in an adaptive case-based application there is a need to handle routine tasks.

As we have mentioned, Oracle BPM Suite already handles routine processes, with all levels of complexity, very well. By extending this offering to include the notion of a "case" and a case event model, you can now build adaptive case management applications that can leverage BPM for routine tasks within a case. Since it is layered on top of BPM, ACM is able to enjoy the same benefits as BPM and provide case analytics, opportunities for optimization, and automation, all this without constraining the knowledge worker with predefined structured processes.

As you will see in the following chapters, Oracle ACM builds on Oracle BPM and therefore fully leverages BPM and SOA stacks to provide a powerful case management framework. This allows for creating case management applications that span enterprise business applications. This is possible because of the powerful adapter framework available in the BPM stack with over 200 connectivity options to a large variety of business applications.

7.3 Oracle BPM/ACM Usage Characteristics

With the strong support of the Oracle SOA Suite integration capabilities, Oracle BPM/ACM is a good candidate to use in an enterprise wide, cross application scenario, where work and information are shared across a group of case workers. Oracle ACM takes care of the end-to-end management and monitoring of all activities, where the activities could be shared across tasks defined in Oracle BPM or in external applications. One of the strong capabilities of Oracle ACM is the ability to act upon events; any change in the lifecycle of a case can result in a new chain of activities to be spawned off.

Oracle BPM/ACM is a middleware product and based upon composition of components via visual programming as will be described later in this chapter and shown in Chapter 17. This product feature, combined with the ability to deliver in an agile way, enables the delivery of (partial) solutions in short cycles. This helps in managing the end-to-end case process and getting a better understanding of the complexity in a gradual way.

Oracle ACM offers different aspects of flexibility at runtime. Adhoc tasks provide the case workers with the ability to start tasks that are not part of the predefined case model. The flexible stakeholder model allows (re)assignment of tasks and responsibilities at runtime. Another level of flexibility is the event-driven capability of BPM/ACM. Rules-based event handlers can be used to alter and adapt the case by making activities available, withdrawing existing activities, change milestone states, etc. Oracle ACM also allows adding on new activities from a library of predefined activities.

Table 7.1 summarizes the usage characteristics of the lifecycle design functionalities for Oracle BPM/ACM.

7.4 Case Management Lifecycle Functionality with Oracle BPM/ACM

In Chapter 3, we introduced the various aspects for the design of a case management solution. Oracle ACM and BPM directly relate to the case lifecycle design category, shown in Figure 7.1.

Focus Area	Product Characteristics			
Case management design area	Life cycle design			
Classification support	All three case management classifications are supported: service request, incident management, and/or investigative solution			
Delivery type: Full implementation or umbrella type	Umbrella type			
Product type	Middleware			
Agility at runtime: ACM or PCM	ACM type: Ad hoc processes as part of a case can be started Flexible stakeholder model Ability to add new event publishing sources			

Table 7.1 Usage Characteristics of Oracle BPM/ACM



Figure 7.1 Case management lifecycle design category. The case lifecycle design category describes the coordination of activities in reaching one of the end goals and consists of three design components: Case flow and process design, rules design, and event design.

When implementing the lifecycle design for a case management solution with Oracle ACM, three products form the core of the solution: ACM, BPM and Oracle Business Rules. Additionally specific Oracle SOA Suite products, such as human tasks and BPEL, will complete the lifecycle area functionality.

7.4.1 Case Flow and Process Design

7.4.1.1 Case Flow Design with Oracle Adaptive Case Management

ACM was introduced in April 2013 as a feature set on top of Oracle BPM. With this release Oracle BPM Suite can now be used not only for structured BPMN-based processes but also for unstructured and adhoc processes. When using Oracle BPM with ACM, it is now possible to create knowledge-worker driven, event-based and content-focused processes, with the focus on reaching an end goal by acting on events instead following a predefined path.

7.4.1.1.1 Support for Structured and Unstructured Work

Actions performed by case workers to progress the case are represented by activities. An activity can be modeled as a structured process, a task, or as a custom implementation. The ability to define activities using these different implementations makes it possible to provide case workers with a catalog of activities to perform. These activities, some of which could be structured processes, can be performed in any order and any number of times as the case worker deems appropriate, allowing a mix of structured and adhoc or unstructured work.

7.4.1.1.2 Key Features

At a high level, Oracle ACM contains the following key features:

- The case is driven by events, originating from both internal and external origin.
- The case is a container for data, content, and relationships with other cases.
- Activities are the predictable part of the case.
- Audit trails capture every event that occurs and actions that are taken during the lifetime of a case.

- Case policies allow putting in controls to help guide the case worker through the case progression.
- A flexible stakeholder model that can be changed at runtime.

7.4.1.1.3 Key Concepts

Following are key concepts in ACM:

- Case: A case is the logical container of data, content, history of events, and actions taken,
- *Case model*: Defines the structure of a case type. This includes case attributes such as name, category, priority, due date, etc. It also includes definitions of case milestones, stakeholders, and structured and unstructured content that the case depends on.
- *Case instance*: A specific instance of a case type and is a collection of documents, data, and case activities that are used to process the case and audit the progress of the case.
- *Case folder*: Case folders are specific locations in the configured content management system where all documents are stored. These folders can either be generic or instance specific.
- *Case data*: the data objects used by the case instance during the case progression. Case data can either be internal (stored in line with the case instance and managed by Oracle ACM) or external (managed by external applications, referenced by Oracle ACM).
- *Case lifecycle*: Case lifecycle tracks the progression of a case instance from start to close and can have states such as active, suspended, aborted or closed,
- *Milestones*: Milestones help track the progression of the case and are used to mark a specific point in the case progression.
- *Case activity*: Case activities allow case workers to perform work in the context of a case. These activities can be BPM processes, workflow tasks, or any custom processing using Java.
- *Case event:* Case events are events that are raised as a consequence of changes to the case. Events are raised for case state changes, activity changes, milestone changes, stakeholder changes, data changes, and document changes.

7.4.1.2 Process Modeling

Oracle BPM features a comprehensive, browser-based composer that business analysts can use to model business processes. The composer provides all the aspects of modeling functions, which allows a process modeler to not only draw out a process model that reflects the real-world business process but also adds implementation artifacts to it, which means right from the browser, depending on appropriate privileges, one can model, simulate, test, and deploy a process to the process engine.

The web-based composer, as shown in Figure 7.2, is complemented by the *BPM Studio* and is part of the *Oracle JDeveloper* IDE. This provides a development environment more suitable for technical developers. A typical BPM application is modeled by business analysts and includes enough details for the developer to take over. The developers then add technical details such as service invocation, complex data mapping, etc., and move the project to a standard code promotion cycle. A key aspect of the BPM project development lifecycle is that both the business analysts and developers work on the same project through a shared repository that is part of a fully collaborative business IT development environment.

Oracle BPM implements BPMN 2.0 standard; which not only defines the notations but also execution semantics. Oracle BPMN engine executes these BPMN 2.0 models directly without having to generate any Java or other intermediate execution code. This means that what you have modeled is what is executed, "what you see is what you execute."

RACLE Business	Process Comp	oser		Preferences	Help *	Log Out Welcome, Ja	mes Cooper 🔘
CCDR > 🕄 CCDCase		Editing	-				🗵 🕹
oject Home							Close Project
CCDCase 🕄				Created on 5:53 PM by JC	OOPER	Recent activity	Snapshot
🚊 5:53 PM by jcooper	🏦 Private Project	🗇 Language: English Edit		More infor	mation	Unpublished changes o addet, 0 modified, 0 deleted	
Processes (7)	Views +	Search		+ New		Creating Project. Published by posper on 5.53 PM.	cesa.bpmn
Human Tasks (2) Web Forms (0) Business Components (49)		WriteOffChargeProcess		iotifyCustomerProcess d: NotifyCustomerProcess	8		FromMerchantProce FromCustomerProce cess.bpmn
∰ Simulations (0) IIIE Business Indicators (0) ⊡ KPIs (0) All (59)		RequestDocumentFrom		lotifyMerchantProcess d: NotifyMerchantProcess	ā	 ・ NotifyCustomerPro ・ ト NotifyCustomerPro ・ ト InitiateChargeback ・ ト PrepareCaseProce 	cess.bpmn Process.bpmn Is.bpmn
		PrepareCaseProcess		equestDocumentFrom d: lequestDocumentFromMerc antProcess	Û	Review Charge Disp	uteTask task Task task .bi
		InitiateChargebackProc				Organization Manage project roles and bu Manage your Milestones	siness parameters

Figure 7.2 Oracle BPM composer.

7.4.1.2.1 Work Management

The human workflow service engine handles all aspects of human work assignment within a BPM process. Oracle BPM provides commonly used workflow patterns as first-class activities in the BPM modeling palette. Hierarchy-based approval workflow and group voting are some examples of these workflow patterns. Using Oracle Business Rules with human workflow rule-based dynamic work assignment can be defined.

7.4.2 Event Design

Case progression in Oracle ACM is event driven. Every change to any of the key elements of a case such as data, documents, activities, milestones, and stakeholders generates appropriate events. Case events are also generated for any case state changes such as the start of a case, close of a case, suspension of a case, etc. In addition to these events generated by the case engine, Oracle ACM also supports user-defined events that can be used to register external events in the case.

7.4.3 Rules Design

As we have seen in Section 3.3.2.3.2, different rule types support the case management execution:

- Case lifecycle decision-making
- Entitlement determination

- Progress measurement
- Data entry validation
- Information discovery

We focus in this chapter on the first rule type, case lifecycle decision-making, since this relates directly to the execution of a case in the Oracle ACM/BPM products. The other rule type functionality is handled in Chapter 10.

Oracle Business Rules is a service component in Oracle BPM Suite that provides decision services to the BPMN process for flow control, task assignment, determination of participants, etc. The rules engine is an inference-based rules engine that allows definition of complex rules using easy-to-use decision tables, verbal rules, and if-then pattern matching.

7.4.4 Life Cycle Design Product Mapping to Oracle BPM/ACM/SOA Suite

We finish the lifecycle design section with a product mapping of the case management lifecycle functionality toward the Oracle BPM/ACM/SOA Suite product sets. In Chapter 5, a drawing technique was introduced that enables us to make a translation from required functionalities towards Oracle products.

The case lifecycle design category contains three main design components:

- Case flows and processes
- Event handling
- Rules support

In the mapping, we include the case lifecycle decision-making rule type. The remaining rule types are described in more detail in Chapter 10.

7.4.4.1 Case Flow and Process Mapping

In Figure 7.3, we summarize the capability mapping toward Oracle BPM/ACM and SOA Suite products. All capabilities in the case flow and process design components relate to only a few Oracle products. Most products in Figure 7.3 have been introduced already earlier in this chapter. Products that have not been introduced will be summarized in the following description.

7.4.4.1.1 Lifecycle Management and Rules Support Case Lifecycle Decision-Making

Both the Oracle ACM and BPM products support the lifecycle management, with Oracle ACM having a more event-based approach and Oracle BPM following a structured predefined path. This functionality is strongly supported by Oracle Business Rules for all decisions to be made during the lifecycle of the case.

7.4.4.1.2 Human-to-Application (H2A)

As part of the case execution, human activities can be initiated. Both ACM and BPM can be the initiating actor for human activities. The human workflow component of Oracle BPMN Suite takes care of the actual execution of the human activity.

	Technical capabilities	ACM/BPM	Oracle rules centric products	Oracle SOA	suite	Oracle webcenter
Case lifecycle mapping to the oracle service cloud Case flows/processes	(Lifecycle management)	Oracle ACM A Oracle BPM			SOA suite O integration	
Service request Incident management	(H2A orchestration 💫	Oracle BPM A Oracle ACM A		(Human workflow A	OSB A	
Investigative handling	A2A orchestration	Oracle BPM		Oracle BPEL		
	H2H collaboration	Oracle ACM				Real-time collaboration
	B2B orchestration			Oracle B2B		
Rules support Case lifecycle decisioning Other rules functionalities are detailed in chapter 10	Business rules 💫		Oracle business rules 💫			
Event support	(Messaging A)			Oracle OEP	Event delivery network]

Figure 7.3 Oracle Lifecycle mapping to Oracle BPM/ACM/SOA Suite products.
7.4.4.1.3 Application-to-Application (A2A)

From within the case execution, automated tasks can also be initiated. This is handled by either the SOA Suite Business Process Execution Language (BPEL) product or by BPM.

BPEL was created in an effort to standardize process execution, just like web services helped in standardizing communication between applications. The standard is defined as part of the OASIS standard group.

Both BPM and BPEL can be used to define processes and both can contain automated and human activities. BPEL is a block-structured, top-down process model that is ideal for straightthrough processing (STP), which is a typical A2A integration flow. When the process also includes H2A capabilities or cannot be easily modeled as top-down flow and requires more of a directed flow graph, BPMN is a better choice. Mixing both BPM and BPEL is a good way to separate pure technical flows from business-aimed processes.

7.4.4.1.4 Human-to-Human H2H Collaboration (H2H)

Oracle WebCenter Real-Time Collaboration supplies the end users with direct interaction capabilities. The main functionalities of this product are instant messaging, presence, chat rooms, and web and voice conferencing.

Oracle ACM supports simple H2H capabilities with the commenting feature on the case level, this enables users to share insights and thoughts around the case execution.

7.4.4.1.5 Business-to-Business Orchestration (B2B)

The Oracle B2B adapter handles complex automated business transactions with business partners.

7.4.4.2 Event Handling Mapping

Event handling is one of the key features in Oracle ACM. As described earlier in the key features for ACM, the case execution is driven by events, originating from both internal and external origin. The *event delivery network* (EDN) is responsible for delivering events within the BPM/ACM/SOA application. Each application component in the BPM/ACM/SOA application can both start events and listen to events that are delivered via the EDN. The ACM product listens and acts upon these events via the business rules application component. *Oracle Event Processing (OEP)* listens to data streams, and has the ability to identify specific situations that require follow-up action, for instance initiate activities in ACM and BPM. In Section 10.3.2, more detail is provided for the OEP product.

7.5 Case Management Interaction Functionality with Oracle SOA Suite

In Chapter 3, we introduced the case management interaction design category, shown in Figure 7.4. We will look here at the portal and channel functionalities the SOA Suite offers. The dashboard functionality will be described later in Chapter 10.

In this section, we will focus on what Oracle products support the portal and the channel design for an Oracle BPM/ACM solution. The dashboard functionality will be described later in Chapter 10.



Figure 7.4 Case management interaction design category. The case interaction design enables maximizing the support for the case work force. This design area contains three design parts: Portal design, dashboard design, and channel design.

7.5.1 Portal Design: 360-Degree View of the Case

For a case worker to effectively process the case to completion with the desired outcome, he needs to have all the information—both data-based and content-based information in one integrated workspace.

For the portal functionality, the following options exist to create a 360-degree view of the case with Oracle products:

- Out-of-the-box functionality via the Oracle BPM/ACM portal
- Application Development Framework (ADF), customized functionality with the aid of the case management/BPM and human task application programming interface (API)

7.5.1.1 Out-of-the Box Functionality via the Oracle BPM/ACM Portal

Oracle ACM provides through a web-based, easy-to-use user interface that gives, at a glance, the complete state of the case—from available and completed activities, state of case data, audit trail to current state of the milestones, shown in Figure 7.5.

7.5.1.2 ADF, Customized Functionality with the Aid of the Case Management/BPM and Human Task APIs

Oracle BPM suite provides a comprehensive set of APIs that can be used to build customized user interface that either complements or replaces the out-of-the-box BPM workspace application. These APIs are available as Java APIs, SOAP/HTTP Web services and REST. The Java API is the full set of APIs, which lets you interact with and manage cases, processes, and tasks. This same Java API set is also exposed as SOAP/HTTP web services. A smaller subset, but complete from the perspective of building next generation mobile applications, is available as REST services.

The out-of-the-box BPM workspace is built using these same sets of APIs. Similarly the new Oracle BPM task management mobile application is built using these same REST services based upon the Oracle Mobile Application Framework.



Figure 7.5 Case management portal with the Oracle BPM/ACM workspace.

7.5.2 Channel Design

The Oracle BPM/ACM and SOA Suite provide support for access via different channels.

Oracle ADF delivers the standardized user interface functionality within the Oracle SOA Suite. ADF is based upon the Java Server Faces (JSF) specifications and contains a visual and declarative development. ADF delivers support for usage in multiple channel environments, such as browsers and mobiles. Oracle BPM/ACM delivers presentation of human task forms via ADF technology.

Mobile Application Framework (MAF) is the device-independent development framework used for creating mobile applications using standard development tools. Developers use Java, HTML5, and declarative components that are part of the MAF to design and build mobile applications that can run on both iOS and Android devices. An integral part of MAF is the built-in security for authentication, authorization, and encryption support. With MAF you only develop your mobile application once and run it on leading device platforms without any change.

Oracle BPM uses MAF to provide a task management application on iOS and Android devices. With this application users can act on their assigned tasks on their mobile devices. The application is designed to work offline in case the BPM server is not accessible.

7.6 Case Management Information Functionality with Oracle BPM/ACM/SOA Suite

The last functional category from Chapter 3 is the case information design (Figure 7.6).

Oracle BPM/ACM delivers limited support for the information design, as can be seen in Table 7.2. Oracle BPM/ACM is aimed at delivering a flexible case management solution in an enterprise wide, cross-application scenario. Oracle BPM/ACM is not aimed at a specific market and therefore does not include case supporting and people relationship data models.



Figure 7.6 Case management information design category. The case information design deals with all sorts of data and documentation flowing through the case and maps the relationships between all persons and/or assets involved in the case. The case information design category contains three design components, data design, document and media design, and people relationship design.

Information Design	Functionality	<i>Supported by BPM/ACM/SOA</i>	More Information in Section
Data design	Case execution data	Yes	7.6.1
	Case supporting data	No	_
	Tactical related information	Yes	7.6.1
Document and media design	Case input management	Yes	7.6.2
	Case output management	Yes	7.6.2
People relationship design	Internal stakeholders	No	—
	External stakeholders	No	_

Table 7.2 Information Support for Case Management with Oracle BPM/ACM

7.6.1 Case Execution Data: Detailed Audit Trail

Oracle ACM tracks every event that occurs in the context of a case and provides a detailed audit trail of these events from the point the case is started, as shown in Figure 7.7.

The audit trail can be protected using permission labels, and allows control of who is allowed to see which events in the audit trail. For instance, the case owner may decide to allow only case participants with "confidential" clearance to view that a particular "document added…" event has occurred. Oracle ACM provides fine-grained access control over different elements of a case. Access control is managed by using permission labels, for instance "confidential" can be a permission label. This label can be attached to any of the elements (data, activities, milestones, documents, and audit trail events). So, if a case worker decides to protect an audit entry and allow only those that have confidential access, he/she can attach the "confidential" label to that particular event in the audit trail. Any stakeholder other than the ones having "confidential" access is not able to see that entry in the audit trail. You can find more details on using permission labels in Chapter 17.



Figure 7.7 Case audit trail.

7.6.2 Document and Media Design

Oracle BPM/ACM deliver support for content management out-of-the-box. Oracle BPM uses Oracle WebCenter Content for storing task attachments. Oracle ACM uses WebCenter Content as its primary content store for all case documents.

Oracle WebCenter Content is an enterprise grade content management system. It provides a web-based interface for managing all types of content such as documents and media content. It also provides plug-ins for desktop applications such as Microsoft Office, which allow you to contribute content directly from these applications.

With WebCenter Content, searching across different types of content is easy and access to content can be secured using a comprehensive security model consisting of security groups, accounts, roles, and permissions.

WebCenter Content comes with conversion services that can be used to convert supported content to various formats. For instance, you can configure that all MS Word documents have a web-viewable version of the document generated as a PDF file.

7.7 Cross-Functional Case Management Capabilities

Functionality shared across all other functional areas is contained in the case management crossfunctional category, as shown in Figure 7.8.



Figure 7.8 Case management cross-functional design category. The cross-functionality design category contains the integration design and the security design that both support the other functional design categories.

7.7.1 Case Management Integration Functionality with Oracle SOA Suite

The BPM/ACM product is built on top of the Oracle SOA Suite. The SOA integration capabilities enable the BPM/ACM applications to act in an enterprise wide, cross application scenario. In Figure 7.3, the Oracle SOA Suite integration product is mapped across all capabilities. Different integration technologies are used to enable integration both internally and externally. The *Oracle Service Bus* (*OSB*) delivers integration capabilities with external sources, and contains all standard service bus functionalities like transformation, routing, secured, and reliable delivery with high performance.

The *mediator* is the internal service bus used between the application components that are part of the BPM/ACM/SOA application. *Adapters* support both the OSB and the mediator with standard-ized connections to external applications, such as databases, ERP systems, and cloud applications.

7.7.2 Case Management Security Functionality with Oracle

Oracle ACM defines a notion of a "stakeholder" who has access to the case and can perform various actions to progress the case. Stakeholders' access to the case can be controlled via userdefined permission labels.

7.7.2.1 Case Stakeholders

A case stakeholder is defined as a role with one or more users, groups, or other roles as members of this special stakeholder role. The stakeholder-based user model can be defined during design as part of the case model. It can be changed at runtime, for a specific case instance, by adding new stakeholder roles or changing the composition of an existing stakeholder model.

7.7.2.1 Permissions

Case permissions are defined as permission tags that can be attached to different elements of a case such as data, documents, activities, and audit-trail entries. Permission tags are user-defined strings

such as "public," "top secret," etc. Access control policies are defined for each of these tags by the case administrator or security administrator using Oracle Platform Security Services framework to control what each permission label means.

7.8 Conclusion

Oracle ACM is built on top of the already existing Oracle BPM/SOA product stack and is a middleware based product. The combination of Oracle products SOA, BPM, and ACM brings together integration capabilities, predictable BPM based processes and event based case management capabilities. These product features make Oracle ACM a good candidate for usage in an enterprise wide and cross application scenario. In this scenario, ACM takes responsibility for the end-to-end management and monitoring of all activities within the context of a case. These activities can be BPM processes, workflow tasks, and even other cases. Since ACM is layered on top of the SOA infrastructure, these case activities can also easily interact with external applications using the powerful adapter framework. All other functionalities, interaction, and information design can be delivered from other products that reside in the Oracle SOA Suite and database product sets.

Chapter 8

Siebel

8.1 Introduction

In this chapter, we will describe the Oracle Siebel product in relation to case management. The goals of this chapter are as follows:

- Give a high-level overview of Oracle Siebel
- Describe the usage characteristics of Oracle Siebel for case management functionality
- Describe how Oracle Siebel has strong support for case management information and the case management lifecycle design
- Show how the Siebel case management solution can be extended into different industry verticals
- Give an example of an industry vertical with the public sector case management solution
- Give an overview of supporting Oracle products

We start this chapter with an overview and usage characteristics of the Oracle Siebel product. Then we describe the product feature of Oracle Siebel according to the case management functional design as presented earlier in Chapter 3, with the main focus on the lifecycle and information functionality.

In Chapter 18, a tutorial is presented on Oracle Siebel where the product implementation steps will be extensively discussed.

8.2 Oracle Siebel Overview

8.2.1 History

Siebel started in 1993 as a system for sales force automation, from which it expanded into the customer relationship management (CRM) area. It is now one of the market leaders in this area. The aim of CRM systems is to organize, automate, and synchronize sales efforts and customer support for companies. Siebel extended the CRM capabilities with case management functionality, which will be the focus in this chapter.

8.2.2 Horizontal and Verticals

In the Siebel product, a distinction is made between functionality used in every industry domain, also known as horizontal functionality, and functionality specifically tailored for a specific industry domain, these are known as verticals.

Examples of *horizontal* functionality are sales, marketing, and analytics-related functionality.

Siebel made specializations for different industries to accommodate the differences in the way these industries deliver services to their customers. These industry solutions are called *verticals*, which use the Siebel's standard CRM capabilities as the base functionality and are based on best practices within each industry. Here is a subset of the industries for which Siebel delivers *vertical* solutions:

- Public sector
- Communications and utilities
- Banking and capital markets
- Insurance and healthcare
- Financial services

8.2.3 Configuration versus Customization

The Oracle Siebel product is a package-based solution. The design and the implementation of the product are based upon *configuration* and *customization*.

Configuration is adjusting the package toward the needs of the business without changing the package. This involves determining what data objects and elements are needed, setting up process models and state models toward the business requirements, determining what fields are needed on a specific user interfaces, and setting up the security model. This activity will not change the behavior of the package; it only composes the package toward the business requirements.

Customization is done when the business requirements do not align with the abilities of the package, and the package behavior needs to be changed. This activity should be kept to a minimum since customization poses extra risks to the maintainability of the application, such as the risk of noncompatibility during upgrades. Another factor favoring for keeping the customization to a minimum is the cost of operation. "Plain vanilla" implementation, with no customization, is the preferred option from a maintainability perspective, since this implementation has the lowest cost of operation and less dependency of specific personnel.

8.2.4 Extensions of Functionality

Every solution in Siebel, whether horizontal or vertical, can be extended to adjust to the specific customer situation. These extensions can be done in every level of the application, to name a few, in the data model, the preconfigured processes, approvals and assessments, and the state models. As shown in Figure 8.1 the extensions built on top of the existing CRM functionality and the horizontals and verticals inherit all the functionalities from the lower levels.

8.2.5 Siebel Specific Terminology

The Siebel product contains a set of building blocks that determine the behavior of the application, as shown in Figure 8.2. Each building block can be configured and extended during the design and implementation of the application.

Table 8.1 lists the building block groupings, to which case management design area this relates and in which section more detail is given of the design of these components.



Figure 8.1 Siebel extension model.

8.2.6 Market Example: Law Enforcement and Policing Operations Solution

Throughout this chapter, we will use the law enforcement and policing operations solutions as examples to show how the different functional design areas can be extended to accommodate a specific tailored industry solution.

The Siebel Public Sector vertical is one of the most sophisticated verticals within the Siebel CRM stack. Capgemini extended the public sector case management functionality into an integrated policing solution reflecting the different types of cases that are known within law enforcement organizations (https://www.capgemini.com/public-sector/transform-police-t-police). The extended solution contains a case management environment specialized for the law enforcement and policing operations and contains preconfigured functionalities that are based on best practices of law enforcement organizations from around the world. These extended features will speed up the realization and roll out of leading Oracle technology and reduce operational costs. In the case management, design descriptions for the Siebel product details from this extension solution will be used to give an insight to the usage of the Siebel product.

8.3 Oracle Siebel Usage Characteristics

Siebel case management is a package-based solution and delivers an end-to-end case management solution tailored at a large variety of industries. The Siebel product delivers a full solution stack, containing data, processes, and user interfaces.

The focus of the Siebel solution is primarily on delivering services to customers and is based upon the case management classification service request. In some industries, such as insurance and the solution for justice and public safety the case management solution has a focus toward the incident management classifications. Both classifications, service request and incident management, are supported in Siebel by investigation handling functionality. The Siebel CRM-based case management solution has two key features, the preconfigured CRM based data model and the extensibility of the solution in all design areas.

Within the Siebel case management solution, all scenarios for the case are defined upfront in the data model, the preconfigured processes, approvals and assessments, and the state models, with all its complexities. The Siebel product aligns mostly with the production case management (PCM) type of case management. Some aspects could be changed at runtime, such as changing the state model, but most design components are fixed.

Table 8.2 summarizes the usage characteristics for Oracle Siebel.



Figure 8.2 Siebel technical overview.

Building Block Grouping	Description	Case Management Design Area	More Information in Section
User interface	The user interface contains the end-user interaction functionality	Case interaction design	8.6.1
Business functionality	This area contains the information objects determining the CRM and case management data model	Case information design	8.4.1
Information This is the core Siebel repository Case information design		8.4.1	
Business services	A business service is a reusable component within Siebel that combines functionality within the system to simplify and minimize the amount of duplicated codes. It can be used as part of the integration with external systems, but also internally to standardize functionality such as calculations	Case lifecycle Design and case integration design	8.5.2
Lifecycle management	This contains the main functionality used to determine the lifecycle of a Siebel case	Case lifecycle Design	8.5.1
Security	This area contains access and record level security	Case roles and authorization design	8.7.1

Table 8.1 Siebel Building Blocks

Table 8.2Siebel Usage Characteristics

Focus Area	Product Characteristics	
Case management design area	Information and life cycle design	
Classification support	All three case management classifications are supported: service request, incident management, and/or investigative solution	
Delivery type: Full implementation or umbrella type	Full implementation	
Product type	Package based	
Agility at runtime: ACM or PCM	PCM type: The solution with all its aspects is defined upfront. Flexibility at runtime of the state model is possible	

8.4 Case Management Information Functionality with Oracle Siebel

Since the data model is one of the key features in the Siebel product, we start the functional description for Siebel with the case information design. The design components for the information design category are shown in Figure 8.3. For a full description of the case information design category, see Chapter 3.

The Siebel product gives full support for all case management information design components as shown in Table 8.3.

8.4.1 People Relationship Design and Data Design

The key component of Siebel is the CRM-based data model. This CRM model is based upon the party model, which is a generic data model that links together person-related information such



Figure 8.3 Case management information design category. The case information design deals with all sorts of data and documentation flowing through the case and maps the relationships between all persons and/or assets involved in the case. The case information design category contains three design components, data design, document and media design and people relationship design.

Table 8.3	Information	Design Support for	Case Mana	gement with	Oracle Siebel	

Information Design	Functionality	Supported by Siebel	More Information in Section
Data design	Case execution data	Yes	8.4.1
	Case supporting data	Yes	8.4.1
	Tactical-related information	Yes	8.4.1
Document and media	Case input management	Yes	8.4.2
design	Case output management	Yes	8.4.2
People relationship	Internal stakeholders	Yes	8.4.1
design	External stakeholders	Yes	8.4.1

as personal details, organization structures, and security profiles. With its open extension-based structure, the Siebel CRM data model is designed for structuring and linking additional information, such as individuals with events, agreements, goods, and services. This enables Siebel to act as a CRM-based case management system for a broad variety of tasks where the purpose is to establish links between different information items.

The design of this part requires mapping the business-required information toward the Siebel CRM data model, where specific attention is given to the building blocks contained in the *business functionality* as shown in Figure 8.2.

- Business Object: A business object corresponds to the main concepts dealing with the context of the application. In case management, this corresponds to concepts such as "person," "suspect," "offence," or "coercive measure." A business object is used to manage the information related to the specific composed information that is needed for a specific vertical or extension, and is mainly designed via configuration.
- Business Component (Fields): A business component (and related fields) is used to combine several of the underlying data tables for retrieval and managing of the information. Usually design in this area is done via configuration.
- Information: The information grouping contains the Siebel building blocks table and columns and is part of the core Siebel repository. Tables (and columns) are used to store all case execution data and the tactical-related information. Design in this area usually relates to providing access to audit trails.

Extending the Data Model, an example based upon the Law Enforcement and Policing Operations use case:

As mentioned in Section 8.2.6, the police case extends the Siebel Public Sector vertical. The data model for the Siebel Public Sector vertical is known as the public party model. To deal with police cases, specific information needs to be added to the already existing public party model. The Siebel Public Sector vertical is enriched with the following new information objects: Offenses, offenders, victims, witnesses, suspects, locations, properties and evidence, and coercive measures. From an audit perspective, besides an audit trail, it is crucial that the usage of data is also stored. For instance, for a sensitive crime case data related to a VIP it needs to be logged as to who requested this data and what was done with this data.

One specific important characteristic of the police information model relates to the fact that three different police case types, incidents, leads, and cases need to be kept in the information model:

- *Incidents* deal with violations of the law, for which no investigation needs to be set up. Examples of incidents are traffic accidents, traffic offenses, and asocial behavior.
- A *lead* gives an indication of possible crime activities. At the time a lead is brought in, no crime activity has been measured yet. Leads might turn into police cases at a later point in time.
- A *case* deals with the investigation related to a crime that has been committed. In order to get a complete picture of the situation, cases, leads, and incidents can be combined together. This helps in finding and arresting the suspects and enables a successful handover to the prosecutor.

These three main information entities (incidents, cases, and leads) of the solution are the pillars of the law enforcement and policing operations capabilities of the solution. They interact with



Figure 8.4 Shared information model for cases, leads, and incidents.

each other to support the law enforcement organizations to combat crime and terrorism as well as enhance the public safety and service, and emergency management. The information models for the police case types, incidents, cases, and leads, each have separate aspects but also partially overlap one another, as can be seen in Figure 8.4.

8.4.2 Document and Media Design

Out of the box Siebel offers attachment functionality to add more detail to a case record by adding relevant documents and other external media. This information is stored as part of the Siebel application or via an external document management system. In both options, Siebel offers the functionality to add metadata and unique identifications for later search and easy retrieval.

8.5 Case Management Life Cycle Functionality with Oracle Siebel

In Chapter 3, we introduced the various aspects for the design of a case management solution. We will look here at the case lifecycle design category for the Siebel product, as shown in Figure 8.5.

Specifically we will look at the case flow and process design functionality within Siebel, as this is the main lifecycle supporting area.

8.5.1 Case Flow and Process Design with Oracle Siebel

Siebel supports the case flow and processes design with three main functionalities: the state model, workflow management, and the assignment management.



Figure 8.5 Case management lifecycle design category. The case lifecycle design category describes the coordination of activities in reaching one of the end goals and consists of three design components: Case flow and process design, rules design and event design.

8.5.1.1 State Model

The state model is one of the key elements in the lifecycle of a case in Siebel. This model describes the allowed states that can exist in the lifetime of the case and the possible transitions between the different case states, also known as the case flow. The state model can also be more complex with detailed subcase states and state flows going back and forward. An example of a complex state flow is shown in Figure 8.7. Changing a case state may be subject to certain conditions, such as data values. Siebel supports defining constraints on case state changes as part of the state model. For instance, in Figure 8.6 going from *active* to *submitted* is an allowed state transition, whereas going from *rejected* to *submitted* is not an allowed state transition.

8.5.1.2 Workflow Management

The Siebel Workflow Management helps defining the different workflows required in the lifecycle of a case. Workflow models can be bound to a specific state or can be started regardless of any state value. This makes it possible to activate (or deactivate) certain activities based upon a moment in the case lifecycle.

The workflow contains all functionalities required to define a process:

- Orchestration of tasks
- Task management, both automated and human oriented
- Routing and assignment management
- Escalation
- Input/output management
- Policy management
- Tasks user interfaces

8.5.1.3 Assignment Management

Assignment management supports routing the work activities to the most appropriate case workers. Business rules support in the determination of this work activity assignment. The assignment management does this by matching candidates, such as employees, positions, or organizations, to predefined and user-configurable assignment objects like a case, an incident, or a lead. To assign the most qualified candidate to each object, the assignment manager applies assignment rules that are specifically defined for each candidate

8.5.1.4 Extending the Life Cycle Management, with the Law Enforcement and Policing Operations Example

The law enforcement and policing operations extend the Siebel Public Sector vertical, and contain three different types of functionalities, incidents, cases, and leads, as discussed earlier in the extension example for the data model. Even though these functionalities are different, commonalities such as information objects and some processes and activities are shared, as shown in Figure 8.6.

As an example of extending the activities in Siebel, we will look here at investigative lead management. Leads are used to capture all kinds of clues, hints, and tips gathered by the public and agents in the field. Most of this information, in relation to a committed or foreseen crime, can often not be used as evidence. As a result of these leads, follow-up investigations might be started, which may result in evidences that can be used to start a new case and/or may be linked to already existing cases. Leads are handled as "information cases" in which the process of validating, analyzing, enriching, and proofing of information is managed. The ultimate goal of a lead is to promote it as evidence to support a case. Investigative lead management supports processes like an approval process and a sophisticated state model to manage its own process. It can hold information in itself or be linked to information in cases, incidents, or other leads. This requires new and/or extended processes, related to the lead management information objects.

Supporting the crime investigation process different state models are detailed, of which an example is presented in Figure 8.7. These state models define the possible routes a police case can encounter and guide the team through a series of steps that might lead to a solution. The case



Figure 8.6 Linking cases, leads, and incidents together.



Figure 8.7 Police investigation state model.

owner, i.e., the police officer leading an investigation, can alter the state model, and consequently change the flow of activities. These state models are customizable in which the handling of a case can be tailored in a specific situation.

8.5.2 Rules Design

We identified in Chapter 3 different types of rules functionalities. The case lifecycle decisionmaking rule type is part of the state model functionality, as described in the previous paragraph. The remaining rule types are described in more detail in Chapter 10. Out of the box Siebel gives rule support for the remaining rule types; however when rules tend to get more complex, an external rule engine or product is a more preferred option.

8.5.3 Event Design

Events can be detected within Siebel on application, applet or business component object level. For details, see the topics mentioned in Section 8.2.6. The Siebel event model is user event based. The event originates from the browser-based application and passes the event toward the applet. The applet in turn passes on the event toward the business component. From these event points functionalities, such as business service or a workflow, can be called.

8.5.4 Life Cycle Design Product Mapping to Oracle Siebel and Related Products

We finish the lifecycle design section with a product mapping of the case management lifecycle functionality toward the Oracle Siebel product sets, based upon the drawing technique presented earlier in Chapter 5. Here we will make a distinction between the "Siebel single entity" scenario and the "Siebel in a wider application landscape" scenario. In the first scenario, the "Siebel single entity" scenario solution is fully covered by the Siebel product. In the second scenario, the Siebel scenario is either part of a larger, end-to-end process, or additional system support needs to be delivered for which Siebel does not offer the required functionality. The product capability mapping for both scenarios is presented in Figure 8.8.



Figure 8.8 Lifecycle management Siebel mapping.

In Figure 8.2, we presented the Siebel building blocks. The lifecycle management functionalities, workflow, assignment, and the state models are mapped toward the case management lifecycle capabilities in Figure 8.8.

The "Siebel in a wider application landscape" scenario covers a (noncomplete) list of scenarios that could be encountered in an IT landscape:

- Integration with external applications is required to call functionality or to share information.
- Application-to-Application (A2A) capabilities are required across applications.
- A workflow is started in an external application.
- Complex rules management is required to support the Siebel functionality.

In the "Siebel in a wider application landscape," the Oracle SOA Suite is an important addition for connecting Siebel and the "wider application landscape" and providing the additional functionality in areas that are not the core functionality of the Siebel product. Examples are the SOA Suite integration capabilities and the addition of BPEL for situations where straight through processing (STP) is required.

8.6 Case Management Interaction Functionality with Oracle Siebel

In Chapter 3, we introduced the case management interaction design category, as shown in Figure 8.9. We will explain here briefly the portal and channel functionalities Siebel offers in this area. The dashboard functionality will be described later in Chapter 10.

8.6.1 Portal and Channel Design

As explained earlier in this chapter, the key component of Siebel is the CRM-based data model. All functionality in the Siebel data model can be made available in the Siebel portal overview for the users. This functionality is out of the box, available, and can be tailored toward usage and role



Figure 8.9 Case management interaction design category. The case interaction design enables maximizing the support for the case work force. This design area contains three design parts: Portal design, dashboard design and channel design.

based access. As with other functionalities within Siebel, the user interface part can be extended in order to meet the requirements of the business.

As can be seen in Figure 8.2, the user interface is based upon a composition of applet-based building blocks that directly relate to the business functionality grouping containing the CRM and case management information objects.

Only recently in the Siebel stack a new user interface technology is added, open UI, which is based upon open standards and technologies such as HTML, CSS, jQuery, and AJAX.

Out of the box Siebel supports browser-based access. When more channel types are required for access with the application, functionality can be exposed via the Siebel integration framework as web services that can be picked up by other channels such as mobile.

8.7 Cross-Functional Case Management Capabilities

Functionality shared across all other functional areas is contained in the case management cross-functional category as shown in Figure 8.10.

8.7.1 Case Management Integration Functionality with Oracle SOA Suite

As described in Section 8.4.4, in the "Siebel in a wider application landscape" scenario integration with external applications is required. The Oracle SOA Suite integration capabilities, BPEL, human workflow, and B2B products support Siebel with functionality that is not the core in the Siebel product. A short description of these products is given in Chapter 7.

8.7.2 Case Management Roles and Authorization Functionality with Oracle Siebel

Within the Siebel product, two main functionalities can be configured in relation to roles and authorization; access management on the user interface level and on data level.



Figure 8.10 Case management cross-functional design category. The cross-functionality design category contains the integration design and the roles and authorization design that both support the other functional design categories.

User interfaces access management and responsibilities:

View-level access control determines which parts of the case management solution a user can access, based on the responsibilities assigned to that user. In Siebel, these responsibilities are known as functions. Based upon defined responsibilities, a user gets access to a specific set of data. An employee assigned to one responsibility might not have access to parts of the case management solution associated with another set of responsibilities. Each user's primary responsibility also controls the user's default screen tab layout and tasks.

Data-related record level access management:

Record-level access control assigns permissions to individual data items within the case management solution. This enables for instance administrators to authorize specific authenticated users who have "view-only access" to particular sensitive data records.

Siebel uses three types of record-level access: *position*, *organization*, and *access group*. When a particular position, organization, or access group is assigned to a data record, only employees within that position, organization, or access group are authorized to view that record:

- A position represents a place in the organizational structure, e.g., a specific job. Typically, a single employee occupies a position; however, it is possible for multiple employees to share a position. Position access allows Siebel customers to classify users based on hierarchy, which can be used for dedicated access to data. For example, a unit chief would have access to data that one of his officers can also access; the same applies to other users who report to the same unit chief.
- Similarly, an *organization*, such as a division or department of an organization, is a grouping of positions that maps to the physical hierarchy of an organization. Employees assigned to a position within a certain organization are granted access to data that has been given to this specific organization. Restrictions can be placed on data to avoid data being accessed from outside the organization.
- An access group is a less structured collection of users or group of users, such as a task force. Groups can be based on some common attributes of users, or created on a temporary basis, pulling together users from different organizations and granting them access to the same data.

8.8 Conclusion

Siebel is a package based solution with the emphasis on CRM. One of the key features of Siebel is the preconfigured CRM-based data model and the extensibility of the solution in all design areas. Siebel extended the CRM capabilities with case management functionality into different industry and cross-industry specific solutions, known as verticals and horizontals. We showed in this chapter how the industry solution public sector case management could be extended into the law enforcement and policing operations solution. Siebel delivers a complete full solution stack, containing data, processes, and user interfaces. The solution can be used in isolation, the "Siebel single entity" scenario and can be used in collaboration with other applications, the "Siebel in a wider application landscape" scenario.

Siebel delivers all solution components in the "Siebel single entity" scenario. The Oracle Suite delivers additional integration capabilities in the "Siebel in a wider application landscape" scenario.

Chapter 9

Oracle Service Cloud

9.1 Introduction

In this chapter, we will describe the Oracle Service cloud in relation to case management. The goals of this chapter are as follows:

- Give a high-level overview of Oracle Service cloud
- Describe the usage characteristics of Oracle Service cloud for case management functionality
- Describe how Oracle Service Cloud strongly supports the case management interaction functionality

We start this chapter with an overview and usage characteristics of the Oracle Service Cloud. Subsequently, we describe its product features according to the case management functional design as presented earlier in Chapter 3, starting with interaction functionality.

9.2 Oracle Service Cloud Overview

9.2.1 History

The relationship between the customer and a product company has always been dominated by questions and issues around the products delivered by these companies. These questions can be asked before a product is acquired and will help the customer during the decision process toward the purchase of the product. If the product has been bought, the customer can have questions about product characteristics or wants to report and discuss issues regarding the product. During the early 1990s, a large part of this communication was done via e-mail. RightNow jumped into this market by delivering customer services via a software-as-a-service (SAAS) platform aimed at lowering the amount of e-mail inquiries for a company. In this solution RightNow delivered websites with strong knowledge capabilities in order to deliver customers services via a self-service model. This platform helped to accelerate winning new customers, maximize customer retention, and improve operational efficiency. Oracle acquired RightNow technologies in 2011, for its cloud-based customer service capabilities.



Figure 9.1 Oracle Service Cloud overview.

9.2.2 Basic Functionality

The product stack evolved into a customer experience (CX) stack and contains the products shown in Figure 9.1.

Oracle RightNow Web Experience provides rich user interface where customers can research, purchase, and solve issues for the products they are interested in. This is delivered to any device.

Oracle RightNow Social Experience provides support for social interactions, for example, via Facebook.

Oracle RightNow Contact Center Experience delivers support for call centers improving and standardizing the interaction with customers.

Oracle RightNow Policy Automation is also known as Oracle Policy Automation (OPA) and supports the self-service with complex rules capabilities, for instance, for the determination of eligibilities. OPA will be described in more detail in Chapter 10.

Oracle RightNow Engage delivers the analytics of all customer interactions in order to improve the customer interactions. This product also delivers the knowledge base and is the heart of the RightNow solution. It enables all service request-related information requests.

Oracle Rightnow Service Experience Platform is the cloud-based platform that integrates the aforementioned service cloud functionalities.

9.2.3 Cloud-Based Management and Usage

The Oracle Service Cloud is a SAAS platform and runs in the cloud to which customers can subscribe. This subscription makes it possible to use and, depending on the role of the user, manage an instance of this cloud. Managing the cloud is done via a localized Oracle Service Cloud client that is placed on a user's computer that is allowed to perform management tasks.

9.2.4 Extensions of Functionality

All components in the Oracle Service Cloud can be extended to include additional specialized functionality. This can include, for example, fields to the data model, the addition of rules, or the adjustment of the customer portal. The Oracle Service Cloud client enables configuration and extension of the application.

9.3 Oracle Service Cloud Usage Characteristics

The Oracle Service Cloud is a SAAS solution that enables interactions with customers around products and is aiming at delivering answers to questions or resolve product issues. These interactions are supported by a knowledge base delivering information through self-service interactions. Call center interactions with customers can be included when the customer requires more information. RightNow is targeted at delivering services for customers, based upon the service request classification, with a strong investigation focus through the knowledge base. Even though delivering support to product issues is named *incidents* in the Service Cloud, these interactions still follow the service request classification paradigm.

The product is delivered via a subscription model to customers and contains a full solution stack, with user interfaces delivered via a multitude of channels, processes, search functionality, call center support, and social media monitoring amongst others.

In Section 4.4, we introduced different ways companies can position themselves to meet the expectations of the customers. Three modus operandi are known in this area: *operational excellence, customer intimacy, and product leadership.* The Oracle Service Cloud aligns with the *product leadership* operating model and contains a state-of-the-art knowledge base enabling searching through the product catalog helping customers to understand the products. This helps the customer to make a measured decision before making a purchase and in the after sales support it helps the customer with queries and issues regarding the product.

The Oracle Service Cloud is a predefined product, with extension options in most functionality areas. Once delivered to the customers, the functionality cannot be changed at run-time, except for the continuous improvement of answers in the knowledge base and the adaption of rules through a rules modeler at run-time. The service request work is highly predictable, even though searching through the knowledge base can be complex. This makes the solution a production case management (PCM) type of case management.

Table 9.1 summarizes the usage characteristics of Oracle Service cloud.

Focus Area	Product Characteristics	
Case management design area	Information and interaction design	
Classification support	Two case management classifications are supported: service request supported via investigations	
Delivery type: Full implementation or umbrella type	Full implementation	
Product type	Cloud based (SAAS)	
Agility at runtime: ACM or PCM	PCM type: Most of the functionality of the service request is defined upfront. The inclusion of Right Now policy automation (OPA) delivers flexibility at runtime. The continuous improvement of answers enhances the quality of the service request	

Table 9.1 Service Cloud Usage Characteristics

9.4 Case Management Interaction Functionality with Oracle Service Cloud

Bringing service through all channels is the core competence of Oracle Service cloud and for this reason it excels in interaction functionality. In Chapter 3, we introduced the case management interaction design category, as shown in Figure 9.2.

9.4.1 Portal Design

The Oracle Service Cloud delivers support for the two main user groups, the customers and the contact center.

The *customers* are supported with a portal that contains all the aspects to get access to the specific product data, get in contact with support agents, and deliver feedback on the experiences:

- Self-service pages to raise questions and get automated answers, aimed at the ease of finding the right information
- Be able to engage with a contact center through different channels when more information is needed
- Show related information and experiences from other users
- Get in contact with communities to interact on product experiences
- Share experiences

A reference implementation is provided out of the box by the Oracle Service Cloud. This reference implementation shows the best practices for customer support and speeds up the case interaction design. The customer portal can be tailored toward the look-and-feel of the company and be personalized toward the needs of the customer, for which an example is provided in Figure 9.3.

The *contact center* is supported with a user agent desktop. This user agent desktop is aimed at delivering a full overview of the customer and the products in order to deliver a fast and good advice for the customer. Even though the service cloud itself is delivered through the cloud, the agents are supported with a client-based application. The user agent desktop can be configured



Figure 9.2 Case management interaction design category. The case Interaction design enables maximizing the support for the case work force. This design area contains three design parts: portal design, dashboard design, and channel design.



Figure 9.3 Customer portal based upon the Oracle Service Cloud.

toward the needs of the agent. Based upon the role of the agent, he/she gets access to certain functionality. The agent desktop contains features such as the following:

- The *workspace* is the view of the user agent in his/her daily work. Both the page layout and the content on the page, such as contacts, incidents, and answers can be adjusted to tailor for the agents' needs.
- Workspace rules help dynamically adjust the page display and behavior according to specific situations.
- Agent scripts and desktop workflow support the agents with a page flow to enter (and acquire) the information in the right order, enabling finding quickly a resolution for the customers' issues.

9.4.2 Channel Design

The Oracle Service cloud is designed to support a large variety of multichannel interactions with customers. This area is growing with new channels opening up:

- The *customer portal* can be accessed through a rich portal or a mobile customer portal
- Direct interaction between the customer and the contact center via chat, e-mail, or phone, for which an example is shown in Figure 9.4



Figure 9.4 Channel option example—contact us.

- Social experience via a variety of communities, such as Facebook, Twitter, or Google
- Broadcasting information via campaigns, mailings, and feedback surveys

Everything is aimed in solving the incident or service request at low cost, favoring channels like self-service and through communities. But if one-to-one interaction is necessary, it offers e-mail support, chat, phone, and cobrowsing, a technology aimed at sharing the customer browser with the contact center. Everything is connected with the knowledge component that makes it possible to deliver the same answers through all channels.

9.4.3 Dashboard Design

The following aspects of the Service Cloud can be monitored:

Customer satisfaction and *workload of the customer agents* are provided in the *Oracle RightNow Analytics Cloud Service*. The quality of the knowledge base needs to be monitored in a continuous improvement loop, as will be described later in Section 9.7.1. The continuous improvement is supported with reporting of top-search keywords and customer reviews.

9.5 Case Management Information Functionality with the Oracle Service Cloud

The information design of the Oracle Service Cloud is aimed at delivering the customers the best answers and information for their question. The design components for the information design category are shown in Figure 9.5. For a full description of the case information design category, see Chapter 3.



Figure 9.5 Case management information design category. The case information design deals with all sorts of data and documentation flowing through the case and maps the relationships between all persons and/or assets involved in the case. The case information design category contains three design components: data design, document and media design, and people relationship design.

Information Design	Functionality	Supported by Service Cloud	More Information in Section
Data design	Case execution data	Yes	9.5.1
	Case supporting data	Yes	9.5.1
	Tactical-related information	Yes	9.5.1
Document and media design	Case input management	Yes	9.5.3
	Case output management	Yes	9.5.3
People relationship design	Internal stakeholders	Yes	9.5.2
	External stakeholders	Yes	9.5.2

Table 9.2	Information Design Support for Case Management with the Oracle
Service Clo	bud

The Service Cloud provides full support for all case management information design components as shown in Table 9.2.

9.5.1 Data Design

The basis for the Oracle Service Cloud is the knowledge base information model. This knowledge base model contains three main parts:

- 1. *Customer relationship model* (*CRM*) functionality, with information objects such as contacts, staff accounts, and organizations.
- 2. A product catalog where assets and products are stored.

3. A *service supporting part* that provides support for incidents, opportunities, and answers. An *incident* relates to a question or request for support. An *opportunity* contains the information about a possible sale with a customer. *Answers* are in response to questions raised by the customer.

The data model of the Oracle Service Cloud can be extended to tailor the application towards the specific needs of the business. This can include the addition of fields to the existing data model and the creation of custom objects. All new information types, both fields and objects, are available in all of the Oracle Service Cloud functionality, such as in workflows and workspaces.

Answering questions from customers based upon a set of search keywords is one of key capabilities of the Service Cloud. The management of answers in the Service Cloud, in the most basic form, is entering an *answer* and a *summary* of the answer. The Service Cloud uses the content of the answers to extract the search words that are used to provide the customer with the right response. As can be seen in Figure 9.6, the management of answers can contain more facets, such as combining the answer into product and categories, adding relationships with other answers, and the addition of attachments. The addition of keywords is only needed to put emphasis on specific words.

Managing the knowledge base is a continuous effort and is adjusted with improved knowledge from new customer–support interactions, as shown in Figure 9.7. The Service Cloud configuration starts with an initial seed of the knowledge with answers known to be important for customers. Once the Service Cloud is put into production, knowledge from the customer–support interactions can be used to add new answers, refine existing ones, and improve the organization of the knowledge base.

The management of the knowledge base contains the following activities:

- Defining and refining answers based upon customer interactions
- Organizing answers into categories
- Building relationships between answers to enable customers obtain quicker insight in product features
- Reporting on customer interactions and top-searches provides insight in the improvement areas for the knowledge base



Figure 9.6 Management of answers.



Figure 9.7 Knowledge base continuous improvement.

9.5.2 People Relationship Design

The Oracle Service Cloud CRM part is based upon two groups, staff and contacts.

The *staff* persons manage the Service Cloud and interact with the customers. *Contacts* are the customers asking information or services through the service requests. Both staff and contact can be organized into different organization levels. Additionally, the staff persons can be grouped into expertise areas such as products and level of expertise. This grouping is used for routing incident to the appropriate group.

A contact can be associated with assets; typically this is used to identify the products the customer has bought and provides the ability to deliver services and answers tailored for that product.

9.5.3 Document and Media Design

The document and media design for the Oracle Service Cloud relates to the product catalog and supports the customer with additional information about the products, such as demo videos, user guides, and product-specific images. These documents and media can be related to the answers as shown earlier in Figure 9.6.

9.6 Case Management Life Cycle Functionality with Oracle Service Cloud

In Chapter 3, we introduced the various aspects for the design of a case management solution. We will look at the case life cycle design category for the Service cloud, as shown in Figure 9.8.

Specifically, we will look at the way the Service Cloud supports the customer with self-service capabilities and, in situations where the question is too complex, provide additional contact center interaction support.



Figure 9.8 Case management life cycle design category. The case life cycle design category describes the coordination of activities in reaching one of the end goals and consists of three design components: case flow and process design, rules design, and event design.

9.6.1 Case Flow and Process Design

The Oracle Service Cloud aims to support the customers with the best answers for their questions. These answers help the customer to resolve an issue with a product or help them in making a thorough decision for purchasing the product. The Service Cloud delivers a customer interaction according to the service request classification. As described in Section 4.4, the service request case contains two main flows, a *triage* workflow and, where needed, a follow-up *service request case handling*. Both service request phases are supported by investigations. Figure 9.9 shows the service request phase in relation to the Oracle Service Cloud products.

The starting point for a customer, in the triage workflow, is the self-service web portal where the initial question is entered. With the support of the knowledge base, answers are provided to the customer. The customer finishes the interaction when the provided information is sufficient.

When the query turns out to be too complex to obtain a satisfying answer, additional support can be delivered in a direct interaction with the contact center. The service request case handling



Figure 9.9 Oracle Service Cloud product usage in a service request.

is now started. At the same time, the customer can engage in discussions on different social media to get an answer. In this case handling, the customer is guided toward the appropriate answer to resolve his/her query or issue.

All information in the customer interaction, with the self-service pages, the contact center, and the social media, is stored for later usage. This information helps in improving the future customer interactions and can help to improve the answers in the knowledge base.

9.6.2 Rules Design

In Section 3.3.2.3.2, we discussed the different rules type that can support a case management solution. The Oracle Service Cloud delivers out-of-the-box support for two rules types, the *case life cycle decision-making* and the *entitlement determination*.

The entitlement determination will be handled in more detail in Chapter 10.

In order to support the *case life cycle decision-making*, a rules engine is available in the Oracle Service Cloud to automate the assignment of service requests (incident) from customers to specific contact center groups or persons. This assignment process can be based upon the characteristics of the service request, such as the source of entrance (for instance chat, portal, and e-mail), time created, product type, and the severity. Figure 9.10 shows an example of a rule based upon the *time-created* characteristic. In this assignment process, the rules engine can also update the service request state model to the next stage. Rules are organized in a rule base, which is a grouping of rules around data objects and can be applied to standard data objects such as contact, organization, task, incident, and answer.

9.6.3 Event Design

Events are raised in the Oracle Service Cloud based upon object event handlers. These handlers act upon a change happening in the data in one of the standard or custom objects in the database. When an event is raised, the object handlers can create or automate tasks based on the triggering event.

Staff Management	Search Edit Incident Rule		
 Departation Appearance Set Configuration Set Configuration Set Configuration Set Configuration Set Configuration Set Configuration Mathematic Configuration Set Co	Source Source	Eding Rabin Rules 2 & Projects Not fand Add Dytinis Add Dytinis Add Dytinis Add Dytinis Add Adden. Then Add Adden. Then Add Adden. Else T T T T T T T T T T T T T	
01.043×		innerent interest	[Manning]

Figure 9.10 Oracle Service Cloud rules design.

For example, when an incident is updated an object event handler can invoke an external system service to update specific data, send an e-mail or notification to the customer, and create a manual task. The technology behind the object event handlers is the PHP scripting language.

9.6.4 Life Cycle Design Product Mapping to Oracle Service Cloud

We will summarize the life cycle design paragraph with a product mapping of the case management life cycle functionality toward the Oracle Service Cloud product components. This is based upon the drawing technique presented earlier in Chapter 5. Figure 9.11 summarizes the product functionalities part of the Oracle Service cloud.

On the left-hand side in Figure 9.11, the core case management capabilities are shown that are supported by the Oracle Service Cloud:

The *service-request* case flows and processes are supported by the Service Cloud functionalities *self-service flow* and the *contact center*. The overall flow supporting the case handling by the contact center is a simple milestone based *state model*.

Cloud integration and *channel integration* functionalities support the out-of-the-box integration with cloud products and all channels supported by the Oracle Service Cloud.

The *rules support* area contains three components:

- 1. *Business rules* support the assignment and routing of requests in the case life cycle decision-making.
- 2. *RightNow Policy Automation*, also known as *Oracle Policy Automation* (OPA), supports complex entitlement determinations, for example, guarantee-related determinations, is the problem covered by the guarantee policy?
- 3. The knowledge base is the core capability in the search functionality.

Event support is supported with the *object event handler* functionality that operates on top of the Service Cloud data model.

9.7 Cross-Functional Case Management Capabilities

Functionalities shared across all other functional areas is contained in the case management cross-functional category as shown in Figure 9.12.

9.7.1 Integration

The Oracle Service Cloud has an out-of-the-box integration channel and cloud integration capabilities. The Service Cloud interacts with all channels previously described in Section 9.4.2. The Service Cloud is also delivered with cloud integration capabilities with a number of standard Oracle applications, such as ATG, Sales Cloud, and Social Relationship Management. When more complex integration is required, the Oracle SOA Suite can be added to the solution.

9.7.2 Case Management Roles and Authorization Functionality

Profiles are the mechanism used in the Service Cloud to control which areas can be accessed with the Service Cloud. Profiles are created for customers and for internal personnel such as agents.
	Technical capabilities Service cloud building blocks
Case lifecycle mapping to the oracle service cloud Case flows/processes	Lifecycle management 💫 State model 🔿
Service request	H2A orchstration A Self service flow O Cloud integration O
	A2A orchstration A2A or
	H2H collaboration A Contact center O Channel integration O
Rules support Case lifecycle decision- making	Business Rules
Entitlement determination	Rightnow policy automation (OPA)
Information discovery	Knowledge base
Other rules functionalities are detailed in Chapter 10	
Event support	Messaging A Object event handler

Figure 9.11 Life cycle management Oracle Service Cloud mapping.





Internal personnel profiles are used to determine the level of interaction with customers, such as who can engage with what customers, who can assign incidents, and who is allowed to perform chats with customers. Administrating type of personnel is allowed to create rules and administrate the knowledge base among other functionalities.

For *customers*, the profiles relate to the data security. Some data can be viewed publicly; other data is private, this data is only available for a customer when they have signed in to the portal.

9.8 Conclusion

The Oracle Service Cloud delivers an application for supporting all product-related interactions between the customer and a company. The Service Cloud enables organizations to position themselves in the area of product leadership. It contains a state-of-the-art knowledge base that helps customers understand the products. The customer uses self-service web pages to search for the answer they need. If the question is too complex, a contact center can provide additional interaction support.

The application has strong capabilities in the portal and channel capabilities. The Oracle Service Cloud supports a large variety of channels:

- The *customer portal* is accessible through a rich portal and a mobile customer portal.
- Direct interaction between the customer and the contact center is supported via chat, e-mail, or phone.
- The customer can interact with other customers through a variety of *communities*, such as Facebook, Twitter, or Google.
- Broadcasting information via campaigns, mailings, and feedback surveys is the last option.

Oracle Service Cloud is a SAAS platform and is enabled through subscription. No installation is required to use the product except for the software to configure the service cloud and for the contact center agent desktop.

The functionality of the Service Cloud contains predefined functionalities that can be extended in some areas. The main extension functionalities are in the data model, where additional custom fields and objects can be added and the customer portal that can be tailored toward the company website.

Chapter 10

Intelligence in Relation to the Case Execution

10.1 Introduction

In this chapter the usage of intelligence supporting case execution is described.

Intelligence supports the stakeholders with the information to make the right decisions during case activities. This intelligence can support the execution of a single case, relate to the governance of cases that currently are executed or to make forecasts and predictions based upon experiences with other cases.

In the previous chapters, we described the three Oracle case products, Oracle (Business Process Management (BPM)/Adaptive Case Management (ACM), Oracle Siebel, and Oracle Service Cloud. In this chapter we describe Oracle technologies that can support the intelligence needs of Oracle BPM/ACM, Oracle Siebel, and Oracle Service Cloud with functionalities that are not core for these products.

We start with summarizing the intelligence requirements needed in the case execution as defined earlier in Chapter 3, and map these toward a choice of Oracle products that are best suited for supporting the intelligence needs for the case execution.

10.2 Case Intelligence Design

In Chapter 3, we described the different categories of a case management design. The intelligence needs at run-time relate to the design areas life cycle, information and the interaction design categories, as shown in Figure 10.1.



Figure 10.1 Case management design components that support intelligence design (in white).

10.2.1 Intelligence Determined by Rules Design

In the case life cycle design category, the rules design determines how decisions and checks are performed at operational level. A summary of the life cycle rule design functionalities is provided in the following:

Case life cycle decision-making

This relates to the life cycle management functionality determining the execution path of a case.

- Entitlement determination The entitlement decision entails the eligibility of the customer on a specific service or claim. This is typically guided by policies and regulations and can change on a regular basis.
- Progress measurement This relates to measuring the progress of a case. This involves measuring the case state against predefined progress indicators.
- Data entry validation Data entered into the case by any of the participants can have validation requirements attached to it, for instance, based upon regulations and policies.
- Information discovery Searching through structured and unstructured data that relate to the case execution can deliver some new or changed insights toward the output and behavior of a case.

10.2.2 Intelligence Determined by Event Design

The event design is used to determine how to handle special events. Specific attention will be given here to the monitoring of incident risk parameters, as described in detail in Section 4.5. This involves measuring case/process related data against predefined risk parameters, in order to predict the likelihood of an incident in the near future.

10.2.3 Intelligence Determined by Dashboard Design

In the case interaction design of a case, the dashboard design is the area where intelligence is delivered to manage the tactical and strategic information needed by the different stakeholders:

- Tactical dashboards support key performance indicators (KPI) management, workload management, and help in providing insight in customer satisfaction.
- Strategic dashboards are used for improvements in the area of case and process execution improvement, the case related products, and the decision support.

10.2.4 Intelligence Information Levels

In the case information design we presented following three information levels that determine the different intelligence needs by the different stakeholders:

- The *operational* level supports in real time the work on a single case.
- The *tactical* level is aimed at improving the efficiency, customer satisfaction, and controlling case-related KPIs.
- The strategic level enables looking back in time and investigating process and product improvements and forecasts.

10.3 Oracle Intelligence Products

Oracle technologies can support the intelligence functionalities that are described in the previous part in this chapter. In Table 10.1 the main Oracle technologies are mapped against these intelligence functionalities. In this table, several Oracle products are checked against the case management intelligence functionalities related to the different information levels, operational, tactical, and strategic.

The second column in this table shows where more detail related to the Oracle product can be found.

In the upcoming paragraphs we will look in more detail into the Oracle products that have not been described so far. We will look into the scope of the product, how the product relates to case management, and optionally the technical architecture.

We will not provide details for the Oracle Business Intelligence (BI) product in this book. Oracle BI is a product on its own and is distantly related to case management and many other books have been written already. The Oracle BI product delivers a rich set of functionalities such as ad hoc query and analysis, dashboards, enterprise reporting, scorecards, Online Analytical Processing (OLAP), and predictive analytics. In relation to case management, BI supports the stakeholders with case/process improvement, product improvement, forecasts, and trend analysis.

10.3.1 Oracle Business Analytic Monitoring

Oracle business activity monitoring (BAM) is a powerful business-friendly tool to support governance type of intelligence. Oracle acquired BAM as part of the PeopleSoft takeover in 2006. It was a hidden gem and came out as a surprise. Having its first implementation in .Net, it was

		Operational						Tactical		Strategic			
Oracle Product	Where to Find More Details	Case Life Cycle Decision- Making	Entitlement Determination	Progress Measurement	Data Entry Validation	Information Discovery	Monitoring Risk Parameters	Workload, Efficiency Improvement	KPI	Customer Satisfaction	Case and Process Improvement	Product Improvement	Forecasts, Trend Analysis
Oracle BPM/ ACM	Chapter 7	x	X	Х				Х					
Oracle Siebel	Chapter 8	x	Х	Х	х			Х	x		Х		
Oracle Service Cloud	Chapter 9	x	х	Х	х			Х	x	Х	х	Х	х
Oracle Business Rules	Chapter 7	х	х	Х	х								
Oracle BAM	Section 10.2.1						X	Х	x		Х		
Oracle OEP	Section 10.2.2						X						
OPA	Section 10.2.3		Х		х								
Oracle Endeca	Section 10.2.4					X							
Oracle BI											Х	х	х
Oracle RTD	Section 10.3.5	x									x		

 Table 10.1
 Oracle Products Supporting Intelligence Needs

subsequently transferred to a Java implementation as part of the Oracle Fusion Middleware 11G. Now in the Oracle Fusion Middleware 12C version it is completely reengineered in order to be completely integrated with the BPM, Service Oriented Architecture (SOA), and BI products, and at the same time to be highly scalable and easily managed by means of the implementation on top of the Oracle WebLogic Server.

10.3.1.1 Functionality

BAM is aimed at providing real time insight in tactical and operational intelligence or "right" time intelligence as named by Oracle. Figure 10.2 provides an overview of the components that help deliver this real time insight:

In the *presentation* layer the following functional components are shown that deliver the insight to the end users:

- *Dashboards* deliver real time insight through a large variety of page compositions with different chart forms combined.
- *Alerts* are used to start follow up activities, such as starting a case or process, based upon a specific situation happening.
- Reports deliver insight via standardized reports delivered at specific times.

The *composition* layer combines together the information required by the functionality exposed in the presentation layer:

• *Workload insight* shows the case and process execution timings in order to deliver insight in the efficiency of the operation.



Figure 10.2 BAM functional components.

- *KPI insight* combines the real-time execution and content data and verifies these against predefined KPIs. These KPIs can be related to efficiency, regulatory, and/or customer-related parameters.
- The *monitoring risk parameters* relates to measuring case- and process-related data against predefined risk parameters (as described in Section 10.2.2).

The *integration* layer is used to receive or request information from the related applications placed in the application layer. This information is passed to the composition layer where the data composition takes place.

In the *application* layer, the application types are shown that deliver the information, the related BPM, ACM, and/or BI applications and external sources.

The *data* layer contains data storage and retrieval functionality. The data entered in the integration layer is stored in the BAM data store. BAM also offers the option to use data that is stored elsewhere in external data stores. A logical view delivers the option to combine both the BAM data store and the external view.

10.3.1.2 How Does BAM Relate to Case Management?

BAM mainly relates to the tactical type of information, as part of case management dashboard design, but also provides some aspects of operational and strategic information:

- The *tactical* information contains the main purpose for the BAM product where direct insight in KPIs and workload is provided through the BAM Dashboards. In the *operational* area, risk analysis provides functionality for monitoring risk areas. This relates to the description given in Section 10.2.2.
- BAM provides in the *strategic* area, insight where the handling of the cases and processes can be improved.

Figure 10.3 shows an example of a BAM dashboard showing KPIs related to the onboarding of partners.



Figure 10.3 Example of a BAM dashboard, related to KPIs for work order management.

10.3.1.3 Technical Architecture

Oracle BAM is based upon already existing Oracle Fusion Middleware building blocks. The main technologies for BAM are as follows:

- Oracle Coherence, the in-memory grid database.
- Oracle Event Processing (OEP) aimed at listening at data streams and described in more detail in Section 10.3.2.
- Application Development Framework (ADF), the standard user interface across the complete Oracle Fusion Middleware and applications stack.
- The Oracle database is used to keep the meta data store (MDS), the repository where design and run-time artifacts are stored.

The BAM technical architecture, shown in Figure 10.4, contains two main parts, the *design* part and the *run-time* aspects.

The design time aspect supports the business with a browser-based solution to create the workload and KPI models, queries, dashboard, and alerts that are used in BAM to present the relevant information. These models and definitions are stored in the MDS, which is the shared repository for BPM, ACM, and BAM.

At run time, the data is delivered from both external data sources and the Fusion Middleware products BPM, SOA, and BI. The BPM, SOA, and BI data is delivered via native and predefined bindings. The external sources deliver their information via database adapters; web services calls or queues towards BAM. Data from both sources is picked up by (the Oracle coherence based) BAM persistence engine. The persistence engine then passes the information to the continuous query service (CQS), which is a query engine that has the ability to listen to a data stream, both parts of the persistence engine as well as already stored data. Every time a



Figure 10.4 BAM technical architecture.

data change occurs, the CQS engine investigates which queries (and related dashboards) are affected by this change. The CQS engine subsequently forward this information to the reportcaching engine, which pushes the result to all (open) dashboards. The CQS also pushes the changes to alert services related to the change and in turn initiates an action, for instance the start of a new process.

10.3.2 Oracle Event Processing

Oracle Event Processing (OEP) listens to data streams and filters out situations that require follow-up activities. These situations can be a unique event or a complex pattern of occurrences happening over a predefined timeframe originating from different sources. When OEP detects a specific situation, follow-up actions can be initiated, such as starting activities in ACM and BPM.

10.3.2.1 Functionality

Oracle OEP is based upon a four-step model, *listen*, *filter*, *compose*, and *send*. Figure 10.5 gives an overview of the different components involved:

- The four-step process *listen*, *filter*, *compose*, and *send* is part of the *composition layer*. The *listen* step receives data from different application sources, via the integration layer. These sources can be applications, social media, and also all connected devices (Internet of Things).
- In the *filter* step, the events are selected that are considered for follow up action.
- The *compose* step combines information together into a message.
- As a last step the messages are sent to applications that are interested in these events, via the channel layer.



Figure 10.5 OEP functional components.

10.3.2.2 How Does OEP Relate to Case Management?

Oracle OEP is tailored at supporting the monitoring part in an incident type of case, as is described in Section 10.2.2.

This can range from offense of health and safety regulations, police-related incidents, to financial fraud–related incidents. Here, two examples from different markets where OEP listens to a data stream and filters out the situations that require further investigation are given.

Example 1: Off-loading products from a truck into a warehouse

In logistics, radio frequency identification (RFID) represents the new standard for the identification and tracking of products. Each product, for example, a box or a crate, is provided with an RFID tag containing a unique identification, which can be read by RFID portals or readers. The movement of the products can be measured by combining the data from the different RFID readers. These RFID tags help support to check the bill of lading against the actual products that are moved from the truck into the warehouse. Another application of RFID tracking is checking the expected off-loading zone against actual loading zone. If a case enters a wrong zone, then an alert, such as a text message to a manager, should be sent out directly. For instance, frozen meat should be in a low temperature zone, but when it is moved into a low temperature zone that is in maintenance, with a higher temperature than allowed according to regulations, an alert should be raised. This alert can be picked up by the OEP engine, which is listening to the RFID application. Since this is a health regulations violation, this might be a reason to start a case or to investigate the situation.

Example 2: Credit card fraud

This example is based upon personal experience of one of the authors. Some years ago, in 2003, at Oracle Open World in Paris, we went to a pizza restaurant at the Champs-Élysées to have some dinner after an exhausting day at the booth. I paid the bill for all the crew with my credit card. In the process of paying, the waiter walked with my credit card to the back and apparently copied my credit card. Two days later I received a phone call from my credit card company that the last 2 days my credit card had been spent at several locations across the globe, Paris, New York, Singapore, and others. The investigation team suspected fraud and wanted to verify if this behavior was correct. Since I did not travel around the world in those last days, my credit card was blocked and all false payments were returned. OEP listens here to the transaction data streams.

The OEP engine listened here to the credit card transaction stream. For this specific fraud example, the filtering part looks at the number of locations for every card ID over a period of, for instance, 24 h. When the number of locations is larger than one, a follow-up step is initiated that checks the locations involved and the feasibility of these locations happening within a certain timeframe. The filtering process can be enriched with complex rule facilities to verify the likelihood of fraud or credit card theft.

Technical Architecture

The main flow of a OEP architecture contains, as described earlier, four main parts, *listen*, *filter*, *compose*, and *send*.

The Oracle OEP implementation is based upon a flow diagram, the event processing network (EPN), in which stages, also named components, are assembled together. Every step in the EPN can receive events (event sink) or send events (event source), which means that at any stage in the flow, filtered events can be enriched and that information also can be sent out to related applications.

The implementation of the *listen* and *send* components is based upon adapters that have by default a queue based and web services type of implementation.

The *filter* and *compose* functionality is implemented in Oracle OEP as stages or components. For the *compose* functionality two options are available, the OEP channel functionality, which is a simple mapping of the received data in the right format or via specifically tailored Java beans.

The most exciting feature is the part of the *filter* functionality where with Oracle Continuous Query Language (CQL), querying of the data stream can be done. CQL is very much like the query language known from the database Sequential Query Language (SQL), with one very big difference, the ability to query data against the passage of time. With CQL we are able write code against time windows looking for an event that has specific properties or even more interesting for a combination of events that attracts our attention. This can be based upon a fixed time window, for example, by counting the amount of events coming in per time frame or starting the range-based query from the moment an event with certain specifics passes.

Here, an example of a query that could be used in the credit card fraud as described in the Oracle Open World at the Champs-Élysées example is given:

```
select distinct(cardId), count(distinct location)
from CreditCardTransaction [range 24 hours]
group by cardId
having count(distinct location) > 1
```

This query listens to a credit card data stream and looks for multiple locations over a range of 24 h.

10.3.3 Oracle Policy Automation

Oracle Policy Automation (OPA) is a business rules engine that is targeted at the handling of complex rules. OPA has its roots in the governmental market where it supports in transforming laws, regulations, and policies into rules in order to determine benefit entitlements and payment amounts. Later in the book, in Chapter 13, we provide an example around this topic. Even though OPA still is strong in the governmental market, the product can be used in any other markets.

10.3.3.1 Functionality

Figure 10.6 gives an overview of the main components of the OPA product, in relation to the products and application layers.

The heart of OPA is the *rules engine* known as Oracle Determinations Engine and is part of the composition layer. Both the design time and run time of OPA have some unique features:

Design-time features:

One of the key features of OPA is the way rules can be designed. Word and Excel can be used to transform the text-based laws, regulations, and policies into rules. A business analyst can do this activity and no support from developers is needed in this phase. A rules canonical data model ensures the uniqueness of the data definitions of the facts used in all the rule definitions.

Run-time features:

OPA often works in environments where both the rules and the customer situation are complex. The end goal of the rules execution is to determine a decision for the customer based upon the customer situation, such as the determination of an entitlement, defining of obligations with the customer, and/or determining a claim decision. The basis for the determination is a version of the laws, regulations, and policies and the customer situation, valid for a specific timeframe. When one of these variables changes, either the rules of the customer



Figure 10.6 OPA functional components.

situation or a recalculation of the entitlement is required. *Temporal reasoning* enables a redetermination and a recalculation of the decision based upon the changed conditions in either the rules or the customer situation. OPA ensures that this redetermination is time aware, meaning that the recalculation is done for the timeframe where the change in rules or customer situation is applicable.

During the execution of the rules, different decisions are made. In some situations, due to compliance or contractual reasons, the decisions need to be stored. OPA enables audit trails containing the decisions on different levels, only the end decision, every in-between decision in the rule execution, or only specific in-between decisions.

OPA can be exposed via a web portal and as a web service. External applications, such as a case management system, usually include the rules execution via *web services* by means of Oracle Determinations Server.

Another way to use the OPA rules engine is as a *portal* through the Oracle Web Determinations product, part of the OPA product stack. This delivers a portal for assessing online a rule set by the customers. The customer is guided through the rules execution in an intelligent way: based upon the information they provide and the outcomes of the in-between results, rules are executed and a conclusion is presented in the portal.

10.3.3.2 How Does OPA Relate to Case Management?

As discussed in Chapter 4, both the *service request* and *incident management* case classifications include *investigative handling* cases in order to determine complex decisions to be made in a case.

Typically, functionalities such as *entitlements determination* are complex and require detailed investigation. OPA can off-load a part of the of *entitlements determination* work by automating the rules execution. This speeds up the efficiency of the case execution and helps improve the quality of the entitlement determination by standardizing the execution.

10.3.4 Oracle Endeca Information Discovery

Oracle Endeca is an e-commerce product that contains intelligence functionality and is beneficial in case management situations where information discovery functionality across structured and unstructured data is needed. Endeca specializes in analyzing and providing business intelligence for unstructured data. One of the unique features of the Endeca product is the guided search functionality, which enables users to access information based upon classifications, and drill down into the data by looking at the data from different views and using any combination of attributes.

10.3.4.1 Functionality

On a high level, the Endeca product contains three main functionalities: integration with structured and unstructured data sources, storage of the acquired information, and end-user information discovery functionality. Figure 10.7 summarizes the main functionalities in relation to the three main Endeca product components:

The *data integration* functionality is contained in the integration and application layers and is supported by the Oracle Endeca Information Discovery Integrator product component.



Figure 10.7 Endeca functional components.

The Integrator product is designed to load data into the Endeca server. The data stores that are fed into Endeca can be of any type, ranging from the database, Enterprise Resource Planning (ERP) systems, Microsoft Excel (XLS), Acrobat Portable Document Format (PDF), and unstructured data originating from Twitter and Facebook. One of the guiding principles of Endeca is "No data left behind," meaning we never know when we can use it, just store it.

The *data storage* functionality is part of the data layer and is supported by Endeca server. Endeca stores its data in a different paradigm than what is used in standard relational databases. In relational environments, predefined structures, such as tables, columns, and relationships determine a fix data model. Data that is entered into Endeca has all sorts of structure that can change over time. This can range from a simple record with attributes to EXtensible Markup Language (XML) structures containing a hierarchy. Upfront, not all (or no) relationships between the data objects are known. In order to challenge the data model knowledge gap, all data that enters the Endeca data is stored as records containing attributes that are simple key-value pairs. Endeca created a query language that enables searching through these key-value pair records and deliver the analytics needed for the end-user information discovery functionality.

The *end-user information discovery* functionality is part of the composition and channel and presentation layer, where Oracle Endeca Information Discovery Studio delivers this functionality. The power of Endeca is to combine all sorts of data across all records (vertical search), and at the same time enabling the enrichment of information by looking at the different data structures related to the queried objects. The Discovery Studio supports the end user with guided search functionalities and has capabilities to drill down and present the information in all sorts of formats.

10.3.4.2 How Does Endeca Information Discovery Relate to Case Management?

Endeca Information Discovery delivers intelligence that is useful in two case management classifications, service request and investigative management; see Chapter 4 for more details.

The *investigation* case management area is the obvious area where Endeca has added value. Specifically, when information is brought together from a large range of structured and unstructured sources, combinations can be made that were not clear before and can lead to new insights. A good example is a police investigation where a combination of police source information from a crime scene, a crime overview of the neighborhood, and social media information and can lead to a breakthrough in a case.

In the service request area, Endeca supports customers finding information in websites. Endeca helps narrowing down the information the customer is looking for with a guided filtering methodology.

This typically takes place in the triage part of a service request. The triage is the preparation phase of customer contact. This phase finishes with a decision whether to proceed with the request. This can lead into a case for further investigation. See Section 4.4.6.2 for more information.

Another example in the *service request* area is providing information about the customer to the company who are about to sell a product to the customer. When the product vendor is about to commit to an obligation that can span over multiple months or years, such as insurance, the company wants to know more about experiences with this customer, experiences that have the same customer profile or who live in the same neighborhood. For instance, if the potential customer wants to have a car insurance but lives in an area with exceeding car theft, the insurance can be denied or have a higher price attached to it.

10.3.5 Oracle Real-Time Decisions

Oracle Real-Time Decisions (RTD) is tailored for optimizing decision-making during customer interactions. RTD is not just a "static" rule engine but adds knowledge of all previous interactions to narrow down the choices that are available in that specific situation. Typical use cases for RTD include call center support, marketing improvement by learning, tailor campaigns on web pages for clients, and drill down toward product choices on the web based upon customer profiles and previous customer behaviors. The RTD base product is a marketing application, but this can be extended in any region.

10.3.5.1 Functionality

RTD enables business users to define performance goals, choices, and rules to support decisionmaking. RTD then adds predictive analytics on top to leverage previously encountered similar situations, and with that help narrow down the choices available for the client.

On a high level the RTD product contains functionalities as shown in Figure 10.8.

The decision engine is the main component of RTD. This is where information and decision components are brought together, based upon which a decision is made. The result of a decision is a weighted list of choices, which enable the customer and/or the business to make a justified decision.

The decision components, based upon which decisions are made, consists of performance goals, choices, rules, and models. As soon is the application is brought into production, the learning part starts and every decision is stored for later reference. The models "learn" from these decisions and customer behavior and take this information into account when the next decision is requested.

10.3.5.1.1 How Does RTD Relate to Case Management?

RTD is typically useful in service request type of cases. As described in Section 4.4, the service request case consist of two main flows, a triage workflow type of pattern and the service request case itself. The added value of RTD is in the triage part where there is intense communication between the first line of contact and the customer. In this area we need to be able to narrow down



Figure 10.8 RTD high level overview.



Figure 10.9 Usage of RTD in service request triage.

the choices for the customer. This in order to decide if the customer likes to proceed or that the response was adequate and no further activities need to be done. RTD and specifically the predictive analytics enable improving the efficiency in this area, providing a selected set of choices based upon which customers can make their decision. RTD also improves the quality of the choice by tailoring the choices for the client and learning from previous experiences with other clients. As a result, RTD helps lowering the initial costs during the initial customer service request stage and provides the customer with a better advice.

In Figure 10.9 encircled is the area in which RTD can support the assessment part.

10.4 Summary

In this chapter we focused on the usage of intelligence during the case execution and the implementation with Oracle products. The different stakeholders during the execution of a case use intelligence in order to make decisions. These decisions support the stakeholders on the following three levels:

- 1. To support the execution of a single case (operational information level)
- 2. To manage the governance of cases that currently are executed (tactical information level)
- 3. To make forecasts and predictions based upon experiences with cases (strategic information level)

The intelligence design is done in the three functional case design categories and specifically in the following design components:

- 1. The rules design component (life cycle design area)
- 2. The three information levels, operational, tactical, and strategic (information design category)
- 3. The dashboard design (interaction design categories)

We looked at different Oracle products that support the intelligence functionalities in the different intelligence-related design areas. The relation between the intelligence functionalities is summarized in Table 10.1.

238 Oracle Case Management Solutions

Oracle BAM can be used in all three information areas, but the main focus for BAM is the governance-related functionality. In this area BAM delivers dashboards that give insight in workload management and KPIs.

Oracle OEP listens to data streams and filters out situations that require follow-up activities. OEP supports case management with the monitoring part in incident classifications.

OPA is a business rules engine that is targeted at the handling of complex rules in order to determine entitlements in service request classifications.

Oracle Endeca supports case investigations with information discovery functionality across structured and unstructured data.

Oracle BI supports the case stakeholders with strategic information aimed at case and process improvement, product improvement, forecasts, and trend analysis.

Oracle RTD supports service request classifications in the triage during the interaction with the customer. RTD supports the customer to narrow down the choices for the customer.

Chapter 11

Concluding Remarks

In this section we discussed the different Oracle technologies supporting case management, Oracle Business Processes Management (BPM)/Adaptive Case Management (ACM), Siebel, and the Oracle Service Cloud. Each of these products is tailored for different aspects of case management, when comparing the following usage characteristics:

- Main focus of functionality—What are the main case management design categories and classifications that the product relates to?
- Delivery type—Is the product delivered as a single entity or an umbrella type of solution?
- Product type—Is the product middleware, package based or cloud based?
- Agility at runtime—What agility does the product offer at runtime, does the product relate to ACM or production case management (PCM)?
- *Oracle ACM* is part of the Oracle Service Oriented Architecture (SOA)/BPM suite and is based upon a middleware based product stack. Strong capabilities of Oracle BPM/ACM and SOA are supports for the event based lifecycle design and the integration capabilities. With these characteristics Oracle BPM/ACM is a good choice for an umbrella type of application used in an enterprise wide and cross application environment. This umbrella type of application takes responsibility for the end-to-end management and monitoring of all activities within the context of a case.
- *Oracle Siebel* is a package based solution and is based upon customer relationship management (CRM) functionality. Key elements in the Siebel solution are the preconfigured CRM based data model and the extensibility of the solution. Siebel extended the CRM capabilities with case management functionality into different industries and cross-industry specific solutions. The Siebel product can be used in isolation, as a solution containing all case management functionality, and can be used in collaboration with other applications in a wider end-to-end scenario.
- The *Oracle Service Cloud* is a cloud-based application aimed at delivering support for product related interactions with customers. Key element in the service cloud solution is the knowledge that supports both self-service search and customer contact center interactions. The product enables companies to position themselves in the area of product leadership.

The service cloud has strong capabilities in the portal and channel areas, where it enables the self-service portal through browsers and mobile, direct interaction with the contact center via a variety of channels and customer interactions with social communities.

Table 11.1 summarizes the characteristics of the Oracle case management product, as described earlier in this book.

All solutions based upon Oracle BPM/ACM, Siebel, and the Oracle service cloud require some level of intelligence at operational, tactical, and strategic level. Oracle BPM/ACM, Siebel and

Focus Area	Oracle BPM/ACM	Siebel	Oracle Service Cloud
Case management design area	Life cycle design	Information and life cycle design	Information and Interaction design
Classification support	All three case management classifications are supported, service request, incident management, and/or investigative solution	All three case management classifications are supported: service request, incident management, and/or investigative solution	Two case management classifications are supported: service request supported via linvestigations.
Delivery type: Full implementation or umbrella type	Umbrella type	Full implementation	Full implementation
Product type	Middleware	Package based	Cloud based (SAAS)
Agility at runtime: ACM or PCM	ACM type: Ad-hoc processes as part of a case can be started Flexible stakeholder model Ability to add new event publishing sources	PCM type: The solution with all its aspects is defined upfront. Flexibility at run-time of the state-model is possible	PCM type: Most functionality of the service request is defined upfront. The inclusion of OPA (RightNow policy automation) delivers flexibility at run-time. The continuous improvement of answers improves the quality of the service request.

 Table 11.1
 Usage Characteristics of Oracle Case Management Products

the Oracle Service Cloud all provide some level of support for intelligence functionality. When a specialized intelligence functionality is required (this is not the core of one of these products) additional products can be added to the solution. We looked at Oracle products that implement the case management intelligence functionalities:

- Support for the execution of a single case (operational information level) is provided by the products Oracle policy automation (OPA), Oracle business rules, Oracle Endeca and Oracle real time decisions (RTD).
- For managing the governance of cases that currently are executed (tactical information level) Oracle BAM is the de facto product in the Oracle landscape.
- To make forecasts and predictions based upon experiences with cases (strategic information level) is supported by Oracle business intelligence (BI).

CASE MANAGEMENT EXAMPLES



Introduction

In this section, we provide four examples to explain how case management can support industry processes. The aim of these examples is to show the usage of the case management solution framework in practice, including the determination of classification type; discuss the specific design components; provide the technical solution mapping; and map their functionality toward Oracle products. The examples are simplified for the sake of brevity.

Each example chapter starts with a description of the specific industry process and the related business challenges. Then we look at the solution details to implement the example based upon the case management solution framework described earlier in (Section 2—Case Management Solution Framework). In the solution description, we start with the design aspects of a case management solution as described earlier in Section 2.

The design of a case management solution contains three main business design areas:

- The case life cycle design, which supports the coordination of activities in achieving the end goals
- The case information design, which deals with all sorts of data, documentation, and media flowing through the case
- The case user Interface design, which maximizes the support for the case work force via portals, dashboards, and a variety of channels

These three business areas are interrelated and supported by the cross-functional design areas integration and roles and authorization.

In every example description, we look at different design components for which the example provides specific characteristics. Figure S.1 gives an overview of the case management examples and the design components that are highlighted in the different chapters.

Each chapter concludes with a mapping toward Oracle Products that can be used to implement this specific example.

		Case Li	ife Cycle .	Design	Case Information Design			Case Interaction Design		
Case Management Example	Short Description	Case Flow/ Process	Rules	Events	Document/ Media	Data	People Relationship	Portal	Dashboard	Channel
Mortgage request	A client requests a mortgage loan to finance the property he/she wants to buy									
Welfare request	Provisioning of welfare service to the citizens, such as child benefit and pensions									
Crime to court/ criminal investigation	The "crime to court" deals with all the activities from the moment that a crime has been committed until the moment somebody is brought to trial and is convicted									
Insurance (property) claim handling	Handling of a property claim entered by a customer									

Figure S.1 Case examples, design overview

Chapter 12

Mortgage Request Handling

12.1 Introduction

The goal of this chapter is to:

- Use the mortgage request process as an example of a case management solution based upon a service request classification in combination with some investigation cases
- Show how the mortgage request related business challenges *customer*, *regulations*, and *financial risk* impact the case management solution
- Show examples of the case management design components: lifecycle design, information design, people relationship design, and portal design
- Determine the solution for the mortgage request handling, and determine the Oracle products needed to implement the solution
- Show how the choice for a service request related operating model impacts the required functionality, technology choices, and supporting Oracle products.

12.2 Reading Guide

This chapter contains two main parts, the first part describes the mortgage request handling process including the main business challenges. The second part determines the solution for mortgages based upon the case management solution framework and contains a mapping towards Oracle products.

12.3 Mortgage Request and Business Challenges

12.3.1 Mortgage Request

The mortgage request handling process is used as an example of a predictable case management solution with a high level of customer/stakeholder interaction. This process is dominated by interactions between the customer, the mortgage provider, and other verification related stakeholders. These interactions are required to judge the customer situation, estimate the value of the property the customer likes to buy, evaluate what product options and variations are available, and check what is affordable for the customer. Even though the mortgage product itself can be complex, the process towards an agreement on a mortgage loan follows a sequential and predictable path containing sales, origination, and fulfillment. The mortgage provider starts a series of internal and external investigations in order to check the identity, solvability, and trust ability of the customer and the value of the property. After discussions with the customer are ended and the customer investigations turn out to be positive, a contract can be created and passed by the notary. After signing the contract, the funds can be made available.

Deviations in the standard flow may cause a cancelation of the process or a return to a previous process step. Nonpredictable events that impact the mortgage handling can originate from both the customer and the mortgage provider. The customer may for instance decide not to buy the house and stop the procedure. The mortgage provider can decide that a particular type of property ground leasehold is not allowed, and therefore the mortgage application will be terminated from the mortgage provider's side. Going back some steps in the mortgage a request process can be the result of incorrect entered data. A customer states at the start of the mortgage product types and variations. When during these discussions the employer income statement arrives and shows that the customer earns €34,500 annually this will not fundamentally change the proceedings. If however the statement shows that the customer's salary is €27,500 annually, the mortgage process will go one or several steps back, or go back to the initial advice stage.

12.3.2 Business Challenges

Three main areas pose business challenges for the mortgage institutes:

- 1. How are we going to serve our customers.
- 2. How to deal with new regulations imposed on the way mortgages are dealt with.
- 3. How to minimize the probability of default.

12.3.2.1 Serving the Customers

Mortgages always used to be a one-size-fits-all offering. The only difference between the mortgage providers was related to the interest rates. The last 10–20 years shows a shift towards three different ways of how customers want to be serviced, and how mortgage lenders position themselves to meet the customer needs:

- Should they go for *customer intimacy* and deliver customized products.
- Should they aim at *product leadership* in which they provide state-of-the art products.
- Should they focus on *operational excellence* and deliver a standardized product for a low price.

12.3.2.2 New Regulations

As a result of the financial crisis stricter regulations are imposed on the financial market. One of them being "Banks will be held accountable for making sure that borrowers have the ability to repay their loans, a sharp contrast from the previous housing bubble where some banks failed to check applicants' income or assets before they issued a loan." as is presented in an article in the MoneyNews website (http://www.moneynews.com/Personal-Finance/mortgage-banks-lender-regulations/2013/12/06/id/540393/).

Another regulation requirement is that end-to-end auditing is required. End-to-end auditing requires that every activity and decision in the mortgage request process needs to be recorded for future reference by authorities.

From a regulatory perspective the financial institutes are required to "know your customer." Knowing your customer, or verifying the identity of the customers, has a negative connotation. This verification includes checking the customer's involvement in criminal activities, such as fraud, money laundering, or terrorist financing.

12.3.2.3 Minimizing the Probability of Default

When a loan has been granted to a customer there is always a chance that this customer is not able to meet his/her debt obligation. In financial terms this is known as "probability of default." A financial risk profile of the customer gives an insight in the probability of a default. This risk profile is created in the mortgage origination phase of the mortgage request process.

12.4 Deriving a Solution Based upon the Case Management Solution Framework

12.4.1 Introduction

In Section II we presented the three steps leading toward a solution: determine the case management classification, the required functionality, and the required technical capabilities. We will focus specifically at the impact different operating models have on the solution. In addition we will provide a mapping towards Oracle products.

12.4.2 Determine Case Management Classification

The case structure for the mortgage request handling consists of multiple cases based upon two case management classifications, the *service request* and *investigations*.

- 1. *Service request*: A client requests a mortgage to finance the property he/she wants to buy, followed by interactions between the customer, the mortgage provider, and other related stakeholders to determine the outcome. In its simplest form, the outcome for a mortgage request by a customer is approved or denied. After approval different options arise relating to product varieties, duration of the mortgage, and payment schedules.
- 2. *Investigations*: As part of the origination and the fulfillment phase different investigations are started, such as the investigation of the customer financial risk profile. Each investigation is a separate case and connected to the service request case.

12.4.3 Determine Functional Components and Needs

Here we will focus on case management design components for which the mortgage request process provides good examples: the *case flow and process design*, *data design*, *people relationship design*, *portal design*, and *integration design* as shown in Figure 12.1.



Figure 12.1 Solution topics (in white) discussed in this chapter.

The requirements for the mortgage lending process depend on the strategy that is chosen as to how to position the mortgage product in the market. In Section 4.4.4 we discussed the three strategy choices for market leadership: *customer centricity, product leadership,* and *operational excellence*. The choice for a specific strategy does not impact the mortgage process flow design, but impacts on how other design components are fulfilled. Table 12.1 shows the requirements related to the choice for a specific strategy and on which design component this has an impact.

12.4.3.1 Case Flow and Process Design

As is described in Section 3.3.1, the case lifecycle design category describes the coordination of activities in reaching one of the end goals and contains three design components, case flow and process design, rules design and event design.

On a high level the mortgage request process contains the following parts: Sales, origination, and fulfillment. The lifecycle of the mortgage request is dominated by a milestone based execution of activities. Prepopulating a list of activities per milestone enables the mortgage (case) workers to choose the most appropriate task/activity needed at a certain moment (Figure 12.2).

The *sales* phase in the mortgage lending process is the phase where the customer shows his interest in a mortgage and the first contact is made, even an early "no" might be given in this phase. This phase relates to the service request triage phase as presented in Figure 4.1, and is detailed further according to the triage model in Figure 4.2.

Origination is the process of creating the *mortgage*. During this process the customer delivers all information needed for the mortgage provider in order to make a decision, subject to proof of all required information by the client and other third parties in the next phase (e.g., income statements, valuation reports, etc.). During this phase additional investigations might be started in order to validate the request.

The *fulfillment* phase is the last phase in the mortgage process whereby the mortgage offer is finalized, all checks are verified, the customer signs the contract, and the money is released.

Both the origination and the fulfillment are milestones in the service request handling case flow and can be detailed according to the case flow and state model in Figures 4.2 and 4.4.

Strategy Choice	Requirements	Impact on Mortgage Request Solution Requirements	Focus on Design Category/ Component
Customer centricity	The customer wants a specialized offering, specifically tailored to his needs	Ensure that the involved sales person has all information of the customer available	Customer interaction/portal Cross functional/integration Customer information/data and PRM
		Give the customer insight in all proceedings of the case	Customer interaction/portal Cross functional/integration and security Customer information/data
		Include direct communication with the customer	Customer interaction/ channel
		Management steering via dashboards, KPIs related to: Mortgage process progress (minimize customer waiting time) Customer feedback (how are we doing)	Customer interaction/ dashboard
Product leadership	Responding rapidly to changing market situations and changing customer needs. For example, currently property values are lower than the mortgage amounts, therefore customers expect a new way of refinancing their mortgage	Be able to enter a broad range of products through a rich user interface	Customer interaction/portal Cross functional/integration Customer information/data

Table 12.1 Strategy Types and Impact on Solution

(Continued)

Strategy Choice	Requirements	Impact on Mortgage Request Solution Requirements	Focus on Design Category/ Component
		Implement complex calculation models supporting customer wishes.	Customer lifecycle/rules design
		Fast support for new technologies	Cross functional/integration
		Management steering via dashboards, KPIs related to product type sales	Customer interaction/ dashboard
Operational excellence	Fast, cheap, and no hassle The McDonald's type of offering, which is completely predictable with no surprises and because of the scale of delivery it can be made cheap	Automate as much as possible.	Customer lifecycle/case flow process
		Automate rules	Customer lifecycle/rules design
		Minimize customer contact	Customer lifecycle/case flow process Customer interaction/portal and channel
		Management steering via dashboards, KPIs on: Amount of completely automated mortgage requests Amount of exception handlings and cost involved	Customer interaction/ dashboard

Table 12.1 (Continued) Strategy Types and Impact on Solution



Figure 12.2 Mortgages request process flow—high level.

The process toward an agreement is highly predictable with a few exceptions originating from internal or external events:

- As described earlier in the *mortgage request* description there could be moments in the chain of activities where the process needs to go back one or more steps that can originate from both parties
- Fraud suspicion immediately stops the progress of the mortgage request process.

12.4.3.2 Case Information Design

As is described in Section 3.3.3, the case information design deals with all sorts of data and documentation flowing through the case, and maps the relationships between all persons and/or assets involved in the case. The case information design category contains three design components, data design, document and media design, and people relationship management design.

12.4.3.2.1 Data Design

During the mortgage lending request multiple persons can be working on the same mortgage request case. These persons originate from different departments and all have different views and activities. In order to be able to service the customer in a better way and at the same time also fulfill the compliance requirement a "one customer view" is required. This provides a complete overview of the customer including all products in use, know your customer, financial risk profile, and history of all activities and contacts. Providing a "one customer view" to all stakeholders ensures that no mistakes are made in the handling of the customer request. Another advantage is ensuring that questions are raised only once to the customer, this avoids irritation of the customer and reduces the risk of errors due to double entry of data. This data is provided to the stakeholders via a variety of channels.

Figure 12.3 provides an overview of the data objects needed when working with a mortgage request. These data objects are combined in folders related to the information needs of the mortgage workers:

- In the "*mortgage request*" area all the data related to the mortgage request are kept. This includes the involved persons, discussions, (intermediate) decisions, and delivered documents.
- The "customer details" area gives a full overview of the customer in relation to the mortgage request. It starts with the accounts details such as place of residence and (bank) account details. In related parties the persons or businesses, related to the mortgage request, are listed. The mortgage worker needs to have an overview of all products acquired at the mortgage provider that are currently used (current "live products.") During the mortgage request insight in the risk profile is needed, for instance to determine the chance of "default." This risk profile is updated on a continuous basis for instance with the payment behavior of the customer.



Figure 12.3 One customer view.

- The "*know your customer*" area contains all information related to possible criminal activities by the customer. This information is required by the regulator and gives an overview of the checks that are performed in the past including all external information that can point to criminal behavior.
- The "*mortgage request process*" area contains all information that gives a full overview of the current mortgage request process. It contains information about the current status of the request, outstanding actions by the stakeholders, lists the persons involved in this mortgage request, gives an overview of the discussions with the customer, and related documents.
- The "*history*" area contains all previous experiences with the customer. It provides an overview of the *previously acquired products* with the mortgage provider, what *decisions* are made in the relation with the customer, for instance was a request ever refused, and an overview of the *complaints* filed by the customer. It also contains an overview of all *arrangements* made with the customer. The *payment history* is an indicator for the risk profile of the customer. The "*involved persons*" area gives an overview of all persons working for the mortgage provider who were involved with this customer.
- The "mortgage policies and regulations" area is based upon the laws and regulations that the mortgage provider should abide by.

The *one customer view* is based upon data originating from a variety of data sources from within the system landscape or provided by external parties. The next step after the case management information design is the inclusion of the technical design areas "integration" and "security." Specifically related to the security design is that not everyone involved in the mortgage lending case is allowed to view and/or change the related data. Sensible information such as "know your customer" may only be viewed and/or changed by the mortgage owner.

12.4.3.2.2 People Relationship Design

The mortgage request process is a good example of a process with a complex people relationship design. During a mortgage request multiple persons are involved in the execution; the customer, persons who represent the customers (intermediaries), stakeholders within the mortgage provider, and external persons involved in investigations. Figure 12.4 shows the persons involved and the relationships between these persons. Three main stakeholders' groups can be distinguished; customer related, all stakeholders related to the mortgage provider, and external stakeholders.

The *customer* has direct contact with the mortgage provider or arranges an intermediary to execute all activities with the mortgage provider. When the contract is signed the customer has contact with a notary who legally witnesses the sign off. The customer also needs to contact an insurance company to arrange an insurance for the property.

At the *mortgage provider* two classes of employees are responsible for all interactions. The front office is responsible for all customer contact. The middle office is responsible for all internal contacts for arranging the mortgage and externally for acquiring services required for creating the mortgage loan.

The *external* stakeholders are external specialists who provide specific services that are needed during the mortgage loan request process, such as checking for fraud and estimating the actual value of the property.



Figure 12.4 Mortgage stakeholders.

12.4.3.3 Case Interaction Design

As is described in Section 3.3.4, the case interaction design enables maximizing the support for the case work force and interactions with the customers. This design area contains three design parts, portal design, dashboard design, and channel design.

12.4.3.3.1 Portal Design

The portal design supporting the customer with the mortgage request depends on the choice for a strategy model as described in Table 12.1. The strategy choices *customer centricity*, *product leadership*, and *operational excellence* have different behaviors toward the interaction with the customer.

For *customer centricity* the portal design is focused at delivering insight in the progress of the mortgage request and share the in-between results with the customer. The portal (and the related channel) also needs to support direct interactions with the customer to enable a good relationship with the customer.

For the *product leadership* strategy the focus is at delivering a rich portal that enables the customer to make combinations of mortgage lending products and the customer situation with new innovative products. The portal should focus on delivering both content and calculation methods and must have the ability to include, in an agile way, new campaigns and products.

In the *operational excellence* situation the portal is aimed at providing the customer a simple means to run through the standardized mortgage request process. Even though this strategy is aimed at delivering low cost operations, the option to contact a service employee directly in the portal should be part of the process.

12.4.4 Determine Solution Mapping

12.4.4.1 Impact of a Mortgage Request Operating Model on Solution Capabilities

The main part of the solution for the mortgage request is to build the lifecycle and customer interaction design categories. For these two design categories the difference in the strategy models is most significant as can be seen in Figures 12.5 and 12.6.

In Figure 12.5 the lifecycle capability of a mortgage request is shown related to the three operating models. The strategy determines what orchestration capabilities should be used in the solution; direct human interaction (H2H), human-to-application interaction (H2A) or full-scale application-to-application interaction (A2A). In the *customer centricity* strategy there is a need for direct collaboration in the mortgage case that requires the need for a human to human (H2H) capability. The *product leadership* strategy requires interaction between the customer and case system; this relies on the human-to-application (H2A) capability. The *operational excellence* strategy on the other hand relies completely on an automated flow guided by application-to-application (A2A).

In Figure 12.6 the different capabilities are shown for the case interaction category. For two design components, the channel and the case portal, the choice for a strategy makes a difference.

All operating models require channel and presentation technical capabilities. First of all access to information is provided via a *web browser* and *electronic channels*. This information is provided via an overview with information coming from different sources such as customer information, case progress, and product information. This is supported by the *user interface integration capability*.

The *customer centricity strategy* requires support from the channel capabilities. In order to provide the customers with tools to enable direct contact at any time and any place, support for *mobile devices* and *apps* is needed.

The *product leadership strategy* additionally is supported by a *user interface* capability in the presentation design acting as an entry form for the mortgage request by the customer. The case request flow for the product leadership is supported by the lifecycle capability *human to application* (*H2A*).

The *operational excellence* operating model in addition needs support for a standardized page flow (*user interface process*).



Figure 12.5 Operating model impact on mortgage lifecycle capabilities.


Figure 12.6 Operating model impact on mortgage interaction capabilities.

12.4.4.2 Mapping Mortgage Request Operating Models to Oracle Products

With the knowledge of the required capabilities for the mortgage lifecycle and the interaction design categories we are able to determine the main Oracle products that can support the mort-gage request functionality.

12.4.4.2.1 Customer Centricity

In order to implement the mortgage customer centricity functionality three Oracle product suites are needed as is shown in Figure 12.7.

Oracle BPM/ACM: Oracle adaptive case management leads the execution of a mortgage request with the customer centricity model and guides the customer through the phases sales, origination, and fulfillment. BPM processes are used to support the case lifecycle with task execution.

Oracle SOA Suite: The Oracle SOA Suite delivers support in three areas—integration, portal, and dashboards.

The Oracle SOA Suite integration capabilities deliver support for all integration activities within the hybrid landscape where the mortgage request operates. Oracle application development framework (ADF) and Oracle WebCenter support the portal functionality. Business activity monitoring (BAM) supports the dashboard functionality, where customer satisfaction is measured.

Other Oracle products: Oracle communicator is used for direct interaction connectivity with the customer.

12.4.4.2.2 Product Leadership

The implementation of the operating model *product leadership* is based upon three Oracle product suites, of which Oracle customer experience (CX) is the main group, as is shown in Figure 12.8.

Oracle CX: Oracle CX delivers in this operating model the main product stack. Oracle service cloud is responsible for steering and content delivery through the product search by the customer. Oracle Endeca delivers sophisticated functionality for delivering information discovery type of functionality as is needed during the search for the best mortgage (related) product.

Oracle SOA Suite: The Oracle SOA Suite integration capabilities deliver support for all integration activities within the hybrid landscape where the mortgage request operates. The Oracle BAM functionality provides dashboard functionality where Key Performance Indicators (KPIs) related to product type sales can be measured.

Other Oracle products: Oracle policy automation (OPA) is an optional component for delivering functionality if complex calculations and rules need to be determined in the mortgage request process.

12.4.4.2.3 Operational Excellence

The product stack for *operational excellence* is the same as used for the *customer centricity* operational model as can be seen in Figures 12.7 and 12.9. The usage of the BPM/ACM suite products in the operational excellence operational model differs significantly from how these products are used in the customer centricity situation.







Figure 12.8 Oracle products supporting the product leadership lifecycle and interaction design.



Figure 12.9 Oracle products supporting the operational excellence lifecycle and interaction design.

The operational excellence operating model is aimed at delivering a standardized model for a low price. This places BPM in the lead, aimed at delivering efficiency through a repeatable and automated workflow. ACM will only be included when an exception situation occurs that requires human intervention.

12.5 Conclusion

In this chapter the mortgage request handling process is used as an example of a service request with a high level of customer/stakeholder interaction.

The business challenges for mortgage request handling relate to *customer*, *regulations*, and *financial risk*. The customer related business challenge deals with how mortgage lenders position themselves, and determines the operating model that is going to be used:

- Should they go for *customer intimacy* and deliver customized products.
- Should they aim at *product leadership* in which they provide state-of-the art products.
- Should they focus on *operational excellence* and deliver a standardized product for a low price.

The impact of the three operating models on the design and technical solution mapping is investigated for the case lifecycle design category and case interaction design category. The following conclusions can be drawn for the operating models:

- For all three operating models the same high-level mortgage process models is used.
- The *customer intimacy* operating model:
 - Interaction with the customers is a key functionality.
 - Support for mobile devices and apps is needed.
 - The Oracle supporting platform is the Oracle BPM/ACM product suite and the SOA suite for integration.
 - Oracle ACM leads the execution of a mortgage request.
- The *product leadership* operating model:
 - The focus is at delivering a rich portal that enables the customer to make combinations of mortgage lending products and the customer situation with new innovative products.
 - Interaction between the customer and a case system is required.
 - The Oracle supporting platform is the Oracle CX product suite with the main products Oracle RightNow for the portal and content delivery and Endeca for information discovery.
- The operational excellence operating model:
 - Is aimed at delivering a standardized model for a low price.
 - Needs support for application to application orchestration and a standardized page flow.
 - The Oracle supporting platform is the Oracle BPM/ACM product suite and the SOA suite for integration.
 - Oracle BPM is in the lead, ACM will only be included when an exception situation occurs.

Chapter 13

Citizen Services in the Public Sector

13.1 Introduction

The goal of this chapter is to:

- Use the welfare request process as an example of a service request supported by complex rules management;
- Show how the welfare request related business challenges for the main three stakeholders, *government*, *citizens*, and *welfare organization* are *cost*, *quality*, *regulations*, and *flexibility* impact the case management solution;
- Focus on case management design components for which the welfare request process provides good examples: case Flow/process design and rules design;
- Determine the solution for the welfare request handling and show that the solution can be fully automated for the handling of high volume standard situations, and at the same time supports knowledge workers act upon complex nonstandard situations.
- Show how the combination of Oracle BPM/ACM and Oracle policy automation supports this solution.

13.2 Reading Guide

This chapter contains two main parts, the first deals with the welfare request process and gives an insight in the business challenges for the main stakeholders. The second part determines the solution for welfare request based upon the case management solution framework and contains a mapping toward Oracle products.

13.3 Welfare Request and Business Challenges

13.3.1 Welfare Request Handling

The handling of a welfare request deals with the provisioning of welfare service to the citizens, and involves products such as child benefits and pensions. Typically a welfare request process deals with a high volume of requests originating from a large customer base. In this process laws and regulations determine the end result, the entitlement. The customer expects to be serviced in the best way with no errors and limited contact.

Social welfare providers support a large customer base. Part of this work is done by hand, which impacts the speed of delivery of a request and can influence the quality of service. The service workers make decisions on "their best effort." Unfortunately this way of working might result in two welfare specialists dealing with the same type of request for two different customers come to a different conclusion, which can differ in the eligibility, duration, and height of the welfare benefit. In most service organizations working practice is silo based, meaning that only a select group of service workers have knowledge of one product, such as child benefit. Downside of this practice is that overlap in knowledge areas is not regular.

The functionality that is needed to support the welfare request is customer relationship management (CRM) in an extreme form. The welfare organization needs information about every citizen in a country or state, all their relationships, history of salaries, welfare schemes, and all decisions made related to the laws at the time the decision was made. This functionality puts a high demand on the system landscape, for instance the required storage needed to store all citizen information.

A functionality that will be described in detail later in this chapter is around "time travel." Decisions around providing welfare to a citizen are based upon information known at a certain moment in time. If it appears that the provided information, such as age, was not correct at that moment, whether caused by a data input error or fraud, the calculation of all welfare schemes for that person need to be recalculated. Needless to say that this is a complex job where all the information needs to be available and changes can be made in retrospect.

In order to fight against fraud and crime, systems of the welfare organization and other citizen providers need to be integrated, within a country but also across borders. For instance the integration between the welfare and the tax system provides an insight in a person's work history during the period he/she was also receiving welfare. This can lead to a renewed insight that the person was not entitled to receive welfare and might initiate a fraud investigation case against this person.

13.3.2 Business Challenges

The three main stakeholders in this area are *government and related politics* making the rules of engagement around the welfare services, the *welfare organization* providing the welfare services and, of course, the *citizens*. The operating model has been the same for decades, and is mainly based upon human activities executed by welfare case workers supported by some level of automation, and executed in segregation from other citizen service providers.

For the *government and related politics* the service operation needs to be more efficient, in control, adaptable to changes, and at the lowest costs. Especially the last reason, cost reduction, is an important driver for changes as a result of the financial crisis. One constraint towards cost reduction is that the service request handling needs to be done within the limits of the regulations.

Citizens request to be serviced better and faster, as they are used to in the current consumeroriented society. As a result the relationship between the welfare service provider and the citizens is changing. Citizens demand online access to their own data and the ability to make changes to this data, through a variety of channels. Predictability of the delivered service is also high on the wish list of the citizens, which includes better upfront predictions of the expected result, timelines, in between updates, and minimum contact as possible. The Dutch Social Insurance Bank, the Sociale Verzekeringsbank (SVB), The Dutch welfare organization, even made their slogan based upon this dogma, "The best customer contact is no contact," meaning we know your situation, and are able to make the best decision with minimum contact.

The *welfare organizations* realize they need to adapt as a result of the changing environment. At the same time they need to deliver high quality work as people depend on the excellent operation of the citizen service providers. Reputation and trust of the citizens are therefore the key words for the operation of the citizen service organization.

13.4 Deriving a Solution Based upon the Case Management Solution Framework

13.4.1 Introduction

The business challenges for the welfare organizations, provided in Section 13.3.2 lead to a renewed solution in which automated case management will play a larger role. Here we provide an outline of the main parts forming the case management type of a solution and focus specifically on the case management design areas case flow/process design and rules design. The structure of this chapter is according the case management solution framework as presented in Section II. We will finish this chapter with a mapping towards Oracle products.

13.4.2 Determine Case Management Classification

The Welfare request can be classified as a *service request*. Citizens claim a welfare request via different channels such as on-line, via a letter or go to a welfare office. Once the request is entered the case manager will determine if the request is legitimate. This result is communicated with the customer. A negative result can turn into a discussion with the customers and a second intake investigation might take place. A positive decision will result into a follow-up *investigation cases* that lead to an entitlement decision. Input for the investigation of the welfare request is the citizen situation, which includes job(s) the customer has/had, the personal situation of the persons, owned assets, previous decisions, criminal records, laws applicable to the welfare request, etc.

13.4.3 Determine Functional Components/Needs

For this example we will focus on case management design components for which the welfare request process provides good examples: the *case flow/process design* and *rules* design as shown in Figure 13.1.

13.4.3.1 Case Life Cycle Design

As is described in Section 3.3.2, the case lifecycle design category describes the coordination of activities in reaching one of the end goals and consists of three design components, case flow and process design, rules design and event design.



Figure 13.1 Solution topics (in white) discussed in this chapter.

13.4.3.1.1 Case Flow and Process Design

The first two processes in the welfare services value chain, *receive a customer event* and *determine entitlements*, as shown in Figure 13.2, cover the handling of a welfare request from a customer to the translation of this request in an entitlement. The last step in the value chain, *prepare financial consequences*, prepares the execution of the financial consequences as a result of this entitlement. The servicing is not part of the welfare request, and handles all arrangements made for a customer once the entitlement is in place.

Receive customer event: The receive customer event business process, as shown in Figure 13.3, describes receiving a benefit request via different channels and the actions taken by the case manager for the handling of the welfare request. In the first part of the process complexity and severity of the request are investigated resulting in follow up actions and priority of execution. This part is also known as triage as presented in Figure 4.2. The case manager registers the request and customer data as part of a task and checks if the data is complete. If not, the case manager contacts the customer for more information. Then the case manager creates a welfare request case and adds the task to the welfare request case. This is typically a workflow type operation and can be implemented as a process.



Figure 13.2 Welfare services value chain.



Citizen Services in the Public Sector

267

Figure 13.3 Receive customer event.

Determine entitlement: Both the determine entitlement and prepare financial consequences are milestones in the service request handling case flow, as presented in Figure 4.3. During the assessment of an entitlement, based upon the citizen request, the eligibility, duration, and height of a welfare benefit are determined by the case manager, as shown in Figure 13.4. If a review of a case is necessary, the case will be delegated to a reviewer (another case manager or a handler). Finally, the decision is sent to the customer and all relevant customer information and the decision on the entitlement is sent to the financial manager. After this activity the follow up step to prepare financial consequences can be started.

13.4.3.1.2 Rules Design

13.4.3.1.2.1 The Know is Separated from the Flow Decision management is related to laws and regulations and tends to change more frequently than the welfare entitlement process flow. The concept "the know is separated from the flow" relates to keeping the design and implementation of the process (the flow) and the rules (the know) separate. The rules for the entitlement decision process can be very complex. This complexity not only originates from the several laws related to the welfare request but also relates to the combination of information that impact the decision; such as current and historical personal situation, specific personal exceptions, the history of claim request, and previous decisions. Decision management needs to be supported by an engine that provides rules management in a business friendly way, but also enables "time-travel" related calculations (see Section 13.4.3.1.2.2).

13.4.3.1.2.2 Multireality and Temporal Reasoning or Time Travel As mentioned earlier, any decision made by the welfare provider is based upon the *reality* known to the provider at a certain moment of time. This reality is used as input for the welfare entitlement, and determines the amount of money to be paid to the customer within specific time intervals, as is shown in Figure 13.5. Multiple parties are involved in the determination of a welfare request: the citizen, the welfare provider, and official registration systems. All of these parties have their own view on the reality. Which reality is the correct one is the prime input for the decision making process. When during the process it becomes clear that the chosen reality was not the correct one, a decision can be made to change the reality to a different one and leads to a recalculation (or revoking) of the provided welfare scheme. Changing a reality as the basis for a welfare scheme can imply changing the past, the present, and the future.

The system needs to be able to store multiple versions of the reality, the source of the reality and an audit trail of the decisions made during the process.

Example

A citizen informs the welfare provider that he will move abroad from May 1 onward. This will impact his social services (products) he has been granted. His child welfare will stop and a "gap" will occur in his pension. The citizen request of the financial implications is calculated based on May 1st. For this, the system must be able do a provisional calculation based on the "truth" of the citizen. The final calculation only can take place when the official relocation message is received electronically from the municipality. The financial implications calculation will now be based on the "truth" of the municipality. When after six months it becomes clear that the citizen actually moved on June 15 instead of May 1, then a new reality based upon this delayed moving date should be calculated.







Figure 13.5 Multireality and temporal reasoning.

13.4.4 Determine Solution Mapping

As we have seen so far in the welfare request process overview and the business challenges for the solution, the case management solution should contain the following main characteristics:

- A predictable milestone based process model
- A high quality overview of the customer
- A complex rules engine, which enables business changes on a frequent basis and recalculation backward in time.

Another solution determining topic is the automation of the handling of high volume standard situations, and at the same time enabling knowledge workers to act upon complex nonstandard situations. One of the key characteristics of welfare request handling is the high volume of requests originating from the large customer base, including all citizens in a country or state. A large part of these welfare requests consists of predictable work that can be handled as straight through processing. Straight through processing or STP is the concept that a request can be handled without any human intervention. This will speed up the work and improve the predictability of the end result. Human intervention in the welfare request case handling is only needed in exceptional areas such as the questions stated as follows:

- Is the customer information complete? [No]
- Do fraud specific indicators provide an indication of a suspicion of fraud? [Yes]
- Do specific personal exceptions apply to the welfare rules? [Yes]
- Is the welfare request related to a change in an existing welfare request, for which the information base needs to be changed? [Yes]

The STP flow is responsible for orchestration of the execution. The actual functionality execution resides in the welfare systems, whether these are legacy (Cobol) systems or new Service Oriented Architecture (SOA) enabled systems. In order to be part of a STP flow, the welfare systems should be able to communicate with the STP flow in a standardized way. As more and more systems originating from other citizen service providers need to be connected to this STP flow an overall communication model needs to be agreed across all providers. This puts high demands on the integration capabilities of the landscape, with high requirements on availability and failover.

13.4.4.1 Determine Solution Capabilities

In the design paragraph Section 13.4.3 we looked at two design components, case flow/process flows and rules. We will determine the solution capabilities for these two design components in this paragraph.

In Figure 13.6 the solution mapping is shown for the case lifecycle design category containing the case flow/processes and rules support and specifically look at the difference between the *large volume STP handling* and the *complex nonstandard situations* handling.

The *large volume STP handling* is aimed at delivering a service with as less human intervention as possible, and contains a predictable execution path. Application to application (A2A) type of orchestration handling supports a process flow that does not require human intervention and



Figure 13.6 Welfare request solution mapping.

requires a predictable flow. Requesting child benefit is a highly standardized process, with predictable steps and activities executed automatically, and only in exception situations specific activities by welfare specialists. Figures 13.7 and 13.8 provide an overview of the Oracle products for the two scenarios. These figures are based upon the mapping structure as described in Section 5.5 and shown in Figure 5.11.

In the *complex nonstandard situations* the handling is executed by case workers, supported by system activities (human to application, H2A support) and contains direct interaction with the customers and other case workers (human to human, H2H support). These nonstandard situations can be determined directly at the start of a case or can be started as an exception flow from the STP process.

For both the STP and nonstandard situations business rules capabilities are crucial for determining the entitlement as discussed earlier in Section 13.4.3.1.2. This support is needed for both the entitlement determination and recalculations as a result of a change in the basis or reality of the entitlement.

13.4.4.2 Mapping the Welfare Request to Oracle Products

The *large volume STP handling* and the *complex nonstandard situations* have different solution characteristics as we have seen in the previous section. This has an impact on the choices for Oracle products to be used for these two scenarios.

13.4.4.2.1 Large Volume STP Handling

For the STP scenario the focus is on application to application (A2A) integration, and is implemented by Oracle BPM (and/or BPEL). Since we know nonstandard situations can be started directly at the start or as the result of an exception in the STP flow, the starting point for ACM is added in Figure 13.7.

A crucial part in this product overview is SOA suite integration part since multiple internal and external systems need to be integrated to deliver a complete customer and case overview enabling a correct entitlement decision.

Oracle policy automation (OPA) is the rules engine that enables the complex entitlement decision process, and also ensures support for time travel related functionality as described in paragraph Section 13.4.3.1.2.

13.4.4.2.2 Complex Nonstandard Situations

The same product stack is used as in the STP situation. For nonstandard situations ACM is in the lead, as opposed to the STP situation where BPM was in the lead. The SOA suite integration is again a crucial part for delivering all information from different internal and external systems (Figure 13.8).

The OPA product is here used for delivering the time-travel functionality (Section 13.4.3.1.2) for recalculation of entitlements, and investigating previous decisions.



Figure 13.7 Oracle products supporting welfare request STP.



Figure 13.8 Oracle products supporting welfare request nonstandard situations.

13.5 Conclusion

In this chapter we have seen that the welfare request handling represents a case management example with a high volume of requests for a large customer base. These cases are based upon a predictable milestone process model and a rules engine dealing with complex regulations and laws. In order to ensure excellent delivery of services to their customers, the welfare provider needs to have access to customer information with a high degree of quality originating from both internal and external systems.

The welfare request related business challenges for the main stakeholders, government, citizens, and welfare organization, are *cost*, *quality*, *regulations*, and *flexibility*. Each of these business challenges has an impact on the case management solution.

The welfare request process case solution has different dynamics for the predictable requests and the complex nonstandard situations. The high volume standard cases are aimed at delivering a service with no human intervention via straight through processing (STP), ensuring lower costs. For the complex nonstandard situations the welfare request requires the work to be done by case workers in an adaptive case management manner.

In the solution mapping and Oracle product determination we have looked at the difference in implementation for the STP and the nonstandard scenarios. The STP scenario is fully automated with Oracle BPM. The nonstandard situations are based upon Oracle ACM. The solution relies heavily on the integration capabilities of the SOA suite integration components, ensuring the quality of information originating from internal and external systems. OPA is used for the complex determination of entitlements, or the even more complex recalculation of entitlements based upon the time-travel concept.

Chapter 14

From Crime to Court

14.1 Introduction

The goal of this chapter is to:

- Use the crime-to-court process as an example of a case management solution based upon an incident management case in combination with investigation cases and connected to many other cases.
- Show how the business challenges for the police force related to crime investigation, *globalization*, *effectiveness of the organization*, *technology*, *social values*, *crime patterns*, and *cost* impact the case management solution and IT landscape.
- Show examples for the case management design components for which the crime-to-court process provides good examples: case flow and process design, event design, document and media design, and portal design.
- Determine the solution for the police investigation handling and show the difference between the *full-blown* implementation scenario and the *umbrella* scenario; each scenario requires a different set of Oracle products.

14.2 Reading Guide

This chapter contains two main parts, the *police investigation and business challenges* and the *solution.* In the first part, an example of a police investigation is described and insight is given in the main business challenges. The second part determines the solution for a police investigation case based upon the case management solution framework and contains a mapping toward Oracle products.

14.3 Police Investigation and Business Challenges

14.3.1 Police Investigation

The crime-to-court process deals with all activities from the moment a law is broken until the conviction of a guilty person. Police investigation and court handling form the main parts of the crime-to-court process. The police investigation part contains unstructured processes and intensive interaction management with involved persons and is also event driven. The second part of the crime-to-court process, after hand over from the police to the prosecutor, deals with the court handling activities. This part is document driven, contains cross-departmental handling, and is a structured and predictable process. In this chapter, we will focus on the police investigation end-to-end process dealing with all the activities from the moment that a crime has been committed until the moment somebody is handed over to court.

The starting point for a police investigation is when the law has been broken or suspicion that a crime took place (or will take place in the near future). The term suspicion of a crime is also known as a lead. When an actual crime has been committed, a motive or reason is needed to create a police case, a prosecutor case, and court case. The lead is preliminarily investigated and if proof is found or more suspicion arises, follow-up investigations are needed. A police case is a container of multiple investigation cases and also can be linked to already existing police investigation cases. Each crime type will lead to a specific set of activities in the case, sometimes very predictable and predefined, but often depending on different circumstances and very unpredictable (led by cause and effect). When the police collect enough evidence, the case will be forwarded to the prosecutor, who will prepare a court case to prosecute the suspect. After preparing the case, the prosecutor will go to court, where the judge decides whether the suspect is guilty based upon the evidence of the prosecutor and the defense of the lawyer. This end-to-end process is simplified in Figure 14.1.

As an example, we will use a police case with the motive of a serious crime. It starts with a phone call of a citizen who reports a burglary at a warehouse. The call will be routed to the incident room (command control) of the police. The officer will open an *incident* to report the crime, circumstances, and location details. Then the officer determines the details of the incident and the priority based upon the provided information. The incident will be assigned to a patrol officer, who is briefed while he is driving to the crime scene. At the crime scene the officer does his investigation, collects more details like interviews with victims and/or witnesses, reports the stolen goods, the offences, suspect description, and the way the burglary took place. The incident level will be raised and automatically a crime case is created when the police officer locates a body on the crime scene. As this type of work is outside the responsibility of the patrol officer, he will hand over the responsibility of the investigations to other officers who also will be assigned to this incident and case. All information that is collected at the crime scene will be recorded as part of the incident, but now all the incident details also become part of the crime case. All collected



Figure 14.1 End-to-end crime-to-court process.

information that is found at the crime scene, such as property information and evidence, is stored properly, according to standard procedures and regulations. After the investigation is finished at the crime scene, the incident will be closed. All further investigations necessary to "build the case" will be part of the crime case. Often these kinds of cases cause an early involvement of the prosecutor, since decisions for further investigation or arrests need approvals from the district authority. After thorough investigations, a suspect is identified and can be arrested when enough evidence is available. This requires an arrest warrant and needs approval from a judge. To fulfill and administer this request, a warrant case is created and needs to be prepared, handled, and approved. This warrant case becomes part of the crime case. After arresting the suspect, the investigation officer will start the interrogation. Depending on the outcome, the officer can decide to proceed and hand over the case to the prosecutor. The investigation officer can also convince the suspect to cooperate with the police and offer him a possible reduction of his sentence. During the investigation of a case, a lot of events might impact this case. Media attention and public opinion can cause the police team to increase in size. There are also various time restrictions during a case, such as a suspect can only be held in custody for a predefined amount of time. When the police force has finished the investigation and plausible evidence is available against a suspect, the case can be forwarded to the prosecutor. The main goal for the prosecutor is to build a case that will hold in court and will prove that the suspect is guilty. In order to build the case, the prosecutor stays in contact with the police to obtain all the required evidence to prove the suspect is guilty. This can result in new research questions and possibly could lead to new arrests. When the prosecutor has finished the case, he is ready to go to court and present the case to the judge. At this time, the opposing lawyer starts preparing his defense for the suspect. He/she bases the defense on the information he gets from the prosecutor and the suspect. Based upon the information of the prosecutor and the defending lawyer, the judge will decide whether the suspect is guilty and pronounces the sentence.

We will finish with some observations that will be used later in the design paragraph (Section 14.4.3):

- Different police cases (with different case types) are linked together
- Involvement of multiple police officers require good hand over moments and communication
- The research is dominated by events altering the course of case
- Both structured and unstructured information can be brought into the case
- External communication to, for example, the press influences the research

14.3.2 Business Challenges for the Police Force Related to Crime Investigation

Police forces worldwide are facing increased difficulties in fighting crime as criminal organizations become advanced and work together on a larger scale. The following business challenges push police forces to evolve into a more effective and efficient organization.

14.3.2.1 Globalization

People are communicating more on a global level. The world market is changing into one marketplace and many people and companies are collaborating globally on a daily basis. Many police forces however are still focused on local level and do not have an extradition treaty with all countries. A criminal in one country can be a free man in another. In some cases this is even possible

between states within a country. In order to fight criminals that work internationally police forces need to work together. Sharing information across police forces enable finding the culprits and the evidence needed to make the arrests. Combining shared information can also help to prevent crime. Using information on a global scale will allow police forces to detect patterns and detect criminal activities much faster. The business challenge around information sharing is the need for a trusted IT environment across departments and jurisdictions. Such a trusted IT environment increases the efficiency and effectiveness both locally and globally.

14.3.2.2 Effectiveness of the Police Organization

The challenges in the area of efficiency are mainly focused on the work that needs to be done in the office. A large part of an officer's working day is spent in the office doing administrative tasks. Most of these tasks require manual data input, which due to the lack of the ability to delegate work leads to higher execution times. Another fact attributing to the amount of work spent in the office is related to the usage of many systems required to find the right information needed in a specific case. This reduces the efficiency, but also the effectiveness as the right information is not always available at the right time and at the right place. Departments working in isolation are another cause for not having the right information. Not all departments want to share the information they have or the information of other departments is not easily accessible. In order to improve the success rate of court cases, the IT systems should be more supportive for the police officers. This can be done, for example, by making the police officers aware when new events happened in a case.

14.3.2.3 Technology

People are connected via social media on a larger scale and personal information is accessed easily via Internet. This increases the availability of privacy-related information for criminals. The vulnerability of people has also increased by the fact that most people cannot keep up with the latest technologies and the knowledge of how to protect themselves digitally. This creates opportunities for criminals and at the same time increases the work for the police. The opening of the digital world can also provide law enforcement with extra information to detect social patterns and prevent crime. As criminals adapt to these changes, this puts more pressure on the needs of the police and its partners.

14.3.2.4 Social Values Are Changing Rapidly

People around the world can share videos and images instantly; this changes social values faster than ever before. Ten years ago only journalists could publish information that was available for a big public; today an 8-year-old kid can put material online that can be viewed and shared around the world instantly. Companies are also using the availability of this information to improve their business. This does not always comply with privacy laws, in which area the police force also is involved to protect the larger audience.

14.3.2.5 Adaption to Social and Crime Patterns

Another driver is the current inability to analyze and act on criminal and emergency type of information. Due to the earlier described technological changes and the changes in society, more interesting data has become available to the police. This could drastically improve the ability to predict and prevent crime. The current IT systems landscape faces the challenge to include all new

information sources such as social media and lacks the ability to identify trends and patterns to anticipate and prevent incidents.

14.3.2.6 National Budgets Are Being Reduced

Besides the previously described challenges, the current financial crisis pushes police forces to lower the cost of operation and increase the effectiveness of the systems. The police needs to become more effective in fighting crime and spend less time and money on administrative tasks. From prevention perspective, the police needs to be more visible in society, both on the streets and in the digital world.

14.4 Solution Design Aspects

14.4.1 Introduction

In Section 14.3.2, we described the current challenges the police forces are facing, and in which areas change is needed. Here we will address these challenges in the described solution, specifically tailored to the police investigation as part of the crime-to-court chain. We will use the case management solution framework as presented in Section II to describe the solution, and consequently will map the solution toward Oracle products.

14.4.2 Determine Case Management Classification

The police investigation process can be classified as a combination of *incident management* and *investigation* type of cases. As a result of an offense a police case is started and multiple investigations can be started. The targeted end result is an arrest with proven evidence for handing over to the prosecutor. The path toward this result is unpredictable in behavior and contains unstructured processes. Special events, such as new evidence or witnesses giving new information, can influence the course of the investigations. These special events can even have an impact after a person is convicted and the case is closed, which can lead to a release of the person and a reopening of the case.

The second part of the crime-to-court process starts after the police handed over the results of the investigation to the prosecutor. This process has the characteristics of a *service request* and is largely supported by additional *investigations*. The police force requests the prosecutor for a court case against a suspect. This court handling process is document workflow driven, guided by strict regulations, and largely predictable in the execution.

14.4.3 Determine Functional Components/Needs

In this design paragraph we will focus specifically on the design components *case flow and process design, event design, document and media design, and portal design as shown in Figure 14.2. In the design descriptions we will touch the integration and security design.*

14.4.3.1 Case Life Cycle Design

As is described in Section 3.3.2, The case life cycle design category describes the coordination of activities in reaching one of the end goals and contains three design components, case flow and process design, rules design, and event design.



Figure 14.2 Solution topics (in white) discussed in this chapter.

14.4.3.1.1 Case Flow and Process Design

14.4.3.1.1.1 Police Investigation High-Level Flow In the flow diagram Figure 14.3, the different steps of a police investigation case are described. The high-level process flow is based upon a predictable milestone-based process. Three process phases, start, finalize, and close case, are highly predictable in execution; each phase is closed with a milestone. Each milestone can only be closed when the exit criteria are met and are driven by regulations. For instance, closing off the case can only be done when the handover of the case to the prosecutor was successful.

The second step in the process, investigate case, is dominated by a high degree of flexibility and unpredictability. In this phase, the police is trying the find enough information, which is the evidence for an arrest and to go to trial. These research events, such as new evidence or witnesses who testify, can lead to different insights and changed research directions.

14.4.3.1.1.2 Linking Police Cases Together Different types of cases exist in the context of a police investigation, which can be visualized in a hierarchy tree. As part of a police case, several case types can be created and connected together. Some cases can conditionally lead to another case and are connected together. In Figure 14.4, a partial overview of different police case types is shown. Some case types can be more administrative, while others have a primary focus on investigating in the field. Each of these case types has its own way of working, with data needed, roles attached, and activities that need to be done. On the left hand side of Figure 14.4,



Figure 14.3 Police investigation main process.



Figure 14.4 Police case type overview.

the "initiator types" of police cases are shown, which can be either automated via an application, an incident, a lead, or by the court.

In Section 14.3.1, we provided an example of a combination of multiple cases to support the burglary and homicide cases.

14.4.3.1.1.3 State Model One of the elements that support the predictable life cycle of a case is the state model. This model describes the states existing in the lifetime of a case, but also the possible transition between the different case states. The state model of a police investigation is a complex model with detailed sub-case states and state flows going back and forward, as shown in Figure 14.5. Each area in the model indicates a specific case status and substatus. When, for example, the case status is closed, the substatus is either "ready for prosecution" or "at prosecution". The status can only switch between statuses that are next to each other in the diagram. In the same example, when a case has the status "closed" and the substatus "ready for prosecution," the case can either be moved forward to "at prosecution" or moved back to "rejected". After that the case can be made active again and made ready for prosecution. The "happy flow" for a case would go from the case statuses "active" and "submitted" through the submitted substatuses



Figure 14.5 Police investigation state model.

"intake," "screening," "ready for approval," and "approved." Finally, in the "closed" status, the case will go through the substatuses of "ready for prosecution" and "at prosecution."

14.4.3.1.2 Event Design

The course of an investigation can be altered by events interfering with the police investigation. These events can originate by means of human intervention or via automated tooling monitoring (special) events.

14.4.3.1.2.1 Human Intervention All police officers can intervene in the police investigation. The police officer owning the case can decide to terminate an investigation toward a specific suspect as a result of another suspect confessing to be guilty. The police officer owning the case is also the person to intervene in the police investigation as a result of time-related checks. A person is only allowed to be in custody for a certain amount of time, exceeding this time limit is only allowed when reasons are brought in and justified with the prosecutor. Police officers can bring in new evidence or gather new information via witnesses that can influence the flow of the investigation. Another type of intervention can be brought in via related cases. As discussed before, in Section 14.3.1, cases are linked together, a breakthrough in one of the cases can also invoke a change in other cases. A type of event that is not impacting the investigation flow is a request for update by one of the stakeholders and this only results in information handed over. For audit reasons, these requests are stored.

14.4.3.1.2.2 Automated Intervention Automated monitoring of risk- and performancerelated parameters could also impact the investigation flow. Monitoring of transactions on the stock market might at one point indicate fraudulent behavior of somebody who is already under investigation in other fraud investigation cases. Monitoring performance-related parameters might impact the team size and setup. When the progress of some police investigations is not high enough, police officers might be transferred to those cases to speed up the investigations.

14.4.3.2 Case Information Design

As is described in Section 3.3.3, the case information design deals with all sorts of data and documentation flowing through the case and maps the relationships between all persons and/or assets involved in the case. The case information design category contains three design components, data design, document and media design, and people relationship management design.

In a police investigation, all three topics in information design, document and media design, information design, and people relationship design, are important. We already highlighted the people relationship design and the data design in the example (Chapter 12) and will focus here briefly on one area, the document and media design.

14.4.3.2.1 Document and Media Design

In a police investigation, a distinction can be made between the input and output material.

The *input design* is a combination of both structured and unstructured data. This can be movies, interviews, sound tapes, social media information, and physical evidence, which all need further investigation on accurateness and validity for the investigation. The main purpose of this material is to build evidence for a specific case. For future reference this material also might be of interest. This requires an easy way of retrieving data related to the evidence material, such as creating adding tags or keywords to the material reference in a database.

The *output design* is focused on the deliverables that is needed during the handover toward the prosecutor. Since the handover deliverables are strictly guided by regulations, this process is highly structured.

14.4.3.3 Case Interaction Design

As is described in Section 3.3.4, the case interaction design enables maximizing the support for the case work force and interactions with the customers. This design area contains three design parts, portal design, dashboard design, and channel design.

We will use the police investigation process as an example to show examples of *portal* design.

14.4.3.3.1 Portal Design

During the police investigation multiple persons are involved, different cases are linked together, document and media are collected, and officers are working on activities. The police officers working on the case can originate from different departments, jurisdictions, and sometimes even across borders. Each stakeholder must have a view on the activities related to their responsibilities in the case.

In Section 14.3.1, we provided an example of a police case related to a burglary, which also turns out to be a homicide. In that example, the list of involved persons, cases linked together, documents and media, and activities is already growing:

- Involved persons: More than 10 different roles could be identified—citizens, incident room, police officer (triage), patrol officer, victims, witnesses, crime officers, crime agents, prosecutor, district authority, judge, suspects, and media.
- Linked cases: In the example, five cases are described—incident (burglary), crime case, search warrant, warrant case, and court case.
- Documents and media: Report crime scene, interrogation reports, and evidence as inputs for the case. After an arrest, the case file is handed over to the prosecutor.



Figure 14.6 Police 360° case overview.

An overarching police portal is needed that provides insight and access to the different cases, based upon the roles of the police officers. On a case-by-case level, a holistic overview of the required functionality is shown in Figure 14.6.

All functionalities shown in Figure 14.6 have view–access functionality and provide functionality for making changes to the case. In Sections 14.4.3.1.1 and 14.4.3.2, we already discussed the *involved persons, documents and media*, and *linked incidents/cases*. Another functionality the portal should provide is direct access to a police officer at the crime scene via a variety of channels. The officer needs to be helped with all sorts of information related to the crime scene before he/she gets there. And once an officer arrives at the crime scene, he/she needs to able to send recorded material directly to the office and link the material to the case. The last functionality that the portal should provide is give insight in the progress of the cases and the related milestone and activities and also provide means to the police officer to intervene in the case. This functionality is crucial for the handover moment toward the prosecutor. A police case can be lost or withdrawn from court when the delivered information is not complete, inadequate, incorrect, or obtained in a nonlegal manner. The case/process completion status gives insight in activities that (still) need to be done.

Two fictitious examples of a police portal are shown in the following. First in Figure 14.7, the entry point for a police officer is shown and contains the cases he/she has access to, the outstanding tasks, and all sorts of news relevant to the police officer.

The next example in Figure 14.8 shows a view on a specific police case, containing all detailed information a police officer requires. On the right hand side of this portal, measurement information is shown that informs the police officer on the progress of the case. The *case completion*



Figure 14.7 Police portal example.



Figure 14.8 Police case view example.

status provides the officer with a percentage of completed material based upon predefined output parameters. The *case processes* part informs the officer of the processes that are finished, currently in progress, and still to be done.

14.4.4 Deriving a Solution Based upon the Case Management Solution Framework

In this solution mapping part we will look at two solution parts that are important for the police investigation solution, the case life cycle design supporting part and the part related to the case interaction design. The first area supports the flow of activities in all related police investigation cases and the second area supports the actual police officers on the ground with information, involved persons, documentation, related cases, and activities. Both areas are highly supported by integration capabilities. Before we describe the solution mappings for these design categories, we will discuss two solution choices for the inclusion of case management in the landscape, the fullblown solution, and the umbrella type of solution.

14.4.4.1 Full-Blown or Umbrella Type of Solution

In this chapter we have shown the complexity of a police investigation case:

- A large amount of police stakeholders originate from different departments, jurisdictions, and (international) organizations. Each stakeholder must have a view on the activities related to their responsibilities in the case.
- A complex case structure containing a hierarchy of investigation cases and linked to other police cases.
- Support from a variety of systems sometimes across organization borders.
- Input determined by a variety of documents, media, and physical evidence.
- Informal processes related to internal communication.

And (near) future needs

- Support for new technologies
- Ability to investigate social media

The current IT landscape does not entirely support all these requirements for the police force as stated earlier in Section 14.3.2. In the upcoming paragraphs, the capabilities required to build a renewed solution are presented, and a mapping toward what Oracle products could be used.

We will use the police Investigation example to show how to regain control over case supporting functionality in a complex and scattered landscape with two solution scenarios, the *full-blown solution* or the *umbrella type of solution*. These two extremes require different ways of delivery, product set, and timelines.

The *full-blown* solution includes all design categories into the solution, case life cycle, case information, and case interaction. In the full-blown scenario, the case management system takes control over all activities related to the case execution. This requires a replacement of the current case supporting systems and migration of all the data. In Figure 14.9, a simplified overview is given of the full-blown scenario. On the left hand side the starting point is shown before inclusion of



Figure 14.9 Full-blown scenario.

the solution. Even though in the full-blown scenario the solution takes responsibility for the entire case execution, the solution will not include all systems in the landscape, requiring some level of system integration with external systems.

The *umbrella* type of solution only includes the case life cycle category functionality into the solution. In this scenario, the case system takes responsibility for the end-to-end control of the case execution, and leaves the "low" level case activities in the existing systems. The data related to the case also stays in the existing systems. The data "flowing" through the case is the bare minimum required to be in control of the case and spawn off activities in the related systems. Optionally, also the case management portal and dashboard functionality can be included in the solution, when the functionality is not already available in the police investigation application landscape. The portal functionality is used to ensure a complete overview for all police officers. The dashboard functionality is required by management to get an understanding of the performance of the police force. Integration support is crucial for this scenario and all communication between the case umbrella system and the supporting internal and external systems needs to be ensured by integration. Figure 14.10 shows a simplified overview of the three phases for regaining control in the landscape. The first step is to get in control by means of an umbrella type of application on top of the existing applications and processes. This enables control in the landscape and adds capabilities to measure the performance of the overall process and singular activities. From the moment the end-to-end control via case management is in place, the improvement analysis can start. Already, during the control phase dashboards can be utilized, for identifying bottlenecks and delays in the process, enabling for root cause analysis and eventually supporting process improvement.



Figure 14.10 Umbrella scenario.

The full-blown and the umbrella scenario are not exclusive in usage. Often the umbrella scenario is used to rollout a full-blown scenario in a phased way.

14.4.4.2 Solution Mapping

14.4.4.2.1 Life Cycle Category Capability Mapping

Both the full-blown scenario and the umbrella scenario require the same capabilities in the life cycle category, but there is a difference in the execution location of these capabilities, as is summarized in Table 14.1.

The *full-blown scenario* provides all functionalities for the solution, except for the applicationto-application (A2A) integration capabilities; this usually is provided by an external integration backbone and enables communication with external systems.

The *umbrella scenario* provides full support for the capabilities, life cycle management, A2A, and business rules supporting event handling. The other required capabilities, human to application (H2A), human to human (H2H), and business rules are either provided by the umbrella solution or part of the existing systems. As an example, checking evidence is part of the end-to-end police investigating. The umbrella application initiates the evidence validation at the forensic department; the workflow related to the evidence validation is executed by the forensic supporting applications, including all work related to the H2A, H2H, and business rules capabilities.

14.4.4.2.2 Case Interaction and Information Capabilities

The police investigation is a good example of the need for the interaction and information requirements. As stated earlier, a large amount of police officers from different departments, jurisdictions, and (international) organizations can work on the same case. During the execution of a police investigation, different media, documents, information, and knowledge around suspects and victims are

Case Design Component	Capability	Full-Blown Scenario	Umbrella Scenario
Case flows and processes	Life cycle management	Provided by solution	Provided by solution
	A2A orchestration	Additional integration support needed	Provided by solution of crucial importance for this scenario
	H2A orchestration	Provided by solution	Provided by solution or part of existing systems
	H2H collaboration	Provided by solution	Provided by solution or part of existing systems
Rules support	Business rules	Provided by solution	Provided by solution or part of existing systems
Event handling	Business rules	Provided by solution	Provided by solution

 Table 14.1
 Lifecycle Capabilities Usage by the Scenarios



Figure 14.11 Police case overview required interaction and information capabilities.

shared. This needs to be presented to the police officers in a standardized way according to the roles and responsibilities. The full-blown scenario contains all functionality required in the interaction and information categories. For the umbrella scenario, the interaction and information functionality is optional, with the portal design component being the most likely functionality to be included.

In Figures 14.7 and 14.8, we provided portal examples for the police force showing a portal overview and a case-specific page. In Figure 14.11, the design components and required capabilities are mapped against the case-specific page.

Four design categories determine the portal solution as shown in Figure 14.11:

- (1) The information design:
 - The page as designed in the Information design category
 - (1.1) Case content data delivers the specific-case data
 - (1.2) People relationship data gives insight in all persons involved and their relationships
 - (1.3) Case execution data is related to the progress of the case execution
 - (1.4) Documents and media provides insight in all documents and media objects or physical evidence brought into the cases

For the different information functionalities the technical capabilities are added, for example, the case execution data, showing progress bars, is determined based upon aggregated data.

(2) The case Interaction design:

The case portal page contains different user interfaces (user interface integration) and input functionalities for new and changed information via simple entry forms or more complex user interface orchestration (user interface process).

(3) Security design:

The functionality and information presented on a page is made available to the police officers according to their roles and responsibilities.

(4) Integration design:

The integration design relates to the provisioning of data to the portal and supports the channel communication. For the presentation of data on a portal additionally in the back-end, integration with different applications and the content management system needs to be ensured.

14.4.4.3 Mapping Police Investigation to Oracle Products

The full-blown and umbrella type of solutions have different characteristics and as a result also map to different Oracle products. The full-blown scenario is Siebel based and the umbrella scenario is Oracle ACM based.

14.4.4.3.1 Full-Blown Scenario Implementation with Oracle Siebel

The full-blown scenario is based upon a set of Siebel products and can optionally be supported by the WebCenter product Suite and the SOA Suite integration product as is shown in Figure 14.12.

The main Oracle product used in the full-blown scenario is the *Siebel case management product*, and specifically the police vertical, for which Capgemini created a specific implementation called "transform police" or "t-police" (http://www.capgemini.com/public-sector/transform-police-t-police, Capgemini Public Sector, 2015, Transform Police (t-Police)). Siebel Customer Relationship Management (CRM) is the choice for people relationship design and takes care of information of all involved persons in the case such as the police officers, victims, and suspects. The user interaction category is a discussion area for the Siebel implementation. Siebel open User Interface (UI) delivers a rich user interface and is tightly connected to Siebel. The integration capabilities with other



Figure 14.12 Full-blown scenario implementation with Oracle Siebel.


Figure 14.13 Umbrella scenario implementation with Oracle BPM/ACM.

applications, however, are minimal and as such can be used in a Siebel-only scenario. Nevertheless data that is available within Siebel based on integration can be visualized true the Siebel open UI. In situations where the portal needs to provide information from different sources, the *WebCenter portal* is a better choice, since this is tailored to deliver information in a rich user interface from a variety of data sources in a multichannel environment. Looking at the amount of documents and media information that needs to be stored in a police case environment, an external document management system is required and is implemented with *WebCenter Content*. The *SOA Suite* integration capabilities deliver the integration needed on all levels in the solution architecture.

14.4.3.3.2 Umbrella Scenario Implementation with Oracle BPM/ACM

This scenario is dominated by Oracle BPM/ACM, existing applications, SOA Suite integration software, and optionally the Oracle WebCenter Suite supporting portal and document management, as can be seen in Figure 14.13.

The combination of *Oracle ACM* and *Oracle BPM* delivers the backbone for the umbrella life cycle functionality and stores all the end-to-end case execution data. The *existing systems* are guided by BPM/ACM and execute the "low" level case activities and also take responsibility for storing all case content data, documents and media, and the people relationship information. The existing systems could be replaced over time with new application software, while still keeping the end-to-end process up-and-running with the overarching BPM/ACM solution.

The SOA Suite integration capabilities deliver the integration needed on all levels in the solution architecture and has an important role in the architecture enabling all communication between the BPM/ACM umbrella functionality and the existing systems. *WebCenter portal* delivers the 360° overview of the case across all systems in a rich user interface. *WebCenter Content* adds centralized document management capability to the landscape.

14.5 Conclusion

The crime to court process contains two parts, a police investigation and a court-handling process. The first process, the police investigation is an incident management classification and is supported by multilevel investigations; the second process, the court handling has the characteristics of a *service request* and is supported by additional *investigations*. In this chapter we have looked at the dynamics of the police investigation.

Specific attention is given to the research phase of the police investigation. This process is unpredictable as a result of events, such as new evidence added to the case or altering the flow of the process. During all phases, new cases can be started as a result of this police investigation and these cases are linked together in order to build the case to go to court. Even though the research phase is unpredictable, the overall police investigation process is milestone based and guided by regulations.

Police investigation-related business challenges are *globalization*, *effectiveness of the organization*, *technology*, *social values*, *crime pattern*, and *costs*. Each of these business challenges has impact on the case management solution.

During the police investigation, multiple persons are working on the case, all needing insight and access to the case information, according to their roles and responsibilities. We have seen the different aspects that need to be accessible via a police portal in order to support the police investigation: the involved persons, documents and media, linked incidents and cases, insight in the case progress, and the related security. We have discussed the two scenarios for including case management in the landscape, *full-blown scenario* and the *umbrella scenario*, and looked at the respective implementation with Oracle products.

The *full-blown scenario* moves all new and existing functionality into one solution, with Oracle products as the Siebel police vertical and Siebel CRM as the base part of the solution.

The *umbrella scenario* provides support for end-to-end control across the landscape, but leaves the "low" level case activities in the existing applications. The Oracle BPM/ACM product, in combination with the Oracle SOA Integration, mainly supports the umbrella scenario.

Chapter 15

End-to-End Insurance Claim Management

15.1 Introduction

The goal of this chapter is to:

- Use the insurance claim management process, for property claims, as an example of a case management solution based upon an incident management case occasionally supported by some investigation cases.
- Show how the insurance claim–related business challenges on *financial, customer facing*, and *the supporting IT landscape* impact the case management solution.
- Show examples of the case management design components: case flow/process design, rules design, channel design, and dashboard design.
- Determine the solution for the end-to-end insurance claim case and show that end-to-end insurance claim case execution can be automated when it is supported by intelligence related to assessing the claim and fraud checks.
- Show that the overarching case flow is supported by Oracle BPM/ACM, and that the rules functionality is supported by Oracle Policy Automation (OPA) and Oracle Endeca Information Discovery.

15.2 Reading Guide

In the first part of this chapter, the insurance claim process and business challenges are summarized. The second part determines the solution for end-to-end insurance claim management based upon the *case management solution framework* and contains a mapping toward Oracle products.

15.3 Insurance Claims and Challenges

15.3.1 Insurance Claims

There are two categories of claims in general insurance: *property* claims and *injury* claims. Each type has its handling challenges, but for both claim types the claim handling is considered to be the "first moment of contact with the customer" and as such is the moment of truth for the relation between the insurance company and the customer.

Property or material claims like collision damages are typically straightforward. The loss can be determined fairly easy from a combination of a damage report, usually filled in by the involved persons, and a police report giving direct insight into whom to blame and liable to pay for the damages. The client can be offered an indemnity payment or can be serviced by a body shop paid for by the insurer. The time between filing and settling the claim is usually short and tends to vary from hours to a few days.

Injury related claims tend to be much more complex and less transparent. These claims have a higher level of sensitivity and deal with privacy of medical information and contain complex rules for determining the potential loss of claims. Injury claims are usually long-running claims: time between filing and closing a claim can be years. In case of disability, the claim can be open for a long period, often at least a year until finally the grade of permanent disability is determined.

In this chapter we focus on the property-related insurance claims.

The customer can enter a claim via different channels. Traditionally, the most common one used has been filling in a standard form that is sent via mail to the insurance company. Now, more channels are used such as filling in a form via the web or via a call center. A channel that is gaining popularity is the insurance app, which enables entering a claim at the scene where the incident happened. Once the claim is received, the incident is entered into the internal claim-related system and is handled via a combination of manual and automated activities. Based on the insurance agreement, a proposal is made that is communicated back to the customer. If the customer does not accept the offer, (new) discussions will start. These discussions continue up until the moment that both parties come to an agreement. Once the proposal is accepted, follow-up work can be started. When the insurance company offers end-to-end handling of car damages, the entire process of outsourced activities needs to be managed and monitored as part of the claim case.

15.3.2 Business Challenges Related to Insurance Claims

There are three main challenges insurance companies face in the insurance area, *financial*, *customer facing*, and the *supporting IT landscape*.

15.3.2.1 Financial

The insurance market has changed dramatically over the years. With the growth of Internet, all sorts of price-comparison tools introduce a decrease in price levels, which pushed the insurance companies into the red. The consumer loyalty is shrinking and as a result, customers are shopping around for a better-priced insurance contract. The financial crisis impacted both the property and casualty insurers like any other financial institutes and in some countries only government support prevented bankruptcy. In the *financial* area, the first and foremost challenge is lowering the operational costs. Claims management is one of the most important processes within an insurance company. Since the insurance price levels are at a low range, claims management is the place where money can be saved. Another financial challenge that needs to be challenged is claim fraud.

Insurance companies are looking for new ways to be more profitable. One of these ways is to develop new business models in the property claims area. These business models can be based upon a partner network where all the damages of the claim are being handled, such as repairing of the car in a garage that has a partnership with the insurance company. Another new topic in the insurance world is dealing with intelligence based upon "big data," for instance for the investigation of fraud trends. Since these data sets can contain all sorts of privacy-related data, this is a sensitive topic.

15.3.2.2 Customer Facing

The handling of claims is the moment of truth for the customer: it is where an insurer can retain or lose the client. The dilemma for insurance companies is looking for efficiency and driving down cost (operational excellence) while serving the customer in a way that he experiences as personal and fair. This is where social media can act as a big booster or a big crasher. Usually the extremes on both ends of the customer experience spectrum are posted on social media with the negative experiences taking up the majority of the reports. In order to collect and leverage these experiences, insurance companies need to monitor the social media and act directly when the claim experiences move toward the negative end.

15.3.2.3 Supporting IT Landscape

The previous business challenges pose challenges on the current IT landscape, in some cases still supported by old technologies in a scattered landscape. This makes it hard to provide the flexibility by the business and ensure stricter compliance rules defined by financial authorities.

15.4 Deriving a Solution Based upon the Case Management Solution Framework

15.4.1 Introduction

The solution description will be based upon the case management solution framework as described in Section 15.4.2, determining the case management classification, functionality, and the technical solution where we look specifically at the life cycle support, with the following topics:

- Fully automated support for the end-to-end claim handling and at the same time delivering support for knowledge workers in exception cases
- Inclusion of new business models in the end-to-end handling
- Different aspects of rule support for the insurance claim handling

We finish this chapter with a mapping toward Oracle products.

15.4.2 Case Management Classification

The claims request can be classified as an *incident management* type of case. It can be debated if the classification service request also fits, since the customer reports damage to his/her property and requests insurance claim compensation. The incident management process model relates more to



Figure 15.1 Solution topics (in white) discussed in this paragraph.

end-to-end claims management than the service request process model. The focus of the incident management process model is to solve the problem for the customer and detect who is guilty, resulting in a conclusion as to who has to pay the claim. For some of the property claims, inclusion of additional *investigations* might be needed, for instance, in case fraud is suspected.

15.4.3 Determine Functional Components and Needs

We will focus in this paragraph on the end-to-end claim management design components *case flow and process design, rules design, channel design, and dashboard design* as shown in Figure 15.1.

15.4.3.1 Case Life Cycle Design

As is described in Section 3.3.2, the case life cycle design category describes the coordination of activities in reaching one of the end goals and contains three design components, case flow and process design, rules design, and event design.

15.4.3.1.1 Case Flow and Process Design

On a high level, the insurance claim handling is a straightforward three-step process, and is executed in sequence. In Chapter 14 we showed a complex life cycle model for an incident management type of case. What makes the life cycle description for the end-to-end claim handling specific is that it should be able to include new business models and be fully automated with the inclusion of human interaction in deviating situations.

The high level structure of the insurance claim process contains three parts as presented in Figure 15.2:

- Record claim, where the customer files the claim through one of the channels
- Execute claim assessment, which consists of a fraud assessment, a value assessment, and a liability assessment among others
- Settle claim; in this phase an agreement between the involved parties is made and executed to settle the claim



Figure 15.2 Claims management high-level process.

New business models help insurance companies with new means to be profitable. One of the new business models is expanding the chain of activities by including incident solving services via body shops paid for by the insurer.

For instance, a car insurer delivers additional services for settling damages as a result of accidents. For this service the insurer has included towing companies, garages, and rental companies in their network. When a customer reports a car incident via his mobile insurance app, the insurer directly arranges a towing company to tow the car to the nearest garage, arrange a rental car for the customer, and have the garage investigate the costs and do the repairs. The insurance company can also make an offer to the other party involved in the accident for servicing their damaged car.

The impact of the inclusion of such a business model into the offering is that the overarching claims case now should also include the management of all activities executed by the related companies.

15.4.3.2 Rules Design

Intelligence is needed in the claim assessment phase to decide the validity of the claim. As described in the previous paragraph, the aim is to automate the case as much as possible. Two types of intelligence work can be distinguished:

1. Assessing the claim against policies:

One part of the claim assessment is checking the claim against the policies, which is predictable work. This work includes activities such as checking that the terms and conditions are met, verifying that the amount equals the agreed amount, and validate that the claim data is not out of date.

2. Fraud checks:

The key element in fighting fraud is getting an indication of fraud even before the claim is settled. The claim can be flagged ensuring the money should not be paid up until the fraud investigation is done and no problem is detected. Based on information from all claims, a scorecard can be derived that gives an indication of fraud. Such a scorecard is populated with information originating from the insurance company but can also be enriched from external sources. The scorecard indicators need to be renewed with new claims experiences on a regular basis, new fraud tricks, and social media–related discussions. These fraud checks can be automated with a combination of rules engines and information discovery type of applications and can be part of the straight-through processing (STP) processing. The challenge in this area is that the data used in these investigations can be privacy related and sensitive data. This might pose a reputation risk if this gets into the press.

15.4.3.3 Case User Interface Design

As is described in Section 3.3.4, the case interaction design enables maximizing the support for the case work force. This design area contains three design parts: portal design, dashboard design, and channel design.

We will use the insurance claim process as an example to show examples of *channel design* and *dashboard design*.

15.4.3.3.1 Customer Centricity Design—Channel Design

Channel support is important for both the insurer and the customer. For the insurer it is a means to cut down operating costs by simplifying and standardizing the access. For the customers, it gives access where and when they need it. Enabling entering the claim directly at the scene of the incident gives a head start in the claim execution process and the ability to engage with the customer in the earliest stage possible. As can be seen in Figure 15.3, the *record claim* process step is the area where the customer enters the claim. This area needs to be set up in such a way that claims can be entered via different channels without recreating different claim processes for every channel.

15.4.3.3.2 Dashboard Design

Three levels of information deliver intelligence across the lifespan of cases:

- At an *operational* level in real time supporting a (single) case with the right knowledge
- At a *tactical* level during "process alive" time aimed at improving the efficiency
- At a *strategic* level when looking back in time when investigating for process, product and knowledge improvement

Dashboards deliver information at real time on a *tactical* level related to the workload and key performance indicators (KPIs) of the insurance company stakeholders.

Workload information is crucial in the relation with the customers. As stated before, claim handling is the "first moment of contact" with the customer. The workload should be balanced with costs involved and allowed waiting time for the customer. A continuous investigation is needed to keep this balance in the right shape. Seasonal influences can have an impact on the amount of claims that are entered. This needs to be fed into the workload planning and be presented in the dashboards at real time.

All stakeholders have *KPIs* attached to the operation of the claim handling. This can be on a departmental micro level or related to the overall claim operation at macro level. KPIs on a departmental level are related to the efficiency of the operation. KPIs on the overall claim operation relate to effectiveness (customer satisfaction). Figure 15.4 shows a dashboard for work order



Figure 15.3 Channel access in record claim.



Figure 15.4 Governance with dashboards.

management from another market, but with the same characteristics as needed in the claims example. These work orders are related to the delivery of a service to a customer at their homes. The customer needs to stay at home for the delivery of the service, for which a time window is agreed. This process should be balanced against optimizing the costs and acting within the agreed time window. This dashboard shows the daily influx of work orders against the amount of finished work orders, and gives insight in the work orders that are outside one of the agreed parameters.

15.4.4 Determine Solution Mapping

In this solution mapping we simplify the case by only looking at the overarching flow of the endto-end insurance claim handling, the related rules management and assume the existing systems stay in place.

15.4.4.1 Determine Solution Capabilities

Most of the property claims are standard claims and have predictable behavior and expected outcomes. These standard claims have the potential to be fully automated. This will help lower the cost of operation and increase the efficiency of the process execution. The prerequisites for moving the current claim management execution in an automated process is that the supporting applications are able to expose functionality as services via a well-defined integration backbone and that automated intelligence engines can be made part of the automated process. These intelligence engines support the assessment of the claim and check for possible fraud. When one of these checks shows suspicious behavior, knowledge workers take over the control of the execution of the claim and start investigations. Additional integration requirements are related to expanding the process with management and monitoring of new outsourced activities, such as car repairs.

15.4.4.1.1 Case Flows and Process Capabilities

The capabilities for the end-to-end insurance claim are summarized in Figure 15.5. The aim of the standard claim handling is to automate this fully, as described above. This requires application-to-application (A2A) capabilities that connect all functionalities exposed in the existing legacy application without any human intervention. When further investigation is needed, the knowledge workers needs access to the different applications guided by a case life cycle and is supported by human-to-application (H2A) interaction. The system should support flexibility for expanding the end-to-end claim management chain with new business models, where work is outsourced to other companies. This area needs to support both A2A integration capabilities and H2A. A2A capabilities are used for the activities in the end-to-end automated chain. H2A capabilities are needed in situations where additional approvals need to be given.



Figure 15.5 Insurance claim life cycle capabilities.

15.4.4.1.2 Rules Capabilities

Rules support is needed in two areas with different characteristics.

In the *entitlement determination* used for the assessment of the claim against policies, the rules validation is based upon transactional data, originating from the customer and the insurer. The business rules characteristics are based upon a predefined set of policies and regulations that can be complex.

In *information discovery*, used for fraud investigation, a business rules engine uses both transaction data and unstructured data to investigate indications of fraud. The business rule characteristics for fraud handling are based upon a scorecard that is derived from previous known fraud cases, new fraud tricks, and social media–related discussions. This scorecard is changed on a continuous basis and needs intelligence to be able to identify new trends.

Integration capabilities support all the application connectivity between the claims management overarching business process management (BPM) and adaptive case management (ACM) processes and the existing landscape including new outsourced activities and connectivity with social media.

15.4.4.2 Mapping the End-to-End Insurance Claim Handling to Oracle Products

Three main Oracle product groups support the end-to-end claims management solution, Oracle BPM/ACM, Oracle rules-centric products, and the Oracle service-oriented architecture (SOA) integration product stack, as shown in Figure 15.6. The products on the right, in white, are the solution-supporting systems.

The case flows and processes are supported by the combination of Oracle BPM and ACM. BPM supports the standard claim handling; ACM supports the activities required when a claim needs further investigation by a knowledge worker.

For the rules support, two products are used supporting the different requirements. Oracle Policy Automation (OPA) supports *assessing the claim against policies* functionality. OPA delivers a business-friendly rules management tool for complex rules management.

The fraud analysis is executed by the Endeca Information Discovery product. Endeca Information Discovery is used for storing and retrieving unstructured information and contains analysis functionality that is optimized for data discovery. The input for fraud analysis is a scorecard that is compiled together from a combination of different sources such as:

- Transactional claim data delivered by the customer and the insurer
- Unstructured data that originates from different sources such as social media
- Customer related data, such an area and age
- Insured product data

Oracle SOA suite integration supports all the application connectivity within the solution. In this solution, the aim is to automate the claims management as much as possible. This requires an integration backbone that should be robust and reliable.



Figure 15.6 Insurance claim Oracle product mapping.

15.5 Conclusion

In this chapter, we looked into the management of insurance property claims. This is an example of a predictable case management solution, with an intake via different channels. The majority of cases can be handled fully automated. During the automated execution, intelligence helps assessing the claim and looks at possible suspicion of fraud. When these intelligence checks indicate a suspicion of fraud, insurance knowledge workers perform further investigation.

The case structure of the property claims is based upon the incident management classification. In situations in which the claim is not valid or fraud is suspected, investigations are started.

The inclusion of new business models in the claims handling process can help the insurance companies to be more profitable. We have seen an example of such a new business model in the property claim area, with the inclusion of services in the claim process that help solve the incident damage, such as making a car repair handling. The impact of this diversification of services is that the incident case handling is expanded with management of all these activities.

The insurance claims-related business challenges are related to the areas of *financial, customer facing, and the supporting IT landscape*. Each of these business challenges has an impact on the case management solution.

In the solution part, we looked at automating the claims management as much as possible in order to improve cost effectiveness and increase the efficiency. In situations where more investigation is required, knowledge workers take over control over the execution of the claim. Oracle BPM supports the automated part and Oracle ACM supports the knowledge worker part. The automated handling of claim management could be extended by the inclusion of outsourced activities supporting new business models. Another typical part of the solution is the inclusion of two different types of rules engines, one supporting the *assessment of the claim against policies* functionality and the other *fraud analysis*. OPA supports the policy assessment whereas the fraud analysis is supported by Oracle Endeca.

Chapter 16

Concluding Remarks

We provided four examples in this part to show how the case management solution framework is used and under what conditions the different case management-supporting Oracle products can be used in the solution.

The first two examples are of the service request classification and the other two examples are incident management–based case management classifications.

The first example of a service request is the *mortgage handling* in which a client requests a mortgage loan to finance the property he/she wants to buy. This process is dominated by interactions between the customer, the mortgage provider, and other verification-related stakeholders. Even though the mortgage product can be complex, the process toward an agreement on a mortgage loan follows a sequential and predictable path from sales to origination and ends with fulfillment. During this process the mortgage (case) worker sometimes needs to choose the most appropriate task/activity from a preset list. The Oracle products supporting the solution depend on the operating model that is chosen:

- The *customer intimacy* operating model:
 - Interaction with the customers is a key functionality
 - Oracle Adaptive Case Management leads the execution of a mortgage request
- The *product leadership* operating model:
 - The focus is at delivering a rich portal that enables the customer to make combinations
 of mortgage lending products and the customer situation with new innovative products
 - The Oracle supporting platform is the Oracle Customer Experience (CX) product suite with the main products Oracle Service Cloud for the portal and content delivery and Endeca for information discovery
- The operational excellence operating model:
 - This is aimed at delivering a standardized model for a low price
 - The Oracle supporting platform is the Oracle BPM/ACM product suite and the serviceoriented architecture (SOA) suite for integration

The second example of a service request is the handling of a *welfare request* in the public sector, for instance, the handling of child benefit or pensions. The customer expects to be serviced in the best way with no errors and limited contact. The prerequisite for dealing with this type of requests is

that the welfare provider needs to have a complete insight and understanding of the situation of the customer. The execution path of the entitlement process is sequential and can be automated for most situations. Rules management can support the determination of the eligibility, duration, and amount of an entitlement in this automated process. Welfare (case) workers can choose the appropriate tasks from a preset list where the process is nonautomated.

The automated part of the solution can be filled in with Oracle BPM. The nonstandard situations are based upon Oracle ACM. The solution relies on the integration capabilities of the SOA suite integration components. Oracle Policy Automation (OPA) is used for the complex determination of entitlements.

The first incident management example is the *crime-to-court* process. This process contains two main parts, which have completely different dynamics. The first part is the police investigation, and is unpredictable in behavior. In this police investigation part, a lot of investigations may be initiated to reach a successful end. Special events, such as new evidence or witnesses giving new information, can influence the course of the investigations. The second part of the crime-to-court process is when the police has handed over the results of the investigation to the prosecutor. The court handling process is document workflow driven and largely predictable in the execution.

We discussed two implementation types for the solution. The *full-blown scenario* moves all new and existing functionality into one solution, this could be implemented perfectly with the Siebel police vertical, containing all functionality to support the police investigation. The *umbrella scenario* provides support for end-to-end control across the landscape, but leaves the "low" level case activities in the existing applications. The Oracle ACM/BPM product, in combination with the Oracle SOA integration, is a good choice for this umbrella scenario.

The second incident related example is "insurance claim handling." Insurance claim handling is considered to be the "first moment of contact" with the customer, and as such is the moment of truth for the relation between the insurance company and the customer. The flexibility that is needed in the end-to-end claim handling is the inclusion of new business models, such as including servicing the repair of the car, via body shops and garages, paid for by the insurer as part of the claim handling. As a consequence of the large volume of claim requests and the focus on operating costs, a large amount of these types of claims is executed in an automated way. Even though the process is highly automated, the focus should stay on the customer, timely delivery of results, and meeting (or exceeding) expectations. In this automated process, different parameters are checked that might indicate fraud. Oracle BPM supports the highly automated part and Oracle ACM supports the knowledge worker part. Rules engines such as OPA support the policy assessment. For fraud assessment, Oracle Endeca could be used.

TUTORIAL

V

Introduction

In this section, we provide two tutorials for case management products: Oracle BPM/ACM and Oracle Siebel.

In the tutorial for Oracle BPM/ACM, we provide a deep insight into the Oracle adaptive case management (ACM) product with a hands-on tutorial. This hands-on tutorial is based on a credit card charge dispute scenario.

The tutorial for Oracle Siebel is related to an approach of how to design and implement a package-based case management solution using the Siebel application. The way the application is used will be shown based upon a police investigation example implemented with the *Law Enforcement and Policing Operations* extension.

Chapter 17

Oracle BPM/ACM Tutorial

17.1 Introduction

In the previous chapters we showed the different aspects of case management, some examples of where it is applicable and also how to design and architect a case management solution, and looked at Oracle case management technologies. Let us make it a bit more real through a hands-on tutorial using a credit card charge dispute scenario, that a quite a few of us who use credit cards have gone through at least once in his or her life.

The goals of this chapter are as follows:

- Do a deep dive into the Oracle Adaptive Case Management (ACM) product
- Provide a hands-on experience with Oracle ACM
- Focus on the case management life cycle design category to show the strength of Oracle ACM

17.2 Credit Card Charge Dispute: Use Case

17.2.1 Use Case Design Topics

In Chapter 3 we introduced the design components needed to describe the case management functionality. In this chapter we will use a simplified credit card charge dispute case where some case management design areas are highlighted as shown in Figure 17.1. In order to get a good understanding of the Oracle ACM product, we will focus mainly on the case life cycle design category and will touch upon the case information design category topics.

17.2.2 Use Case Description

You receive your monthly credit card statement, open it, and realize the balance due is not what you were expecting. An amount is charged that does not match with the price of the product you recently purchased. There are a couple of choices on how to report this issue—report it directly from the credit card company's website or just pick up the phone and call their customer care number. In either case they register your complaint and temporarily remove the charge from your credit card until they investigate why this happened. After a few weeks, you get a letter letting you know that the charge was indeed a wrong charge made by the merchant and that it has been corrected and that the issue has been resolved in your favor.



Figure 17.1 Case management design categories and components used in the ACM tutorial.

What you were involved in was a typical case management scenario that can be classified as an *incident management case* as described in Chapter 4. You as the *subject* of the case were the one that initiated the case. To resolve your case, a number of persons, participants that have a stake in resolving the case in the most effective manner (stakeholders), performed different actions (activities) to progress the case from its start until its closure. This will involve collaboration with departments within the company as well as with external entities, such as the merchant whose charge is under dispute. Let's take a behind-the-scenes look at how this case was processed. Since the objective of this tutorial is to give you a hands-on experience with Oracle ACM and not a lesson in credit card dispute handling we will keep the process simple.

Let us look at the steps that describe our example use case.

- 1. Based on the complaint, a case is opened by the customer service representative (CSR).
- 2. The CSR reviews the case and prepares it for processing by adding relevant information. She requests a signed letter from the customer declaring that the charge being disputed was not made by him and also asks for the copy of the invoice.
- 3. Once the letter is received from the customer, the CSR updates the case by adding the document to the case.
- 4. If the amount disputed is less than \$25 and the CSR concludes, based on the documents and information received from the customer, that this charge is in fact not valid, she has the authority to write-off the charge.
- 5. Up until now the case has been in its initial phase. With the receipt of the letter, it can now move to the next phase. The next, *analysis*, phase is where an analyst is assigned to review and investigate the disputed charge.
- 6. As part of his review the analyst reviews the transactions on the customer's credit card.
- 7. Once satisfied he has all the information, he requests the merchant to provide a copy of the sales draft for the charge.
- 8. The analyst receives the copy of the sales draft that he adds to the case.
- 9. Based on the sales draft and the documents provided by the customer, the analyst concludes that the charge indeed is wrong.
- 10. The analyst initiates a chargeback on the merchant for the extra amount charged and gives the merchant 30 days to contest the chargeback.

- 11. The merchant does not contest the chargeback.
- 12. The case is closed in favor of the customer and he is informed about the final outcome.
- 13. During the process, the customer may call to get an update on the dispute.
- 14. During the process fraud may be detected and a follow up fraud case is opened.

In Section 4.5, we presented the standard flow for the incident management classification. The characteristics of the *credit card dispute* align with an incident *management* as shown in Table 17.1.

In Figure 17.2 we added the roles *customer* and *shop owner* and simplified the model toward the current *credit card dispute* case.

Alignment	Incident Management Characteristics	Specific Alignment Details for Credit Card Dispute Case
1	Incident handling is aimed at solving the problem and finding the root cause	The credit card dispute process is aimed at solving a balance due misalignment and communicates with the customer and shop owner to identify who is to blame, the customer, the shop owner, or the credit card company
1	Initiation of an incident may be unpredictable	The customer discovers an error in his credit card statement and reports this at the credit card company. An incident like this can happen, but not on a regular basis
J	The high level execution of an incident is predictable, but certain parts of the execution are highly influenced by events	The process follows a standard flow containing assessment, investigation, and communication with the customer and the shop owner
		Events, like fraud, can happen during the case execution, but have a low probability
1	Both internal and external communication is key	External communication with the customer and the shop owner are key in the process
		Internal communication between the analysts can help finding the root cause quicker
✓	Data related to an incident is of both structured and unstructured in nature	The input for the data is structured. Receipts or other proof can be unstructured data
✓	Intelligence is needed to solve the case	Combining customer and shop owner information helps to solve the case. Monitoring intelligence provides information for possible fraud
✓	(Optionally) incident may be started by a customer	The incident is started by a customer
✓	(Optionally) regulatory requirements are often important	Regulatory requirements are not described in this case, but in a "real" credit card dispute it is important

 Table 17.1
 Alignment of the Credit Card Dispute with Incident Management









17.2.3 Credit Card Charge Dispute Life Cycle Design

In Chapter 3 we introduced the various aspects for the design of a case management solution. We will look here at the case life cycle design category for the credit card charge dispute as shown in Figure 17.3. The case life cycle design category describes the coordination of activities in reaching one of the end goals and consists of three design components: case flow and process design, rules design, and event design.

17.2.3.1 Case Flow and Process Design

A typical unstructured process does not follow any predefined path and therefore, using any type of flow diagram during analysis may not be very useful. In our use case, the unstructured part relates to the sequence (or lack thereof) of activities performed within a high-level structured sequence of milestones. A mind map can help to graphically capture these requirements.

For the purpose of this hands-on tutorial a mind map is used to capture the requirements of the use case.

Let us begin by answering some questions.

What activities are being performed?

The first things you need to identify are all the different activities that are being performed throughout the life of the case. Here the key activities (not in any particular order):

- Prepare case—Initialization activity for setting up the case for processing.
- Review charge dispute—Review the details of the dispute to ensure all required information is provided for further processing.
- Request documents from customer—Request customer to send one or more supporting documents necessary for resolving the dispute.
- Write-off charge—Initiated by the case owner to accept the dispute and write-off the charge without having to go through detailed investigation. Typically allowed only for disputed amounts less than \$25.
- Investigate charge—Initiate an investigation to resolve the dispute.
- Request document from merchant—Request the merchant to submit supporting documents required to resolve the case.

- Initiate chargeback—Initiate the process to charge the merchant for the disputed charge if the investigation shows that it was the merchant's fault.
- Notify customer—Send customer various notifications regarding the dispute case.
- Notify merchant—Send merchant various notifications regarding the dispute case.

These activities are shown in the figure below:



What are the important milestones to track?

It is important to identify milestones in the case life cycle that can be used to assess how the case is progressing. In our use case scenario, the following milestones seem to be important to track:

- Initial case review completed
- Analysis started
- Merchant interaction initiated
- Analysis completed

At this point we have identified what actions need to be performed, and what important milestones will help us track the case.



Adding the milestones to the mind map, we see the following in the figure below:

17.2.3.2 Rules Design

What policies govern the case?

We now have a good idea of the different activities that may be performed to progress the case from start to finish. We also know which milestones are important to us. Although case work tends to be lot less structured and a bit more ad hoc as compared to a traditional process, it is not completely without controls. There are rules and policies that control which activities are more appropriate given the state of the case at any point in time.

For instance, the activity to write-off a charge should only be available if the amount being disputed is less that a specific amount and if the charge falls under specific types of charge. Similarly, there are dependencies between activities. For example, "request documents from merchant" should be available only after the "merchant interaction" milestone has been reached.

320 Oracle Case Management Solutions

For our example, here are the set of rules that need to be applied to this case:

1. Activity "review charge dispute" is repeatable and is required to be done at least once and should be available only after the case initialization is complete as shown in the following figure:



2. Activity "request documents from customer" is optional and can be repeatable. It depends on the activity "review charge dispute" being started as depicted in the following figure:



3. Activity "write-off charge" is nonrepeatable and optional and should only be available after "review charge dispute" is completed and only when the disputed amount is less than \$25 as shown below:



4. Activity "investigate charge" shown below is optional and can only be done once. It should only be available after analysis has started.



5. Activity "initiate chargeback" shown below is optional and nonrepeatable and should only be available once analysis is complete and the result of the investigation indicates it was the merchant who was at fault.



6. Activity "notify customer" shown below is optional and repeatable and has no dependencies.



7. Activity "notify merchant" shown below is optional and repeatable and has the same dependencies as "initiate chargeback".



17.2.3.3 Event Design

What external events can affect the case progression?

Events drive how a case progresses. Events can be both internal events as well as external events. Internal events are those that are raised because of actions being performed on the case. For instance, uploading a supporting document is an event, which may result in an action being taken to review the uploaded document. Similarly, the completion of an activity raises an event that in turn could cause a milestone to be reached. We have already addressed which internal events are of interest to us and how they should be handled. We did this by analyzing dependencies between activities, milestones, etc.

External events on the other hand are the ones that occur outside the control of the case engine. In our use case, we have one external event that needs to be handled as part of the case processing; the event is that of a customer calling to get an update on the dispute. This would typically result in the case owner communicating the status of the case via an email.

322 Oracle Case Management Solutions

Another event that can occur outside of the case that will impact the case is when a possible fraud is detected. Fraud detection will typically happen in different systems altogether and this needs to be communicated to the case so that appropriate actions are taken. In our use case, if a possible fraud is detected, a new case should be started for the fraud and the current case should be suspended until a decision is made regarding the fraud. This is explained in the following figure:



17.2.4 Credit Card Charge Dispute Information Design

In this paragraph we will look at a high level of information requirements needed to create the Oracle ACM implementation for the credit card charge dispute. Case management information design category is shown in the following figure:



The case information design deals with all sorts of data and documentation flowing through the case and maps the relationships between all persons and/or assets involved in the case. The case information design category contains three design components, data design, document, and media design and people relationship design.

17.2.4.1 People Relationship Design

Who will be participating in the case?

In the simplified scenario, we have three main stakeholders—the CSR who handles the service call and creates the case, the analyst who works on the case to resolve it, and a case manager who

oversees dispute management cases. Each of these stakeholders may also have different levels of privileges. For instance the case manager will have full access to the case, whereas the CSR may only be granted view-only access to the case. There will be other people who will be indirectly participating for performing specific tasks assigned to them. But they typically will not have access to the case itself as explained in the following figure:



17.2.4.2 Data Design and Document Management Design

What information does the case require and how is it used?

Case processing is information-centric. Source of information can be structured data, typically provided by applications and databases, and unstructured data such as documents, images, videos, etc.

In our simplified use case, the following data items will be needed:

- Dispute—Provides details of the dispute registered, such as the customer ID, disputed transaction details, the date the dispute was registered, credit card number associated with the transaction, status of the dispute, resolution of the dispute, etc.
- Customer—Customer account details
- Credit card account—Credit card information including transaction history
- Investigation report—Details of the dispute investigation outcome

The data can be categorized based on its usage as follows:

- Reference only—Used only as reference by case workers and stakeholders
- Read-only case data—Used in the case as part of rules and policies, data expressions, input to activities, etc., and is not updated by the case engine
- Case data—Same as read-only case data but it gets updated by the case engine as the case progresses

Our use case data can be defined using the categorization shown in the figure below:



NOTE: Read-only case data can be marked as "external," which results in the case calling a user-defined handler to fetch the data from an external source as and when needed. The details of this are beyond the scope of this book.

The completed mind map, shown in the figure below, captures the analysis. We will use this in the next chapter as the input for our design.



17.3 Installation and Tutorial Files

In this section you will implement the credit card charge dispute use case we analyzed in the previous paragraph. Before beginning with this tutorial, download and install Oracle BPM 12.1.3. This software is available on http://www.oracle.com/technetwork/middleware/bpm/downloads/ index.html. This is a "quickstart" installer, which allows you to run business process management (BPM) and adaptive case management (ACM) from within the Oracle JDeveloper, without having to do a separate server installation. Follow the instructions from the installation guide available at http://docs.oracle.com/middleware/1213/core/SOAQS/index.html.

All tutorial material needed in this chapter, the data definitions in XSD and test XML files, are presented in Appendix 17.A. These files and a resulting complete set of tutorial files are also available on the following website: http://oraclecasemanagementsolutions.com/.

17.4 Key Elements of Oracle Adaptive Case Management

We have introduced Oracle ACM already in Chapter 7. This chapter goes one level deeper and describes the details of all ACM-related functionalities. Oracle ACM provides a comprehensive framework for building highly flexible and dynamic business process applications. It extends Oracle BPM Suite's scope to address unstructured ad hoc processes by introducing the notion of a *case* as shown in the following figure:



17.4.1 Case

A case is a container that holds the state of the case and includes the following:

- The overall state, such as whether the case is active, suspended, or closed
- The current state of the milestones
- Data needed for progressing the case
- Documents and other content
- Case Stakeholders and their access privileges
- Details about relationships with other cases

The case engine in Oracle BPM Suite is responsible for the life cycle of all cases. The state of the case changes when participants in the case perform activities relevant to the case. Internal and external events also can cause changes in the case state.

Designing and implementing a case, therefore, involves identifying these various elements of the case and defining the details for each. Before we start designing the case model for our "credit card charge dispute" use case, let us take a deeper look at these key elements.

17.4.2 Case Activities

Activities are actions performed on the case by either the user or the system with a goal to progress the case to its completion. These actions may be performed by the case participants or by the system itself.

17.4.2.1 Activity Implementation

Activities in Oracle ACM can be implemented using any of the following:

- Business process model and notation (BPMN)—Activities can be implemented as structured BPMN processes. These can use the full set of BPMN activities to model the activity implementation.
- Human tasks—Human tasks, one of the core components of Oracle BPM, can be used as an activity implementation and is suitable for activities that require work assignment and tracking. A BPM process is not required for this type of activity.
- Custom—If BPM and human task are not suitable implementation options, Oracle ACM allows adding a custom activity implementation. This implementation is a Java class that implements a specific Java interface. This allows you to provide any kind of activity. An example of such an activity implementation could be a Java class that creates a calendar invite using one of the calendar protocols such as CalDAV.

17.4.2.2 Activity Properties

Table 17.2 describes the various properties that define a case activity.

17.4.3 Case Data

Case data refers to data that is needed by the case for the following functionalities:

- Rules conditions used in the case life cycle determination
- Expression evaluation—To determine various case property values such as case title, case category, stakeholder members, and folder names for content storage

All instances of case data objects, by default, are stored as part of the case (internal case data) and managed by Oracle ACM.

17.4.3.1 Data Definition

Case data can be of simple types such as string, integer, and date. In most situations, the case data will be a complex type, with its own structure.

Property	Description
Name	Name of the activity
Manual	The user will initiate the activity from the case workspace. The case workspace is described in more detail in Section 17.6.4.1
Automatic	The case engine will initiate the activity whenever the activity is available for use
Required	When this property is set, the activity is required to be initiated and completed
	The result of this setting is that the case cannot be closed normally; it can only be force-closed
Repeatable	When this property is set, the activity can be initiated any number of times with an instance of a case. Without this property being set, an activity can be initiated only once during the case's lifetime
Conditional	If this property is set, the activity will have to be explicitly made available. Typically, this is done with the support of rules actions. If the activity is marked as unconditional, it is available as soon as the case is started. As a result, the case engine initiates automatic and unconditional activities as soon as the case starts
Input	Defines the data items that will be passed as input to the activity implementation. The input can either come from one or more of the <i>case data</i> items or can be captured from the user using an <i>activity form</i>
Output	Defines the data items returned by the activity implementation after successful completion. One or more of these data items can be configured to update the corresponding <i>case data</i>

Table 17.2 Activity Properti	ies
------------------------------	-----

Case data structure can be defined in two ways:

- Using the BPM business components catalog—you can use existing business objects or define new ones and use them as types for the case data
- Using XML Schema definitions

17.4.3.2 External Case Data

Some data, used in the case, originates from external sources, such as an ERP application. This data is not stored in the internal case data store, but is only referenced. This is in order to avoid the data getting stale or out of sync with the actual data. For example, a case data object may refer to a service request created in the CRM application. When the service request is updated via the CRM application, the corresponding case data object needs to be refreshed with the changes.

This can be a challenge to implement, since the CRM application will typically be unaware of the case.

Oracle ACM provides the ability to mark case data items as being external. Whenever the case needs the data, for example, to evaluate a rule, it will call a registered Java class that implements a specific interface. The implementation of this Java class is up to the case developer and will typically handle getting the data from appropriate sources. So for our service request example, you would declare the case data to be external. During case processing, every time the case needs any data from the service request, it will call the Java class, which in turn will fetch the data from the CRM application. This way, there is no danger of keeping two copies of data in sync.

17.4.4 Case Stakeholders

A case can have different groups of people working on it at various points in time during its lifetime. These groups of people are called stakeholders. Oracle ACM provides a flexible stakeholder model that allows you to define the different stakeholders at design time. It also allows changing of the model at runtime. This runtime change to the stakeholder model is limited to the particular instance of a case.

Stakeholder membership can be a combination of individuals, groups, application roles, and even BPM process roles. Stakeholder member definitions can be either named or dynamic. There are following three ways a member can be added dynamically to the stakeholder:

- Using an expression—you can specify an XPATH expression on a case data item that will evaluate to a valid member (individual and group role, or process role)
- Programmatically, at startup—you can add members to one or more stakeholders during startup by setting up the case parameters appropriately
- At runtime, using the case workspace

17.4.5 Case Documents

Content such as Word files, PDF files, images, and videos is a key part of any case. Case progression can be influenced by content operations such as upload, delete, and update.

Oracle ACM provides basic content operations that include file *upload*, *file check-out*, *file check-in*, and *deletion* of a file. It integrates with a content management system (CMS) for managing the document storage. By default, the documents are stored in the same database as the ACM persistent store. This mode is useful during initial development only. It is recommended to configure a CMS—either the preintegrated Oracle WebCenter Content or any CMIS (content management interoperability services) compliant CMS, such as Alfresco.

17.4.6 Rules and Policies

With Oracle ACM, it is possible to model a case without any rules and policies. In such cases, all activities are made available to all stakeholders at any time during the case. This may be fine where the nature of the work is truly ad hoc and where full flexibility is required. But in most use cases there is a need for some control and governance and the ability to guide the case workers through the case processing.
This is where rules and policies help by controlling which activities to make available at what point in time. Rules can also be based on the state of one or more milestones, changes in state of the case, changes to data, contents of the data, etc. Using a combination of conditions, one can put meaningful controls in the case without constraining the case workers' ability to use their knowledge and best judgment to process the case.

Oracle ACM uses the Oracle Business Rules engine as the rules engine and automatically creates a rules dictionary appropriately seeded with case functions, case metadata, and case data as fact types.

Most of the rules you define will make use of the different events that the case engine raises to determine the actions to take.

17.4.7 Case Events

The Oracle Adaptive Case engine at its core is driven by events. Events can be either internal or external. Internal events are generated as a result of changes to the case. The case engine generates the following types of events:

- Life cycle events
- Activity events
- Milestone events
- Document events
- Data events
- User events

Most of these events further define subtypes that indicate the specific event that has occurred. Table 17.3 lists all the events and their subtypes.

17.5 Implement the Case with Oracle ACM

Now that you have a good overview of the key case elements in Oracle ACM, let us identify these elements in our use case and use them to define our "credit card charge dispute" case model. We will use the following series of steps to design, deploy, and test a case management solution based upon Oracle BPM/ACM:

- Design the case management solution
 - Create case project
 - Configure case properties
 - Create case data model
 - Set up content configuration
 - Define user events
 - Define stakeholders and permissions
 - Define case activities
 - Define case life cycle decision-making rules
- Deploy
- Test

Table 17.3Case Events List

Event Type	Subtype	Event Raised When
Life cycle	STARTED	Case is created
	COMPLETED	Case is completed
	UPDATED	Case header is updated (typically via API)
	DUEDATE_REACHED	Case due date is reached
	SUSPENDED	Case is suspended, either manually or automatically via rules
	RESUMED	Case is resumed from its suspended state
	REOPENED	A closed case reopened
	ABORTED	Case is aborted
Activity	ACTIVATED	An activity is marked as being available for instantiation either by the user or via rules
	COMPLETED	An instantiated activity completes successfully
	FAULTED	An instantiated activity aborts due to an exception
Milestone	REACHED	Milestone is marked as having been reached, either by the user or via rules
	REVOKED	A milestone that was previously marked as having been reached is now set to be open again
	DEADLINE_REACHED	Due date associated with the milestone is reached
Document	ADDED	A document is added to the case
	DELETED	A document is deleted from the case
	MODIFIED	A checked-out document is checked-in
Data	No subtype	Case data is modified
User	No subtype	A user event is raised

17.5.1 Create the Case Management Project

Start JDeveloper 12.1.3 that you installed as per instructions at the beginning of this chapter:

- 1. Create a new application workspace by selecting *File* \rightarrow *New* \rightarrow *Application*.
- 2. In the New Gallery window shown below, select BPM application

Q			
<u>C</u> ategories:	Items:	Show All Descriptions	
Applications Connections Deployment Descriptors	🔁 Java Desktop Application	-	
	ADF Fusion Web Application Application from EAR File		
- Deployment Profiles - Diagrams			
—Java	Application Template		
	BPM Application Creates a BPM application. Th BPM project This project has	e application consists of one also SOA technology	
BPM Tier	🔁 Custom Application		
Activity Guide Business Components	Database Application		
-Case Management	🔁 Extension Application		
Business Tier	🔁 Java EE Web Application		
-ADF Business Components 	OEP Application		
-Contexts and Dependency	🔁 Service Bus Application		

3. Name the application "CCDApplication" (CCD being short for credit card dispute) and choose the appropriate directory for the application. Use *acmbook.tutorial.ccd* as the *application package prefix*. This is shown in the figure below.

Name your applica	tion			5	
	Application Name:				
Application Name	CCDApplication				
Project Name	Directory:				
 Project SOA Settings 	/home/prasen/WorkProje	cts/jdeveloper/acm-bo	ok-solution	Browse	
	Application Package Prefi	x			
	acmbook tutorial.ccd				

In step 2 of the *Create BPM Application* wizard, enter project name as *CCDCase* and click on *Next*. Make sure you select "BPM" from the "Project Features" list.

4. In step 3 of the wizard, keep the default composite name and select *Composite with Case Management* and click *Finish* as shown below.

Application Name	Composite Name:
<u>Project Name</u>	CCDCase Start from: Standard Composite SOA Template
	Empty Composite Composite With BPEL Process Composite With Human Task Composite With Subprocess Composite With Spring Composite With Mediator Composite With Mediator Composite With Gase Management Composite With Gase Management Composite With Business Rule

5. Accept the default in the Create Case Management window shown below and Click OK.

🔞 Create	Case Manageme	ent	
Create Case	Management		
Name: Namespace:	CCDCase http://xmlns.oracle	e.com/CCDCase	
Help		ок	Cancel

This will create a new case project and open the case management definition page shown below for editing.

🖉 CCDCase 👘						
General Data & Documents	Case Mana Specify the	gement - CCDCas case information be	se low and use t	the tabs on the left.		
Stakeholders & Permissions Translation	Title:	Plain Text 🔻				
	Summary:	Plain Text 🔻				0
	Priority.	3 (normal) *				
	Category:	Plain Text 💌				
	🗆 Due Time					
	Duration: By Va	lue 🔹 0 Year Calendar By Value 👻	s 0 Months 0	Days 0:0:0	2 2	Ø
	Milestones				+/)	604
	Name	Can be l	Revoked	Duration Type	Duration	
	🗉 Outcomes					►/×
	Name					

17.5.2 Define the Case Properties

The case management definition page has multiple tabs, each for defining different properties and elements of the case. Let us start with the *General* tab.

17.5.2.1 Title

This defines the title that will be displayed in the case UI for a given case instance. The title can be a simple string value or a translation. Translations can be parameterized, which means not only can you create a translation table for multiple languages but you can also create dynamic text that is evaluated at runtime. This allows you to define meaningful titles such as "Credit Charge Dispute Case for John Doe," where the "John Doe" part is inserted at runtime based on an XPATH expression.

For now, select *Plaint Text* and type in "Credit Card Charge Dispute Case." Once you have defined the case data we will update this to use a translation.

17.5.2.2 Summary

This defines a brief summary of the case. This can be plain text or can be a parameterized translation just like the title, described previously.

For this tutorial, leave the summary blank.

17.5.2.3 Priority

Defines the case priority ranging from 1 (highest) to 5 (lowest) and is useful in querying cases. It can also be useful in defining rules based on the case priority.

Set the priority to 3.

17.5.2.4 Category

This can be used to categorize the case. For example, you could categorize the dispute cases based on the card holder's status such as gold, platinum, etc. The value can be either a literal string or can be a parameterized translation.

For now, leave this field empty.

17.5.2.5 Due Time

You can define when a case becomes due by defining the duration. The duration can either be a fixed value or can be derived at runtime from case data using an XPATH expression. The expression should result in a valid XML duration format. The due time is calculated based on the case start time. If *Use Business Calendar* is checked and an *Organization Unit* is chosen, then the time computation will consider appropriate business holidays defined for that organization unit, to arrive at the due time.

Set the duration to 7 days and keep the *Use Business Calendar* unchecked and *Organization Unit* empty.

17.5.2.6 Milestones

Milestones help to track the progression of the case. We identified the following milestones in our analysis in the previous chapter:

- Initial case review completed
- Analysis started
- Merchant interaction initiated
- Analysis completed

Add an entry for each milestone by clicking on the \oplus icon and name them as per Table 17.4. Each of the columns is explained here:

- *Name*: This is the internal name for the milestone. This name is used in rules for managing the state of the milestone. If you need to see the milestones in a particular order in the Case UI, you need to ensure they are appropriately named. The case UI sorts milestones alphabetically. For our use case, prefix each name with A01, A02, etc., to ensure the milestones are displayed in the desired order.
- Display Name: This is displayed in the case UI.
- *Can be Revoked*: If checked, the user can reopen a milestone that has been completed.
- Duration: If set, this marks the milestone as being overdue when the specified duration is complete (Table 17.4).

17.5.2.7 Outcomes

Outcomes are predefined set of possible outcomes of the case. These are used while closing the case indicating how the case was closed.

Internal Name	Display Name	Can Be Revoked	Duration
A01_InitialCase ReviewCompleted	Initial case review completed	No	3 days
A02_AnalysisStarted	Analysis started	No	No duration keep the default values 0 years, 0 months, 0 days
A03_Merchant InteractionStarted	Merchant interaction started	No	No duration
A04_Analysis Completed	Analysis completed	Yes	5 Days

Table 17.4 Milestones Definition

Table 17.5Outcomes Definition

Outcome	Display Name
CaseResolvedNoDispute	Case Resolved-charge is valid
CaseResolvedMerchantAtFault	Case Resolved-merchant at fault
CaseResolvedOurFault	Case Resolved—our fault

Add the following outcomes (Table 17.5):

Once you have completed the outcome definitions, you are done with general case properties and your *General* tab should be like the figure below:

Plain Text •	run caro charge Dispute Case		
Summary: Plain Text 💌			
Priority: 3 (normal) *			
Category: Plain Text 💌			
🗟 Due Time			
Duration: By Value O Year Use Business Calcodae	i 0 Months 7 Days 0:0:0		
Duration: By Value O Year Use Business Calendar Drganizational Unit: By Value Hilestones	0 Months 7 Days 0:0:0		↓ /X ∧
Duration: By Value Use Business Calendar Drganizational Unit: By Value Milestones Name	0 Months 7 Days 0:0:0	Duration Type	ouration
Duration: By Value Que Business Calendar Use Business Calendar Drganizational Unit: By Value Ritestones Name Initial Case Review Completed	0 Months 7 Days 0:0.0	Duration Type By Value	Duration O Years 0 Months 3 Days 0.0.0
Duration: By Value Q Year Use Business Calendar Drganizational Unit: By Value Hilestones Name Initial Case Review Completed Analysis Started	10 Months 7 Days 0:0 0	Duration Type By Value By Value	Duration O Years 0 Months 3 Days 0.0.0 O Years 0 Months 0 Days 0.0.0
Duration: By Value Que Business Calendar Use Business Calendar Drganizational Unit: By Value Name Name Initia Case Review Completed Analysis Started Marchant Interaction Started	Can be Revoked	Duration Type By Value By Value By Value	Duration O Years 0 Months 3 Days 0:0:0 O Years 0 Months 0 Days 0:0:0 O Years 0 Months 0 Days 0:0:0
Duration: By Value Q Year Use Business Calendar Drganizational Unit: By Value Kitestones Nane Initial Case Review Completed Analysis Started Merchant Interaction Started Analysic Completed	Can be Revoked	Duration Type By Value By Value By Value By Value By Value	Duration O Years 0 Months 3 Days 0.00 0 Years 0 Months 0 Days 0.00 0 Years 0 Months 0 Days 0.00 0 Years 0 Months 0 Days 0.00
Duration: By Value Duration: By Value Use Business Calendar Drganizational Unit: By Value	s 0 Months 7 Days 0:0:0		
Duration: By Value Vear Use Business Calendar Drganizational Unit: By Value V Name Initial Case Review Completed Analysis Started Marchant Interaction Started Analysis Completed	Can be Revoked	Duration Type By Value By Value By Value By Value	Duration O Years 0 Months 3 Days 0.0.0 O Years 0 Months 0 Days 0.0.0 O Years 0 Months 0 Days 0.0.0 O Years 0 Months 5 Days 0.0.0

17.5.3 Define the Data Model

17.5.3.1 Identify the Case Data Objects

During the analysis of the use case, we identified the different data elements that will be needed during the processing of the case as shown in the following figure:



You can use this analysis to further determine data elements that need to be defined in the case. Let us start with the obvious ones—*dispute* and *investigation report* are definitely required to be defined in the case as they will be updated during the case processing.

You also have *customer* and *credit card account*, which are primarily reference data that will be displayed on the user interface to give the case worker the appropriate contextual information about the case. *Credit card account* data need not be defined as case data since it is not referred directly in the case. On the other hand, the *customer* data may be required as case data since attributes from it may be used in the case. For example, you can define the case title to include the name of the customer in the title. This means that you will need to provide an XPATH expression that evaluates to the name of the customer. Another example would be the need to categorize the case based on the status of the customer. Both of these attributes will typically be available in the *customer* data.

This narrows down the list of case data objects to include in the case definition

- Dispute
- Investigation report
- Customer

The following sections describe the definition of these data objects with Oracle ACM.

17.5.3.2 Define the Case Data Objects

To create the case data definitions, switch to the *Data & Documents* tab from the *General* tab that you have been working on until now. Clicking on the **4** icon will open the window for adding a new case data definition.

Table 17.6 describes this pop-up window:

🛞 Create P	Parameter	
Name:	Parameter	
Display Name:	Parameter	
Type:	999 int	•
	999 decimal	
	99E double	
	🖄 dateTime	
<u>H</u> elp	♦ duration	el
-	base64Binary	
	TCaseAction	
	Case Case	~
	Case	-

Field	Description	
Name	Name of the case data. This name is used in your expressions and as fact names in Oracle Business Rules	
Display Name	This will be used to label the data in the user interface	
Туре	 Defines the type of the case data. The type can be any of the basic types such as <i>string, int,</i> and <i>date.</i> Complex types can also be used. When using complex types, there are three options 1. Use existing business object from the business components in the project 2. Create a new business object based on an existing XML Schema 3. Create a new business object from scratch In the case of options 2 and 3, the object definition ends up being stored in the business components library 	

 Table 17.6
 Description of New Case Data Creation Pop-Up

17.5.3.2.1 Dispute Data Definition

Create the dispute data object using option 3 (from scratch) described earlier. Use the following values (Table 17.7):

- 1. Click on the \clubsuit icon, that is alongside the *Type* field, to open the *Create Business Object* window. Name the object as *DisputeBO*.
- 2. Click on the *Search* icon 🔍 to open the *Browse Modules* window. This will show all the different modules available in the business catalog.
- 3. Click on the New icon Sto create a new module named CCDBusinessObjects.

The data object creation process so far looks like presented in the following figure:

		at a dec 1 di di il	ter	×			Editable Is External
	Name:	DisputeDO					
3	Display Name:	Dispute Details					
Flex	Type:	0	Create	Business Obj	ect	×	+ / X 3
Nan		Business Object	DisputeBO	Ø Search:	Browse Modules	×	Ipdatable
Docume	Help	Destination Module:		Search Results:			
ecify the e parent	e document lo it folder must	Help		B-	Creat	e Module	
Parent F	Folder:	By Value By Value By Value			cebusinessoupers		

- 4. Select this new module and click on OK.
- 5. Click on *OK* to close the window and click *OK* again to complete.
- 6. At this point you have defined a new case data called *DisputeDO* but have not yet defined the structure of this object. You do that in the following steps.

Field	Value
Name	DisputeDO
Display Name	Dispute details
Editable	Yes
Is External	No (unchecked)

 Table 17.7
 Values for Dispute Data Object Creation

7. Expand the *Business Components* folder in the *Projects* panel shown below. Within it you should see the *CCDBusinessObjects* module you just created. Expand this folder to see the *DisputeBO* object. Click on the *Refresh* icon if you do not see the business object.



Double-click on the *DisputeBO* object to open it. Oracle BPM allows you to define complex data models and provides features such as the ability to define abstract data definitions and use object inheritance. It also allows you to define methods in the object using Groovy scripts; this is outside the scope of this tutorial. For this tutorial you will define a concrete object without inheritance and without any methods. Click on the *Add* icon \clubsuit to add new attributes as defined in the figure below. The arrows $\Uparrow \ p$ help to arrange the fields in the correct order. Press the *Save* button after finishing the work.

Description:			Edit Documentation
Abstract: Parent: Attributes			ት ତି ት X
Name	Туре	Not Null	Value
disputeld	abc string		4
creationDate	🖄 dateTime		
customerid	abc string		
disputeAmount	(1999) decimal		
transactionReferenceNumber	999 long		
merchantid	abc string		
status	abc string		
transactionDate	🖄 dateTime		
	and the second se	-0-0	

17.5.3.2.2 Investigation Report Data Definition

Create the investigation report data object just as you created the dispute data object. Name the case data object *InvestigationReportDO* and provide the display name as *Investigation Report*.

For the type of this object, create a new type names *InvestigationReportBO* under the *CCDBusinessObject* module. The following figure shows the structure of this data object.

Description:			Edit Documentation
Abstract: Parent:	None> ▼		4 L L 1
Name	Туре	Not Null	Value
disputeld	abc string		
outcome	abc string		
	abc string		
analystid			
analystid startDate	🖄 dateTime		
analystid startDate endDate	🖄 dateTime 🖄 dateTime		

Press the Save button 🗊 after finishing the work.

17.5.3.2.3 Customer Data Definition

Customer data is a good candidate for being defined as an external data. Any changes to the customer details in the database that is the source of truth for customer information should be available to the case. If the data is not defined to be external then the copy of the customer data in the Case will quickly become stale depending on the duration of the case and the frequency of changes to the customer information.

Implementing external data handler is an advanced topic and beyond the scope of this book, so, for this tutorial, you define the data as being internal; same as the previous two objects. For this object, instead of creating the data definition from scratch, you use a predefined schema (customer.xsd). See Section 17.3 for the locations from where this schema can be obtained.

- 1. Add a new case data with the name *CustomerDO* and display name as *Customer Details*.
- 2. When defining the type, click on the *Add* icon **+**.
- 3. Name the business object as CustomerBO in the Create Business Object window.
- 4. Select the destination module to be CCDBusinessObjects.
- 5. Select based on external schema. This will enable the Search icon for loading the XSD.
- 6. Click on the *Search* icon to open the *Type Chooser* window.
- 7. Click on the *Import Schema File* icon 🔀 to open the file browser window. Browse to the location where you downloaded the customer.xsd file and select it.

8. In the *Localize Files* pop-up window shown below, keep the default options and click *OK*. You should see the customer.xsd under the *Project Schema Files* tree in the type explorer.

🕲 🗉 Type Chooser		
		56
Type Explorer Project Schema Files CCDCaseRules_CCDCaseRulesMessag CCDCaseRules_CCDCaseRulesTypes.x DecisionServicePrimitiveTypes.xsd Customer.xsd Customer TCustomer DisputeBO.xsd InvestigationReportBO.xsd	eTypes.xsd sd	
Type: {http://www.mycompany.com/ns/CCDCase	}Customer	
Show Detailed Node Information		
Help	ОК	Cancel

9. Select the first node, *Customer*, under the customer.xsd as shown in the figure above. Click *OK* on the open windows to complete the case data creation. Make sure you leave the *Is External* checkbox unchecked.

Press the *Save* button **I** after finishing the work.

You now have the case data defined and it can be used in expressions and rules in the following parts of the tutorial.

17.5.3.3 Create Data Forms

The case data objects you defined can be viewed and updated at runtime from the BPM Workspace. This is possible by creating data forms. Data forms are autogenerated ADF applications that are embedded in the BPM Workspace. Oracle Application Development Framework (ADF) is the facto user interface standard within the Oracle Fusion Middleware and is based upon the J2EE Java Server Faces standard. The following steps walk you through creating a single data form application that will have a page (and an ADF task flow) for each case data object:

- 1. Select the Data & Documents tab in the case definition page.
- 2. Select Dispute Details data object and click on the create form icon and select NewProject. Name the project CCDCaseDataForms and click OK to complete. This will create a new project that contains an ADF task flow with a single ADF page called DisputeDO.jspx. You can modify the layout and add other UI controls, validations, etc., to your specific requirements. For the purpose of this tutorial, the default form layout is used. Save all projects and close the DisputeDO.jspx, DisputeDO.xml, and DataControls.dcx files.
- 3. Navigate back to the *CCDCase* definition, select the *Investigation Report* object and click on the *Create Form* icon. This time around, select *Select Existing Project*. In the pop-up that opens, select the *CCDCaseDataForms* project and click *OK* to complete.
- 4. Modify the generated *InvestigationReportDO.jspx* ADF page to remove the *Save* and *Reset* buttons by selecting the button and deleting it. This will disable updating the data from the workspace. This way you can enforce that the investigation report is only updated via an activity. There are a number of more elegant options for doing the same, such as making all fields read-only and conditionally disabling the *Save* and *Restore* buttons based on logged in user. But for this exercise, let us keep it simple.
- 5. Repeat steps 3 and 4 for the *Customer Details* data object. This form should also disallow updates.

17.5.3.4 Define Flex Fields

The case engine stores all the case data objects as XML in the case database along with other case instance metadata. This makes it difficult for queries that require attributes from these data objects as part of the query criteria. For example, if you wanted to find all *CLOSED* cases for **CustomerDO.customerId** equal to *110220* and **InvestigationReportDO.outcome** equal to *VALID_CHARGE*, you would have to first fetch all cases with status equal to *CLOSED* and then for each case, inspect the XML for the corresponding data object to find the required cases. This is obviously not a very efficient way of querying cases. Oracle ACM provides a way to map XML elements in the Case data to user-defined fields, called **Flex Fields**.

Flex fields are a mechanism by which you can expose selected data from the case data objects as named fields in the case. These can then be used in the query criteria when fetching cases. The case engine provides support for defining 20 such fields for each string, number, and date types with a maximum of 60 such fields across these types. The values for these fields are provided via XPATH expressions using the appropriate case data objects.

Although querying cases using the Case API is beyond the scope of this book, we will add a couple of Flex Fields to see how to define them. In the following steps you add two flex fields—one for the customer ID and another for the investigation report outcome.

- 1. Click the add + icon for *Flex Fields* in the *Data & Documents* tab.
- 2. In the *Create Flex Field* window, enter the name of the flex field as *CustomerId*, choose the type as *String* and leave the *Column Name* to the default selected.
- 3. For the source file, click on the *Express Builder* icon a to open the expression builder.

4. Expand the *CustomerDO* node in the *Variables* panel, select *CustomerId* and click on *Insert into Expression* to copy it into the expression panel as shown in the figure below.

Expression:	50 1	50
\$CustomerDO/ns3:Customer/ns3:CustomerId		
	A Insert Into Expression	
Variables	Functions	
	String Functions Compare-ignore-case Compare-ignore-case Concat	
Description:		
An XPath expression of the variable		

Click on *OK* to close the window and click *OK* again to complete.

Note: The namespace prefix (ns3) may be different in your project.

5. Repeat the steps for defining a flex field named *InvestigationOutome*. Define the type as *String* and for the expression, select the *outcome* attribute from the *InvestigationReportDO* variable.

Press the *Save* button **I** after finishing the work.

This completes your data definition. At this stage your *Data & Documents* tab should look something like this as shown in the figure below:

General Data & Documents	B Data 🔶 🖉 🖇						
User Events	Add data which will be used to process the case.						
Stakeholders & Permissions	Name Type Editab						
Translation	Dispute Details	🐻 Disp	R DisputeBO				
	Investigation Report	linve:	stigationReportBO	>			
	Customer Details	Cust					
	Elex Fields			+/×3			
	Manage custom fields						
	Name	Туре	Source	Updatable			
	CustomerId	te String	String SCustomerDO/ns3 Customer/ns3 Customerid				
	InvestigationOutcome	atd String	\$InvestigationReportDO/ns2:InvestigationReportBO/ns2:outcon	ne 🗌			
	Documents						
	Specify the document loc The parent folder must b	ation. The c e exist alrea	ase instance folder will be a direct sub folder of the parent fold ady.	er.			
	Parent Folder:	By Value	•				
	Case Instance Folder:	By Value	•				
	Create Case Instan	ce Folder					
	Document Proper	ties		4			

17.5.4 Content Configuration

Oracle ACM provides out-of-the-box integration with Oracle WebCenter Content, a Content Management System (CMS) for managing content acquired, generated, or used as reference by a case. For development purposes, a built-in database-based content storage is provided with the Oracle BPM/ACM product. This content storage does not have CMS features such as check-in and check-out and is only meant to be used as a development configuration. In this tutorial, we will use this integrated content server in JDeveloper. If you do have an instance of Webcenter Content that you can use, you can configure the BPM server to use that. Please refer to https://docs.oracle.com/middleware/1213/bpm/bpm-develop/case_mgmt_bpmpd.htm#CIHJBCAB for steps to configure CMS integration for ACM.

In case you plan to use a CMS such as Oracle WebCenter Content, follow these steps to set up your case.

- 1. In the *Data & Documents* tab, enter a name for the parent folder under which all the content for this case type will be stored. Use the name *CCD Case Content*.
- 2. Select *Create Case Instance Folder*. This will store content to the named instance-specific folder. If this is not checked, content for all instances will be stored under the parent folder.
- 3. For the *Case Instance Folder*, select *By Expression* and click on the *Expression Builder* icon to define the expression that will evaluate to a value that will be used to create the instance-specific folder.
- 4. Use the value of *disputeId* as the name of the instance folder as shown in the figure below.

insert fragments from the fragment editors below the Exp	pression field.
Expression:	0 m m
\$DisputeDO/nsl:DisputeBO/nsl:disputeId	
ns Ins	ert Into Expression
Variables	Functions
	String Functions
Description	
An XPath expression of the variable	

Note: If you use WebCenter Content as the CMS, you need to enable the *Framework Folders* feature in it. The parent folder should be relative to the root folder.

Name	Display Name
ProbableFraudDetected	Probable fraud detected
CustomerRequestsUpdate	Update requested by customer

Table 17.8 User Events

17.5.5 Define User Events

For this use case, there are two predefined user events that the user can raise via the BPM Workspace:

- A probable fraud is detected while investigating the case
- Customer requests an update on his case

You define these two events in the *User Events* tab. Add the two events given in Table 17.8 by clicking on the *Add* icon.

17.5.6 Define the Stakeholders and Permissions

We have identified following three main stakeholders in this use case:

- Case manager—person (or persons) responsible for the overall progress of the case
 - Has access to all aspects of the case
- Customer service representatives
 - Can only view the case and cannot take any actions
- Dispute analysts—persons responsible for investigating disputes
 - Can view the case and can perform activities relevant to solving the dispute

Before you define the stakeholders, let us do a quick primer on how stakeholders and permissions are used in Oracle ACM.

17.5.6.1 Application Roles

Oracle ACM leverages the same roles and permission infrastructure as the rest of the Oracle Fusion Middleware stack and that is handled by a component called Oracle Platform Security Services (OPSS). OPSS manages application-specific and global role definitions and corresponding permissions in a policy store. These roles are called application roles and are used by all components such as J2EE applications, ADF applications, WebCenter portals, SOA Suite, BPM Suite, and, of course, ACM. Application roles allow creating user groups specific to the needs of an application without having to make changes to the configured lightweight directory access protocol (LDAP) directory such as MS Active Directory. These application roles can be comprised of individual users, groups defined in the directory, as well as other application roles.

17.5.6.2 Application Policies

Application policies define authorization policies that control access to resources in an application. Authorization policies define the resource to be controlled, the action(s) to be authorized, and the Java class that implements the permission implementation. Policies are typically granted to application roles. These can also be granted to individual users or groups defined in the directory such as LDAP.

17.5.6.3 Stakeholders

Stakeholders in a case are created as application roles and are associated with a special J2EE application called *OracleBPMProcessRolesApp*. These roles are created in the same way the process roles corresponding to the swim lanes in a BPMN process are created. Stakeholders can have any combination of user, groups, application roles, and process roles as members. An important difference between stakeholder roles and other application roles such as process roles is that the membership for stakeholders is defined in the case database and not the OPSS policy store. This is to allow runtime and instance-specific changes to the stakeholder model. If you do define members to a stakeholder via OPSS, the case engine will not obey it. Stakeholder roles are created as application roles primarily to allow access control using permissions.

17.5.6.4 Permission Labels

Oracle ACM provides access control to various aspects of the case using the notion of a permission label. Permission labels are strings and denote the type of permission such as "Highly Confidential" and "Public—Read-Only." These can be any arbitrary labels. These labels can then be attached to various elements in the case. In version 12.1.3 (the version used in this book), only activities can have a label attached at design time. For all other elements, the case API can be used to attach labels at runtime.

For each label, three application roles are created—one for each action (READ, UDPATE, and INVOKE). For example, if you have label called "CONFIDENTIAL," then the three roles will be CONFIDENTIAL.READ.Role, CONFIDENTIAL.UPDATE.Role, and CONFIDENTIAL. INVOKE.Role. To associate permission labels with stakeholders, you add the appropriate role, group, or user to appropriate permission label role.

The following sections describe the steps to define the stakeholders and permission for our use case.

17.5.6.5 Define Roles

Before you create the stakeholders, application roles need to be created that will define the memberships. These roles can then be used to define stakeholder members, for assigning tasks, and for defining swim lanes in BPMN processes. 1. From your *Applications* tab in JDeveloper, open the *Organization* node under *BPM* as defined below.



- 2. Click on the Add icon to open the dialog for the Role name.
- 3. Enter the value for *Name* of the role as *Analysts*.
- 4. Repeat steps 2 and 3 for adding the CSR role.

Note: As a best practice, role memberships should not be done at design time. You will do that once we deploy the application.

17.5.6.6 Define Stakeholders

- 1. Open the CCDCase case definition and select the Stakeholders & Permissions tab.
- 2. Add the following stakeholders by clicking on the Add icon for Stakeholders (Table 17.9):
- 3. Click on the Add icon next to Dispute Analysts stakeholder and select Add Process Role.

Name	Display Name
Stakeholder_DisputeAnalysts	Dispute analysts
Stakeholder_CSR	Customer service representatives
Stakeholder_CaseManager	Case manager

Table 17.9 Stakeholder Definition

4. You now see a new member with type *Process Role* added in the table as shown below. Click in the *Value* column and then on the icon at the right in the column to select the *Analysts* process role you defined earlier.

🖃 Dispute Analysts				/×
Member Type	Valu	іе Туре	Value	
Process Role	By V	alue		
Customer Service I Member Type	Represen Va	Selec ProcessOwner ProcessReview CaseOwner	t Role ver	्र स्व
Permissions		CaseManager Stakeholder_C Stakeholder_D StakeHolder_C	aseManager isputeAnalysts SR	+/>

- 5. Repeat step 4 for customer service representatives. This time select the CSR role.
- 6. For the case manager role select *Add User* rather than *Add Process Role* like you did in the previous steps. In the *Value* column enter *weblogic*.

This completes the stakeholder definition.

Note: You can assign a case manager dynamically at runtime when you start the case. The WebLogic user is only used as the default case manager. This is explained in Section 17.6 in the later part of this chapter.

17.5.6.7 Define Permissions

All Oracle ACM projects, when created, are seeded with two default permission labels:

- Public
- Restricted

Name	Display Name		
PublicRO	Public read-only		
FullAccess	Full access		
InvestigatorsAccess	Restricted access-investigators		

Table 17.10 Permission Definition

As discussed earlier, these labels can be arbitrarily named and can be anything you want them to be. You define your own after removing the default labels.

- 1. Select the *Public* permission and delete it. Repeat that for *Restricted*.
- 2. Click on the Add icon to add the following new permissions (Table 17.10):

This completes your design time definition of the permissions. You define the stakeholder permissions using the Oracle Enterprise Manager Fusion Middleware Control, after you deploy the case application project.

17.5.7 Define Case Activities

The analysis in the previous chapter identified a number of activities that may be performed on the case. Along with the activities, their properties were also identified. This will be used to define the activities in the case. Refer to the analysis chapter for each activity that was identified. Table 17.11 summarizes the activity properties and will help you in creating the activity definitions.

Table 17.11 contains a property and type, which was not discussed in the analysis. This indicates the implementation of the activity. This can be either a *BPMN* process or a *human task*. The decision on whether to use BPMN instead of human task in case of work assignment activities depends on the following:

- Need for pre- and/or postprocessing in regards to the task.
- Need for handling human task outcomes. In the 12.1.3 version, the outcome from taskbased activities cannot be saved to case data. You need to wrap that in a BPMN process so that the outcome can be processed in BPMN.
- Need to sequence tasks—task leads to another task or tasks in a predetermined sequence (structured flow).

The column *Conditional* is set to *Yes* if the analysis determined that this activity depends on one or more events.

The following sections describe the definition and implementation of the activities listed in Table 17.11. In order to keep the tutorial simple but still show all the important aspects of ACM, you will be creating very simple implementations of these activities. We will provide in the following paragraphs an example for a BPMN and a human task activity type.

Activity Name	Туре	Auto/Manual	Reqd.?	Conditional?	Repeat	Input Types	Output Types	Save Output?
PrepareCase	BPMN	Auto	Yes	No	No	DisputeBO	DisputeBO, Investigation ReportBO	Yes
ReviewCharge Dispute	HT	Manual	Yes	Yes	Yes	DisputeBO	DisputeBO	Yes
RequestDocuments FromCustomer	BPMN	Manual	No	Yes	Yes	DisputeBO, CustomerBO	None	No
WriteOffCharge	BPMN	Manual	No	Yes	No	DisputeBO	DisputeBO	Yes
InvestigateCharge	HT	Manual	Yes	Yes	No	DisputeBO, Investigation ReportBO	Investigation ReportBO	Yes
RequestDocuments FromMerchant	BPMN	Manual	No	Yes	Yes	DisputeBO	None	No
InitiateChargeBack	BPMN	Manual	No	Yes	No	DisputeBO	None	No
NotifyCustomer	BPMN	Manual	No	No	Yes	DisputeBO CustomerBO	None	No
Notify Merchant	BPMN	Manual	No	Yes	Yes	DisputeBO	None	No

 Table 17.11
 Case Activities and Their Properties

17.5.7.1 Create a BPMN-Based Activity

Start with the *PrepareCase* activity implementation. This activity is automatic and unconditional, which means it will be activated as soon as the case starts. And since it is unconditional, it will be initiated as soon as it is activated. Activities like these are useful for doing any initial processing before the case workers start working on the case. It is also a good way to update or initialize one or more case data objects. In your simplified implementation, you do the following:

- Update the status of the dispute to "ASSIGNED."
- Initialize the investigation report to be returned to the case.

To create a BPMN-based case activity, you need to first create a BPMN process and then create an activity from this process. The following section describes the steps using *PrepareCase* as an example.

Create PrepareCase BPMN process:

1. In the *Projects* browser, right-click on *BPM*→*BPMN Processes* and select *New*→*BPMN 2.0 Process...* as shown in the following figure:



2. Name the process PrepareCaseProcess and accept the other defaults and click on Next.

Case activities implemented as BPMN processes can be either synchronous or asynchronous. If a process is modeled as an asynchronous process, make sure the process has an end message event (that is, it is not a one-way asynchronous process); otherwise the activity will be treated by the case as "active" perpetually.

3. The process you are creating should take dispute details as an input and return the updated dispute details back to the case.

At the *Arguments* step in the *BPMN 2.0 Process Wizard* window, click on the *Add* icon to add an input argument. Enter the argument name as *disputeDetails* and select *Browse...* in the *Type* dropdown list.

4. Select *DisputeBO* from the list shown below.

BPMN 2.0 Proc	ess Wi	zard	
Arguments			
Definition Arguments Initial Implementatio Advanced	Input Argume Name:	Output ents Defini Edit Ai disputeD	tion 🖉 / 🗙
	Type: Help	G Brow	se

- 5. Switch to the *Output* tab and repeat the above steps. Name the argument *updatedDisput-eDetails*. Add another argument and name it *investigationReport* and for the type, chose *InvestigationReportBO* from the type *List of Values*.
- 6. Click on *Finish* to create the BPMN process. You should see a simple process with a start and end message event.

In the following steps you add a process data object to hold the dispute details received as input and then update status attribute in the object.

- 7. In the *PrepareCaseProcess* structure panel, right-click on the *Process Data Object* and select *New*.
- 8. In the *Create Data Object* pop-up, enter the name as *disputeDetailsDO* and select *DisputBO* as the type (select *Browse* to show the type *Browser*). Keep other defaults and click *OK* to complete.
- 9. Repeat step 8 for *investigationReportDO* of type *InvestigationReportBO*.
- 10. Right-click on the *Start* node and select *Properties* to open the properties window. Switch to the *Implementation* tab.
- Click on Data Associations and drag disputeDetails from under Arguments to disputeDetailsDO under Data Objects. Click OK to complete.

12. Drag and drop the *Script* task from the component ribbon above the process model to the process between the *Start* and the *End* nodes as shown below.



- 13. In the properties window for the Script, enter the name as *Update Data*. Switch to the *Implementation* tab and click on *Data Association*. In the *Data Association* window, expand the *disputeDetailsDO*. Right-click on the attribute named *status* under *disputeDetailsDO* and select *Expression*, to open the *Expression Builder*. Type in "ASSIGNED" (including the enclosing quotes) in the *Expression*: field. Click *OK*.
- 14. Drag *PrepareCaseProcess→Data Objects→disputeDetailsDO→disputeId* from the left-hand side to *PrepareCaseProcess→Data Objects→investigationReportDO→disputeId* on the right-hand side as shown below.





15. Click OK to complete. Your process should look like this:

- 16. The last step copies the data to the output so that it can be returned to case. Rightclick on the *End* node and select *Properties*. Switch to *Implementation* tab and click on *Data Associations*. In the *Data Associations* window, drag *disputeDetailsDO* from under *Data Objects* to *udpatedDisputeDetails* under *Arguments*. Similarly drag *investigationReportDO* to *investigationReport*.
- 17. Click OK to complete and click on Save All to save the project.

Create PrepareCase Activity:

Now that the BPMN process for PrepareCase is complete, you can define the activity for it.

- 1. Right-click on the *PrepareCaseProcess* under *CCDCase→BPM→BPMN Processes* in the *Projects* browser panel. In the pop-up menu, select *Promote as Case Activity*. Enter the name as *PrepareCase* and display name as *Prepare Case*. This will open the PrepareCase activity page.
- 2. Refer to Table 17.11 and complete the definition.
- 3. Make sure you check *Save Output as case data* for both output parameters. Your completed activity definition should look like this:

General Advanced	Case Activity - Prepa Specify the Case Activi	ty information below.			
	O Manually Activated () Auto	omatically Activated 📓			
	Required				
	🔲 Repeatable				
	Conditionally available				
	Permission:		Q,		
	🗆 Case Activity Input				60
	Name	Source	Save user input in case	Case Data Name	
	disputeDetails	Case Data	(N/A)	Dispute Details	
	🗏 Case Activity Output				69
	Name		Save output as case data	Case Data Name	
	updatedDisputeDetails		4	Dispute Details	
	investigationPenort		9	Investigation Report	

This completes the definition for the PrepareCase activity. Similarly, using Table 17.11, create activities that are of type BPMN.

17.5.7.2 Create a Human Task-Based Case Activity

We have identified two activities that are based on human task. The following steps describe, using the InvestigateCharge as an example, how to create a human task–based activity. A human task is comprised of two parts: a task definition and a task form. You start with the task definition. *Create InvestigateChargeTask human task*:

1. Make sure *CCDCase* in the *Projects* browser is selected and press CTRL+N to open the *New Gallery* window. In the search box, type in "Human." This will filter down the list to two items. Select the *Human Task (Service Components)* item as shown below.

<u>C</u> ategories:	_	<u>I</u> tems:		Show	All Description
Faults	^	ADF Task Flow	w Based o	on Humar	n Task (JSF)
Service Components Tests Transformations/Tra Web Tier HTML JSF JSF/Facelets JSP Servlets All Items	ņ	Human Task Launches the which you de option, you m within a SOA Navigator.	k (Servic Human 1 fine a hur nust selec project in	e Compo Fask Crea nan task. It a SOA J the Appl	onents) ation wizard, in To enable this project or a file ication

- 2. This will open the *Create Human Task* pop-up window. Enter the name of the task as *InvestigateChargeTask*. Keep other defaults and click *OK* to complete. This will open the *InvestigateChargeTask* task definition.
- 3. Select the *Assignment* tab and drag the *SingleParticipant* component from the *WorkflowEditor* palette into *Stage1* box as shown below.



- 4. Double-click on the Stage1. Participant1 to open the Edit Participant Type window.
- 5. Click on the *Add* button and select *Add Application Role*. In the value column type in *CCDCase.Analysts* as shown below.

General Advanced:	Type: Single	•	Label: Stagel.Participant1	
	Build a list of participa Let participants mai Auto assign task to	nts using: Name nually claim the ta a single User	sk Assignment Pattern :	east Busy 💌 🥖
	Specify attributes us Participant Names	ing: () <u>V</u> alue-ba	ed () Rule-based	+· X
	Identification Type	Data Type	Value	15 8934
	Application Role	By Name	CCDCase.Analysts	

- 6. Click *OK* to complete and *Save All* to save.
- 7. In the following steps you define the data elements that will be used by the task. Refer to Table 17.11 for the input parameters for the *InvestigateCharge* activity.
- 8. Select the *Data* tab and click the *Add* icon and select *Add other parameter* from the list to open the *Add Task Parameter* window. Make sure *Variable* is selected and select *Element*. Click on the search corresponding to the element field to open the *Type Chooser* window.
- 9. Select *Project Schema Files* →*DisputeBO.xsd*→*DisputeBO* and click *OK*. Back in the *Add Task Parameter* window, select *Editable via Worklist* and click *OK* to complete.
- 10. Repeat steps 8 and 9 and this time select *Project Schema Files* \rightarrow *InvestigationReportBO. xsd* \rightarrow *InvestigationReportBO.*
- 11. Select the General tab to define the task properties. For Task Title, enter Investigate Charge Dispute # and then select the Expression Editor icon a to open the Expression Builder window. In the expression builder, select task:task→task:payload→ns1:DisputeBO→ns1:disp uteId and click on Insert Into Expression to copy the expression. Click OK to complete.

Note: The name space *ns1* may be different in your project.

- 12. Enter Investigate Credit Card Charge for Description.
- 13. Click on the *Search* icon for *Outcomes* and unselect *APPROVE* and *REJECT* and select *OK* from the list of outcomes. Click on *OK* to close the *Outcomes Dialog* window.
- 14. In the *Application Context* enter *OracleBPMProcessRolesApp*. This defines the application stripe to use for application roles. This is the default application stripe for all BPM roles. Make sure the name is correctly entered; otherwise your task will not get assigned correctly.
- 15. Your task definition is complete at this point.

356 Oracle Case Management Solutions

Create Task Form for InvestigateCharge Task:

Most tasks require some form to be presented to the assignee to be completed as part of the task. You can create these forms in a number of ways—autogenerate, use task-form wizard, build from scratch or use web forms in the browser-based BPM composer. For the purpose of this tutorial you will use the autogenerate option.

In the *InvestigateChargeTask.task* page, click on *Form* and select *Auto-Generate Task Form* to open the *Create Project* window. Enter the name of the project as *InvestigateChargeTaskForm* and click *OK* to complete. This will create a new ADF project that will contain the autogenerated task form and associated artifacts. Wait for the taskDetails1.jspx page to be created and click *Save All* to save all the projects. Typically you would rearrange the fields in the page to be more presentable. For now, you keep the default layout.

17.5.7.2.1 Create Investigate Charge Activity

- 1. In the *Projects* browser select *CCDCase→BPM→Human Tasks→InvestigateChargeTask.task*, then right-click and *Promote as Case Activity*. Enter the *Name* as *InvestigateCharge* and *Display Name* as *Investigate Charge*.
- 2. In the InvestigateCharge activity definition page, select *Required*. Change the source of disputeBO and investigationReportBO case activity input to case data.
- 3. Click on the *Search* icon for the *Permission* and select InvestigatorAccess permission. This will restrict this activity to only those users that have InvestigatorAccess permission granted. You will be setting this up later in the exercise using the Fusion Middleware Control.
- 4. Select *Save output as case data* for both the case activity output parameters. Your completed definition should look like this:

Advanced	Specify the Case Activity Information below.				
	Manually Activated Automatically Activated Manually Manual				
	₩ Required				
	E Repeatable				
	Conditionally available				
	Permission: InvestigatorsAccess	Q.			
	🛛 Case Activity Input		2 ⁰		8
	Name	Source	Save user input in case	Case Data Name	
	dispute80	Case Data	(N/A)	Dispute Details	
		Case Data	(M/A)	Investigation Report	
	investigationReportBO	Case Data	intervi		
	investigationReportBO	Case Data	iner.		69
	InvestigationReportBO	Case Data	Save output as case data	Case Data Name	8
	InvestigationReportBO	Case Data	Save output as case data	Case Data Name Dispute Details	B

This completes the definition for the *InvestigateCharge* activity. Similarly, create *ReviewChargeDispute* Human Task activity by referring to the Table 7.11 and following the same steps as for *InvestigateCharge* task. For task participant use the Application Role *CCDCase.CSR*.

17.5.8 Define the Rules

Your case definition at this point in the exercise is essentially complete. The case can be deployed now and used. But you will only see activities that you have defined as being unconditional. All conditional activities will require rules to be defined that will enable (activate) these activities so that they can be executed (either manually or automatically). If you would like to see the case in action, without the conditional activities, you can follow the instructions in Section 17.6.

In the following sections you define the rules for activating activities as well as changing the milestone status based on the analysis done in the previous chapter. One of the ways to be productive with defining rules is to tabulate all the different dependencies in a spreadsheet. You start with the mind map and look at all the "Depends on" nodes. Create separate tables for each type of event. For example, in the following figure, *Review Charge Dispute* activity depends on an activity to be completed.



As shown below, in this case it is an activity event, whereas it is a milestone event for *the Investigate Charge* activity.



By using this method, you can create tables such as these. Table 17.12 captures all actions to be taken when specific activity events occur.

Start with defining rules for the activity events and then move to milestone and user events.

Activity Event	Activity Name	Additional Condition	Action	Name
Completed	Prepare case	_	Activate	Review charge dispute
Completed	Review charge dispute	Dispute amount ≤ 25	Activate	Write-off charge
Completed	Review charge dispute	_	Reach milestone	Initial review completed (A01)
Activated	Review charge dispute	_	Activate	Request document from customer
Activated	Investigate charge	—	Reach milestone	Analysis started
Activated	Initiate chargeback	_	Reach milestone	Merchant interaction started
Completed	Investigate charge	_	Reach milestone	Analysis completed

Table 17.12 Activity Event Actions

17.5.8.1 Activity Event Rules

The Table 17.12 is best defined using a decision table in Oracle Business Rules. Following steps guide you through defining this decision table.

1. Before you begin, you need to make sure the rules dictionary is updated with the names of activities and milestones.

There is an issue with the 12.1.3 release where the dictionary is updated only when changes to case data are made. So to force a dictionary update, add a dummy case data, save the project, delete the dummy case data, and then save the project again as shown below.

🕷 CCDCase				
nts	🖻 Data		8	/ X @
	Add data which will be used	to process the case.		
ermissions	Name	Туре	Editable	Is External
	Dispute Details	DisputeBO	v	
	Investigation Report	InvestigationReportBO	v	
	Customer Details	Customer80	¥	
	DummyCaseData	als string		

- 2. From the *Projects* browser select *CCDCase→BPM→Business Rules→CCDCaseRules.rules* and double-click to open it in the Oracle Business Rules editor.
- 3. Click on the *CCDCaseRules* under *Rule Sets* and click on the *Add* icon in the *Decision Tables* column.
- 4. Click on *Decision Table 1*, change the name to *Activity Event Rules*, and press *ENTER*. The following describes the rule for the first row in Table 17.12.
- 5. Click on *Add* icon and select *Condition*. This will add a row for the new condition. Doubleclick in the *CI* cell under *Conditions* column heading to open a drop-down with a list of *Value Options*. In the text field, enter *CaseActivityEvent*. This will filter the list to show the *CaseActivityEvent* fact. Select *CaseActivityEvent*→*activityName*.
- 6. Click on the "?" in the R1 column, select COMPLETED, and click OK.
- 7. Click on the *Add* icon and select *Condition*. Double-click on *C2* and select *CaseActivityEvent*→*activityName*.
- 8. Select Activities from the Value Set drop-down list as shown below.

🔯 Overview 🛛 🧼 General Rule	s 🛛 🖉 Verbal Rules	Actvivity Event Rules ×
💐 60 Activity Event Rule	5	
C2 R1: ?		▼ 🗉 Local Value Set 💌
<u>Conditions</u> C1 CaseActivityEvent.activityEvent	R1 t	Create Value Set
C2 CaseActivityEvent.activityNam	e	Milestones
		 Outcomes Data
Y Confilm Production		UserDefinedEvents Activities

•	<u>C</u> onditions	R1	
C1 CaseA	ctivityEvent.activityEvent		COMPLETED
C2 CaseA	ctivityEvent.activityName		PrepareCase
×	Conflict <u>R</u> esolution	PrepareClase Specify Value as Don't Care otherwise NotifyCustomer InitiateChargeBack ReviewChargeDispute Ø PrepareCase NotifyMerchant InvestigateCharge WriteOffCharge	
•	Actions	RequestDocumentsFromMerchant	
	<insert action=""></insert>	RequestDocumentsFromCustomer	
			OK Cancel

Click on the "?" under R1 and select PrepareCase as shown below.

- 9. Click on the *Add* icon and select *Action→call*. This will add a new row under the *Actions* column on the bottom half of the table.
- 10. Double-click on *A1* under *Actions* and select *activateActivity*. In the *Arguments* table, click in the *Value* column select *Activity.ReviewChargeDispute* from the drop-down list as shown below. Go to the bottom-end of the drop-down list to locate the *Activities* object.

Activ	ities	ReviewChargeDispute	I.
⊨ }…∎	Red	questDocumentsFromMerchant	
÷	Red	questDocumentsFromCustomer	
÷	Mil	estones	
÷	Ou	tcomes	
÷a	Us	erDefinedEvents	
ėe	Act	tivities	
		NotifyCustomer	
-	a	InitiateChargeBack	
		ReviewChargeDispute	
-		PrepareCase	
		NotifyMerchant	
		InvestigateCharge	
		WriteOffCharge	
	a	RequestDocumentsFromMerchant	
1		RequestDocumentsFromCustomer	

Select the Parameterized option.

- 11. Unselect Always Selected.
- 12. Back in the decision table, select *A1-R1* cell.

13. Your first rule, "activate *ReviewChargeDispute* when *PrepareCase* completes", shown below, is now defined.

Settings	😨 Overview 🔷 General	Rules 🖉 Verbal Rules 🔯 Activ	ty Event Rules X		
Facts	ə 🔍 🏍 Activity Event R	iles		4·X A = 0 < 3 H·H·B	3 2 2 3 • 8
CO GIODAIS	• <u>C</u> onditions	R1		-	
E Value Sets	C1 CaseActivityEvent.activit.		COMPLETED		
a Links	C2 CaseActivityEvent.activit		PrepareCase		
Decision Functions					
@ Translations					
Test.					
Data Explorer					
a Business Phrases					
Rule Sets 💠 💥					
@ CCDCaseBulas					
W CEDCaseRules	× Conflict Resolution				
	- Actions				
	A1 call activateActivity		2 5		
			Averables Content have	- Discuss	

14. Now define the rule for the second row in the *Activity Event* table, which is "Activate WriteOff Charge when ReviewChargeDispute is completed and if dispute amount is < = 25."

Click on the Add icon and select Rule.

- 15. It is a good practice to use a value set for ranges such as "more than 10 but less than 25." Moreover, when using *Decision Tables*, value sets are required for selecting values in the rules column. Select *Value Sets* in the sidebar of the rules editor. Click on the *Add* icon and select *Range Value Set*. This will create a new value set named "Value Set 1." Select this value set and edit it by clicking on the *Edit* icon to open the *Edit Range Value Set* window.
- 16. Change the *Name* to *WriteOffRange*. Click on the *Add* icon to insert a row in the *Range Values* table. This will add a row with endpoint as 0; change the value from 0 to *1*. Add another row and change the value of the first endpoint to *26*. You now have the range defined:

ata Type: Int				
Include Disallowed Values in Tes				
	ts			
escription:				
riteOffRangeWriteOffRange				
ange Values:				4 3
Endpoint Included Endpoint Allo	wed in Action	s Range	Alias	Description
EB 26	4	>=26	>=26	
EB 1 🗹	1	[126)	(126)	
🖼 -Infinity 🕑	1	<1	<1	
Bi 1 V Bi -Infinity V	2	[126) <1	(126) <1	

- 17. Select the *CCDCaseRules* under *Rule Sets* and you should see the decision table you have been working on. Since we already have conditions defined for *activityEvent* and *activity-Name*, you do not need to add new conditions. All you need to do is add a new rule that checks for the activity name to be *ReviewChargeDispute*. To do that, click on the *Add* button at the top and select *Rule*. This will add a new column *R2*. Select the cell *C1-R2* and it should open a drop-down list with activity event types. Select *COMPLETED*. In the *C2-R2* cell, select *ReviewChargeDispute*.
- 18. Now you need to define the action to be taken for this new rule. Since you already defined an action for activating an activity (*A1*), you change the value for this rule without having to add a new action. Click the cell *A1-R2* and select *Activities→WriteOffCharge*. Make sure you select the A1-R2 row so that it has a check marked for the cell.
- 19. This rule is still missing one condition. Writing off a charge is only allowed if the disputed amount is less than or equal to 25. So you need to add a new condition for this. Click on the *Add* icon and select *Condition*. In the new *C3* row, double-click to open the drop-down and select *DisputBOType*→*disputeAmount*. From the drop-down list for value sets, select the *WriteOffRange*. Click in the *C3-R1* cell and select *Specify Value as Don't Care*, since this particular condition is not relevant for the *R1* rule. In the *C3-R2* cell, select the *[1..26]* range (which is greater than or equal to 1 and less than or equal to 25). This completes the rule definition for the second dependency.

Repeat this for the remaining rows in the *Activity Events* table. Once you have completed all the rules for the activity events, your decision table should look as shown in the figure below.

Londitions	R1	R2	83	84	85	86
1 CaseActivityEvent activityEvent	COMPLETED	COMPLETED	ACTIVATED	ACTIVATED	COMPLETED	ACTIVATED
2 CaseActivityEvent.activityName	PrepareCase	ReviewChargeDispute	ReviewChargeDispute	InitiateChargeback	InvestigateCharge	InvestigateCharge
3 DisputeBOType.disputeAmount	-	(126)		-	-	
Actions						
Actigns 1 call activateActivity	V	e	V		0	
Actions call activateActivity activityName:String	✓ Activities.ReviewChargeDisp	C Activities. WriteOffCharge	V Activities. RequestDocument	Activities. Review Charge Disp	Activities.ReviewChargeDisp.	Activities. ReviewChargeDis
Actigns 1 cell activateActivity activityName:String 2 cell reachMilestone	♥ Activities.ReviewChargeDisp	♥ Activities.WriteOffCharge	♥ Activities.RequestDocument.	C Activities. Review Charge Disp	Activities. ReviewChargeDisp	. Activities.ReviewChargeDi:
Actigns Coll activateActivity activityName:String Coll reachWilestone milestoneName:Song	☑ Activities.ReviewChargeDisp I Milestones.A02_AnalysisSta	♥ Activities.WriteOffCharge ♥ Milestones.A01_InitialCaseR.	♥ Activities.RequestDocument. ♥ Milestones.A01_InitialCaseR.	 Activities.ReviewChargeDisp [√] Milestones.A03_Merchantint	Activities. ReviewChargeDisp Milestones. A04_AnalysisCo	Activities. ReviewChargeDia V Milestones. A02_AnalysisS

17.5.8.2 Milestone Event Rules

Following the same instructions as those for activity event rules, create a new decision named *Milestone Event Rules*. Use Table 17.13 to define your rules.

Here are some hints to help you with the creating these rules.

- 1. Use the fact named *CaseMilestoneEvent*. The relevant attributes in this fact are *milestoneEvent* for the event raised (*REACHED*, *REVOKED*, etc.) and *milestone*, which contains the name of the milestone.
- 2. For milestone names, make sure you select the Milestone value set.
- 3. Create a value set called *InvestigationReportOutcomes* that has the following values: *MerchantFault*, *OurFault*, and *CustomerFault*.

Table 17.13Milestone Events

Milestone Event	Milestone	Additional Condition	Action	Name
Reached	Initial review completed	_	Activate	Investigate charge
Reached	Merchant interaction started	_	Activate	Request documents from merchant
Reached	Analysis complete	Result = Merchant fault	Activate	Initiate chargeback
Reached	Analysis complete	Result = Merchant fault	Activate	Notify merchant

- 4. For the additional condition (result = merchant fault), use the fact *InvestigationReportBOType*. *outcome*. Make sure you select the *InvestigationReportOutcomes* as the value set for this condition.
- 5. You will need to add more than one action for activating activities.

Once you complete your rules definition for the milestone events, your decision tables should look like this:

T RT HEACHED	TMilestoneEvent	X	(
 <u>C</u>onditions 	RL	R2	R3
C1 CaseMilestoneEvent.milesto	PEACHED	REACHED	REACHED
2 CaseMilestoneEvent.milestone	A01_InitialCaseReviewCompleted	A03_MerchantinteractionStarted	A04_AnalysisCompleted
3 InvestigationReportBOType.o		(+)	"MerchantFault"
▼ Artions			
1 call activateActivity	9		(V)
activityName:String	Activities.InvestigateCharge	Activities.RequestDocumentFromMerchant	Activities. InitiateChargeback
2 call activateActivity			v

17.5.8.3 User Event Rules

We identified two user events that the user can raise while processing the case. The actions that need to be taken are defined in Table 17.14.

Define the rules for these two events by creating another decision table. This time, name it *User Event Rules*. Use the *fact UserDefinedEvent.eventName* in the conditions and make sure you select *UserDefinedEvents* as the value set for the rules.

Table 17.14 User Event Actions

Event Name	Conditions	Actions
Update requested by customer	—	Notify customer with updates
Probable fraud detected	_	Suspend case Notify customer Initiate fraud case

Note: For the purpose of this use case, you need to only define the rule for "Notify Customer."

Once you are done with defining the rules, you are ready for deploying and testing your first casebased application.

17.5.9 Resulting Application

The architecture of the Oracle SOA Suite is based upon the service composite architecture (SCA). Every SOA, BPM, and ACM application is combined together in SOA composites, based upon SCA. These composites are composed together with the different Oracle SOA-based technologies, such as BPMN activities, human tasks, rules, BPEL, events, adapters, and (of course) ACM. The resulting application is packaged and installed on the Oracle SOA Suite on top of the WebLogic Server.

The resulting composite for the CCDCase application is shown in the figure below.



Every Oracle ACM application is exposed via web services containing a standard set of operations that can be used to control the case execution, such as startCase, closeCase, and attainMilestone, as shown in the figure below.

CCDCase.service		CCDCase
Operations:		
startCase attainMilestone revokeMilestone abortCase suspendCase resumeCase closeCase reopenCase		

In the next chapter we will use this service to start the CCDCase and work on a credit card dispute case.

17.6 Deploy and Test

This section describes the steps for deploying the application and then using it.

17.6.1 Prepare Server for Deployment and Test

Before you can deploy your application, you need to start the integrated WebLogic Server.

17.6.1.1 Start the Integrated WebLogic Server

Navigate to $Run \rightarrow Start$ Server instance from the JDeveloper menu. If this is the first time you are starting the server, you will be prompted to provide the administrator name, password, and the port number for the server. Keep the default port number (7101) and enter the user name as *weblogic* and password as *welcome1*. The server should start in about 5–8 minutes (the first time it will take a lot longer as it will create a new domain first and then start the server). Once it starts up, you should see the message SOA Platform is running and accepting requests in the Running: IntegratedWeblogicServer – Log panel as shown below.

Running: IntegratedWebLogicServer - Log 🔹 🗉
Q. Actions *
<pre>cum 7, 2014 5144138 PM FDD > dettices <genvers <="" dea_modals="" pre=""> Comment ucrauitscure 1s now listening on 12/10.111/102 for protocols 110ps, tss; (usps; nttps; > cum 7, 2014 514438 PM FDD > dettices <genvers <="" dea_modals="" pre=""> Comment Ucrauitscure 1s now listening on 22019;7180:938:2424:44ff;1416174418:7101 for protocols 10ps, 13s; (laps, st cum 7, 2014 514438 PM FDD > dettices <genvers <="" dea_modals="" pre=""> Comment Ucrauitscure 1s now listening on 12019;7180:938:2462:44ff;14161741817201 for protocols 10ps, 13s; (laps, st cum 7, 2014 514438 PM FDD > dettices <genvers <="" dea_modals="" pre=""> Comment Ucrauitscure [1]* is now listening on 100:00:00:01:7100:17201 for protocols 10ps, 13s; (laps, nttps; > cum 7, 2014 514438 PM FDD > dettices <genvers <="" dea_modals="" pre=""> Comment Ucrauitscure [3]* is now listening on 100:00:00:01:7100:17202 for protocols 10ps, 13s; (laps, nttps; > cum 7, 2014 514438 PM FDD > dettices <genvers <="" dea_modals="" pre=""> Comment Ucrauitscure[3]* is now listening on 100:00:01:01:01:01:01:7102 for protocols 10ps, 13s; (laps, nttps; > cum 7, 2014 514438 PM FDD > dettices <genvers <="" dea_modals="" pre=""> Comment Ucrauitscure[3]* is now listening on 1600:01:01:01:01:01:01:01:01 for protocols 10ps, 13s; (laps, nttps; > cum 7, 2014 514438 PM FDD > dettices <genvers <="" dea_modals="" pre=""> Comment Ucrauitscure [3]* is now listening on 1680:01:01:01:01:01:01:01:01 for protocols 10ps, 13s; (laps, nttps; > cum 7, 2014 514438 PM FDD > dettices </genvers></genvers></genvers></genvers></genvers></genvers></genvers></genvers></pre> defaultscure Cervers dEA_MODALS Comment Ucrauitscure Cervers
Messages Extensions - 🚯 Running: IntegratedWebLogicServer - 📦 Business Rules - 💌
Note: If you do not see the panel, press <Control+Shift+L>

17.6.1.2 Enable Development Mode

The default audit-level setting for the SOA infrastructure server is production, which reduces the amount of information logged as part of the instance audit trail. This is optimal for most production deployments but is not very useful during development. Change the audit level to development so that you can get detail audit trail for the case instances you create.

- 1. Using your web browser, login to http://localhost:7101/em as *weblogic* user with *welcome1* as the password
- 2. In the *Target Navigation* panel on the left, expand the *SOA* folder and select *soa-infra* (*DefaultServer*). On the *soa-infra* page, click on the *SOA Insfrastructure* drop-down list and select *SOA Administration→Common Properties* as shown in the figure below.



3. In the SOA infrastructure common properties page, select *Development* from the *Audit Level* drop-down list and click the *Apply* button on the top right of the page.

17.6.1.3 Create Test Users

When the integrated server is installed, three users are created, one of them being WebLogic—the administrator. To test your case application, you need a few users that will be added to the different process roles you defined while modeling the case. You create the users at this time and add them to the appropriate process roles after the case application is deployed.

- 1. In the *Target Navigation* panel, select *WebLogic Domain*→*DefaultDomain*.
- 2. In the *DefaultDomain* page that is displayed, select *Security→Users and Groups* from the *WebLogic Domain* drop-down list.
- 3. Make sure you have the *Users* tab selected and click on the *Create* button to open the *Create A New User* form. Use the following values to create the users (Table 17.15):

Name	Description	Provider	Password
jcooper	James Cooper	DefaultAuthenticator	welcome1
jstein	John Steinbeck	DefaultAuthenticator	welcome1
wshake	William Shakespeare	DefaultAuthenticator	welcome1

Table 17.15 Test Users

17.6.2 Deploy Case Application

Now that the server is running and configured, you can deploy the case application. There are following four different deployments that you need to do, though the order in which they are done is not important:

- CCD case—case composite project
- CCD case data forms—the J2EE application that contains the data forms
- Review charge dispute task form—the J2EE application that contains the form for the review charge dispute task
- Investigate charge task form—the J2EE application that contains the form for the investigate charge task

17.6.2.1 Deploy the Case Composite

To deploy the CCDCase composite application, follow these steps:

- 1. In the *Projects* browser in JDeveloper, select the *CCDCase* project and right-click. In the pop-up menu, select *Deploy*→*CCDCase*
- 2. Select Deploy to Application Server and click on Next.
- 3. Set the *New Revision ID* to 1.0 and select *Overwrite any existing composites with the same revision ID* and uncheck *Keep running instances after redeployment*. Click on *Next* to continue.
- 4. Click Next in the Task flow Deployment step without selecting anything.
- 5. Select Integrated WebLogicServer and click Next and then Finish on the next step.
- 6. As shown below this will initiate the deployment process. Watch the *Deployment* tab in the log window to see the deployment process. Once the deployment is complete you should see the *Deployment finished* message.

In second second second second	
Deployment - Log	
Q	
(06:57:53 AN) Deployment 31 (06:57:53 AN) Function 15 (06:57:53 AN) (06:57:53 AN) Running dependency (06:57:53 AN) Deploying profile (06:57:53 AN) Deploying profile (06:57:53 AN) Deploying profile (06:57:53 AN) Deploying scale (06:58:10 AN) Properties arrive ar	<pre>inted Gweblogic 12.x), analysis - for thus SDA Project 'CCDCass.jpr' to '1.0' e to <u>AbsoArchicLarisievelDestZebumerk/CCDCassLeterner</u> [http://b.0.2.15:1201] use_retJ.0.jar to partition 'defmult' on server DefmultServer [http://b.0.2.15:1201] mome_oracle/jdevelDepr/wymork/CCDAppl:cation/CCDCass/deploy/scm_CCDCass_rev1.0.jar fTP request for deployeent CTP request for deployeent Extinct to nest10.0.2.15; port1701 Bjoyment descriptor mome from the server, response code=200 ed archive scc.CCCDCass_rev1.0.jar with 0 warning/severe messages to partition "defmult" on server DefmultServer [http://10.0.2.15:17101] pipoyent 44 seconds pipoyent 44 seconds</pre>

17.6.2.2 Deploy Case Data Forms Application

The case data form application is a J2EE application and should be deployed from the *Application* menu.

- 1. Select Application→Deploy→CCDCaseDataForms...
- 2. Select *Deploy to Application Server* and click on *Next*.

3. Select *IntegratedWebLogicServer* and click on *Next* and then click on *Finish* to begin deployment. If you see the *Oracle Deployment Configuration* pop-up show up, click on *Deploy* to continue as shown below.

Deployment - Lo	99
Q	
[06:59:38 AM] [06:59:38 AM] [06:59:38 AM] [06:59:38 AM] [06:59:38 AM] [06:59:45 AM] [07:59:46 AM] [07:00:02 AM] [07:00:02 AM] [07:00:20 AM] [07:00:20 AM] [07:00:20 AM] [07:00:20 AM] [07:00:20 AM]	Deployment started Target platform is (Weblogic 12.x). Retrieving existing application information Running dependency analysis Building Deploying 2 profiles Wrote Web Application Nodule to <u>/home/oracle/ideveloper/mywork/CCDApplication/CCDCaseDataForms.wmr</u> Wrote Web Application Nodule to <u>/home/oracle/ideveloper/mywork/CCDApplication/deploy/CCDCaseDataForms.wmr</u> Wrote Web Application Deploying 0 data source(s) to the server Deploying Application [Deployer:149J92]Operation 'deploy' on application "CCDCaseDataForms' is in progress on "DefaultServer". [Deployer:149J92]Operation 'deploy' on application "CCDCaseDataForms" has succeeded on "DefaultServer". Application Deployed Successfully. The following URL context root(s) were defined and can be used as a starting point to test your application: /case/CCDCaseDataForms/CC

17.6.2.3 Deploy Task Form Applications

The last part of the deployment is the deploying of the task forms you generated. The process for deploying these is the same as that for the case data forms since these forms are also standard J2EE applications based on ADF.

Follow the instruction in the previous section to deploy the following:

- Review charge dispute task form
- Investigate charge task form

17.6.3 Create a Case Instance

You now have everything deployed and are now ready to test the application.

As described in Section 17.5.9, every Oracle ACM application is controlled via a standard set of web service operations. A new case is created by invoking the startCase operation on the case service. Any web service client, such as SOAPUI (http://www.soapui.org/), can be used to initiate a case. Your WebLogic Server also provides a simple web service test client that can be used to invoke web services. You use this client for your case.

The startCase operation takes one parameter—startCaseInputMessage, which in turn contains the full case object. All elements in the case object are optional. You can call startCase with an empty case object by just passing in<ns:case/> and a case will still be created. For your first test, you use an XML document defined in Appendix 17.A. See Section 17.3 for the locations where to obtain this schema.

There are two important parts of the case in this test data:

- The stakeholder member—this is one of the many ways to add one or more stakeholders and members at runtime. The test XML is adding *jcooper* as the case manager. The name Stakeholder _ CaseManager corresponds to the name that you gave the case manager stakeholder in the case model.
- Case data—this is how you can create instance of one or more case object on case startup. In this test, you see that there are two data objects. The first is the dispute details and the second is the corresponding customer details.

This is not the only way to create a case. Typically you would either have a mediator, a BPMN, or a BPEL component in the composite wired to the case component. This component would capture basic required information for creating a case and in turn will build the case object using an XSLT and then call the startCase method in the case service.

The following steps describe how to create a new case instance:

1. Using your web browser open the URL http://localhost:7101/soa-infra. You should see a page that looks something like the figure below:



- 2. Select the link *TestCCDCase.service*. This will open the *Web Services Test Client* page and you should see the different operations provided by the case service such as startCase, attainMilestone, and revokeMilestone.
- 3. Click on the *Test* button for the startCase operation to navigate to the test page for this operation. The default view is a form-based view. Since you are going to use a precreated XML input, click on the *Raw Message* button to switch to XML mode.
- 4. Clear the XML in the text box and copy and paste the XML from the test XML document, presented in Section 17.A.2 and click on *Invoke* to call the service.
- 5. Once the call returns successfully, you should see the response from the case that will contain the case ID assigned to the new case as shown below.



17.6.4 Working on the Case

Once a case is created, case participants use the BPM Workspace application to work on their cases. The BPM Workspace out of the box available with Oracle BPM, is a browser-based application that provides all the functions required to participate in business processes—either structured on unstructured. It provides the following key functions:

- Case workspace
- Comprehensive worklist
- Process tracking
- Process dashboard
- Administration

For your use case, you will primarily be using the case workspace. Before you start working on the case, let us take a quick tour of the case workspace. You can find more detailed information on using the BPM Workspace in the Oracle BPM 12.1.3 documentation (https://docs.oracle.com/middleware/1213/bpm/bpm-user/bpmug_ws_gt_strt.htm#BPMUG115).

17.6.4.1 Case Workspace

Oracle ACM provides a web-based user interface, the case workspace, which provides an overview on the complete state of the case and enables executing activities on the case. The figure below shows an example of the case workspace. The areas highlighted and numbered are described in the following:

		Case 4	Process Tracking #10 Deshboards - More	
≡ · [4] Credit Car	d Charge D	Ispute Case ③	Add Comment Raise Event	创 · · · · Actions ·
Activities All • 2	Data Documents	Audit Trail		Milestones 7
Available 🔤 🔹		4	Customer Details Dispute Details Investigation Report	A Initial Case Review Completed Open -
Send Email	customer\$d	1001		in 2 de
Create Task	firstName	John		Analysis Started Open +
Notify Customer	middlename			
Review Charge	lastname	Doe		Merchant Interaction Started Open -
Dispute	socialSecurityNo	999999999		
Completed	nationality	USA		Analysis Completed Open -
Contraction Contraction	maritalStatus	MARRIED		next week on Frida
Completed 16 hours ago	gender	м		L
	status	GOLD		8
3 -	addressLine1	111 No Name St		Stakeholders 🜵 🖉
Cinted Cases	addressLine2			Customer Constant Research Mark
No cases are linked to this case	city	San Francisco		Disconcers CSP
	state	CA		ap consecon
	postaiCode	99999		Case Manager
	country	USA		de jcooper
	homePhone	4154150000		
	workPhone	6506506500		Dispute Analysts
	cellPhone	4084084070		CCDCase Analysts
	EMailAddress	jdoe@emailexample.com		L
	lustomerSince	3/25/2000 85		
				-

- 1. Clicking on this opens the case list view where you can see all the cases that you are a participant in. You can also search for cases using different case attributes and metadata.
- 2. It is the activity panel and shows all available, running, and completed activities. Highlighted activities indicate required activities. The *Add* icon allows case participants to add activities to the list of available activities.
- 3. The *Linked Cases* panel shows all the cases that are linked to this case. You can also create new links using the *Add* button. Links can define one of four predefined relationships; related-to, depends-on, duplicate-of, and subcase. All of these except the subcase can be defined at runtime only. Subcase relationships can be defined either at runtime or design time. Subcases are beyond the scope of this book and hence have not been discussed.
- 4. The center panel is a tabbed panel and shows the details of the different case data. There will be one link for each case data object that has been instantiated. If an instance of a case data

object is not created, there will not be a link for it. For the details to show up, you need to build and deploy a data form, which you have done already.

- 5. Switching the tab to *Documents* allows you to add, update, and download documents from the configured CMS.
- 6. The Audit Trail shows a trail of all the events that have occurred since the start of the case.
- 7. The *Milestones* panel shows all the milestones defined for the case. The case worker can manually update the state of the milestone by switching the *Open* state to *Completed*. If a milestone has a deadline defined, you will see when it is due. For example, in the figure it shows that the *Initial Case Review Completed* milestone is due in 2 days.
- 8. Stakeholders can be managed in the workspace using this panel. Selecting the stakeholder and clicking on the *Edit* button allows you to add new members to the case. Clicking on the *Add* button allows you to add a new stakeholder and the members. Note that any changes done to the stakeholder model at runtime are specific to that case only.
- 9. The *Add Comment* link allows case participants to insert comments in the audit trail. This is very useful to document the progress or any information pertinent to the case.
- 10. The *Raise Event* link allows case participants to raise user-defined events. These events have to be defined in the case model.
- 11. The Actions drop-down list allows case workers to close, suspend, resume, or abort the case.
- 12. Clicking on this icon shows the case instance information such as priority, category, start date, and due date.

17.6.4.2 Update Process Roles

You defined two process roles in your case project—CCDCase.CSR and CCDCase.Analysts. But these did not specify the members. Although you can define the membership at design time, it is better to manage roles using the BPM Workspace. Before you work on the case, add members to these roles. The following steps describe how to do that.

- 1. Open the URL http://localhost:7101/bpm/workspace. This is the URL for the BPM Workspace. Log-in as *weblogic* with password as *welcome1*. Make sure you started the integrated WebLogic Server, as described in Section 17.6.1.1
- 2. Once you have successfully logged in to the workspace, click on the *Administration* link at the top of the page. Click on the *Roles* link, if it is not already selected. You should see the roles administration page:

dministration Areas	Roles	Search		0			19
rganization	Name		Members		Description	Context	
Roles	CCDC	BPM Admin R Case.Process Ow	ole Administrators ner		BPM application admin role, has full privilege for performing any oper	OracleBPMProcessRolesApp OracleBPMProcessRolesApp OracleBPMProcessRolesApp	
Calendars	CCDCA	se Stakeholder	C			OracleBPMProcessRolesApp OracleBPMProcessRolesApp	
Organization Units	CCDC	ase StakeHolder CCDCase Analy	rsts			OracleBPMProcessRolesApp OracleBPMProcessRolesApp	
Varametric Roles	CCDC	CCDCase.C Case.PUBLICRO.P	ISR IE		CCDCase.PUBLICRO.READ.Role	OracleBPMProcessRolesApp OracleBPMProcessRolesApp	
lusiness Parameters	CCDC	Case PUBLICRO LI See PUBLICRO IN	IP_, W		CCDCase.PUBLICRO.IPOATE.Role CCDCase.PUBLICRO.INVOKE.Role CCDCase.PUBLICRO.CLOSE.Role	OracleBPMProcessRolesApp OracleBPMProcessRolesApp OracleBPMProcessRolesApp	
ex Fields	CCDC	ase.PUBLICRO.F	0 5		CCDCase.PUBLICRO.FORCE_CLOSE.Role CCDCase.FULLACCESS.READ.Role	OracleBPMProcessRolesApp OracleBPMProcessRolesApp	
ublic Flex Fields							
votected flex fields	8						
ak Administration	[
pproval Groups							
and Configurations							
rask configurations							

- 3. Click on the *CCDCase.Analysts* in the *Roles* table. This will open up the *Details: CCDCase. Analysts* panel.
- 4. Click on the *Create* icon **[™]** in the *Members* table to open the identity browser where you can search for roles, groups, and users to add.
- 5. Change the selection from *All* to *Users* and enter *jstein* in the search field and click on *search*. You should see the user jstein that you added earlier. Select the row and click on *OK* to add.
- 6. Click on Save to save the changes.
- 7. Repeat steps 3 through 6. This time, choose *CCDCase.CSR* and add the user *wshake* to it. Make sure you click on *Save* when done.
- 8. Log-out of the workspace.

17.6.4.3 Grant Investigator Access Permission

Earlier in the exercise you associated the permission *InvestigatorsAccess* to the *Investigate Charge* activity so that only appropriately privileged users can initiate this activity. You define the users that have this privilege using the Fusion Middleware Control.

- 1. Open the Fusion Middleware Control in the browser by navigating to http://localhost:7101/em.
- 2. Login as *weblogic* with password as *welcome1*.
- 3. Open Application Deployments→OracleBPMProcessRolesApp→OracleBPMProcessRolesApp(D efaultServer).
- 4. Right-click and select *Security* \rightarrow *Application Roles* shown below.

ORACLE Enterprise Manager	usion Middleware Co	ntrol 12c	
🚽 WebLogic Domain 👻			
Change Center (2) (2) Changes + (2) Recording +		OracleBPMPro Application Deploy	ment - Start Up
Target Navigation		A Summary	
View •		General	e Active
Internal Applications BPMComposer		Application Type Deployed O	e ear n DefaultServer
DefaultToDoTaskFlow DefaultToDoTaskFlow		Servlets and JSF	Active Sessions 0
		Request Process Request	sing Time (ms) 0 ts (per minute) 0.00
OracleBPMBACServerApp DracleBPMComposerRolesApp GracleBPMCromposerRolesApp GracleBPMProcessRolesApp		Requests (per n Pending Re	ninute) 0.00 quests 0
CracleBPMProcessRolesApp (OracleBPMWorkspace	Home		
B PartnerOnboardingTaskForms B ReviewChargeDisputeTaskForm SECDemoDataForms	Monitoring Control Logs		fest Point http://localhost:7101/Orac
SECDemoTaskForms service-bus SimpleApprovalTaskFlow	Deployments Web Services		ittps://localhost:7102/Ora
 SOATestDemoApp Soa-webapps 	Security MDS Configuration	an i	Application Policies Application Roles

5. In the Search panel, enter the string CCDCase.INVEST and click to search. This should show you the different application roles as shown in the following figure that were automatically created for each of your permissions defined in the case. Each of these roles is used to control different operations such as READ (read document, read data, read audit trail, etc.), UPDATE (update any of the case elements), CLOSE (close case), FORCE_CLOSE (force close a case when there are required activities open), and INVOKE (invoke an activity or raise an event).

Disformation			
An application rol	e CCDCase.INVESTIGATORSAC	CESS.INVOKE.Role has be	een updated.
pplication Rol	es		
Application roles are context of end users	the roles used by security awa accessing the application.	are applications that are sp	pecific to the application. These roles are seeded by applications in si
P To manage users	and groups in the WebLogic Do	main, use the Oracle Wet	bLogic Server Security Provider.
> Policy Store Pr	ovider		
Search			
Enter search keyw	ord for role name to query role	s defined by this applicati	ion. Use application stripe to search if application uses a stripe that is
	Dela Nama Charte		
	Role Name Starts	With CCDCase.INVES	ST 🔘
View - Crea	ite	With CCDCase.INVES	ST 🔘
View 👻 🔄 Crea	ite 📴 Create Like	With CCDCase.INVES	st 🔘
View - Crea	ite	With CCDCase.INVES	ST
View Creation Creation Creation CCDCase.INVEST	IGATORSACCESS.READ.Role	With CCDCase.INVES	ST
View Cres Role Name CCDCase.INVEST CCDCase.INVEST	IGATORSACCESS.READ.Role	With CCDCase.INVES	DE Description Na Description CCDCase.INVESTIGATORSACCESS.READ.Role CCDCase.INVESTIGATORSACCESS.UPDATE.Role
View Crest Crest Concase.INVEST CCDCase.INVEST CCDCase.INVEST	IGATORSACCESS.READ.Role	With CCDCase.INVES	ST Dis Description Dis Description CCDCase.INVESTIGATORSACCESS.READ.Role CCDCase.INVESTIGATORSACCESS.UPDATE.Role CCDCase.INVESTIGATORSACCESS.INVOKE.Role
View Crest CCDCase.INVEST CCDCase.INVEST CCDCase.INVEST	IGATORSACCESS.READ.Role IGATORSACCESS.UPDATE.Role IGATORSACCESS.UVOKE.Role IGATORSACCESS.CUOSE.Role	With CCDCase_INVES Edit Delete	ST Dit Description CCDCase.INVESTIGATORSACCESS.READ.Role CCDCase.INVESTIGATORSACCESS.UPDATE.Role CCDCase.INVESTIGATORSACCESS.UNVOKE.Role CCDCase.INVESTIGATORSACCESS.CLOSE.Role
View Creation Role Name CCDCase.INVEST CCDCase.INVEST CCDCase.INVEST CCDCase.INVEST	IGATORSACCESS.READ.Role IGATORSACCESS.UPDATE.Role IGATORSACCESS.UVOKE.Role IGATORSACCESS.IVVOKE.Role IGATORSACCESS.CLOSE.Role IGATORSACCESS.FORCE_CLOS	With CCDCase_INVES Edit Delete SE.Role	ST Dit Description CCDCase.INVESTIGATORSACCESS.READ.Role CCDCase.INVESTIGATORSACCESS.UPDATE.Role CCDCase.INVESTIGATORSACCESS.UNVOKE.Role CCDCase.INVESTIGATORSACCESS.COSE.Role CCDCase.INVESTIGATORSACCESS.FORCE_CLOSE.Role
View Creation Role Name CCDCase.INVEST CCDCase.INVEST CCDCase.INVEST CCDCase.INVEST	IGATORSACCESS.READ.Role IGATORSACCESS.UPDATE.Role IGATORSACCESS.UPDATE.Role IGATORSACCESS.UPVOKE.Role IGATORSACCESS.CLOSE.Role IGATORSACCESS.FORCE_CLOS	With V CCDCase.INVES	ST Dit Description CCDCase.INVESTIGATORSACCESS.READ.Role CCDCase.INVESTIGATORSACCESS.UPDATE.Role CCDCase.INVESTIGATORSACCESS.UNVOKE.Role CCDCase.INVESTIGATORSACCESS.FORCE_CLOSE.Role CCDCase.INVESTIGATORSACCESS.FORCE_CLOSE.Role
View Creation Conclusion Conclus	IGATORSACCESS.READ.Role	With V CCDCase.INVES	ST Dit Description CCDCase.INVESTIGATORSACCESS.READ.Role CCDCase.INVESTIGATORSACCESS.UPDATE.Role CCDCase.INVESTIGATORSACCESS.UPOKE.Role CCDCase.INVESTIGATORSACCESS.FORCE_CLOSE.Role CCDCase.INVESTIGATORSACCESS.FORCE_CLOSE.Role
View Creation Role Name CCDCase: INVEST CCDCase: INVEST CCDCase: INVEST CCDCase: INVEST CCDCase: INVEST CCDCase: INVEST	IGATORSACCESS. READ. Role IGATORSACCESS. UPDATE. Role IGATORSACCESS. UPDATE. Role IGATORSACCESS. LIVVOKE. Role IGATORSACCESS. CLOSE. Role IGATORSACCESS. FORCE_CLOS OF CCDCase. INVESTIGATO	With CCDCase.INVES Edit Edit SE.Role CRSACCESS.READ.Rol	ST Dit Description CCDCase.INVESTIGATORSACCESS.READ.Role CCDCase.INVESTIGATORSACCESS.UPDATE.Role CCDCase.INVESTIGATORSACCESS.INVOKE.Role CCDCase.INVESTIGATORSACCESS.FORCE_CLOSE.Role CCDCase.INVESTIGATORSACCESS.FORCE_CLOSE.Role

- 6. Select the CCDCase.INVESTIGATORACCESS.INVOKE.Role role and click on Edit.
- 7. In the *Add Principal* pop-up, select *Type* as *User*, and enter *Principal Name* as *jcooper* and click to search.
- 8. As shown below, select the listed user and click OK to add it to the invoke role.

Type	User				
Deleviert	Charles Intitle .				
Principal Name	Starts with	JCOO	p		
Display Name	Starts With V				Q
arched Principals					
Principal			Display Name	Description	
jcooper			James C	Demo User	
Advanced Option					

17.6.4.4 Scenario #1

You use two scenarios to test the application. In the first scenario, a charge of \$34.56 is disputed by John Doe. The case was created using this information in the previous section. James Cooper is the case manager for this case. Here is how the case gets processed.

- 1. James Cooper logs in to the BPM Workspace using his user ID *jcooper* and password as *welcome1*.
- 2. He clicks on the *Case* link to navigate to the case workspace. He sees John Doe's dispute case. He looks at the case details clicking on the *Dispute Details* link. In the dispute details, he changes the status from *ASSIGNED* to *IN PROCESS* and saves it.
- 3. He clicks on the *Audit Trail* link to look at all that has happened to this case since it was created.
- 4. He decides to inform the customer about the case status and initiates the *Notify Customer* activity. Since we have only created dummy processed for these activities, the *NotifyCustomer* finishes immediately. He still sees the *Notify Customer* activity in the list of available activities since this is a repeatable activity and can be initiated any number of times.
- 5. Next he notices a required activity—*Review Charge Details*. As shown below, he initiates this activity and waits for the activity to complete. This activity results in a task being assigned to the CSR role. If notifications were configured for this task, the CSRs would get an email informing them of this new task.

Available 👘 🔫	Press Enter to Search	C All Comments Event •
Send Email Create Task Notify Customer This activity was previously initiated	Activity started: Review Charge Dispute Please review this charge dispute by John Doe. jatein	now
Review Charge Dispute This activity was previously initiated	Activity completed: Notify Customer Hello jatein	3 minutes ago
Review Charge Dispute Started now	Q Data modified: DisputeD0 jstein	4 minutes ago

- 6. Assume, the CSRs receive the e-mail and one of the CSRs, William Shakespeare logs in to the BPM Workspace using his ID *wshake* and password as *welcome1*. He notices a new task in his inbox to review the dispute with ID D0001. He clicks on it to view the details. He finds everything to be in order and decides to add a comment indicating so. He does that by clicking on the *Create* icon in the *Comments* panel and adds the text "All details seem to be in order. OK to proceed with the investigation." He clicks *OK* to complete his task.
- 7. James Cooper checks back on the case to see if the review is completed. He refreshes the case workspace by clicking on the refresh icon and notices that the review has been completed. He clicks on the *Audit Trail* to view the events. He then clicks on the *Comments* link to see if there were any comments added by the CSR and notices William's comment.
- 8. James also notices that the first milestone, *Initial Review Completed*, has been achieved. Since the case has now progressed to the next part of the processing, he also sees a couple of more activities, namely, *Investigate Charge* and *Request Documents from Customer*, are now available.

Note: If you have setup your permissions properly as described earlier, only James should have access to the *Investigate Charge* activity. To test this, log-out and log-in as wshake or as jstein. They should not be seeing this activity at all.

- 9. Since James is satisfied with the information available, he initiates the investigation by clicking on the *Investigate Charge* link. This causes a task being assigned to the analysts to complete the investigation. This also marks the start of the analysis and the *Analysis Started* milestone is marked as having been reached.
- 10. John Steinbeck, one of the analysts, logs in to the BPM Workspace using his ID as *jstein* and password as *welcome1*. He notices the new task for investigating the charge dispute case with ID D0001. He clicks on the task to show the task details. He conducts his investigation and once done he opens the task and sets the *Outcome* to *MerchantFault* (this is case sensitive) in the *Investigation Report BO* section of the task form. He clicks *OK* to complete the investigation task.
- 11. Assume that James receives a notification when the investigation is complete. He checks in on the case clicks on *the Investigation Report* and sees the outcome was that the merchant is at fault. He also notices that since it is a merchant fault, activities for communicating with the merchant and initiating chargeback are now available. He also notices the *Analysis Completed* milestone has been automatically marked as *Completed*.
- 12. James initiates the chargeback process by clicking on the *Initiate Charge-back* activity link.
- 13. Once the chargeback process is complete, he decides to close the case by clicking on the *Action* drop-down list and selecting *Close Case*. He selects the outcome as "Case Resolved Merchant at Fault" and clicks *Close Case* to complete the case.

17.6.4.5 Scenario #2

In the second scenario, the amount of dispute is \$20 and the case owner decides to write-off the charge.

Use the same test XML as for scenario #1 but change the dispute amount to 20 and the dispute ID to D0002.

Run through this scenario by yourself using scenario #1 as a reference.

17.7 Summary

In the first part you learned how to analyze a case using a mind map diagram by identifying all the different activities that may be required to process the case. The mind map serves the purpose of graphically showing all the related elements of case and their dependencies.

In the second part you saw how to convert the mind map into an implementation design using simple tools such as XSL tables for tabulating all events and corresponding actions. You used these tables later on to create rules using Oracle Business Rules. You built the complete case application by creating activities based on BPMN process and human tasks and deployed and tested it all from within Oracle JDeveloper.

17.A Appendix

17.A.1 customer.xsd

```
<?xml version = `1.0' encoding = `UTF-8' ?>
<schema xmlns = "http://www.w3.org/2001/XMLSchema"
xmlns:xsd = "http://www.w3.org/2001/XMLSchema"
    xmlns:tns = "http://www.mycompany.com/ns/CCDCase"
targetNamespace = "http://www.mycompany.com/ns/CCDCase"
    elementFormDefault = "qualified" >
<element name = "Customer" type = "tns:TCustomer"/>
<complexType name = "TCustomer">
<sequence>
<element name = "CustomerId" type = "xsd:integer"/>
<element name = "FirstName" type = "xsd:string"/>
<element name = "Middlename" type = "xsd:string"/>
<element name = "Lastname" type = "xsd:string"/>
<element name = "SocialSecurityNo" type = "xsd:integer"/>
<element name = "Nationality" type = "xsd:string"/>
<element name = "MaritalStatus" type = "xsd:string"/>
<element name = "Gender" type = "xsd:string"/>
<element name = "Status" type = "xsd:string"/>
<element name = "AddressLine1" type = "xsd:string"/>
<element name = "AddressLine2" type = "xsd:string"/>
<element name = "City" type = "xsd:string"/>
<element name = "State" type = "xsd:string"/>
<element name = "PostalCode" type = "xsd:integer"/>
<element name = "Country" type = "xsd:string"/>
<element name = "HomePhone" type = "xsd:integer"/>
<element name = "WorkPhone" type = "xsd:integer"/>
<element name = "CellPhone" type = "xsd:integer"/>
<element name = "EMailAddress" type = "xsd:string"/>
<element name = "CustomerSince" type = "xsd:dateTime"/>
</sequence>
</complexType>
</schema>
```

17.A.2 test-startcase-input.xml

```
<ns2:stakeHolderMembers>
<ns2:stakeHolderMember>
<ns2:stakeHolder>jcooper</ns2:stakeHolder>
<ns2:stakeHolderType>USER</
ns2:stakeHolderType><ns2:applicationContext>OracleBPMProcessRolesApp</
ns2:applicationContext>
</ns2:stakeHolderMember>
</ns2:stakeHolderMembers>
</ns2:stakeHolder>
</ns2:stakeHolders>
<ns2:data>
<ns2:name>DisputeDO</ns2:name>
<ns2:data>
<DisputeB0 xmlns:xsi =http://www.w3.org/2001/XMLSchema-instance</pre>
xmlns = "http://xmlns.oracle.com/bpm/bpmobject/CCDBusinessObjects/
DisputeBO" >
<disputeId>D0001</disputeId>
<creationDate>2014-01-01T00:00:00</creationDate>
<customerId>1001</customerId>
<disputeAmount>34.56</disputeAmount>
<transactionReferenceNumber>2234</transactionReferenceNumber>
<merchantId>BB1134</merchantId>
<transactionDate>2013-12-20T00:00:00</transactionDate>
<creditCardNumber>44445555666667777</creditCardNumber>
<status>NEW</status>
</DisputeBO>
</ns2:data>
</ns2:data>
<ns2:data>
<ns2:name>CustomerDO</ns2:name>
<ns2:data>
<Customer xmlns:xsi =http://www.w3.org/2001/XMLSchema-instance
xmlns = "http://www.mycompany.com/ns/CCDCase">
<CustomerId>1001</CustomerId>
<FirstName>John</FirstName>
<Middlename></Middlename>
<Lastname>Doe</Lastname>
<SocialSecurityNo>999999999</SocialSecurityNo>
<Nationality>USA</Nationality>
<MaritalStatus>MARRIED</MaritalStatus>
<Gender>M</Gender>
<Status>GOLD</Status>
<AddressLine1>111 No Name St</AddressLine1>
<AddressLine2></AddressLine2>
<City>San Francisco</City>
<State>CA</State>
<PostalCode>99999</PostalCode>
<Country>USA</Country>
<HomePhone>4154150000</HomePhone>
<WorkPhone>6506506500</WorkPhone>
<CellPhone>4084084070</CellPhone>
<EMailAddress>jdoe@emailexample.com</EMailAddress>
<CustomerSince>2000-03-25T00:00:00</CustomerSince>
```

</Customer> </ns2:data> </ns2:data> </ns2:case> </ns1:startCaseInputMessage> </soap:Body> </soap:Envelope>

Chapter 18

Siebel Tutorial

18.1 Introduction

Siebel is a package-based solution that contains out-of-the-box case management solutions for specific use cases in different industry domains. Once a decision is made to implement a case management solution with Siebel, different activities need to be done to implement the solution. We will look at two aspects of a Siebel-based implementation scenario; the approach on how to design and implement a package-based case management solution and to use the package. In this chapter we will use a police investigation example implemented with the *law enforcement and policing operations* extension, as described in Chapter 8.

The goals of this chapter are as follows:

- Give a high level overview of the Siebel implementation approach
- Show how the implemented solution is used by the end user
- Provide usage details of the Siebel extension for the law enforcement and policing operations solution

18.2 Siebel Implementation Approach

The Siebel implementation approach is based upon following four phases:

- Choosing the Siebel functionality
- Familiarization
- Design
- Implementation

These phases align with the case management solution framework as presented earlier in Section II. The relation between the Siebel implementation approach and the case management solution framework is shown in Figure 18.1. In the first phase, *choosing the Siebel functionality*, the *case management classification* and the Siebel product type is determined. The *case management functional design* is done in the phases, *familiarization* and *design*. After the case



Figure 18.1 The Siebel implementation approach.

management functional design is finished, the *solution mapping* can be determined as part of the design phase. In the last phase the implementation of the Siebel-based case management application can take place.

18.2.1 Choosing the Siebel Solution Type

In Chapter 8 we discussed the different functionalities Siebel provides. The base functionality is a CRM-based application extended with case management capabilities. Siebel extended the case management functionalities toward different industry and cross-industry-specific solutions, known as verticals and horizontals. The first step in the Siebel implementation approach is to determine which Siebel functionality is going to be used as the basis for the solution. In this stage, the following characteristics are used:

- Related industry domain
- Required processes and case high level definition
- The case management classification, service request, incident management, and/or investigation
- Fit-gap with existing base product, verticals, or horizontals

In this chapter we will look at the Siebel *law enforcement and policing operations* extension to fulfill the requirements of the police force in order to be able to fight criminality more effectively.

The classification of the case management to be used in the solution is of the type incident management. This classification impacts the life cycle and the need for intelligence on different levels. This classification was discussed in more detail in Section 4.5.

18.2.2 Familiarization

The core users, process owners and change champions are acquainted with the case management functionality during the familiarization phase. This helps them to understand, absorb, and document the impact of the changes to existing processes and ways of working. This phase provides the foundation for change management, solution realization, and deployment. Typically this phase has a short duration.

Menu	new	odete	Jave Greate G	ase Cancer	Justy -						
Incident S	ummary:*	Dead person	found in his home d	uring surveillance		Incident	Туре:	Violent Crime	•		
Incident	Number:*	1-3SKS4		Date Created:	Created: 4/7/2014 04:48:55 PN		Incident Status:		Active		
As	sessment;		120	Created By:	AGENT16		Incident Pri		High		
Date Occurred: 4/7/2014 04:48:55 PM		48:55 PM	Date Reported:	4/7/2014 04:48:55 P	M 🖿	Incident Set	verity:	High			
	Location:	Herengracht	1217 📧	Descriptive Location:	After lead, dead boo discovered in nremi	fy ses of]	Owned	ву:*	AGENT16	B 3	
More Info	Offenses	Allegations	Offender-Persons	Victim-Persons	Measures Incid	ent Witnesses	Suspects	Prop	erty Evider	nce Cases	Note
Menu	• Ada	d Delete	Query Search		Type of Measure						
	Туре	of Measure	Last Name	First Name	Gender		Measure Date		Location	Measure S	itatus
Same of the	emises	N	mwegen, van	Henry	M 4/	6/2014 02:00	:00 PM	Herenge	racht 1217	Completed	

Figure 18.2 Screen design for coercive measures within an incident used for PWT and CRP.

The deliverables of this phase are as follows:

- Business vision, including the scope of the project
- The main requirements for the system
- An overview of the to-be processes and a gap analysis with the current processes
- A first draft of the to-be application architecture
- Insight in a data migration plan outlines the data migration needs, phasing, and approach

Two of the most used types of workshops within the familiarization phase are the *package walk through* (*PWT*) and the *conference room pilot* (*CRP*) with the different stakeholders. Both deliver an innovative contribution to this phase. In the *PWT* sessions, the desired functionality is compared with the standard available functionality in the Siebel product. The outcome of the PWT sessions is used to create a first draft of the system. In the *CRP* sessions, this predefined system is evaluated in a more detailed manner and adjusted in an iterative and interactive way.

These sessions are supported by the user interfaces to visualize the way the system supports the functionality. An example of such visualization is shown in Figure 18.2 that supports the sessions used to determine the police investigation functionality.

18.2.3 Design

The design aims at delivering two case management solution framework artifacts, the case management functional design, shown in Figure 18.3, and the solution mapping.

18.2.3.1 Functional Design

Every functional category will be investigated with the stakeholders; this will be used as input for the configuration and customization part in the implementation phase.



Figure 18.3 Case management design categories and components.

For the police investigation, case-specific attention will be given to the following areas: *Case life cycle design category*—As we have seen in Chapter 14, the life cycle management for the police investigation is complex:

- Many investigation types exist.
- Complex state models describe the execution of each investigation type.
 Figure 18.4 shows an example of a complex state model for a lead investigation.
- Different processes support the execution of (allowed) activities during specific states in the state model. Figure 18.5 shows an example of a process around coercive measures against a suspect.
- Both human and automated activities support the process workflows.

All these design aspects need to be designed in relation to the capabilities of the Siebel *law enforcement and policing operations* extension.

Case information design category—The Siebel *law enforcement and policing operations* extension data model already contains information entities that support the work of police investigations, as described earlier in Chapter 8. The main information entities are combined in *incidents, leads,* and *cases*. During the design, these information entities are checked against the needs of the police force and related to the current information sources.

Case interaction design categories—Out of the box Siebel delivers standard portal views for the different stakeholders in the police investigations. The police force needs to have the ability to visualize the relations between persons, cases, activities, documentation, and media. All police stakeholders have different requirements what information they need to have access to and how this should be visualized. In this design, the user interface design is shaped toward the needs of the end users in iterative workshops. Another aspect in this design is investigating the channel types the police need to work with the application. Besides, the out-of-the-box browser-based portal new media are entering the police force, such as mobility and connecting to different social media.

Cross-functional design category—In police investigations, often sensitive information is used that should not be made available to everyone in the police force. The Siebel functionalities, *user interfaces*



Figure 18.4 Complex state model of a lead to support its full life cycle.



Figure 18.5 Example of a process around coercive measures against a suspect.

access management and responsibilities and *Data related record level access management* enable designing *roles and authorization* functionality on both user interface and on data level.

The Siebel application is going to be placed in an already existing complex police application landscape where information needs to be exchanged between the different applications. In this, the design investigation is done around data ownership, related applications, and the way information is shared between Siebel and the other applications.

18.2.3.2 Solution Mapping

As discussed in Chapter 8, two scenarios exist for implementing Siebel in an already existing landscape, the "Siebel single entity" scenario and the "Siebel in a wider application landscape" scenario.

In the "Siebel single entity," all functionality required by the police force for police investigation is contained in the Siebel *law enforcement and policing operations* extension.

In the second scenario "Siebel in a wider application landscape," the Siebel application is only responsible for a part of the end-to-end police investigation and/or additional support is needed in areas where Siebel does not deliver the needed functionality.

Both scenarios have implications on required functionality in related applications, data migration, and the already mentioned integration capabilities of the application.

18.2.4 Implementation

In the implementation phase, the standard application is configured as specified and custom components such as interfaces and data conversion scripts are built. The objective is to deliver the application as intended by the design and ready for testing and validation. Unit testing is an integral part of the implementation process and is carried out by the developers within the development environment. In this phase, test preparation takes place by establishing the various test plans for system test, integration test, and user acceptance test. The test scenarios are created based on the defined functionality during the design phase often captured in use cases.

During the realization of the application, feedback loops are established to involve the police organization early in the process of reviewing the solution. These feedback session aid the adoption process in the organization and enable necessary changes where needed in the application.

The setup of the application is done on two levels: at the business administration level and the technical administration level:

Business Administration level: Within this level the business logic and user access of the application is managed. The business administrators require a good understanding of the business functionalities. The following (noncomplete) list shows examples of topics that are administrated at this level:

- State model: State models determine the lifecycle of the entities in the application. It controls the status change, what statuses are available, who can change them, and when it can be changed. An example of setting up a state model is shown in Figure 18.6.
- Assignment manager: These rules can be managed at business administration level. These rules determine what user is assigned to cases and tasks. Due to its complexity and impact, the assignment rules are often managed at technical administration level.

	itus								4 8 of 10+
teni +	les Deele	Salie Carice Q	untry .						
	Name* PP	Lead Status	Activation *	12/1/2012 08:02	07 PM 😁	Created By	SADIIN		
Business Co	manant # DU	Rised 1	Execution		-	Created Date/Time	10/31/2013 12:03:21 PM	-	
0341144 00	interiment - The	o cesa	20 Sold and a			Contraction Contractions.	Total terms to terms at the		
	terest into		Continents						
	100		5						
	-								
Ando States	Transitions	Categories							
Second III	ben Dehte	Store Manual S	sarch	From Category					1.8.49
Calmoney	From State	To Calegory	To State	Public	Rule Field	Rule Operator Rule	Aller	Bule Exercacion	1.52556
an chargery	Active	to category	Rejected	1 dates	COMP I I I I I		50772		
	Active		Validated	<u> </u>					
			a second s						
1	Closed		Rejected	14					
	Closed Submitted		Rejected Rejected	8					
	Closed Submitted		Rejected Rejected Under Investigation	8					
	Closed Submitted Submitted Under Investigate	a	Rejected Rejected Under Investigation Closed	8					
	Closed Sobrritted Subrritted Under Investigate	a.	Rejected Rejected Under Investigation Clased Rejected	8 8					
	Closed Sobritted Subritted Under Investigate Under Investigate	a. o	Rejected Rejected Under Investigation Cleared Rejected Re	8 8 8 8					
	Closed Sobrritted Subrritted Under Investigate Under Investigate Validated	a. a	Rejected Rejected Closed Close	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8					
	Cosed Submitted Submitted Under Investigate Under Investigate Validated	a n	Rejected Rej	8 8 8 8					
	Consel Closed Submitted Under Investigatio Under Investigatio Validated	a 0	Reported Reported Under Investigation Closed Rejected Rejected Submitted	8 8 8 8 8		8 4 9 8			
	Consel Consel Submitted Under Investigate Under Investigate Vakitated	a a	Reported Reported Under Investigation Clased Reported Reported Submitted	8 H 8 H 8 6 H					
uthorized P	Consel Consel Submitted Under twestSpate Under twestSpate Validated	a a	Reported Reported Under Investigation Closed Reported Submitted	3 H 3 H 3	No Rec	ж « » и то п			
uthorized P	Consed Submitted Submitted Under Investigatis Velicitated Velicitated	n n	Reported Reported Under Investigation Clased Reported Submitted Submitted		No Rec	N 4 N N			

Figure 18.6 Setup screen for state models.

- Responsibilities and positions: A user can have one or more responsibilities and usually one position. A responsibility determines what functionality the user can access and the position determines what data the user can access. The access to screens, views, and applets is managed in the responsibilities assigned to the users.
- *List of values*: For speeding up the end-user work and data quality, purposed pick lists are defined in the system. The values of the pick lists are managed in one big table called the list of values (LOV). Dependency of combined pick lists is also managed in the LOV.

Technical Administration Level: This level of administration and setup is related to the IT side of the application. This level requires an understanding of the system architecture and technical functionalities of the application and the interfaces. Within this level, the application software, databases, and interfaces are managed:

- Workflows and business services: Within Siebel business processes can be automated using workflow and business services. Workflow to control the processes within the application and business services to control the processes outside the application.
- Integration and data migration: Integration with other applications and databases as well as migrating data from external sources into the application database is managed in the technical administration level.

18.3 Using the Police Investigation Application

When the police investigation application is delivered to the police organization, the end users need to be trained into using the application. Here we will show, based upon a police investigation in a murder case, the different steps how the application is used to support the police officers. We use a simplified detective story to show the different activities for which the police case management is used and will show how a lead, an incident, and a case are created within Siebel and how these are linked together. The implementation we use for this police investigation example is based on a position and role set up specifically defined to fit this story and can be adjusted to other needs and requirements.

18.3.1 Storyline

The storyline is taken from one episode of a famous Dutch detective TV series by Baantjer (http:// en.wikipedia.org/wiki/A._C._Baantjer). The author Baantjer used to be a police officer himself and brought knowledge from his own field work into the novels.

In this series, the senior police officer de Cock and his junior sidekick Vledder solve cases with a typical sense of humor known in Amsterdam. The episode we use here is "Healing Masters" (http://www.baantjer.org/). The plot in this episode takes place in a cosmetic surgery clinic and in short deals with tense business relations, a love affair, and a badly treated customer resulting in three homicides. The timeline in Figure 18.7 shows the different activities and events taking place from the moment one of the surgeons has been found murdered. In this picture the three main police case types, the lead, the incident, and the case are plotted. The case starts with the intake of a lead indicating a missing person. When the police officers go to the house of the missing person, they find the body. All information at the crime scene is captured as an incident. The severity of the incidents leads to the creation of the case and is used to store all activities related to the investigation, such as witness statements. The case is also linked to the lead and all Incidents captured during the investigation.

18.3.2 Recording a Lead

The story starts with capturing the information of the interview with the woman that warned the police. This will be captured as a lead *Henry van Nimwegen missing*. This process is called the (easy) intake of a lead. The recording of a lead is a sequential process during which the following information is entered:

- Generic information
- Person, object location, and event (POLE) information
- Operational details
- Lead details

Once the lead is entered, the lead can be further worked on and/or viewed by all authorized personnel in a lead 360° overview. The lead can also be shared with other agencies or organizations via the lead export.

18.3.2.1 Creating the Lead and Entering the Generic Lead Information

The starting point for entering a lead is the lead page as shown in Figure 18.8. From this page, new leads can be created, existing leads can be searched, and the lead can be linked to a case or shared with other agencies or organizations (lead export).

The intake of the lead, shown in Figure 18.9, starts with entering the lead name, which is a short recognizable label. An extensive description of the lead will take place at a later moment.

The date field will be prefilled and cannot be overwritten by the actor. The lead ID is fixed by the system and the actor will also be automatically recorded as the creator of the lead. Lead assignment



Figure 18.7 Timeline for the "Healing Masters" murder case.

										eau status t	and the	2 cf 2*
en •	Delete Save Can	cel Qitry Lease	Sport Link to C	Sase .								
Lead Name:*	Henry van Nimwegen misaiz	9			Person:	•	Lead Priority:	Routine		Source Name:		1
Load Seriel #:		Assigned To:	OFFICER	ND	Object:		Lead Type:	Investigative		Source Type:	Phone	
Lead ID:	1-6293951	Date Assigned:			Location:		Leed Status:	Active	•	Source Organization:		
Date Created:*	4/4/2014 83:58:24 PM	Date Completed			Event		Lead Sub-status	Pending		Relability:		

Figure 18.8 The lead page.

Lead Name:*	Henry van Nimwegen missir	1g		
Lead Serial #:		Assigned To:	OFFICER	Q .:
Lead ID:	1-6293951	Date Assigned:		
Date Created:*	4/4/2014 03:56:24 PM	Date Completed:		Ē

Figure 18.9 Intake of the lead.

ain menioeis					~	ned To:	OFFICER	S
Save					1 - 2 of 2	ssigned:		
Selected						erritory:	West	
Primary	User ID	Last Name	First Name	Position	Relationsh	moleted		
				Adjuster				
¥	OFFICER	Brown	Paul	IPP Case Handler				
						Notes	Approvers	Auto Lead
	111					Notes	Approvers ing Henry van	Auto Leac

Figure 18.10 Multivalue pick list.

will take place manually based on a multivalue pick list capability (**III**), as shown in Figure 18.10. It is possible to assign multiple persons (team) to a lead. One of the assigned officers will be the prime and final responsible person (primary).

Assignments can be done manually, automatically, and semiautomatically. Often the assignment rules are based and set up on specific needs and requirements that fits best to a police force and can be changed or adjusted at any time.

Person:	٩
Object:	•
Location:	Q :
Event:	

Figure 18.11 POLE fragment on the lead page.

18.3.2.2 Entering the Person, Object Location, and Event Information

The person, object location, and event (POLE) information is the basic information that is needed during the initial information gathering, as shown in the POLE fragment on the lead overview page in Figure 18.11.

The *person* can be selected via a single-value pick list. When the person does not exist in the system, a new person can be created.

In the *object* field describes the nonhuman object related to the lead. When the lead deals with a "lost or found" object, this field is entered. In our example, with a missing person, the object field is left empty.

The recording of the *location* information is based on a single value pick list. In this field information about the address, area and Geographic information system (GIS) details can be stored. Area information is useful when a proximity to a location is known, such as and object is found in a public park or near to a lake.

The last POLE detail is *event* and is related to a data and a description. In our example, the event relates to the date Henry van Nimwegen was missing and a description of the circumstances.

18.3.2.3 Recording the Operational Information

The operational information relates to the way the police will handle this lead. Figure 18.12 shows the fields related to the operational information on the lead page.



Figure 18.12 Examples of lead priority values.

Lead Priority:	Routine	•	Source Name:		٩
Lead Type:	Investigative	*	Source Type:	Phone	٠
Lead Status:	Active	•	Source Organization:		٩
Lead Sub-status:	Pending	*	Reliability:		•

Figure 18.13 Operational fragment on the lead page.

Most of the operation information can be stored by making use of the LOV capability. This will speed up the intake and store information in a consistent way. All these LOVs are maintained at run-time via the business administration page "administration-data."

- The *lead priority* gives an indication when the lead will be acted upon.
- The *lead Type* determines who will be assigned to the lead.
- In Figure 18.4, an example of a state model for a lead is shown. The *lead* (*sub*) *status* gives detailed information of the whereabouts of the lead in the state model.
- The source information relates to the origination of the lead as shown in Figure 18.13.

18.3.2.4 Adding More Lead Details

During the investigation on the lead, information becomes available that needs to be attached to the lead investigation. The Siebel police application offers support for linking *attachments, note,* and more *POLE* information to the lead. Incidents and cases, as will be described later in this chapter, can also be linked to the lead. In Figure 18.14, an example of an attachment, the recorded *"Interview with Irene van Moerkapelle,"* is shown.

Menu 🔹 Ne	w Delete Sa	ve Cancel	Query Lead Ex	port			
Lead Name:*	Henry van Nimw	Henry van Nimwegen missing					
Lead Serial #:			Lead ID:	1-6293951			
Case Name:	Dead of healing r	naster 🖭	Date Created:*	4/4/2014 03:56:24 PM			
Case Owner:	OFFICER		Lead Originator:	AGENT16	Lead		
lore Info Activities	Activity Plans	Addresses	Property Attachme	ents Cases Contacts	Evidence		
	elete Save (Query Search	h	Name 💽	New File		
Menu 🔹 De					and a second sec		
Menu 🔹 De	i	Size (In Byte		Modified	Update Fi		

Figure 18.14 Adding an interview in MP3 format to the lead.

Menu * Nev	Deleter Save Cancel	Query Lead Exp	ort						
Lead Name:*	Henry van Nimwegen missing	Contraction of Contraction		Lead Priority;	Routine		Assigned To:	OFFICER	
Lead Serial #:		Lead ID:	1-6293951	Lead Type:	Investigative		Date Assigned		
Case Name:	Dead of healing master 50	Date Created:*	4/4/2014 03:56:24 PM	Lead Status:	Active		Territory;	West	
Case Owner:	OFFICER	Lead Originator:	AGENT16	Lead Sub-status:	Pending	٠	Date Completed:		
Nexu • Ne	w Delete Save Cancel	Query Submit	Recal						
Lead Information Lead Name * Lead Type: Lead Sub-type	V Delete Save Caroon Henry van Nimwegen missing Investigative • Report •	Query Submit	Recall Active 7 Pending 7	Lead Description Unne van Moertapel Minwegen. Her gut Mink had an appointe She wint to clink n His partners stated 1	le comes in to the po feeling is that someth ent 2 days ago whe ext day but he didn't that he is very punct	ice station and sti ing is wrong, re he didn? show show up there edit val.	ates that she is mise up.	sing Henry van	
Lead Information Lead Name:* Lead Name:* Lead Type: Lead Sub-type Lead Priority:	V Delete Save Cancel Henry van Nimwegen missing Investigative • Report • Routine •	Cuerr Submit	Recall Active Pending B	Lead Description Irene van Moertapel Nimwegen. Her gut MM had an appointm She went to clinic n His partners stated to	is comes in to the pole feeling is that someth ent 2 days ago where ext day but he didn't hat he is very puncth	ice station and sti ing is wrong, re he didn't show show up there ett val.	ates that she is misd up. her.	sing Henry van	
Lead Information Lead Name.* Lead Name.* Lead Sub-type Lead Priority: Case Information	V Delete Save Clancel Henry van Nimwegen missing Investigative • Report • Rodine •	Query Submit	Recall Active Pending B	Lead Description Irone van Moerkapel Nimwegen. Her gut Mad an appoint En wind to cinic n His partners stated I Suspect Informatic	le comes in to the po feeling is that someth ent? 3 days ago whe ext day but he didn't i hat he is very puncts	ice station and sti ing is wrong, 's he didn't show show up there eth val.	ates that she is miss up. her.	sing Henry van	
Ead Information Lead Name:* Lead Name:* Lead Name:* Lead Sub-type Lead Plorty: Case Information Case Name:	V Delete Save Cancel Henry van Nimwegen missing Investigative • Report • Routine • Dead of healing master	Query Submit Lead Status Lead Status Lead Status Approval Template	Recoll Active Pending 8 53	Lead Description Fine-van Moritage Him-van Mor	le comes in to the po feeling is that someth ent? 2 days ago when ext day but he didn't i hat he is very punct int as	ice station and sti ing is wrong, rs he didn't show show up there eti val. 533	ates that she is miss up. her. Gro	aing Henry van	
Each Information Lead Information Lead Name:* Lead Nype: Lead Phorpy: Case Information Case Information Case Type:	Celete Sales Cancel Henry van Namvegen missing wostigetive • Report • Routine • Dead of healing master Crismal	Case Senal #	Recall Active Pending 8 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Lead Description The van Noortage Filtewear Mer get With data expedite His partners stated i Suspect Informatic Suspect Informatic Suspect Statu	is comes is to the pole feeling is that someth ent 2 days ago when days to the ddn's is that he is very punct in a s	ice station and sti ing is wrong, in he didn't show show up there etil rat.	ates that she is miss up. her. 3 Group Stat	aing Henry van	
Iterat ** Iter Lead Information Lead Name * Lead Sub-type Lead Sub-type Lead Phinfly Case Information Case Name Case Status	V Delet Save Carcel Henry van Namvegen nissing hvestigative • Report • Routine • Deed of hesing master Criminal Active	Covery Submit Lead Status Lead Sub-status Approval Template Case Serial # Case Number:	Recall Active Pending 8 NY-CRC-1-6372189 1-6372189	Lead Description	Is comes in to the pole feeling is that someth ent 2 days ago when ent 2 days up when ent day but he didn's state of the second source of the state of the second source of the second	ice station and sti ing is wrong, in the ddr3 show show up there etil ast.	ates that she is miss up, her Group Stat	sing Henry van	

Figure 18.15 Providing a 360° overview in the lead overview page.

18.3.2.5 Provide All Relevant Lead Information in the Lead 360° Overview

A complete overview of the lead and all related attachments, notes, lead details, and relations to incidents and cases is available in the lead 360° overview as is shown in Figure 18.15. This view is available to all authorized personnel who need to work on this specific lead or work on related leads, incidents, and/or cases and are searching for commonalities.

18.3.2.6 Sharing the Lead Information via Lead Export

Another important feature is the capability to share a lead and its related information with other agencies and organizations, where police officers are working on other investigations and looking for relations between police investigations. The lead can be shared in different formats like mail, document, and file. The outcome (answer) of the information request normally is stored as an attachment to the lead. The lead export can also be used to hand-over the lead information to intelligence and analytic tools of specialized departments within the own organization (Figure 18.16).



Figure 18.16 The lead export.

	· Distal Lind Li	Contract Contract	Starte.							
Incident Summary *	Dead person found in his f	one during surveillance	e		incident Type	Volent Crime •	Source Name:		0	
Ricident Number *	1-35854	Date Create	d 4/7/2014 04:48:53 P	4 5	Incident Status	Active •	Source Type:	Sarveilance	•	
Assessment		E Createt 0	y AGENTIS		Incident Priority:	High •	Source Organization:		C.	
Date Occurred:	4/7/2014 04:40:55 PM	Date Reporte	d: 4/7/2014 04 48 55 Pi	u =	Incident Severity	нал	Case Name:	Dead of healing master (R	3	
Lecation	Harangracht 1217	E Descriptiv	After lead, dead bod discretiverent in memory	W and a state	Owned By #	AGENT16 RE				
Are lots Offerens	Allegations Offender /	aran Tuto Pesa	nt Beaures Beed		Tanperts Pro	sporty Existence Cause	Autor Autorities Loof	Attachesonis Groups	Crossines 3	alpens Cumluts 💽 1-1=11 🗔
Are late Otherson	Abagahana Ohender /	ariana Wates Parsa	Lend C		tasperts Pro	sperty Existence Cause	Autor Autorities Lood	Attachesonis Groups	Crossines 3	Alpechs Cumluths 💽
Lead ID	Allegalizers Offender /	aran Search ne Lead 39	n: Bessures field Land D e Land Status	Lead 1	Serial # Assi	operty Existence Cause Igned To Terre van Noerkaprile Mil had an appontmen	Description C Description C corres in to the	ettachesente. Groups	Organiteses 3	agent Contacts 💽 1-1art 📖 Case 1
Level ID 6293061 H	Alegature - Oferene - F Lead Na Lead Na erry vát Neweger masong	ersons Status Parso Search me Lead Sp Treestgative	n Resource (Rest Land D e Lead Status Active	Lead 1	Serial # Assi OFFICER	sperty Existence Cases signed To Perre van Naerkapelle Jelf had an appointmen She went to clinc, neet	taxes annume text	Attactionaria Groups ase Name	Creandaness 3	dipeds Controls 💽 1-1aft (Gu Case S
Azes tots Othereas Here: 4 Land 10 42223651 H	Alegature - Ofenne - Lead Na Lead Na Lead Na	intern State Period	n Heatury head Land C - Land Status Active	Lead 1	Serial # Assi	sperify Environme Connec ligned 76 Ferrer van Noerhapelle Jelf had an appointeen Sine went to other ned His partners statud the	Abane Infinition Level Description C corres in to the 12 days age white day to fine ddir. Dead of the the is very gue.	Attateants Groups	Crownikows 3	digardis Conthudis (a) 1-1 ant (a) Conce S
Ause tots Otherass	Allegations Observed 4	intern Plate Para ary Search the Lead Ty Prostigative	er Henneys Been Land C e Land Status Attive	Lead 1	Serial # Assi	spenty Environme Canas ligned To Verre van Noerhapelle kell had an apportnem She went to clim, ned His performe statut the	Anne Annue Leve Description C Corres in to the 12 days aga wh day but he door Dead of he it he to very pun.	attactorative Groups	Crowskees 3	alpeds Controls 💽 1-1art (Li) Cone 1
Land ID Land ID 6222051 H	Alegatives Offenser / Lead Tar Lead Tar Lead Tar	errors Viene Perce erro Storch me Lead 3g Protestigation	er Bennurs Benn Land C u Land Sintes	un Titteren	Serial # Acci OFFICER	eventy Evidence Caxes Append To Permer van Noenfaapelle Helf held en appontemer Sine went to cites med His gestnere sided tha	Annue Annues Lead	Attachments Groups	Crowshoese 3	alpeds Consists (n) 1-1aft (g) Core

Figure 18.17 The incident creation page.

18.3.3 Recording an Incident

In the crime story example, a surveillance car is sent to the home of Henry van Nimwegen after the lead intake. The police officers discover in the home the dead body of Henry van Nimwegen and results in the creation of an incident "Dead person found in his home during surveillance." The original lead is automatically linked and related to the incident. Figure 18.17 shows the incident information on the top part of the incident creation page and the relation with the lead in the lower part of the page.

In the following we explain some standard capabilities around incident management, creating and working on an Incident, offense management, and coercive measures related to the Incident.

18.3.3.1 Creating an Incident

The creation of an incident starts with entering generic information around the incident and contains the following base information:

- Incident summary
- Dates (occurred, created, and reported)
- Location
- Operational information (type, status, priority, severity, and owner)
- Source (name, type, and organization)

This information is typically entered in the system by a police officer at the scene or at the control room as shown in Figure 18.17. Assessment functionality can be used to determine the incident type, priority, and severity. Eventually the Incident will be assigned to an individual officer, a team, or patrol car.

18.3.3.2 Working on an Incident

The various persons working on the incident will capture more details. These persons are the patrol officers at the crime scene and when a more serious crime took place, the detectives and investigators. All information stored at the crime scene will be in real time available to all users who have access (rights) to this incident. The incident page offers functionality to store all sorts of detailed

information related to the incident. A series of tab pages, shown in the middle part of the incident page in Figure 18.17, offer the means to enter information such as the following:

- Offenses
- Person information
- Relations between persons (offender-person, victim-persons), incident witnesses, suspects, and contacts
- Physical related material such as property, evidence, and attachments
- Activities
- Relations with leads and cases

Additionally, it allows you to create a detailed profile of a potential offender whose real identity is not known. This subject profile can then be matched with the actual contact record once the identity is discovered

18.3.3.3 Offence Management

Offence management provides the capability to track and cross reference criminal offences as part of the incident. Offences can be logged as part of the incident and used for reporting purposes and further analysis. The tab page offenses on the incident page provide the offense management functionality as shown in Figure 18.18.

Data about the offenders, victims, force used, and assets can be captured quickly and easily. Officers can follow up on the offence by adding notes or related documents to the record to provide a comprehensive picture of the offence. All of the necessary information for reporting, tracking, and managing offences is easily entered and readily available to officers. The offence management capability enables officers to track alleged offenders and associated offences that arise as part of an incident and record associated arrests, manage property, and file evidence. Officers can capture structured information about the circumstances surrounding the incident using specific categories such as suspicious death, serious assault, robbery, or sexual assault, along

and the second	Courte Sane Courte C	Ervel	Same							
cident Summery *	Dead person found in his home d	Aring surveillance			Incident Type:	Violent Crime ·	Source Name			
Acident Number #	1-35854	Date Created:	4/7/2014-04-42-55-91		Incadent Status	Active •	Seurce Type:	Surveillance		
Assessment		Created By	ACENTIN		Incident Dranthy	Non .	Source Crossilution			
Data Damana		Reds Records	ACCOUNTS ON AN AN AN AN				Part Barrow	the state of the s		
Date Occurrent	empere se es so rei El	Liate Haboose	erm2014-04-46.55.848		rodent severty	ngs •	Case Here:	Deep of nearing man	NP NI	
Location:	Herengracht 1217 ID	Location:	After lead, dead body discovered in premiser	ant -	Owned By #	AGENTIE 90				
Offense flan	e Officere Cade	Natur	Offense Name Offense Catego Class A. Horscide		Location Ty	cx Offense			X 21-34 d 27	Dest
T)	11537	1000	AN A		10	Offense Co	da	Offense Category	Offense Name	
					- M	urder	Group A-Crime Agenst	Person Ma	rther	
					0	8	Class A Homode	Tie .	gligent Manalaughter	
					10	15	Group A Crime Against	Person Ne	gigent Manulaughter	
					0	bacene Material	Group A-Crime Against	Person Ot	iscene Material	_
					0	bscene Malerial 18 de sectod	Group A-Crime Against Group A-Crimes Against	Person Ot If Seciety Op	ecene Naterial enating/Promoting Gambling	
tore Info					03	bocene Material 18 ck.pocket 19	Group A-Crime Against Group A-Crimes Against Group A-Crime Against Group A-Crimes Against	Person Ot I Seciety Op Person Po I Seciety Po	scene litaterial xenting/Prombing Gambling 3-peciet moorashy/Clacena Material	1.41
tore into					0 3 9 3	bscene Material 18 ck-pocket 10 14	Group A-Crime Against Group A-Crimes Against Group A-Crime Against Group A-Crimes Against Group A-Crimes Against	Person Ot it Seciety Op Person Pic it Seciety Per it Seciety Per	socete Material smatting Promoting Cambling A geolet mography/Obscena Material eatibution	1 47 1
tore Info	1 2666 Quer				0 21 23 24 24 24 24 24 24 24 24 24 24 24 24 24	bacene Material 18 ck-pocket 19 Ma 498	Group A-Crime Against Group A-Crimes Against Group A-Crime Against Group A-Crimes Against Group A-Crimes Against Group A-Crimes Against Group A-Crime Against	Person Di If Secarly Di Person Pic If Secarly Pic If Secarly Pic Person Ra	scene Naterial enating/Promoting Gambling %-occlief mography/Obscene Material estibution pp	1 41 1
tore Info	• Sudar Darr	Watter	dian Type		0 3 3 3 4 3 4 8 1	bscene Material 8 di-pociet 70 July 44 ape 80	Group A-Crime Against Group A-Crimes Against Group A-Crimes Against Group A-Crimes Against Group A-Crimes Against Group A-Crime Against	Person Ot If Seciety Op Person Pic If Seciety Pic If Seciety Pic Person Sa Person Sa	acene Vateral exekting/Fronzing Gambing Skipecidet mography/Clascene Material mathemis pe mje Aasaut	1 47 1
tore Info Crisse Name Offense Co	* Norder 10 54 Norder 10	Netiva	dina Type Sub-Type		0 3 4 8 •	bacene Material 8 61. poolett 10 14. 14. 18	Group A-Crime Against Group A-Crimes Against Group A-Crimes Against Group A-Crimes Against Group A-Crimes Against Group A-Crime Against Group A-Crime Against	Person Ot I Society Op Person Po I Society Po I Society Po Person Ra Person Se	scene Vatoreli enating/Hometrig Canteling scenetel mography/Oliscens Material natitution patie Material patients patient	1 art
lore Info	* Marder 19 fe Marder 19 fe Geselded	Nutrie	ellon Type Seb-Type		0 3 8 5 4 8 5 1 1 1 5	boome Material 8 6 6 6 70 70 8 8 8 8 8 8 8	Group A-Crime Against Group A-Crime Against Group A-Crime Against Group A-Crime Against Group A-Crime Against Group A-Crime Against Group A-Crime Against	Person Ot I Seckely Op Person Per I Seckely Po I Seckely Pr Person Ra Person Se H H	scene Malereal exemplifyceneting Ganeting Apeciel mography/Clascene Malerial natiotion gel migra Assault s	1.011

Figure 18.18 Incident offense management.

with descriptive text, to allow detailed information to be captured, and yet still provide for ease in reporting and analyzing crime trends. Additionally, offence management allows officers to track the use of drugs or alcohol related to the offence and record any use of force during the offence, including the use of any weapons.

18.3.3.4 Coercive Measures

Coercive measures functionality is used to ensure that the criminal investigation procedure is not disturbed and prevent the continuance of criminal activity. This functionality is used in both incident and case management. Examples of coercive measures are arrest, detention, seizure, and search of premises and covert coercive measures are telecommunications interception and technical surveillance. The decision to detain a suspect is always made by a competent court of law at the request of the police. Nobody may be detained if this would be unreasonable on account of the nature of the case. The coercive measures management can be found on the incident page as part of the tab page measures as shown in Figure 18.19.

Coercive measures give the police the power to apprehend and arrest suspects but also to submit their cases to a prosecutor for decision. Coercive measures may be used only if this is justifiable in view of the severity of the offence under investigation, the importance of solving the case, and the infringement of the rights of the suspect or others that the usage of coercive measures would entail.

18.3.3.5 Incident Overview

A last feature we would like to mention is the *More Info* feature. The feature shows all relevant information of the incident that a user would like to see in one view. This view is also known as the incident 360° overview, as part of the *More Info* tab page shown in Figure 18.20.

		Delete	bave Create	Case Cancel	Juery				
Incident Su	ummary:*	Dead person	found in his home	during surveillance			Incident	Type: Violent Crime	•
Incident N	Number:*	1-3SKS4		Date Created:	4/7/2014 04:48:55	PM 🗂	Incident St	atus: Active	•
Ass	sessment		C	Created By:	AGENT16		Incident Pri	ority: High	٠
Date 0	Occurred:	4/7/2014 04:4	48:55 PM 🗂	Date Reported:	4/7/2014 04:48:55	PM 🗂	Incident Sev	erity: High	•
	Location:	Herengracht	1217 🕥	Descriptive Location:	After lead, dead b discovered in orei	ody mises of;	Owned	By:* AGENT16	বা
iore Info	Offenses	Allegations	Offender-Persor	15 Victim-Persons	Measures	ident Witnesse	s Suspects	Property Evider	nce Cases
Menu	- Add	Delete	Query Searc	ħ	Type of Measu	re 💌			
	Type	of Measure	Last Name	e First Name	Gender		Measure Date	Location	Measure S
	emises	Ni	imwegen, van	Henry	M	4/6/2014 02:00	0:00 PM	Herengracht 1217	Completed

Figure 18.19 Coercive measures.

	Circl II	Stat.					
Dead person found in his hi	ine during surveillance	1	Incident Type	Notent Crime +	Source Name		1
1-35K54	Date Created	477/2014 04:48:55 PM	Incident Status: 4	-clive •	Source Type	Surveilance .	
	Created By:	AGENTIN	Incident Priority: H	ign •	Source Organization		
4/7/2014 04:48:55 PM	5 Date Reported	4/7/2014 94:48:55 PM	Incident Severity: It	lah •	Case Nors	Dead of healing master 18	1
Herengrachi 1217 3	Descriptive Location	After lead, deed body discovered in memory of	Ownad By 🕷 🔺	GENTIN NT			
Allegations Difference Pe	same Video Persone	Beauty Reduct Witness	e Bolgesta Proper	ty Enidence Cases	Notes Activities La	ds Matheens Groups	Cinumitations Butgists Contacts
							Pricette Int - 1 of te
1-36KS4	Date Created	e 4/7/2014 04 48 55 PM	1 Sub-Statu	6: In-Progress	Source Type	Surveillance •	
Dead person found in his	home during aurveillance		Incident Statu	s: Adlve	Source Name:	R	
Violenti Crime	Created By	AGENITIE	Incident Priorit	y mph	Source	10	
Herengracht 1217	30 Date Reported	4/7/2014 04:45:55 PM	1 Incident Severa	y High	Relability		
4/7/2014 04:45:55 PM	11 Heda rderes	10	Related Incident Nam		Keywords		
	NT Screening Decisi	ant l	Related Incider	*	Dwned By:*	AGENTIS BE	
	Offender Johnson	antion		10 J.			
	and dat more						
	St. Offender Name	5	0 Offender Statu	5			
menry Ninwegen, ver		1 A 1 A 1 A 1 A 1 A 1 A 1 A 1 A 1 A 1 A					
i lienty timwegen, van	Group Here		Group Statu	1			
	1-35534 4772014 54-4555 PM 4772014 54-4555 PM 4772014 54-4555 PM 4772014 54-555 PM 4772014 54-4555 PM 4772014 54-4555 PM	Lobest Lobest	L-35534 Die Creater APC/01/4 64 85 59 M Die Creater Die Creater APC/01/4 64 85 59 M Die Creater Die Creater APC/01/4 64 85 59 M Die Creater Die Creater	1-35524 Date Created: #770214 54:45579 Notert Stanz, # 1-35524 Date Created: #770214 54:45579 Notert Stanz, # 4770214 54:45579 Date Stanz, # A7001 54:45579 Notert Stanz, # 4770214 54:45579 Date Stanz, # A7001 54:45579 Notert Stanz, # 100000 Date Created by ADDITIS Notert Stanz, # Notert Stanz, # 100000 Date Created by ADDITIS Notert Stanz, # Notert Stanz, # 100000 Date Created by ADDITIS Notert Stanz, # Notert Stanz, # 100000 Date Created by ADDITIS Notert Stanz, # Notert Stanz, # 100000 Date Created by ADDITIS Notert Stanz, # Notert Stanz, # 100000 Date Created by ADDITIS Notert Stanz, # Notert Stanz, # 100000 Date Created by ADDITIS Notert Stanz, # Notert Stanz, # 100000 Date Created by ADDITIS Notert Stanz, # Notert Stanz, # 100000 Date Created by ADDITIS Notert Stanz, # Notert Stanz, # 100000 Date Created by ADDITIS Notert Stanz, # Notert Stanz, # 100000 Date Created by ADDITIS Notert Stanz, # Notert Stanz, # 100000 Date Report Notert Stanz, # Notert Stanz, # 10000	1-33534 Date Created 47/0214 64:435 FN In oddert Status, Addie + 1-33534 Date Created 47/0214 64:435 FN In oddert Status, Addie + 1/2024 64:455 FN Date Status, 47/0214 64:435 FN In oddert Status, Fordy, Statu	1-3554 Date Created, Ar7014 54-455 PM Nodert Stanz, Active Active Source Type 1-35524 Date Created, Ar7014 54-455 PM Nodert Stanz, Active Holdert Stanz, Active Source Type 4/70214 54-455 PM Date Stanze, Ar7014 54-455 PM Nodert Stanz, Active Modert Stanz, Active Casa Stanz 4/70214 54-455 PM Date Stanze, Ar7014 54-455 PM Nodert Stanz, Active Ownal By# Active Casa Stanz 1 Descriptive: Anti-stance in reasons at an intervent intervent intervent in reasons at an intervent i	1-355/4 Date Created (no.2014) 64-61.55 PM Nodert Stona, Active • Source Type, Surveitance • 10 Created by ADDITIS Nodert Stona, Active • Source Type, Surveitance • 4/7/2014 64-6155 PM Date Stopane, Andread • Nodert Stona, Active • Source Type, Surveitance • 10 Descriptive, Andread • ADDITIS Nodert Stona, Active • Source Type, Surveitance • 11 Descriptive, Andread • Modert Stona, Active • Source Type, Surveitance • 11 Descriptive, Andread • Modert Stona, Active • Nodert Stona, Active • 12 Descriptive, Andread • Modert Stona, Active • Nodert Stona, Active • 12 Descriptive, Andread • Modert Stona, Active • Nodert Stona, Active • 13 Descriptive, Andread • Modert Stona, Active • Source Type, Surveitance • 13 Descriptive, Andread • Modert Stona, Active • Source Type, Surveitance • 13 Descriptive, Active • Source Type, Surveitance • Source Type, Surveitance • 14 Descriptive, Active • Source Type, Surveitance • Source Type, Surveitance • 15 Descriptive, Active • Source Type, Surveitance • Source Type, Surveitance • 15 Descriptive, Active • Source Type, Surveitance • Source Type, Surveitance •

Figure 18.20 Incident 360° overview.

18.3.4 Recording a Case

We started the police investigation example with the creation of a lead. The discovery of the dead body of Henry van Nimwegen leads to the creation of an incident. From the incident a case can be created, as is shown in Figure 18.21. A case deals with the investigation related to a crime that has been committed.

The Siebel *law enforcement and policing operations* provides a secure electronic case file that is used by officers to manage a crime investigation. It provides a team-based investigation for the case and allows the various tasks to be assigned to the members of the team. A security model ensures that only specified officers, detective, and investigators can access a case. The investigation case



Figure 18.21 Creating a case from an incident.

enal • Reation •	Approval Template: Team		-							
ikation •	Team:		NI 68	Ne Seriel R. 7	NY-CRIC-1-6372109	Sub-Status	Under Investigation			
		OFFICER	80	Category 0	Crime Investigative	Date Opened #	4/7/2014 05:00:331	N 0		
	Last Name	van Ninwagan.	82	FistName	lake	Date Closed				
nderd •	Country		•	Created By:	AGENTIS	Early Referrals	10			
	Info Changed	12	Class	sficator.*	Unclassified	Late Referrals	10			
stgetor +	Measure Period:			investor: E	5	Centact Country	listeriards.			
and defension of the second se										
to make an and the state of	ame Bocial Security	Num Home Phone	Wark Phone	Addes	e Bieth Date	Clisenship	Gender	Email Address	Type	Contact Met
se nueve landate s	ame Bocial Security	Sum Home Phone	Work Phone	Addrey	re Bieth Cale	Citizenskip Netherlanda	Gernher M	Email Address	Type	Contact Met
A DECEMBER OF THE OWNER OWNER OWNER OWNER OWNER OWNER	Reator • Agastar • Areas Aggenerate A	SE Prin Charged Egalitati • Measure Perceit same Ageneret Assistant Agenty Plane me County County Search	Bit Changes Bit Chang	No Dangat Cas Second Parist Monaur Puriot Cas Monaur Puriot Cas Monaur Puriot Cas Monaur Puriot Cas Monaur Monaur	Nith Changed Classification * Isplat Wessure Forest Ith control fore Internet Advity Toria Statutants Evidence Toria Internet Classification Evidence Toria Toria Toria Internet Classification Evidence Toria Toria Toria	Stro Calegos Ho Calegos Caleg	Si Mr Casept Casefrador Undesrie Undes	No Changed Changed	Sto Calego	Str. Calegorie Calego

Figure 18.22 The case overview.

is linked to all of the necessary objects to hold the full set of crime information in one integrated place, thus, making it quicker and easier for officers to find and share information. Figure 18.22 shows an example of the case for the police investigation example we use in this chapter.

In the following we explain some standard capabilities around the case handling, the information held as part of the case file, assignment of work, case serialization, evidence management, involved persons, and the report out.

18.3.4.1 Information Held as Part of the Case File

Similar to the incident management, the case page offers functionality to store all sorts of detailed information related to the case. Tab pages, as shown in the middle part of the case overview in Figure 18.22, provide different pages to enter case related information, such as the following:

- Collected case information
- Interviews, assets, quotes, evidence, property, notes, assessments, and circumstances
- Activities
- Plans, activity plans, arrests, assessments, and approvals
- Persons
- Involved organizations, case team members, identifies, and identification Info
- Related information
- Leads and incidents

18.3.4.2 Assignment of Cases and Ownership

By default, cases are assigned to the user who creates the case. Cases can be assigned to one or more employees. Assignment of cases can occur automatically, based on geographic location, availability, or other criteria. Administrators can set up assignment rules based on various criteria of the case record. Case ownership can also be adjusted manually by adding employees to the case team. The police investigation is strictly guided by laws and needs strict approval functionality for instance with arrests and coercive measures. The application offers dotted line, or

Menu 🔻 Nev	Dele	te Save (Cancel	Quer	Y Check Sp	elling Regene	rati
Case Name:*	Dead of	healing master					
Case Type:	Criminal	8	•	App	roval Template:		
Sub-Type:	Applicat	ion	•		Team:	OFFICER	
SSN:					Last Name:	van Nimweger	1
Priority:	Standar	d	•		Country:		
Assessment:		[g		Info Changed:			
Organization:	Investigator		٠	 Measure Period: 			
More Info Interview	Measur	es Approvers	Act	ivities	Activity Plans	Attachments	
Menu 🔹 🛛 Ad	d Dele	te Query	Search		1	Position	_
Position		Last Name		First	lame Sec	quence Number	
		lara	Bill		2		
IPP Case Manager	[q.] h	lote	L'III		-		

Figure 18.23 Case approval overview.

cross-department, approvers to model organizational approval workflows. A case approver list can be established to obtain approvals from various people from different departments. For example, both the case supervisor and the finance manager can be designated to approve before a payment will be issued. Additionally, it allows substituting or additional approvers to be added to a case as needed. A case agent, with the correct assigned approval roles, can manually add an approver to the list in addition to the predefined approvers already in the system. Figure 18.23 shows an example of an approval list for our police investigation example.

18.3.4.3 Case Serialization

Case serialization provides the ability to standardize case numbering across the agency by generating unique, custom-defined serial numbers for cases and related leads and evidence. When a case record is created in the solution database, a unique serial number is automatically generated and stamped on the record, as shown in Figure 18.24.

In many agencies, the case numbering system is a standard employed to track case files and all related material. Using case serialization, administrators can set the case serial number to include a combination of multiple fields so that the composite number is meaningful to an investigator. For example, case serial numbers can be generated from a combination of fields unique to the case and combined with either a local counter, based on the number of cases in the system, or a universal counter that numbers every record individually. When related records are created, such as leads or evidence that is related to the case, serial numbers will automatically be created in the proper format, utilizing the unique numbering methodology defined by the agency.

Menu - Nev	Delete Save Cancel	Query Check Spe	ling Regenerate All S.	Recall Validation	
Case Name:*	Dead of healing master			Case Number:*	1-6372189
Case Type:	Criminal	Approval Template:		Case Serial #:	NY-CRIC-1-6372189
Sub-Type:	Application •	Team.	OFFICER	Category:	Crime Investigative •
SSN:		Last Name:	van Nimwegen	First Name:	Julius
Priority:	Standard •	Country:		Created By:	AGENT16
Assessment:	(M)	Info Changed:	E	Classification:*	Unclassified •
Organization:	Investigator •	Measure Period:		invest Err.	
More Info Interview	Measures Approvers A	ctivities Activity Plans	Attachments Evidence	Property Contacts	Leads Notes Calendar
Merce .	W Delete Query Sean	ch	Evidence Number	Regenerate All	
11010		the second se			

Figure 18.24 Case serialization.

18.3.4.4 Evidence Management

Evidence management enables agencies to track and manage evidence associated with the cases they are managing. Evidence management supports a two-tier evidence model, on a single case level and across multiple cases. This model enables agencies to track unique physical items in addition to managing these items as evidence in the context of one or more cases. It provides the capability to consolidate evidentiary items across an agency, yet still manage evidence in the context of each individual case. During an investigation, evidence collected at a crime scene, such as a weapon, can be logged as a unique item in the application, and later linked as evidence in multiple cases. Figure 18.25 shows an example of evidence, a bullet found at the crime scene, logged as part of our police investigation example.

								i tatte
New Y	Telefe See Cause	Querry						
Future lane 1	D-IN	And a second		Future Top	Weather	Future Damer	Contrast Int	
Publisher Humber #	1.262201	Case Name #	Tool front incentioning	an Evidence Status	Tubettat .	Louised Ba	and	
Evitence Same	Annemarket DIL 1.777	Case Service	Annamarian A. Pat. 1, 2222	Tustance Darast		Date Langed		
Property ID			and all and a second second					
1.538736754								
e trilo itomures	Actualizes Activity Plana a	Bachments Sudir frail	Contants Ormage	Suspects Holes	Related Evidence Service Re-	aunts SubEvidence		
								4 f af f+
Property and	Contraction of Contra	and the second s						
Transfer I and	Carter Carter	Disea.						
odence information				Description				
Evidence liame:*	Doll	Evidence Type:	Weapons	Description				
Evidence Name:* Created By:	SADARN	Evidence Type Evidence Subtype	Vieapons Framma	Description				
Evidence liane.* Created by: Evidence Status.*	SADAN Salandad	Evidence Type: Evidence Subtype: Evidence Drug Type:	Vvecons Feseres	Descripter				
Evidence liane.* Created By: Evidence Status.* Evidence Substatus:	ELAN SADANA Salavalted	Evidence Type Evidence Sobtype Evidence Drug Type Extended Quantity	Vergens Freema	Description				
Evidence Name.* Created By Evidence Status.* Evidence Substatus Location:	Eule SADAtte Subwithed • RD	Evidence Type Evidence Subtype Evidence Drug Type Extended Quartity Estimated Value	Vespens Fromma	Description				
Evidence Name * Created By: Evidence Status * Evidence Substatus Location: End Date:	Evel SADARI Subvitted • RI D	Evidence Type Evidence Subtype Evidence Drug Type Extended Quartity Estimated Value Receipt M	Vespens Preams	Description				
Evidence Name.* Created By: Evidence Subanture Location: End Date: Shert Date:	SADAN SADAN Submitted • RC RC RC	Evidence Type Evidence Soltype Evidence Drug Type Estimated Quartity Estimated Value Receipt bf. Unit of Measure	Preme	Description				
Evidence Name.* Created By: Evidence Status.* Evidence Substatus: End Date: Start Date: uses Information	SADAN SADAN Submitted • NC D	Evidence Type Evidence Soltype Exidence Drug Type Estimated Quertity Estimated Value Receipt II: Unit of Heasure: Accel Information	VVecuns Freems	Description				
Evidence Name * Created By: Evidence Substance Evidence Substance Evidence Substance Evid Date Start Date unte Information	SADIM SADIM Subridhed Sponfaart Investigate Sponfaart Investigate	Evidence Type Evidence Subtype Evidence Drug Type Estimated Quartity Estimated Value Receipt M: Unit of Measure: Agust Information Agust Product.	Western	Description		Property Type		
Evidence lianes * Created By Evidence Subatrue Location End Date: Start Date: anie Information Case lianes * Case Sianes	SADURS Salantikal	Endence Softype Evidence Softype Exidence Drug Type Extended Quarity: Estimated Value Receipt M Unit of Dessure: Accel Information Accel Product: Accel Product:	Wequess Freema	Description		Property Type Entimates Cauranty		
Evidence Hanne ** Created By Evidence Salatas ** Location End Date: Shart Date: ante Information Case Hanne ** Case Stante:	Solar Boold SADURE SolardBad • • • • • • • • • • • • • • • • • • •	Endence Subyee Evalence Subyee Evalence Duig Type: Estimated Owney Receipt M: Unit of Uneasore Asset Industry Asset Description Asset Description Asset Description		Description Description Asset Parent 8		Property Type Estimates Guardity Unit of Meesure		
Evidence liaens.* Created By Evidence Salatata Location End Date Start Date and Information Case Ilans Case Ilans Case Nature Case Nature Case Nature	Bull SADDRS Sabritest Sabritest Sabritest Sgonfloart Investigate Active 1.22226 Augumentand DU-1.22228	Evidence Type Evidence Sathype Evidence Dustype Extended Quartity Extended Quartity Unit of Neasore Accel Information Asset Peter Asset Descriptor Asset Descriptor Asset Reter	VV980499 Freeme	Description Description Asset Parent # Asset Parent #		Property Type Estimated Guardity Unit of Messarr Estimated Value		
Evidence liaine.* Created By: Dividence Substative Evidence Substative End Date: Start Date: Start Date: Case Name:* Case Name:* Case Name: Case Senia Case Senia Case Senia Case Senia	bull SJ2009 Salashtal • • Riji Significat Investgate 10 Active 122205 Active S2010	Evidence Type Evidence Suttype Evidence Dutype Estenated Guarty Estenated Value Receipt In Und of Vessore Agent Information Asset Product Asset Product Asset Description Asset Transit Asset Transit		Description Asset Parent # Asset Owner		Property Type Estimated Guardity Unit of Measure Estimated Value		

Figure 18.25 Evidence management in the context of a case.

For each evidence record, different activities and documents can be tracked, in accordance with the specific requirements of each case. The application enables agencies to manage the history of activities or actions specific to each case and each piece of evidence. All activities associated with a piece of evidence, such as entry into a property store, submission for forensic tests, or transfer to court during a trial, are securely logged against the evidence. This ensures that its history is known and therefore, the admissibility of the evidence is not compromised.

Agencies can also establish start and end dates for each evidence item and serialize evidence as part of a case and manage the disposal process once the items of evidence no longer need to be held.

18.3.4.5 Involved Persons

A person within a case (or incident) can be a suspect, a victim, an offender, a witness, but also a lawyer or a family member. Within the police investigation, it is important to store all relevant information of a person and visualize it based on the "role" of the person. The person information is available in the *Identification Info* tab page in the case overview, as shown in Figure 18.26.

18.3.4.6 Report Out

When the case is solved and needs to be handed over to the prosecutor, a report out needs to be created of the case content. It is possible to produce a report using different templates, customize the name of the report, change the type (PDF, Doc, RTF, mail or fax) or use other languages (Figure 18.27).



Figure 18.26 Involved persons in the case.


Figure 18.27 Report out of the case.

18.4 Conclusion

In this chapter we have provided a tutorial for a Siebel-based case management solution on two aspects, the Siebel implementation approach and the usage of the application. The Siebel implementation approach contains four phases:

- Choosing the Siebel functionality
- Familiarization
- Design
- Implementation

We aligned these phases with the case management solution framework (CMSF) as presented earlier in Section II.

The CMSF supports the Siebel approach as follows:

- The first step in the Siebel implementation approach is to determine which Siebel functionality is going to be used as the basis for the solution. The CMSF case management classification supports the determination of the Siebel industry solution as required for the solution.
- In the familiarization (PWT workshops) and the design (CRP workshops) the functionality required for the solution is determined. The CMSF case management design categories and components support the determination of the required functionality.

In the second part we looked at the usage of an implementation of the Siebel law enforcement and policing operations solution. In this example we used a detective story to show how a Siebel application in real life is used. The basis for the Siebel law enforcement and policing operations solution is the lead, incident, and case functionality. We provided examples for all three functionalities in the police investigation example provided in the detective story.

Chapter 19

Concluding Remarks

Dear reader, if you are here after having gone through most of the chapters of this book in some detail, then congratulations for having completed the journey and thanks for your time and interest. We, the authors, have put in significant effort and burnt plenty of midnight oil to put together the content that benefits from our extensive work in this and related areas as well as from a lot of specific research that we undertook in support of this book. In spite of all the hard work that goes into writing a book like this, we thoroughly enjoyed the process; we do sincerely hope that you enjoyed reading the book and found the content useful.

When you reflect on the essentials of case management, especially the adaptive version, you will find that this approach of organizing and executing business activities is possibly the closest we can get to the way we, the humans, solve nontrivial problems. Most of human actions are event driven and so is case management. When we are in trouble trying to solve a difficult problem we naturally consult others who may help us or we seek alternate approaches to attack the problem; you have seen in the solution architecture and implementation chapters that case management applications facilitate collaboration among stakeholders and allow new activities or participants to be added during the case progression. We make decisions based on our knowledge and experience keeping within applicable norms of interaction and behavior; these are also the guiding principles of case management application architecture and usage. We are usually goal oriented in a problem-solving situation and consider the details of the fine-grained actions secondary to finding the final solution; goal orientation is also a fundamental aspect of case management. Since case management can handle knowledge work needed to solve difficult business problems not amenable to traditional business process management (BPM), and since the supporting technologies necessary to design and implement case management are available at a high level maturity, it is not a surprise that there is currently a high level of interest in case management; a quick search on the Internet will rapidly establish this observation. Software vendors have, of course, taken this growing interest seriously. While early stage solutions for case management technology platforms were usually from smaller and niche vendors, in recent years larger ISVs such as Oracle have made case management technology one of their key offerings; many System Integrator (SI) firms like eProseed and Capgemini now include case management consulting as part of their core competency; these trends are further pushing broader adoption of case management (Figure 19.1).



Figure 19.1 Expected evolution of traditional BPM and ACM into Generalized BPM.

So, what does future hold for case management? Besides deeper and wider adoption across industries, we expect to see the following:

- 1. A stronger movement toward hybrid solutions, that is, combination of traditional BPM and case management in the same solution balancing efficacy and efficiency; we termed this as "Generalized BPM" approach in this book
- 2. Integrated tooling support from independent software vendors (ISVs) for Generalized BPM solutions, as Oracle has already done in their BPM suite product, and many others are starting to offer similar capabilities
- 3. Tighter integration of content and portal technologies with case management technology platforms for still better handling of structured and unstructured content and for facilitating easier and more productive human interactions
- 4. Inclusion of more analytics of all varieties in case management solutions such as historic trends and real-time and predictive analytics that could provide timely and helpful insights to knowledge workers trying to resolve cases
- 5. More progress toward simplified yet powerful case worker interfaces that mimic natural description of tasks and team interactions
- 6. Better harnessing of knowledge from records of cases solved that could be packaged as assisting modules to inspire and guide knowledge workers.

An interesting thing to note here is perhaps the emerging *battle* between the proponents of adaptive case management (ACM) and BPM. A few years ago there were predictions by the pundits that the market size of ACM will be significantly larger than that of BPM, some going as far as stating that it will be an order of magnitude larger. Also, some of these industry watchers had claimed that ACM will subsume BPM. ACM will provide additional capabilities for managing business activities, thus, substantially increasing the proportion of all business activities that would be digitized—we do not, however, believe that there is enough evidence yet to reasonably predict if ACM market would be order(s) of magnitude larger than BPM or that it will engulf the traditional BPM discipline. ACM solution construction, as we have explained in several ways so far, is event driven. It is technically possible to recreate a process graph (as in traditional BPM) using appropriate event generation and event handling. In other words, it is possible to use ACM technology platform to replicate traditional BPM processes—we do not think this is necessarily a smart idea since wherever traditional BPM processes do a good job, there is no real benefit of introducing ACM. Lately, we are noticing suggestions that BPM will absorb ACM—we do not think this is a useful debate either since ACM characteristics have many differences when compared to traditional BPM. Of course, we could extend the definition of what BPM encompasses, for example, as we have done in defining "Generalized BPM" that essentially combines traditional BPM and ACM leveraging the commonalities while recognizing the difference. And, yes, we do predict that something like Generalized BPM will subsume traditional BPM and ACM in forthcoming years.

As we have remarked earlier in the book, a large portion of business activities contain elements of unpredictability and much of high-value work require nonroutine knowledge work—case management's goal is to deliver solutions to these types of business activities. Hence, it is safe to assume that the popularity of case management will continue to rise for a while and it will eventually become a mainstream strategy for solving business problems. Along with you, dear reader, we are eagerly waiting to participate in this evolution.

We provided two tutorials in this part to show how a case management solution is designed, implemented, and used with the products Oracle BPM/ACM and Oracle Siebel.

In the Oracle BPM/ACM tutorial, we introduced a mind-mapping technique to identify the different activities part of the case. Based upon a credit card charge dispute scenario we showed how to create and test a BPM/ACM application in JDeveloper.

In the Oracle Siebel tutorial we provided an alignment between the case management solution framework and the Siebel implementation approach. In the second part of this tutorial we showed the usage characteristics of the Siebel law enforcement and policing operations solution. We used a police investigation to show the main police supporting functionalities *lead*, *incident*, and *case* are used.

Appendix A: CORA Model Capabilities*

Layer	Cluster	Capability	Description
Channel access	Ultra thin client	Ultra thin client	Capability to access an application by only delivering presentation data to the client and sending user actions (i.e., server-based computing)
	Thin client	Web browser (HTML)	Capability to access an application through a web browser using plain HTML
	Thin client	Mobile device	Capability to access an application installed on a mobile device. (apps)
	Rich client	Web browser (RIA)	Capability to access an application through a web browser providing a desktop-like user interface (rich Internet application)
	Rich client	Workstation	Capability to access an application through a client installed on a work station
	_	Electronic channel	Capability to provide inbound and outbound electronic data exchange from and texternal actors (B2B and B2C)
Presentation	_	User interface	Capability to validate user input and render and format presentation data output
		User interface process	Capability to orchestrate user interactions in a one-time execution span
		User interface integration	Capability to integrate different user interfaces in one view

^{*} The definition of the layers and capabilities described in this chapter are obtained from the CORA model (www.coramodel.com).

Layer	Cluster	Capability	Description
Composition	Orchestration	Human to human	Capability to collaboratively use the same resource, which can consist of multiple actors and multiple time execution spans
	Orchestration	Human to application	Capability to allocate tasks tusers and control the required routing of these tasks
	Orchestration	Application to application	Capability to allocate tasks to applications within an organization and control the required routing of these tasks
	Orchestration	Business to business	Capability to allocate tasks to applications between organizations and control the required routing of these tasks
	Composition	Business component	Capability to compose reuse business logic from one or more IT systems and add (often cross domain) business logic
	Composition	Business entity	Capability to compose business specific cross-domain data objects (i.e., employee, invoice, etc.)
	Composition	Business rules	Capability to compose and execute cross-domain and cross-application business rules
	Composition	Life cycle management	Capability to manage the end-to-end management of all activities in a business process or case
Integration	Synchronous communication	Service mediation	Capability to provide a standard way of describing and calling a function synchronously
	Asynchronous communication	Messaging	Capability to provide a standard way texchange messages asynchronously
	Asynchronous communication	File transfer	Capability to provide a way to exchange (large) messages made up of files asynchronously
	Common	Integration core	Capability to provide core functionality for exchanging data, like transformation, enriching, filter, etc., including capabilities for loading and extracting data and data quality management (e.g., cleansing)

Layer	Cluster	Capability	Description
	Common	Integration metadata	Capability to provide metadata functionality for use during the exchange of data
Application	Make	Make application logic	Capability to provide a set of specific coherent business logic
	Make	Make application entity	Capability to provide (application specific) data objects (i.e., customer and order)
	Buy	Buy application logic	Capability to provide a set of specific coherent business logic
	Buy	Buy application entity	Capability to provide (application specific) data objects (i.e., customer and order)
	Legacy	Legacy application logic	Capability to provide a set of specific coherent business logic
	Legacy	Legacy application entity	Capability to provide (application specific) data objects (i.e., customer and order)
Data	Data access	Data access	Capability to perform operations on data resources, for example, CRUD (as a minimal set)
	Data storage	Master data	Capability to store master data
	Data storage	Transactional data	Capability to store transactional data
	Data storage	Unstructured data	Capability to store unstructured data
	Data storage	Aggregated data	Capability to store aggregated data
	Data storage	Canonical data	Capability to store canonical data
Security	_	Authentication	Capability to check if a user (human or system) is the one it indicates it is by various (combined) means like username and password, chip card, finger print, etc.

Layer	Cluster	Capability	Description
	_	Principal propagation	Capability to forward the user credentials throughout the landscape
	_	Compliancy	Capability to pre- and post-assess the compliancy of activities performed in a system against organizational policies, industry standards, and regulations
	_	Encryption (data at rest)	Capability to encrypt data stored in databases, files, and/or messages
	_	Authorization	Capability to authorize the use of functionality and data, preferably based on role-based access and managed from a life cycle perspective
	_	Single sign-on	Capability to log in one system and then reuse the authentication data to log in into another system automatically
	_	User management	Capability to manage users, groups, authentication details, and authorizations
	_	Logging	Capability to log all kinds of system activities
	_	Archiving	Capability to move data, which is no longer actively used, to separate data storage device for long-term retention
	_	Auditing	Capability to establish whether information systems maintain the integrity of stored and communicated data
IT governance	Service-oriented architecture (SOA) governance	SOA governance	Capability to provide service policies, a service repository, and a service registry. The registry capability facilitates the publication, discovery, and consumption of services. The repository capability provides a broader view that allows tracking of services and maintains its relationships tother assets types

Layer	Cluster	Capability	Description
	_	System management	Capability to manage individual system software assets
	_	System monitoring	Capability to monitor system assets, ranging from database growth to server temperature
	_	Business activity monitoring	Capability to monitor the performance of business processes supported by automation from the level of activities tprocesses
	_	Configuration management	Capability to manage versions of individual software components and documents and their composition intreleases
	_	Change management	Capability to support the submission of a change request, the definition of solutions, change impact analysis, decision process, and implementation, including software transport
	_	Repository management	Capability to store and query metadata, including software versions
	_	Test management	Capability to perform automatic testing
	_	Batch management	Capability to define, schedule, execute, and monitor jobs
	_	Incident management	Capability to restore a normal service operation as quickly as possible and to minimize the impact on business operations

Appendix B: A Solution Characteristic

		"1"	"WI"	"S"				Business				Special Event	Special		
Industry Sector	CM Application	Ţ	Ţ	Т	Automated Tasks	Human Interactions	Integration (EAI, B2B)	Rules/Expert System Help	Regulations/ Statutes	Structured Content	Unstructured Content	Handling/ CEP	Security Considerations	Customer Facing	Communication External
Horizonta	l/Common														
	Employee onboarding	z	z	I	ž	×	¥	z	T	٤	Г	z	-	т	Ч
	Customer onboarding	z	z	I	ž	¥	W	٦	T	Σ	Γ	z	L	т	Т
	AP and AR processing		z	I	Г	¥	M	L	-	т	Γ	z	-	_	Ч
	Facilities management	-	z	I	Γ	т	Γ	z	×	Σ	Γ	z	z	_	Т
	Order/Complaint/ service exception management	_	z	Т	L	т	z	W	L	£	ž	z	z	т	¥
	Proposal and bid management	т	z	_	L	т	Γ	٦	T	т	M	z	z	_	Σ
	Trade and contracts management		z	I	¥	¥	W	٢	Γ	т	Γ	L	¥		¥
	GRC (Compliance tracking and reporting)	I	X	Σ	W	X	н	H	т	т	н	L	ſ	L	W
	Complaints	L	z	I	Т	н	W	W	W	W	Μ	Γ	L	н	Γ
PS + Socia	Il Services														
	Benefits eligibility	V	z	I	W	Μ	М	н	н	Γ	L	z	z	н	L
	Grants management	ν	z	I	W	н	L	W	¥	т	L	z	z	н	Γ
	Tax returns processing	н	z	_	н	L	М	н	н	н	L	z	z	M	Γ
	Unemployment/ Welfare services	ν	z	I	W	н	М	W	M	L	Н	z	z	н	L
															(Continued)

	ion															(par
	Communicat External	-	т	т	т	т	т	т	т		-	Г	Г	X	_	(Continu
	Customer Facing	т	т	т	L	т	т	т	т		т	т	т	н	т	
Special	Security Considerations	z	z	z	Г	z	z	т	н		т	т	т	L	т	
Special Event	Handling/ CEP	z	z	z	т	z	т	т	V		_	_	_	Ţ	¥	
	Unstructured Content	Σ	Σ	Σ	т	т	т	т	т		Σ	Σ	Σ	M		
	Structured Content	т	т	т	-	_	Σ	_	т		т	т	Σ	X	Σ	
	Regulations/ Statutes	т	т	т	т	т	¥	т	т		т	т	т	W	т	
Business	Rules/Expert System Help	Σ	т	т	т	I	¥	т	т		Σ	£	Г	¥	-	
	Integration (EAI, B2B)	_	Г	Σ	-	_	т	т	¥		Σ	Σ	Г	L	Σ	
	Human Interactions	т	т	т	т	т	т	т	т			т	т	т	т	
	Automated Tasks	_	Г	Ч	Σ	_	т	т	т		т	т	Г	F	_	
"S"	Т	-	т	т	Σ	т	т	_	L		т	т	т		Σ	
"WI"	Т	z	z	z	т	Σ	т	т	٤		z		_	т		
"1"	т	т	т	Σ	т	٤	٤	т	т			Σ	т	٤	т	
	CM Application	Information request processing	Patent application processing	Permits & operator licenses	Environmental safety monitoring	Child protection/ Substance Abuser Help/HIV Advisory	Disaster relief	Terror/Epidemic threat handling	Adverse incident tracking in clinical trials	ervices	New account opening	Loan origination	Investor servicing	Dispute resolution	Wealth management	
	Industry Sector									Financial Se						

	Market Ma	000000 W H H H H H H 000000 H H H H H H	000000 W H H H H H H 000000 H H H H H H	00000 X	Markowski H	00000 X <th></th> <th>$\begin{array}{c ccccccccccccccccccccccccccccccccccc$</th>		$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
I X X	I I Z Z	I I I Z Z L Z	т т т т д д д д			и и да и		
	3 3 4							
з х								
т т	т т т т	Ι Ι Ι Σ	Ι Ι Ι Σ Ι Ι Ι Ι	т т т т х т Σ Σ	Ι Ι Ι Ι Ι Ι Ι	Ι Ι Ι Ι Ι Ι Ι Ι	Ι Ι Ι Ι Ι Ι Ι Ι	I I I I I I I I I I I I I I
×	Σ I	ΣΙΙ						
Σ	L Z	Z L Z	<u>z</u> – z	Σ L Σ	<u></u>			3 3 3 3 3 3
:	: т	: т т	: I I	: I I Z	: I I Σ -	Ξ - Ξ	z - z	3 ² ² ² ² ² ² ² ²
	Г			× ×	I Z L L	Z I Z	Z I Z	Z I Z
	ΣΣ	Σ <u>Σ</u> –	L Z Z	Z – – Z Z	I Z Z - Z I	I I Z L Z	I I Z - Z Z	I I Z Z - Z Z
	z –	Ζ - Σ	Z - Z I	Z - <u>Z</u> I Z	Σ - Σ Ι Ζ Ζ	Ζ - Σ Ι Ζ Ζ Ζ	Ζ - Σ Ι Ζ Ζ Ζ	Z -
	тт	<u> </u>	I I I Ž	Σ Ι Ι Ι Σ Ι	I I I Z			
enonganean	M&A services Stock market listing	M&A services Stock market listing Fraud management	M&A services M&A services Stock market listing Fraud management P&C claims processing	M&A services M&A services Stock market listing Fraud management P&C claims processing processing risk management	M&A services M&A services Stock market listing Fraud management P&C claims P&C claims processing Underwriting & risk management New account opening	M&A services Stock market listing Fraud management P&C claims processing processing risk management New account opening Policy management	M&A services Stock market listing Fraud management P&C claims P&C claims processing Underwriting & risk management New account opening Policy management	M&A services M&A services Stock market listing P&C claims P&C claims P&C claims Processing Underwriting & risk management New account opening Policy management
			lusurance	lusurance			Healthcare	Healthcare

Hull B3B System Help Statutes Content Content Content Content Content Content Content External M 1 M 1 H	WI, "I."	WIn "In	WI"		"S"	Automated	Human	Integration	Business Rules/Expert	Regulations/	Structured	Unstructured	Special Event Handling/	<i>Special</i> Security	Customer	Communication
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	CM Application T T T Tasks Interactions (1	T T Tasks Interactions (t	T Tasks Interactions (t	T Tasks Interactions (t	Tasks Interactions (1	Interactions (1	. *	EAI, B2B)	System Help	Statutes	Content	Content	CEP	Considerations	Facing	External
Image: constraint of the state of the st	"Continuum of M H N M H care" coordination H	T Z T S	I Z I	I Z	I	т		т	×	н	т	X	Z	т	т	¥
· ·	Legal contracts L H N M M management	Z Z L L	Σ Σ Ζ Ι	Σ Σ Ζ	W	W		W	L	н	н	Γ	Γ	т	н	Γ
Image: 1 Image: 1 <t< td=""><td>Policy and M N H M M management</td><td>Σ Σ Σ</td><td>× × I Z</td><td>Z Z I</td><td>X</td><td>M</td><td></td><td>¥</td><td>т</td><td>Н</td><td>Σ</td><td></td><td>Γ</td><td>т</td><td>X</td><td>Г</td></t<>	Policy and M N H M M management	Σ Σ Σ	× × I Z	Z Z I	X	M		¥	т	Н	Σ		Γ	т	X	Г
	lice															
	Legal practice L N H H M management, time tracking and billing	Z L	Σ Ι Ι Ζ	х н	Σ I	X		Т	т	Н	т	L	Γ	т	L	Γ
Image: Line state Image: Line state	eDiscovery and H M N H M management	E E E E	E Z Z	Σ Ι Ζ	Σ I	¥		н	н	Н	т	Γ	X	т	L	L
ч ч ч ч ч ч ч ч ч ч ч ч ч ч	Docketing and M N H H H Court proceedings coordination	т т z х	I I Z	н н н	r r	т		z	¥	т	т	L	Ч	т	Γ	Г
	Criminal H H L M H investigation	H H	H L	H V	H	т		¥	W	н	L	т	т	т	L	т
	Mandatory H N H H M information collection (e.g., biometrics)	Σ Ι Ι Ζ Ι	Σ Ι Ι Ζ	х н н	Σ T	M		н	т	Н	L	т	L	т	L	Г

CM Application	L	T	Г. н	Automated Tasks	Human Interactions	Integration (EAI, B2B)	Rules/Expert System Help	Regulations/ Statutes	Structured Content	Unstructured Content	Jpecial Event Handling/ CEP	Security Considerations	Customer Facing	Communication External
tlities		1											-	
Process safety management	٤	т	z	¥	т	ž	т	т	L	Σ	Σ	т	Σ	¥
FERC eTariff (US)	z	z	т	н	z	W	W	н	н	z	z	L	z	W
Project and transmittals management	_	z	т	¥	Σ	Σ	Г	Ч	т	-	L	Γ	£	¥

"I" - Investigative; "IM" - Incident Mgmt; "S" - Service Request or Offer

Index

A

Activity governance, 86, 88 Adaptive case management (ACM), 407-409 agile adaptivity, 49 BPM suite audit trail, 188-189 BPMN 2.0 compliant process, 178 case flow and process design, 180-182 characteristics, 179 cross-functional design category, 189-191 document and media design, 189 event design, 182 features, 180-182 interaction design category, 185-187 product mapping, 183-185 rules design, 182-183 stakeholder, 190 variations and unpredictability, 178 business agility, 16-17 adaptive planning, 41, 43 business capabilities, 38 range and time, 38-39 solution component granularity, 44 solution management complexity, 44 system-theoretic approach, 43 working styles, 44-45 business case business needs and financial benefits, 94-95 cost-benefit-risk, 94 costs, 95-96 dominant forces, 92-93 format, 93 risks, 96-97 CAS (see Complex adaptive systems (CAS)) case activities BPMN (see Business process modelling notation (BPMN)) human task (see Human task-based case activity) implementation, 326, 348-349 properties, 326-327 case elements, 324-326 case management project

CCDApplication, 331 composite selection, 332 editing page, 333 gallery, 331 CCDCase application, 363-365 content configuration, 343-344 credit card charge dispute (see Credit card charge dispute) customer.xsd, 379 data definition, 326-327 data objects customer data, 339-340 data forms, 340-341 definitions, 336 dispute data, 337-338 flex fields, 341-342 identification, 336-337 investigation report, 339 DCM and PCM, 48 characteristics, 67-68 competing design, 62 cost-quality-time, 69 definition, 66 deterministic vs. nondeterministic, 67-69 Google's search engine, 66 Holy Triangle, 61 Iron Triangle, 61 load and deformation characteristics, 65 pure play vendors, 62 purpose case management, 63 quality-cost-time, 63 system classifications, 64-65 traditional ERP, 67 definition, 21-24, 67 deployment and testing case workspace, 369-370 CCDCase composite application, 366 CCDCaseDataForms, 366-367 CSR, 373 dispute details, 373 enable development mode, 365 instance, 367-368 Investigate Charge activity, 374

investigator access permission, 371-372 Notify Customer activity, 373-374 roles administration, 370-371 task form, 367 test users, 365-366 WebLogic Server, 364 documents, 328 event design, 329 external case data, 327-328 interactionism, 47 midsize insurance company, Europe adoption plan, 90-91 BPM model, 89 handling claims, 88 maturity, 91-92 Nile crocodiles and cuttlefish, 48-49 open and adaptive system, 50 vs. PCM collaboration, 73 dimensions, 74-75 event-driven paradigm, 73 features and capabilities, 75 high-quality user interfaces, 76 journey, 70-71 National Defense Executive Reserve Conference, 71 radar chart, 74 record applications, 76 TEI formula, 73 permissions application policies, 345 application roles, 344 definition, 348 labels, 348 properties category, 334 duration, 334 milestones, 334 outcomes, 334 priority, 334 summary, 333 title, 333 rules design activity event actions, 357 Add icon, 360 definition, 356-357 dummy case data, 358 investigate charge activity, 356 milestone events, 361-362 Parameterized option, 359-360 and policies, 328-329 PrepareCase activity, 359-360 review charge dispute, 356, 359-360 user events, 362-363 value set, 358 WriteOffRange, 360

SR-CAS, 50-51 Four-Cs (see Four-Cs principles) hybrid nature, 60-61 KPIs, 61 relevance of elements, 55-56 stakeholders application roles, 345-346 definition, 344-348 OracleBPMProcessRolesApp, 344 program, 328 runtime, 327-328 XPATH expression, 328 target adaptive system, 52 test-startcase-input.xml, 380-381 tutorial files installation, 324-325 usage characteristics, 239, 241 user events, 344 AIIM, see Association for Information and Image Management (AIIM) Application Development Framework (ADF), 186-187, 340-341, 344, 355, 367 Application layer BAM, 228 CORA model cross-functional design areas, 162, 164 design categories and components, 157, 159 high-level mapping, 158-159 information design mapping, 162-163 interaction design, 161-162 lifecycle design, 159-161 Endeca Information Discovery, 234-235 OPA product, 232 Association for Information and Image Management (AIIM), 22

B

BAM, see Business analytic monitoring (BAM) BPMN, see Business process modelling notation (BPMN) BPM suite, see Business process management (BPM) suite Business agility, 38 adaptive planning, 41, 43 business capabilities, 38 range and time, 38-39 solution component granularity, 44 solution management complexity, 44 system-theoretic approach, 43 working styles, 44-45 Business analytic monitoring (BAM) application layer, 228 composition layer, 227-228 dashboard, 228 data layer, 228 implementation, 225, 227 integration layer, 228 presentation layer, 227 technical architecture, 229-230

Business case costs, 94-96 dominant forces, 92-93 format, 93 needs and financial benefits, 94-95 risks, 96-97 Business Process Execution Language (BPEL), 180, 185 Business process management (BPM) suite; see also Adaptive case management (ACM) BPMN 2.0 compliant process, 178 case flow and process design, 180-182 characteristics, 179 cross-functional design, 189-191 event design, 182 features, 180-182 information design, 187-188 audit trail, 188-189 document and media design, 189 interaction design, 185-186 channel design, 187 portal design, 186-187 maturity model architecture, 84, 87 business and strategy, 84 capability domains, 83-84 governance, 88 information, 88 infrastructure, 87 OA&M, 88 organization, 88 projects, portfolios, and services, 88 product mapping, 183-185 rules design, 182-183 stakeholder, 190 variations and unpredictability, 178 Business process modelling notation (BPMN) case activities and properties, 348, 349 creation, 350, 352 credit card rewards dispute resolution, 15, 40-41 data association, 352-353 definition, 353-354 implementation, 348 input type, 350-351 Start and End nodes, 351 Business-to-business (B2B) orchestration, 185

С

Capability maturity model (CMM), 18, 23, 34, 79, 106 BPM (see Business process management (BPM) suite) CMMi, 80 generalized BPM, 80–81 maturity levels, 81–82 SOA, 80 Capability maturity model integration (CMMi), 80–81 CAS, see Complex adaptive systems (CAS) Case management (CM), 14, 16 ACM (see Adaptive case management (ACM)) application of solution, 28 Broker model, 25 delivers value, 24 Forrester's classification, 26-27 knowledge-oriented, 24 self-service models, 24 BPM suite (see Business process management (BPM) suite) British sitcom, 4–5 business dictionary, 5 classifications Forrester case management, 119-120 incident management case (see Incident management case classification) investigative case (see Investigative case classification) principles, 120-121 service request (see Service request classification) CORA model (see Common reference architecture (CORA) model) creating solutions ACM solutions, 27-30 agile methods, 33-34 build/buy, 30, 32-33 case data, 37 case documents, 37 CMMN standard, 34 credit card rewards dispute resolution, 36, 38-41 customer/internal supporting systems, 34 levels, 34-36 credit card rewards disputes BPMN, 11 CSR ticket, 10 real-life variations, 11, 13 traditional BPM model, 11-12, 14 crime-to-court process (see Crime-to-court process) criminal investigation, 7-9 DCM (see Dynamic case management (DCM)) doctor and patient, 5 ecosystem, 18-20 functional design design components (see Case management design components) design template, 116-117 principles, 101-102 goal-driven, 9-10 history, 20-21 knowledge work freedom, 9-10, 17-18 Nordstrom, 4 on-demand collaboration, 9-10 PCM (see Production case management (PCM)) solution framework, 171 uncertainty, 3 WF and BPM, 3

Case management design components cross-functional design aim, 114 integration, 115-116 principles, 114 roles and authorization, 114-115 information design aim, 108 document and media design, 112 operational level, 108-109 people relationship design, 112 principles, 108 strategic level, 110-111 tactical level, 109-110 interaction design, 113 aim, 112 channel functionality, 113-114 dashboard functionality, 113 portal functionality, 113 principles, 112 lifecycle design, 104 aim, 103 case/process flows design, 104-107 event design, 106, 108 principles, 104 rules design, 106 Case management model and notation (CMMN), 23, 34, 106 Case Management Society of America (CMSA), 22 Case management solution framework (CMSF), 99 CBD, see Component based development (CBD) CEP, see Complex event processing (CEP) Channel access layer, 156 Clinical case management, 25 CM, see Case management (CM) CMM, see Capability maturity model (CMM) CMMN, see Case management model and notation (CMMN) CMS, see Content management system (CMS) CMSA, see Case Management Society of America (CMSA) Common reference architecture (CORA) model, 411-414 business rules capabilities, 165, 169 case flows and processes capabilities, 165, 167-168, 170 event handling capabilities, 165 functional decomposition cross-functional design areas, 162, 164 design categories and components, 157, 159 high-level mapping, 158-159 information design mapping, 162–163 interaction design, 161–162 lifecycle design, 159-161 IT landscape, impact of business functionalities, 156 methodological input, 157-158 risk identification, 155 solution architecture, mapping, 157-158 stacked solution layer, 156 technical capability mapping, 156-157, 166, 168

Competitiveness of firms, 5-6 Complex adaptive systems (CAS) characteristics, 54 complicated systems, 53 components, 54-55 individual parts, 52 intentionality and purposefulness, 56 schematic of, 52-53 sense-and-respond manner, 56 subsystems, 54 Complex event processing (CEP), 30, 418-422 Component based development (CBD), 44 Composition layer, 156, 227-228 Content management system (CMS), 20, 328, 343, 369 Continuous Query Language (CQL), 232 CORA model, see Common reference architecture (CORA) model Credit card charge dispute case flow and process design, 317-318 description, 313-316 event design, 321-322 incident management, 315 information design data design, 323-324 document management, 323-324 implementation, 321 people relationship design, 322-323 rules design initiate chargeback, 321 investigate charge, 320-321 milestones, 319-320 notify customer, 321 notify merchant, 321 request documents, 319 review charge dispute, 320 write-off charge, 320 Credit card rewards disputes BPMN, 11 CSR ticket, 10 real-life variations, 11, 13 resolution, 36, 38-41 traditional BPM model, 11–12, 14 Crime-to-court process business challenges crime patterns, 282-283 globalization, 281-282 national budgets, 283 police organization, 282 social values, 282 technology, 282 police investigation (see Police investigations) Criminal investigation, 7-9, 134, 399, 421 CSR, see Customer service representative (CSR) Customer capital, 5-6 Customer centricity, 256-257, 260-261, 302 Customer relationship management (CRM), see Siebel Customer service representative (CSR), 314, 322, 373

D

Data integration, 234-235 Data layer, 156, 162, 228 Data objects customer data, 339-340 data form creation, 340-341 definitions, 336 dispute data, 337-338 flex fields, 341-342 identification, 335-336 investigation report, 339 Data storage, 156, 235 Diamond Credit Card Company, 10-11 Dynamic case management (DCM), 22, 48 characteristics, 67-68 competing design, 62 cost-quality-time, 69 definition, 66 deterministic vs. nondeterministic, 67-69 Google's search engine, 66 Holy Triangle, 61 Iron Triangle, 61 load and deformation characteristics, 65 pure play vendors, 62 purpose case management, 63 quality-cost-time, 63 system classifications, 64-65 traditional ERP, 67

E

Endeca Information Discovery, 306 components, 234 data integration, 234–235 data storage, 235 end-user information, 235 investigation area, 235 service request area, 235 End-user information discovery, 235 Enterprise Resource Planning (ERP), 57–58, 67, 76, 235 Event delivery network (EDN), 185 Event-driven architecture (EDA), 177

F

Flex fields, 341–342 Forrester's classification, 26–27, 119–120 Four-Cs principles capabilities, 58–59 collaboration, 59 context, 57–58 coordination, 59–60 Functional decomposition cross-functional design areas, 162, 164 design categories and components, 157, 159 high-level mapping, 158–159 information design mapping, 162–163 interaction design, 161–162 lifecycle design, 159–161

Н

Holy triangle, 61 Human task–based case activity creation, 354–356 definition, 356 task form, 356 Human-to-human (H2H) collaboration, 185

I

Incident management case classification business challenges, 134 context, 133 crime-to-court process (see Crime-to-court process) design components, 136-138 case flow model, 139-142 case state model, 143-145 phases, 135, 139 determination, 134 industry processes, 134 principles, 135 Insurance claim management, 312 case flow and process design, 300-301 case management classification, 299-300 customer facing, 299 financial, 298-299 injury claims, 298 IT landscape, 299 property claims, 298 rules design, 301 solution mapping case flows and process capabilities, 304-305 entitlement determination, 306 information discovery, 306 integration capabilities, 306 Oracle product, 306-307 user interface design channel design, 302 dashboard design, 302-304 Integration layer, 156, 162-163, 228 Intelligence design BAM application layer, 228 composition layer, 227-228 dashboard, 228 data layer, 228 implementation, 225, 227 integration layer, 228 presentation layer, 227 technical architecture, 229-230 components, 223-224 dashboard design, 225 Endeca Information Discovery, 234-235

event design, 224 information levels, 225-226 OEP credit card fraud, 231 functionality, 230 off-loading products, RFID, 231-232 OPA, 232–234 RTD, 236-237 rules design, 224 InvestigateCharge activity creation, 354-356 definition, 356-357 task form, 356 Investigative case classification business challenges, 145 context, 143, 145 design components, 148-150 case flow model, 146-147 case handling work, 151 case state model, 151-153 loosely coupled interactions, 146 determination, 145 industry processes, 145 principles, 145 Iron triangle, 61

J

Java server faces (JSF), 187, 340

K

Key performance indicators (KPIs), 302 BAM dashboard, 228 case management information levels, 109 case owners and departmental stakeholders, 110 claim handling, 302 operating model, 260 Oracle SOA suite, 261 product leadership lifecycle, 26, 263 tactical dashboards, 113 Knowledge governance, 88

L

Law enforcement and policing operations solution, 195

Μ

Mobile application framework (MAF), 187 Mortgage request handling business challenges customer service, 248 probability of default, 249 regulations, 248–249 case flow and process design, 250, 253 customer intimacy, 311 information design document and media, 253 one customer view, 253-255 people relationship, 255-256 interaction design customer centricity, 256 operational excellence, 257 product leadership, 257 investigations, 249 operating model customer centricity, 257, 260-261 lifecycle capability, 257-258 operational excellence, 257, 261, 263-264 product leadership, 257, 261-262 product leadership, 311 service request, 249

Ν

Nordstrom, 4

0

Object Management Group (OMG), 18, 23 OEP, see Oracle Event Processing (OEP) OOTB case management, see Out-of-the-box (OOTB) case management OPA, see Oracle policy automation (OPA) Operational excellence, 257, 261, 263-264, 311 Operations, administration, and management (OA&M), 83,88 OPSS, see Oracle Platform Security Services (OPSS) Oracle business intelligence, 110, 139, 225 Oracle Event Processing (OEP) credit card fraud, 231 functionality, 230 off-loading products, RFID, 231-232 Oracle Platform Security Services (OPSS), 344 Oracle policy automation (OPA) design and run time features, 232-233 entitlements determination, 233-234 insurance claim, 306-307 product leadership, 261-262 STP, 275-277 Oracle service cloud cross-functional design integration, 220 roles and authorization functionality, 220, 222 functionality, 210 history, 209 information design components, 214-215 data design, 215-217

document and media design, 216–217 people relationship design, 217 interaction design channel design, 213–214 dashboard, 214 portal design, 212–213 life cycle design case flow and process design, 218–219 event design, 219–220 product mapping, 220–221 rules design, 219 usage characteristics, 210–211, 239–241 Out-of-the-box (OOTB) case management, 30

Р

PCM, see Production case management (PCM) Person, object location, and event (POLE) information, 394-395 Police information model, 199-200 Police investigations case flow and process design high-level process, 284 police case type, 284-285 state model, 285-286 case recording evidence management, 403-404 incident, 400 information, 401 involved persons, 404 ownership assignment, 401-402 report out, 404-405 security model, 400-401 serialization, 402-403 classification, 283 end-to-end process, 280-281 event design, 286 full-blown scenario, 312 external systems, 290-291 implementation, 294, 296 interaction and information capabilities, 292-293 lifecycle capabilities usage, 292 incident recording coercive measures, 399 creation, 397 More Info feature, 399-400 offence management, 398-399 persons working, 397-398 information design, 286-287 interaction design completion status, 290 documents and media, 287 functionality, 288 involved persons, 287 linked cases, 287 police case, 288-289 police portal, 288-289

lead recording generic information, 391, 393 interview recording, 395 lead export, 396 operational information, 394-395 POLE, 394 360° overview, 396 storyline, 391-392 umbrella scenario, 312 dashboard functionality, 291 implementation, 295-296 interaction and information capabilities, 292-293 lifecycle capabilities usage, 292 phases, 291 portal functionality, 291 Policy governance, 86, 88 PrepareCase activity creation, 348, 350 data association, 351 definition, 353 implementation, 348 input type, 351-353 Start and End nodes, 352 Presentation layer, 156, 227 Production case management (PCM), 22, 48 characteristics, 67-68 competing design, 62 cost-quality-time, 69 definition, 66 deterministic vs. nondeterministic, 67-69 Google's search engine, 66 Holy Triangle, 61 Iron Triangle, 61 load and deformation characteristics, 65 pure play vendors, 62 purpose case management, 63 quality-cost-time, 63 system classifications, 64-65 traditional ERP, 67 Product leadership, 250-251, 257, 261-262, 264, 311

R

Radio frequency identification (RFID), 231 Range agility, 38–39 Real-Time Decisions (RTD), 236–237 Record-level access, 207 RFID, *see* Radio frequency identification (RFID) RTD, *see* Real-Time Decisions (RTD)

S

SAAS, *see* Software-as-a-service (SAAS) Sense-and-respond complex adaptive systems (SR-CAS) Four-Cs principles capabilities, 58–59 collaboration, 59

context, 57-58 coordination, 59-60 hybrid nature, 60-61 KPIs, 61 relevance of elements, 55-56 Service cloud, see Oracle Service cloud Service-oriented architecture (SOA) suite; see also Adaptive case management (ACM) agile adaptivity, 178 CCD case, 363-364 channel design, 187 customer centricity, 260 customer intimacy, 264 EDA, 177 full-blown scenario, 294 Fusion Middleware, 178 generalized BPM, 86 information support, 188 infrastructure, 366, 368 insurance claim, 306 integration capabilities, 179 integration functionality, 190, 206 interaction design, 260, 262-263 lifecycle management Siebel mapping, 204 operational excellence, 263 Oracle lifecycle mapping, 184 portal design, 186-187 principle, 178 product integration, 190 product leadership lifecycle, 262 umbrella scenario, 295-296 welfare request nonstandard situations, 275, 277 STP, 275-276 Service request classification business challenges, 122-123 context, 120 design components, 125-127 case flow model, 124, 128-129 case handling work, 128, 130-131 case-state model, 132-133 design template, 124 determination, 120-121 industry processes, 121–122 principles, 123 Siebel, 239, 409 building blocks, 194, 197 case flow and process design assignment management, 201-202 law enforcement and policing operations, 202-203 state model, 201-203 workflow management, 201 characteristics, 195-197 configuration vs. customization, 194 CRM, 193 cross functional category, 206-207 extensions function, 194-195

familiarization, 384-385 functional design complex state model, 386-387 cross-functional design, 386, 389 information design, 386 interaction design, 386 life cycle management, 385–386 process around coercive measures, 386, 388 function integration, 206 horizontal and verticals, 194 implementation, 383-384 business administration level, 389-390 technical administration level, 390 information design document and media design, 200 people relationship and data designs, 198-200 interaction design, 205-206 lifecycle design, 204 case flow and process design, 200-203 event design, 203 product mapping, 203-205 rules design, 203 police investigation (see Police investigations) portal and channel design, 205-206 roles and authorization, 206-207 solution mapping, 389 terminology, 194, 196-197 usage characteristics, 239, 241 SOA suite, see Service-oriented architecture (SOA) suite Software-as-a-service (SAAS), 209-211 Solution characteristics, 418-422 SR-CAS, see Sense-and-respond complex adaptive systems (SR-CAS) Stakeholder governance, 86, 88 Straight through processing (STP) application-to-application, 185 welfare request, 273, 275-276 Strength-based case management, 25

Т

Time agility, 39, 44 Total Elemental Interactions (TEI), 73

W

Welfare request process, 311–312 investigation cases, 267 life cycle design decision management, 270 determine entitlement, 270–271 multireality system, 270, 272 receive customer event, 268–269 nonstandard situations, 275, 277 organizations and citizens, 266–267 service request, 267 STP, 273, 275–276 Work flow (WF), 3, 74